

University of South Florida Scholar Commons

Graduate Theses and Dissertations

Graduate School

3-24-2016

Financial Literacy Continuing Professional Education Cognitive Needs Assessment for Florida Small Business Owners

Pearl Dahmen Para

Follow this and additional works at: http://scholarcommons.usf.edu/etd Part of the <u>Adult and Continuing Education and Teaching Commons</u>

Scholar Commons Citation

Para, Pearl Dahmen, "Financial Literacy Continuing Professional Education Cognitive Needs Assessment for Florida Small Business Owners" (2016). *Graduate Theses and Dissertations*. http://scholarcommons.usf.edu/etd/6126

This Dissertation is brought to you for free and open access by the Graduate School at Scholar Commons. It has been accepted for inclusion in Graduate Theses and Dissertations by an authorized administrator of Scholar Commons. For more information, please contact scholarcommons@usf.edu.

Financial Literacy Continuing Professional Education

Cognitive Needs Assessment

for

Florida Small Business Owners

by

Pearl J. Dahmen

A dissertation submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Curriculum and Instruction with an emphasis in Adult Education Department of Leadership, Counseling, Adult, Career, and Higher Education College of Education University of South Florida

> Major Professor: Waynne B. James, Ed. D. Jeffrey D. Kromrey, Ph. D. William H. Young, III, Ed. D. Katerina Annaraud, Ph. D.

> > Date of Approval: March 10, 2016

Keywords: financial knowledge, financial capability, small business, entrepreneur

Copyright © 2016, Pearl J. Dahmen

Dedication

This work is dedicated to a young entrepreneur with the courage to start her own business (Tropicality), my daughter, Becca Dahmen.

It takes a village to raise a Ph.D. (modified African proverb). I want to express my appreciation for the support and encouragement of my entire village: family, friends, classmates, professors, business associates, colleagues, and the myriad individuals who personally participated in my research. I would love to list everyone by name, but fear my memory may fail me. To all of them, I say, "thank you, a thousand times thank you."

There are a few key individuals who had a notable influence on my journey. I do wish to recognize them by name:

To my husband of six years, Greg Para, I owe an overwhelming thank you for sharing his creation, THE GOAT PROGRAM®, and his passion for teaching teens. The journey to my Ph.D. started with our partnership.

To Eileen Rodriguez, Director of the West Central region of the Small Business Development Center at the University of South Florida, I owe my gratitude. Our writing collaboration, and Eileen's professional generosity, inspired me to endure my journey.

To my Sarasota travel buddies and classmates, Heba and Tanya, I give my appreciation, our time together transformed a chore into a joy, provided strength to sustain, and brought fun to the final days of a long intense process.

Acknowledgments

I would like to express my deepest heartfelt gratitude to my committee for guiding and encouraging me throughout this journey, thank you for the support, reassurance, assistance, and advice.

Dr. Waynne James, my major professor, took me under her protective wing, and led with the tenderness, style, and grace that only she possesses. Words cannot express my feelings of appreciation. Thank you for everything; yes, everything. It is with my deepest, and sincerest gratitude, I say, "I would not have done it without you."

Dr. William H. Young shared his insightful words. His contribution was influential in my journey in so many ways.

Dr. Jeffery D. Kromrey was my statistic guru. From the first day of statistic class to the final days of dissertation analytical exploration, he provided the wisdom and guidance to complete the tasks at hand.

Dr. Katerina Annaraud, a jewel from the University of South Florida Sarasota-Manatee campus, was the perfect member rounding out my committee. Her style and flair were a lovely addition to my journey contributing the unique perspective only she could possess.

I must also acknowledge and express thanks for the professional support of the Florida Small Business Development Center Director, Mike Myhre, Associate State Director, Dr. Cheryl Kirby, and FSBDCN Computer Analyst, James Gray, as well as the entire team at the Small Business Development Center at the University of South Florida.

Table of Contents

List of Tablesiv		
List of Figuresvi	ïi	
Abstractvii	ii	
Chapter 1: Introduction	145668812	
Chapter 2: Literature Review	448912345034699124691770	

Evaluation of assessments	62
Summary	65
Chapter 3: Methods	67
Research Design	67
Research questions	67
Variable	68
Population and Sample	69
Instrumentation	71
Instrument development	72
Instrument development steps	77
Final instrument	85
Data Collection	86
Field test	86
Procedures	86
Data Analysis	88
Content validity	89
Evidence demonstrating validity	89
Validation process procedures	93
Construct validity	93
Reliability	98
Descriptive statistics and <i>t</i> tests	99
Analysis of variance (ANOVA)	99
Summary	101
Chapter 4: Findings	102
Research Questions	102
Study Participants	102
Response rates of participants	103
Demographic profile of respondents	104
Analysis of Research Questions	104
Research question 1	104
Research question 2	107
Research question 3	109
Research question 4	103
Research question 5	114
Observations	117
Chapter 5: Summary, Conclusions, Implications and Recommendations	121
Summary	121
Conclusions	122
Implications	124
Private business and government agencies	124
Entrepreneurship programs	124
Small Business Development Center (SBDC) stakeholders	125
Small Business Development Center (SBDC) consultants	125

Florida Legislature	126
Recommendations for Further Research	126
References	129
Appendices	149
Appendix A: Big Five Financial Literacy Survey Questions	150
Appendix B: Institutional Letter of Support	151
Appendix C: Pool of Financial Literacy Questions	152
Appendix D: Panel Members	193
Appendix E: Request for Participation to Potential Panel Members Email	197
Appendix F: Initial Panel Communications	198
Appendix G: Initial Panel Round 1 Online Survey	200
Appendix H: Initial Panel Round 2 Online Survey	202
Appendix I: Validation Panel Communications	281
Appendix J: Verification Panel Communication	285
Appendix K: Validation Panel Round 1 Online Survey	288
Appendix L: Self-assessment Financial Literacy Questions	330
Appendix M: Validation Panel Round 2 Online Survey	331
Appendix N: Verification Panel Online Survey	375
Appendix O: Field Test Communication	396
Appendix P: Field Test Online Survey	
Appendix Q: Final Review Panel Email Communication	404
Appendix R: Final Review Panel Online Survey	
Appendix S: Final Financial Literacy Instrument	
Appendix T. Participant Email Communications	420
Appendix U: IRB Approval Letter	424
Appendix V: Summary of Sources and Permission used to Obtain	
Financial Literacy Question for Dissertation Study	426
	720

About the AuthorE	nd P	'age	Ś
-------------------	------	------	---

List of Tables

Table 1:	Number of Times That Financial Literacy Definitions was Cited in Journal Articles by Author(s) Based in Google Scholar Citations
Table 2:	Financial Literacy Attributes by Domain, Researcher(s) and Journal Article, Report of Dissertation53
Table 3:	Summary and Comparison of Financial Literacy Content Categories Defined by FLEC and FLAT
Table 4:	Profile of Variables Related to the Conduct of the Study
Table 5:	Fernandes, Lynch, and Netemeyer (2014) Correlations of Financial Literacy with Gender, Age, Education, and Self-Efficacy
Table 6:	Fourteen Sources Used to Develop Initial Financial Literacy Question Pool Including Criteria for Selection, Number of Questions (Initial, Eliminated, and Retained) and Participant Age and Originating Country
Table 7:	Research Process Steps to Achieve Content Validation
Table 8:	Expert Panel Demographics78
Table 9:	Selection Criteria #2, Number of Times Researchers Were Cited
Table 10	Descriptive Statistics of Field Test
Table 11:	Factor Pattern Coefficients and Communalities Based on Exploratory Factor Analysis97
Table 12	Analysis of Dependent Variables Skewness and Kurtosis 101
Table 13	Demographic Characteristics of Respondents in the Survey 105
Table 14	Descriptive Analysis Profiling the Financial Literacy of Florida Small Business Owners
Table 15:	Descriptive Analysis Profiling the Financial Literacy by Gender

Table 16: Descriptive Analysis Profiling the Financial Literacy by Age Group	109
Table 17: ANOVA Summary Table of Tested Knowledge for Age Groups	110
Table 18: ANOVA Summary Table of Self-assessed Knowledge for Age Groups	110
Table 19: Tukey's Test Post Hoc Results of Self-assessed Knowledge by Age Group	111
Table 20: Descriptive Analysis Profiling the Financial Literacy by Education Level	112
Table 21: ANOVA Summary Table of Tested Knowledge for Education Level	113
Table 22: Tukey's Test Post Hoc Results of Tested Knowledge by Education Level	114
Table 23: ANOVA Summary Table of Self-assessed Knowledge for Education Level	114
Table 24: Descriptive Analysis Profiling the Financial Literacy by SmallBusiness Classification	115
Table 25: ANOVA Summary Table of Tested Knowledge for Small Business Classification	115
Table 26: Tukey's Test Post Hoc Results of Tested Knowledge by Small Business Classification	116
Table 27: ANOVA Summary Table of Self-assessed Knowledge for Small Business Classification	117
Table 28: Tukey's Test Post Hoc Results of Self-assessed Knowledge by Small Business Classification	117
Table C1: Initial Pool of Financial Literacy Questions for Content Validation Process	152
Table C2: Key to Pool of Financial Literacy Questions Source Codes: Source, Number (n), and References	191
Table D1: Initial Panel Members and Their Areas of Expertise	193
Table D2: Validation Panel Members and Their Areas of Expertise	195

Table D3: Verification Panel Members and Their Areas of Expertise	. 195
Table D4: Final Review Panel Members and Their Areas of Expertise	. 196

List of Figures

Figure 1:	Financial Literacy/Capabilities Model Summarizing the Financial Literacy Attribute Relationships within Financial Capabilities	54
Figure 2:	Scree Plot of Eigenvalues Depicting Decision Point for Number of Factors to Use based on Loading	96
Figure 3:	Histogram of Tested Knowledge Score Percentage Distribution10	06
Figure 4:	Histogram of Self-assessed Knowledge Score Percentage Distribution	07

Abstract

The purpose of this study was to assess the financial literacy continuing profession education cognitive needs of Florida small business owners through exploring their profile. To determine the financial literacy profile, an instrument containing 18 tested knowledge and 5 self-assessed knowledge questions was created. Using a panel of experts, the instrument was developed from previously tested financial literacy questions from several sources.

Data were collected from clients of the West Central Region of the Florida Small Business Development Center at the University of South Florida. The online survey completed by participants included demographic questions to provide data to profile small business owners' financial literacy by gender, age, education level, and small business classification.

The results indicated small business owners have a high financial literacy. There were significant differences found between the financial literacy of men and women. Men's scores were higher for both tested knowledge and self-assessed knowledge. Younger small business owners scored lower than older small business owners. There were significant scoring differences between the highest and lowest levels of education. Tested scores and self-assessed scores increased with higher education levels. Preventure/start-up business owners scored lower than the small-medium enterprise owners. Implications included developing educational programs attentive to women small business owner's needs, as well as newer and/or younger small business owners.

viii

Chapter 1

Introduction

Globally low levels of financial literacy are the source of unease (Lusardi & Mitchell, 2011a, 2011b, 2013, 2014). International policy makers have expressed extreme concern about the prevalent gaps in financial literacy (Atkinson & Messy, 2012). According to Lusardi and Mitchell (2013), the literature indicated large numbers of people around the world were assessed to be financially illiterate. Lusardi and Mitchell (2013) recommend improving the level of financial literacy world-wide as vital to addressing the concerns global leaders, policy makers and business people. The changing landscape of financial markets and individual's requirements for interactions in these markets produced the recent increase of awareness in financial literacy highlighting the need for improvement (Atkinson & Messy, 2012; Donohue, 2011; Hsu, 2011; Hung, Yoong, & Brown, 2012; Lusardi & Mitchell, 2008; Lusardi & Wallace, 2013; Mottola, 2013; Schmeiser & Seligman, 2013). The United States has not been immune to the financial literacy crisis (Lusardi & Mitchell, 2013). Schwab, chairman of the President's Advisory Council on Financial Literacy, commented saying, "We believe the market turmoil and credit crisis of 2008 underscore the critical need for improved financial literacy in the United States" (President's Advisory Council on Financial Literacy [PACFL], 2009, p. VII).

Evaluating financial literacy in the United States, Lusardi and Mitchell (2013) pointed to changes in pensions as one of the major contributors to the current state of

financial literacy. Responsibilities for saving, investing, and decumulating wealth have shifted from corporations and government to workers and retirees. Forty years ago, social security was supplemented with defined-benefit pension plans focused on what a retiree would receive at retirement. Workers relied on employers and did not need the level sophistication and understanding of financial matters that have become essential in today's economy. The current defined-contribution pension plans and individual retirement accounts (IRAs) emphasize what must be invested in the present. These changes in retirement planning demands coupled with the increasing complexity of the financial markets require high levels of financial literacy. Extended life expectancies add the risk of retirees outliving their assets, thus compounding the necessity of improving financial literacy (Finke, Howe, & Huston, 2011; Lusardi, 2011; Lusardi & Beeler, 2007; Lusardi, 2012a; Lusardi & Mitchell, 2007a; Van Rooij, Lusardi, & Alessie, 2007, 2009, 2011). The current generation of retirees, Baby Boomers (born 1946 to 1964), is facing the need for more sophisticated financial literacy (Lusardi & Beeler, 2007; Lusardi & Mitchell, 2007a, 2013, 2014; Klapper, Lusardi, & Panos, 2013; Van Rooij et al., 2011).

Borrowing money has also become more complicated. Consumers have greater access to installment credit and credit cards, mortgages, including sub-prime mortgages, and alternative financial services, such as payday loans, pawn shops, auto title loans, tax refund loans, and rent-to-own shops. Generation X (born 1965 to 1980), Millennial (born 1981 to 2000), and Boomlet (born after 2001) generations must also be concerned about the present as well as the future of their financial well-being (Lusardi, 2012a; 2015; Lusardi & Mitchell, 2013; Novak, 2014). To manage their financial well-

being, an individual must address the long-range money practices of building assets, protecting assets, and borrowing as well as basic money concepts for day-to-day money management (Kunovskava, Cude, & Alexeev, 2014; Lusardi & Mitchell, 2011a, 2011b; Miller, Reichstein, Salas, & Zia, 2014; Van Rooij et al., 2011). The precise definition of financial literacy continues to be open to deliberation. Hung, Parker, and Yoong (2009), Huston (2010), and Remund (2010) concurred that an agreed-upon definition does not exist. Based on the myriad of conceptual definitions of financial literacy, researchers developed models, including several attributes other than knowledge. Financial understanding, ability, skills, aptitude, confidence, experiences, behavior, financial well-being, financial socialization, financial inclusion, and resources were included in various conceptual definitions (Hung et al., 2009; Huston, 2010; Lee & Hanna, 2014; Lusardi, 2011, 2012a, 2012b; Lusardi & Beeler, 2007; Lusardi & Mitchell, 2007a, 2007b, 2008, 2011a, 2011b, 2013, 2014; Lusardi & Scheresberg, 2013; Lusardi & Wallace, 2013; Mason & Watson, 2000; Mottola, 2013; Nicolini, Cude, & Chatterjee, 2013; Robb & Woodyard, 2011; Scheresberg, 2013; Schmeiser & Seligman, 2013; Sherraden, 2010; Woodyard & Robb, 2012).

Financial literacy attributes identified by researchers can be classified into three domains (cognitive, affective, and social). This research focused on the cognitive domain, more specifically knowledge, including both objective and subjective financial literacy. Carlson, Vincent, Hardsty, and Bearden (2009) considered objective knowledge to "reflect what we know" (p. 864) and subjective knowledge to "reflect what we know" (p. 864) and subjective knowledge) is easily measured by the number or percentage of correct responses to financial literacy

knowledge questions; while subjective knowledge (self-assessed) is the self-belief determined through self-assessment, based on personal perception and confidence in their skills and abilities (Fernandes, Lynch, & Netemeyer, 2014; Wilson, Kickul, & Marlino, 2007).

The idea of financial capability has been raised by some researchers. Sherraden and Grinstein-Weiss (2015) proposed financial capability goes beyond financial literacy to include the ability and opportunity to act. Lachance (2014) indicated financial capability represents the ability to manage financial resources effectively empowering "individuals to make informed choices, avoid pitfalls, know where to go for help, and take other actions to improve their present and long-term financial well-being" (p. 17). In other words, financial capability results in financial well-being as an outcome of financial literacy together with an individual's access to resources, financial information, and financial products. This also requires inclusion in financial markets and access to financial services (Buckland, 2014; Hung et al., 2012; Robles, 2014; Sherraden, 2010, Sherraden & Grinstein-Weiss, 2015).

Statement of the Problem

Research focusing on the general United States (U.S.) population agreed financial literacy is low and improvement is imperative to the economy (Lusardi & Mitchell, 2008, 2013, 2014; Lusardi & Tufano, 2009; Lusardi & Wallace, 2013; Mandell & Klein, 2009; Theodos, Kalish, Mckernan, & Radcliffe, 2014). State-by-state analysis identified geographic disparity in financial literacy (Bumcrot, Lin, & Lusardi, 2013). Based on the National Financial Capability Study (NFCS), the State of Florida financial literacy scores ranked in the bottom 20% of the U.S. (FINRA Investor Education

Foundation, 2013). Research related to financial literacy of small business owners is minimal, and no research has been conducted in the U.S. Without a profile of the financial literacy of small business owners, legislators and governmental agency policy makers can only guess at the financial literacy continuing professional education needs of small business owners when developing and funding small business education initiatives.

Statement of the Purpose

The purpose of this study was to assess the financial literacy continuing professional education cognitive needs of Florida small business owners. In addition, financial literacy continuing professional education needs of this group by gender, age, education level, and small business classification have not been investigated. This research adds to the knowledge base and informs governmental agencies such as the Small Business Administration (SBA), American Small Business Development Centers (ASBDC), Florida Small Business Development Centers Network (FSBDCN), and Service Corps of Retired Executives (SCORE) of the educational needs of Florida small business owners (Newman, Ridenour, Newman, & DeMarco, 2003). Non-profit organizations such as SBA, ASBDC, FSBDCN, and SCORE, funded by federal and state government agencies, exist to educate and support small businesses and their owners (American Small Business Development Centers [ASBDC], 2016; Florida Small Business Development Center Network [FSBDCN], 2016a; Small Business Administration [SBA], 2013). This research provides a basis for future programs these organizations might offer to small business owners.

Research Questions

The research questions identified to address the purpose of this study were:

- 1. What is the financial literacy profile of Florida small business owners?
- 2. What are the direction and extent of differences in the financial literacy profile by gender?
- 3. What are the direction and extent of differences in the financial literacy profile by age?
- 4. What are the direction and extent of differences in the financial literacy profile by education level?
- 5. What are the direction and extent of differences in the financial literacy profile by small business classification?

Significance of the Study

The SBA's Office of Advocacy (2016a, 2016b) described Florida's nearly 2.2 million small businesses as drastically impacting Florida's economy; advising Florida small business is crucial to the fiscal condition of the state. A substantial portion of Florida's economy was, and continues to be, dependent on the state's small businesses (Atwater, 2013). According to the 2010 U.S. Census business statistics, 91.5% of Florida businesses reported having less than 20 employees, which put them into the category of a small business. As of February 2015, in the SBA Small Business Profiles for the States and Territories, Florida was ranked number three (behind Texas and California) for starting and maintaining a small businesses in the U.S. (SBA Office of Advocacy, 2016c). In light of over two million small businesses representing 97.5% of Florida's workforce and the state's overall low financial literacy ranking, there was a void in the understanding of the need for financial literacy continuing profession education of Florida small business owners (U.S. Census, 2010). Research covering Florida small business owners' financial literacy ratings of did not exist.

The fiscal year 2015 U.S. Federal Budget had 710 million dollars allocated to the SBA to support small business owners. An additional 15 million dollars of the SBA budget was allocated to entrepreneurial education initiatives (SBA, 2015). In 2013, the State of Florida enacted House Bill No. 2007 (K-20 Education, 2013) allocating four million dollars to the FSBDCN to provide support for Florida small business owners. Federal and Florida budgets provide evidence of the magnitude of federal and Florida dollars being allocated to educate and support small business owners. With this in mind, informing governmental agencies and the Florida legislature of the financial literacy continuing professional education needs of small business owners would provide insights to supporting this group with financial literacy education. Efforts by ASBDC, FSBDCN, SCORE, and the Florida legislature to educate and support small business owners are rooted in goals to grow and improve the Florida economy, and in turn, the U.S. economy.

Entrepreneurial education is at the core of supporting small businesses to grow and flourish (SBA, 2013). Fairlie and Holleran (2012) examined the state of entrepreneurship training for individuals noting governments around the world view promoting entrepreneurship as a national priority. O'Connor (2013) explored the governmental interest in entrepreneurship education advocating achievement of specific economic outcomes as the goal of entrepreneurship education. According to Nunoo and Andoh (2012), financial literacy education was an important component of entrepreneurial education.

This study aimed to contribute the small business owner financial literacy perspective to the body of work examining the complex phenomena of financial literacy.

Focusing on the Florida small business owners' financial literacy continuing professional education cognitive needs with particular attention to differences based on gender, age, education level, and small business classification, this research can potentially provide invaluable information for the Florida legislature and Florida governmental agencies supporting small business owners.

Limitations

This study had certain inherent limitations. This study was conducted in one region of Florida. Recommendations made, as well as the survey results, of this study must be properly understood as being limited to the West Central region of Florida.

This study was administered online. A link to the study's survey was sent via email to potential participants. The prospective participants, clients of the Florida Small Business Development Center Network, represented the segment of the Florida small business population that had chosen to seek consulting and advice from this non-profit organization. The potential participant pool did not include small business owners who had not sought out business consulting, as well as those who had engaged another non-profit organization or private consultant. Generalization to all Florida small business owners may be restricted.

Definition of Terms

The following operational definitions were used for the terminology within this research study:

Financial attitude. The way an individual thinks or feels about money and its use in a financial context.

Financial behavior. The way individuals act or conduct themselves in relation to money and financial transactions.

Financial capability. A broad concept, encompassing knowledge and skills necessary to understand personal financial circumstances together with the motivation to take action, the resources, and the access to financial services required to manage financial resources effectively.

Financial knowledge. Monetary information, understanding, or skill gained from experience or education.

Financial literacy. An individual's ability to use knowledge and skills in conjunction with personal financial attitudes and beliefs to manage personal monetary resources effectively based on accessible economic information and products for lifetime fiscal security.

Financial literacy profile. The combination of an individual's tested financial literacy assessment score and self-assessed financial literacy assessment score representing the individual's grasp of financial information.

Knowledge content categories. Four categories commonly employed to summarize financial concepts when teaching and assessing financial literacy are:

 Money basics. Money basics is the first of four knowledge content categories of financial literacy, and represents concepts necessary to understand and manage money or financial resources. Concepts include: compound interest, time value of money, purchasing power/inflation, spending, and short-term savings.

- 2. Borrowing. Borrowing is the second of four knowledge content categories of financial literacy representing concepts relating to the debt (using money from a financial institution under an agreement to repay it plus a fee for the privilege of using it). Concepts include annual percentage rate (APR), finance charges, loan terms, credit scores; and types of borrowing (e.g., education, home, vehicles).
- Building assets. Building assets is the third of four knowledge content categories of financial literacy representing concepts relating to growing and increasing financial resources. Concepts include long-term-savings accounts/certificates of deposit, stocks, bonds, mutual funds, and annuities.
- 4. Protecting assets. Protecting assets is the fourth of four knowledge content categories of financial literacy, and represents concepts relating to avoiding the possibility of loss of money and/or financial resources. Concepts include: investment product risk (stocks vs. bonds vs. mutual funds), investment diversity, and insurance protection.

Self-assessed financial knowledge. Self-belief or perception of financial knowledge determined through self-assessment, based on personal perception and confidence.

Small business. A small business concern is one that is independently owned and operated, is organized for profit, and is not dominant in its field. Small business owners are often referred to as entrepreneurs. Entrepreneurs are considered individuals establishing and managing a business for the principal purpose of profit and

growth while furthering personal goals. The small business is commonly the primary source of income and will consume the majority of person's time and resources.

Small Business Classifications. Categorizes small businesses based on the number of years in business and number of employees.

- Pre-venture/start-up. Represents a small business classification where individuals are either exploring the possibility of starting a new business (preventure) or have been operating a new business for less than three years. There are no employees.
- 2. Small business. Represents a small business classification where individuals are in business. They have been in business for more than three years and have less than five full-time employees.
- Small-medium size enterprises. Represents a small business classification where a business has been operating for three or more years with five or more full-time employees.

Tested financial knowledge. Financial knowledge assessment determined through test results.

List of Acronyms

Numerous acronyms are commonly used when referring to organizations and studies or instruments in the literature. The following is a list of the acronyms used in this study and what they stand for:

ALP	RAND American Life Panel Study
ASBDC	American Small Business Development Centers
FINRA	Financial Industry Regulatory Authority

FLAT	Financial Literacy Assessment Test
FLEC	Financial Literacy and Education Commission
FSBDCN	Florida Small Business Development Center Network
HRS	Health and Retirement Study
NBER	National Bureau of Economic Research
NFCS	National Financial Capability Study
OECD	Organization for Economic Co-operation and Development
PACFC	President's Advisory Council on Financial Capability
PACFL	President's Advisory Council on Financial Literacy
SBDC	Small Business Development Center
SBA	Small Business Administration
SCORE	Service Corps of Retired Executives

Organization of the Study

This study is organized into five chapters. Chapter 1 includes a statement of the problem, discussion of the purpose of the study, statement of research questions, significance of the study, limitations, definition of terms, a list of acronyms, and organization of the study. Chapter 2 reviews and evaluates pertinent literature related to the history of financial literacy in the United States, the current state of financial literacy (global, United States, Florida, and small business owners), measuring financial literacy (definitions, models, and assessments), and a summary. Chapter 3 details the research methods for the study. This includes the research design, research questions, population and sample, instrumentation, data collection process, data analysis, and a summary. Chapter 4 includes the research questions, response rates of participants,

demographic characteristics of respondents, and analyzes of the financial literacy assessment scores by the five search questions, and observations. Chapter 5 includes a summary of the study, conclusions, implications, and recommendations for further research.

Chapter 2

Literature Review

The purpose of this study was to assess the financial literacy continuing professional education cognitive needs of Florida small business owners. Literature about the history of financial literacy in the United States opens the chapter. Next, the chapter presents the current state of financial literacy, starting with global financial literacy, moving to the United States financial literacy, financial literacy measurement, and a summary.

History of Financial Literacy

Economic crises around the world fueled concern over financial illiteracy as measured internationally by researchers in numerous articles, reports, and working papers (Atkinson & Messy, 2012; Bumcrot et al., 2013; Crossan, Feslier, & Hurnard, 2011; Fornero & Monticone, 2011; Klapper et al., 2013; Lusardi & Mitchell, 2013; Pahnke & Honekamp, 2010; Vieira, 2012; Wolfe-Hayes, 2010). Anxiety over the current state of the United States financial literacy continued to grow until a presidential advisory council was created (President's Advisory Council on Financial Capability [PACFC], 2013a; President's Advisory Council on Financial Literacy [PACFL], 2009).

Sensitivity to financial literacy and the needs of adults to learn financial skills date back to 1795. Davies (1795), a clergyman, discussed the needs of poor agricultural workers in *The Case of Labourers in Husbandry Stated and Considered.* Davies (1795) suggested, "encouraging frugality among them [poor workers]" (p. 98) by teaching the laborers to save more money through creating a budget. As the field of adult education developed, the 1862 Morrill Act authorized the establishment of land grant institutions (Liston & Bivens, 1993; Tschache, 2009). These institutions fostered the creation of home economics as an area of study. Home economics covered understanding household finance. Concepts such as debt, credit, loaning, borrowing, and interest rates, as well as the life skills of making a budget, keeping financial records, and making frugal purchasing choices, were also included in the curriculum (Liston & Bivens, 1993). The 1914 Smith-Lever Act launched the Cooperative Extension Service (Tschache, 2009). The Cooperative Extension Service was established to provide information by bringing learning opportunities to where learners lived in rural and farm areas. The Smith-Hughes National Vocational Education Act of 1917 expanded on the Smith-Lever Act (Tschache, 2009). This allowed land grant institutions' Cooperative Extension Service programs to offer practical vocational education courses, including financial literacy education (Kozup & Hogarth, 2008; Liston & Bivens, 1993; Tschache, 2009).

From the 1930s to the 1990s, education in family resource management, including financial literacy, was sponsored by government agencies and offered by private organizations in communities and the workplace (Fox, Bartholomae, & Lee, 2005; Hopely, 2003; Vitt, Anderson, Kent, Lyter, Siegenthaler, & Ward, 2000). In 1995, the U.S. Department of the Treasury together with 65 public and private organizations assembled the American Savings Education Council with the expressed mission of improving financial literacy (American Savings Education Council, n.d.). In 1998, the Securities and Exchange Commission (SEC) in partnership with nearly 50 private and

public entities targeted improving financial literacy by beginning a campaign to encourage savings and investments (Vitt et al., 2000). By 2000, the United States had over 90 financial educations programs available to adult learners (Hopely, 2003).

The establishment of Financial Literacy and Education Commission (FLEC) under the Fair and Accurate Credit Transactions Act of 2003 marked the start of the U.S. Government's involvement in financial literacy (Financial Literacy and Education Commission [FLEC], 2010). The Secretary of Treasury headed the commission of 19 federal agencies and bureaus to promote financial literacy (FLEC, 2010). For several years, the existing programs functioned with little notice except the agencies that funded them. Agencies providing the programs reported statistics related to the program's impacts, typically with a positive picture of the educational outcomes (Miller et al., 2014; Tippet & Kluvers, 2007). Concerns over financial literacy were elevated to the presidential level in 2008 when President Bush created the President's Advisory Council on Financial Literacy (PACFL) charged with improving financial literacy in the United States (PACFL, 2009).

The United States economic crisis of 2009 escalated the interest in financial literacy. On January 29, 2010, President Obama signed Executive Order 13530 creating the President's Advisory Council on Financial Capability (PACFC) (Executive Order 13520, 2010). PACFC was tasked with solving the problem of financial capability by addressing financial literacy among the American people. The goal of PACFC was to assist Americans in understanding financial matters and making informed financial decisions (PACFC, 2013a). On April 2, 2010, President Obama (2010) issued a Presidential Proclamation calling for April to be the financial literacy month. He linked

financial literacy to the economic health of our nation saying, "Our recent economic crisis was the result of both irresponsible actions on Wall Street, and everyday choices on Main Street" (Obama, 2010, para. 2). The comment fueled funding by government agencies, private organizations, and educational institutions to support research as well as the development and implementation of financial education to combat the low levels of financial literacy in the United States (FINRA Investor Education Foundation, 2016a; Financial Literacy and Education Commission, 2014; National Endowment for Financial Education, 2014).

In 2011, the FLEC's National Strategy Working Group released the *United States National Strategy 2011* (FLEC, 2010) for financial literacy. The intention was to engage all FLEC member agencies as well as non-profit, private, and academic sectors to achieve the following strategic goals: Increase awareness of the access to effective financial education, "1. Determine and integrate core financial competencies, 2. Improve financial education infrastructure, and 3. Identify, enhance, and share effective practices" (FLEC, 2010, p. 2).

Over the past 200 years, billions of government and private dollars were invested in financial education to improve financial literacy in the United States. Evidence of positive increases financial literacy scores was expected. The plethora of research since 2000, covering the state of global and the United States financial literacy paints a picture contrary to this expectation (Fernandes et al., 2014; Huston, 2010; Miller et al., 2014; Vieira, 2012).

Current State of Financial Literacy

Global financial crisis has induced global leaders, major banking companies, financial institutions, economic experts, government agencies, grass-roots consumer/community interest groups, and academia to search for causes of the crisis (Braunstein & Welch, 2002; Gallery & Gallery, 2010). Wolfe-Hayes (2010) suggested there is a connection between the financial crisis and financial literacy:

It is no great surprise to learn that current financial crisis began with sub-prime mortgages that were marketed primarily to those with less income and education—and presumably less financial literacy—than those who were eligible for prime mortgages. Financial literacy clearly has ongoing macroeconomic ramifications. (p. 107)

Concurring with the connection, Hung et al. (2009) encapsulated financial literacy's contribution to the economic downturn in the U.S. stating, "poor financial decision-making appears to be a surprisingly widespread phenomenon with sizeable consequences that build over time and often go unnoticed until a point of crisis" (p. 21). According to the Consumer Financial Protections Bureau's (2013) *Financial Literacy Annual Report,* there is a relationship between the state of the economy and financial literacy. The Consumer Financial Protections Bureau stated, "Financially capable consumers are essential to fully and responsibly harness the financial system's tremendous ability to enhance economic stability" (p. 58). Examples from recent research point to the lack of financial literacy as a contributing factor in the decline of responsible personal financial management. This decline played a significant part in the waning economy (Consumer Financial Protections Bureau, 2013).

Investigating financial literacy in more detail, researchers found women's financial literacy assessment scores were lower than their male counterparts (Agnew &

Carmeron-Agnew, 2015; Atkinson & Messy, 2012; Donohue, 2011; Hsu, 2011; Hung et al., 2012; Lusardi & Mitchell, 2008; Lusardi & Wallace, 2013; Mottola, 2013; Schmeiser & Seligman, 2013). When researchers explored financial literacy by age, they discovered young adults and elderly adults show greater incidence of low financial literacy assessment scores (Beckmann, 2013; Hsu, 2011; Lusardi, 2011, 2012b; Lusardi & Mitchell, 2014; Lusardi, Mitchell, & Curto, 2010; Lusardi & Tufano, 2009; Mandell & Klein, 2007; Pahnke & Honekamp, 2010; Scheresberg, 2013).

Global financial literacy. In 2011, the Global Financial Literacy Excellence Center was founded at the George Washington University School of Businesses to inform policy as well as develop and promote financial literacy programs around the world through scholarship and research. The Global Financial Literacy Excellence Center "mission was driven by the awareness that financial literacy levels around the world have reached a crisis point" (2016a, para. 3). The organization's research, disseminated through journal articles and policy papers, indicated financial literacy was universally low worldwide. The research also highlighted women, ethnic groups, individuals with low income, and those with low education reported lower than average financial literacy assessment scores around the world (Global Financial Literacy Excellence Center, 2016b; Lusardi & Mitchell, 2011a, 2011b, 2014; Lusardi & Wallace, 2013).

The Organisation for Economic Co-operation and Development (OECD) conducted a Pilot Study of 14 countries (Albania, Armenia, British Virgin Islands, Czech Republic, Estonia, Germany, Hungary, Ireland, Malaysia, Norway, Peru, Poland, South Africa, and the United Kingdom) using an instrument developed by the OECD

International Network of Financial Education. The study identified worldwide financial knowledge as low (Atkinson & Messy, 2012). According to the OECD report, Measuring Financial Literacy (Atkinson & Messy, 2012), most people exhibit very basic financial knowledge, but more complex concepts such as compound interest and diversification were limited among the populations of every country. The OECD also found lower scores were prevalent for women and individuals with lower levels of education and income (Atkinson & Messy, 2012). Individual research studies covering numerous countries around the world identified low levels of financial literacy in Australia (Agnew, Bateman, & Thorp, 2013), Brazil (Mendes-da-Silva, Nakamura, & Moraes, 2012; Norvilitis, & Mendes-da-Silva, 2013; Potrich, Vierira, Coronel, & Filho, 2015), Canada (Nicolini et al., 2013), Dominican Republic (Drexler, Fischer, & Schoar, 2010), Europe (Klapper et al., 2013), France (Arrondel, Debbich, & Savignac, 2013), Germany (Pahnke & Honekamp, 2010), Ghana (Ansong, 2011), India (Bonte & Filipiak, 2012; Kumar, Mathur, & Sal, 2014), Italy (Fornero & Monticone, 2011), Japan (Sekita, 2011), Mexico (Hastings & Mitchell, 2011), New Zealand (Crossan et al., 2011), Romania (Beckmann, 2013), South Africa (Shambare & Rungimbana, 2012), Switzerland (Brown & Graf, 2013), and the United Kingdom (Nicolini et al., 2013).

According to the research, global financial literacy was clearly less than deemed adequate to access the sophisticated financial markets around the world. The global economy has been impacted by the low levels of financial literacy. Lusardi and Mitchell (2011b) stressed, "financial illiteracy undermines not only individual retirement security but indeed, the stability of the global financial system more generally" (p. 14). They suggested, "boosting financial literacy skills may well be critically important for economic

and social welfare not only of this generation, but those to come" (p.14). Corroborating Lusardi and Mitchell's (2011b) research, the OECD working paper prepared by Atkinson and Messy (2012) highlighted the significance of the need to improve financial knowledge. Lack of such knowledge existed among a substantial portion of the population in all countries evaluated by the OECD. However, the United States population financial literacy was not included in the OECD study. The financial knowledge questions used in the OECD study cover the same topics used to determine financial literacy in the United States (Allgood & Walstad, 2013; Atkinson & Messy, 2012; Bumcrot et al., 2013; Lusardi, 2011, 2012a, 2012b; Scheresberg, 2013). While the United States financial literacy was included in some global research (Lusardi & Mitchell, 2011a, 2011b), the United States has been singled out as the prime focus in numerous research studies (Allgood & Walstad, 2012; Bumcrot et al., 2013; Buckland, 2011; Lusardi & Mitchell, 2011b, 2014; Lusardi & Scheresberg, 2013; Mandell & Klein, 2009; Mottola, 2013).

United States financial literacy. Financial literacy scores in the United States were comparable to global financial literacy scores (Lusardi & Mitchell, 2011a, 2011b; Nicolini et al., 2013). Lusardi and Mitchell (2011b) "observed low levels of financial literacy in the USA are prevalent elsewhere [Germany, the Netherlands, Sweden, Italy, Japan, and New Zealand]" (p. 503). Various recurring surveys have been conducted in the United States. These surveys included identical or similar questions to assess financial literacy (Allgood & Walstad, 2013; Bumcrot et al, 2013; Lachance, 2014; Lusardi & Mitchell, 2007a, 2007b, 2007c, 2008, 2011a, 2011b, 2013; Mottola, 2013; Nicolini et al., 2013; Robb & Woodyard, 2011; Scheresberg, 2013; Woodyard & Robb,

2012). The primary reoccurring surveys are Jump\$tart Coalition; Health, and Retirement Study (HRS); RAND American Life Panel (ALP); and National Financial Capability Study (NFCS).

These repetitive surveys concentrated on different aged participants. The Jump\$tart Coalition surveys were administered biennially to high school seniors and college students dated back to 1998 (Mandell & Klein, 2007). HRS queried adults over the age of 50 years in 2002, 2004, and 2006 (Lusardi & Mitchell, 2007a, 2008; Schmeiser & Seligman, 2013). The ALP study conducted in 2007 included participants age 18 years and over (Lusardi & Mitchell, 2007b). This survey included a group not covered by the Jump\$tart or HRS surveys, adults between the ages of 25 and 50 years of age. In 2009, FINRA Investor Education Foundation funded the NFCS online to over 25,000 adults age 18 years and older (FINRA Investor Education Foundation, 2016). NFCS was repeated with new participants in 2012 (FINRA Investor Education Foundation, 2016). Responses to the financial literacy questions from these four surveys (Jump\$tart, HRS, ALP, and NFCS) have been the primary source of assessing and evaluating financial literacy in the United States.

Jump\$tart Coalition. The Jump\$tart Coalition survey included 31 questions specifically directed towards financial literacy plus additional demographic questions (Mandell, 2008; Mandell & Klein, 2007). The baseline survey was administered to 1,532 high school seniors during the school year 1997-98. In the first survey, only 10.2% of the respondents answered at least 75% of the questions correctly with an average score of 57.3%. The survey was repeated in 2000, 2002, 2004, 2006, and 2008 with average scores of 51.9%, 50.2%, 52.3%, 52.4%, and 58.3% respectively (Mandell,

2006, 2008). A multitude of empirical articles referencing these results provided the evidence for low financial levels among teens and young adults in the United States (Alhenawi & Elkhal, 2013; Huston, 2010; Knoll & Houts, 2012; Lusardi & Mitchell, 2007a, 2007b, 2007c, 2008, 2011a, 2011b, 2013; Mandell, 2006, 2008; Mandell & Klein, 2007, 2009; Miller et al., 2014; Wolfe-Hayes, 2010). The focus of the Jump\$tart Coalition for Personal Financial Literacy, a non-profit organization based in Washington DC, is to prepare the nation's youth for life-long financial success through education (Jump\$tart Coalition, 2016). Mandell (2006, 2008) and Mandell and Klein's (2007, 2009) Jump\$tart research only addresses financial literacy among high school seniors and college students (young adults) in the United States. Examining the financial literacy questions were added to the Health and Retirement Study (Fonseca, Mullen, Zamarro, & Zissimopoulos, 2012; Lusardi & Mitchell, 2007a, 2007b, 2008, 2011b; Schmeiser & Seligman, 2013).

Health and Retirement Study (HRS). HRS was a longitudinal survey first administered in 1992 by the University of Michigan. The original sample included individuals and their spouses born from 1931 to 1941. Subjects were re-interviewed every two years. New subjects in the early 50's age range were added every six years. The HRS survey collected detailed demographic, asset, health, healthcare, housing, income, and employment data. In 2002, a three-item financial literacy quiz focusing on numeracy and risk assessment was added to the survey. The financial literacy questions continued in the 2004 and 2006 surveys (Lusardi & Mitchell, 2007a, 2007b, 2011b; Schmeiser & Seligman, 2013)

Lusardi and Mitchell (2007a) used the 2004 and 2006 survey results to compare net worth, planning for retirement, and financial literacy, in Baby Boomer Retirement Security: The Roles of Planning, Financial Literacy and Housing Wealth. The findings of this report indicated respondents who planned for retirement had higher levels of wealth at retirement. Lusardi and Mitchell (2008) based findings and conclusions in Planning and Financial Literacy: How Do Women Fare? on the 2004 HRS to posit financial literacy and retirement planning of women was of particular interest. The resulting data analysis showed women's financial literacy below that of men (Lusardi & Mitchell, 2008). Schmeiser and Seligman (2013) used the 2002, 2004, and 2006 HRS data and regression analysis to relate financial literacy scores to financial well-being and financial capability. Several other researchers have used the HRS data or referenced results of data analysis in numerous research articles (Agnew et al., 2013; Bumcrot et al., 2013; Buckland, 2011; Finke & Huston, 2014; Lusardi & Beeler, 2007; Lusardi & Mitchell, 2013; Van Rooij et al., 2009). While Jump\$tart assessed young adults and HRS addressed older adults' financial literacy, the gap between Jump\$tart and HRS was included in the RAND American Life Panel (ALP).

RAND American Life Panel (ALP). The University of Michigan Research Center conducted the original ALP survey. Three financial literacy and planning questions developed by Lusardi and Mitchell (2007b) in collaboration with van Soest (Professor in Econometrics at Tilburg University, the Netherlands) were added to ALP in 2007 (Lusardi & Mitchell, 2007b). Lusardi and Mitchell's (2007b) findings from the ALP were consistent with the results of prior analysis by Lusardi and Mitchell using the HRS. Fonseca et al. (2012) used ALP to explain the gender gap in financial literacy. Focusing
on couples' roles in marriage and decision-making, Fonseca et al. (2012) revealed, "men and women have different production processes for financial literacy" (p. 105). The study linked gender differences to household specialization with men specializing in making household financial decision. To support this specialization, men must acquire financial knowledge while women concentrated on other household functions, which included day-to-day financial responsibilities (Fonseca et al., 2012).

HRS and ALP were existing surveys. Three financial questions were added to both surveys to include a financial literacy assessment section (Fonseca et al., 2012; Lusardi & Mitchell, 2007a, 2007b, 2008, 2011b; Mandell, 2006, 2008; Miller et al., 2014; Schmeiser & Seligman, 2013; Wolfe-Hayes, 2010). NFCS was commissioned as the first national study of the financial literacy of American adults, ages 18 years and older (FINRA, 2016).

National Financial Capability Study (NFCS). Applied Research and Consulting LLC originally administered the NFCS State-by-State survey as an online survey between May and July 2009. Applied Research and Consulting LLC was founded in 1955 to conduct research in the form of public opinion polling, market research, and data analysis for government based organizations and businesses (Applied Research & Consulting LLC, 2016). FINRA Investor Education Foundation funded the survey (FINRA Investor Education Foundation, 2016). A team, led by Lusardi of Dartmouth College, Applied Research and Consulting LLC, the FINRA Investor Education Foundation, and the Office of Financial Education of the U.S. Treasury with input from Copeland of the Employee Benefit Research Institute, the American Institute of Certified Public Accountants, and Willis (a professor at the

University of Michigan), designed the NFCS survey (FINRA Investor Education Foundation, 2009). Over 25,500 U.S. adults 18 years and older were queried (approximately 500 respondents per state, plus the District of Columbia). The state-bystate survey was conducted again in 2012 with a new group of participants (FINRA Investor Education Foundation, 2009, 2013). The data collected from both surveys, weighted for U.S. Census demographics, are available for public use participants (FINRA Investor Education Foundation, 2009, 2013). The NFCS was the source for various research articles covering financial literacy in the United States. There were five financial literacy questions used in the NFCS. Two new questions were added to the three questions originally created for HRS and ALP (Allgood & Walstad, 2013; Lachance, 2014; Lusardi & Mitchell, 2011a, 2011b; Lusardi & Scheresberg, 2013; Mottola, 2013; Nicolini et al., 2013; Robb & Woodyard, 2011; Scheresberg, 2013; Woodard & Robb, 2012). Responses to other NFCS questions covered demographics, financial behaviors, and financial well-being. Researchers used NCFS data to develop statistics and evaluate the condition of financial literacy in the United States (Allgood & Walstad, 2013; Lachance, 2014; Lusardi & Mitchell, 2011a, 2011b; Lusardi & Scheresberg, 2013; Mottola, 2013; National Financial Capability Study [NFCS], 2016a, 2016b; Nicolini et al., 2013; Robb & Woodyard, 2011; Scheresberg, 2013; Woodard & Robb, 2012). Demographic information in the survey enabled researchers to define specific focal points such as age, gender, ethnicity, socio-economic status, and geography. For example, Lusardi (2011) used NFCS data in American's Financial Capability, a National Bureau of Economic Research (NBER) working paper, to paint a troubling picture of American's lack of basic financial literacy.

Robb and Woodyard (2011) examined the 2009 NFCS data concentrating on a composite measure of financial literacy and its impact on personal financial wellness, including financial satisfaction, financial behavior, financial attitudes, and objective measures with financial knowledge being considered an attribute of the overall wellness. The results from this view of the data supported previous literature on financial literacy suggesting the level of financial knowledge impacts financial behavior and financial confidence (Robb & Woodyard, 2011).

Woodyard and Robb (2012) utilized the 2009 NFCS data again to explore gender differences with respect to effects on objective (tested) and subjective (self-perceived) knowledge in terms of self-reported actions. The Woodyard and Robb (2012) research centered on six behaviors: emergency fund savings, credit report, overdraft activity, credit card payoff, retirement account savings, and risk management. Woodyard and Robb's (2012) findings reinforced previous work by Lusardi and Mitchell (2008) using the HRS study identifying women's financial literacy levels lower than men.

Lusardi (2012a) looked at financial decision-making in older adults based on 2009 NFCS data. The findings indicated a worrisome lack of financial literacy in this demographic with severely low levels among women.

Allgood and Walstad (2013) echoed findings that overall literacy of young adults is low. Additionally, Allgood and Walstad (2013) found women and minorities also rank below the mean. In *Geography of Financial Literacy*, Bumcrot et al. (2013) discovered considerable variations within the United States geographically. Again, overall financial literacy among Americans was low. Their research found "the mean financial literacy index values [on a scale of 0 to 5] ranges from 2.75 in Louisiana to 3.30 in New

Hampshire" (p. 7). The lowest financial literacy scores in the U.S. were in the eastern and southern parts of the country.

In the NBER working paper, *Financial Literacy and High-Cost Borrowing in the United States,* Lusardi and Scheresberg (2013), using NFCS data, found "that numeracy and knowledge of basic financial concepts is strongly negatively correlated with high-cost borrowing" professing "financial literacy is important in explaining financial behavior" (p. 3). Mottola (2013), *In Our Best Interest: Women, Financial Literacy, and Credit Card Behavior,* concentrated on women's financial literacy. Using the 2009 NFCS data, Mottola (2013) found women's credit card behaviors were costlier than men, concluding, "that improving characteristics that are mutable-like financial literacy and math skills may represent the most effective and efficient means of eliminating credit card behavior differences between sexes" (p. 13).

Nicolini et al. (2013) included the United States in *Financial Literacy: A Comparative Study Across Four Countries* (United Kingdom, Italy, Canada, and the United States). The United States data were taken from the 2009 NFCS, while the 2006 British Financial Services Authority Financial Capability Baseline Survey was used for the United Kingdom. The 2009 Canadian Financial Capability Survey conducted by Statistics Canada provided Canadian data, and the 2009 Italian Survey on Household Income and Wealth data were used to represent Italy's financial literacy. The results suggested variations in financial literacy levels across countries with inconsistencies in socio-demographic variable effects indicating standardized financial literacy assessments across countries asking identical questions would better facilitate this type of comparison (Nicolini et al., 2013). Scheresberg (2013) only used data from

participants aged 25 to 34 years and found 34% of young adults were ill-equipped to deal with financial responsibilities. The findings also indicated significant differences between women's and men's scores. Women scored very low despite higher levels of education (Scheresberg, 2013).

Lachance (2014) used both 2009 and 2012 NFCS with regression analysis to evaluate financial literacy scores, education, and social networks (neighborhoods). The findings indicate informal mechanisms such as knowledgeable social networks played a role in financial literacy. The impact of financial knowledge development through social networks or neighborhoods was apparent in socioeconomic groups. The lack of natural exposure to informal financial knowledge due to less-educated neighborhoods with lower proportions of college-educated members was evident in financial literacy scores (Lachance, 2014). Lusardi and Mitchell (2014) in *The Economic Importance of Financial Literacy: Theory and Evidence* combined the 2009 and 2012 NFCS results with five other financial literacy assessments to demonstrate the prevalence of inadequate financial literacy around the world. They suggested challenges existed in providing educational tools necessary to improve financial literacy to the level essential for effectively navigating today's complex financial and credit markets.

Lusardi (2011, 2012a), Robb and Woodyard (2011), Woodyard and Robb (2012), Allgood and Walstad (2013), Mottola (2013), Nicolini et al. (2013), Scheresberg (2013), Lachance (2014), and Lusardi and Mitchell (2014) all conducted research using data from the NFCS. Each examined the data from a different perspective with diverse emphasis. Despite evaluating NFCS data from differing viewpoints, all concluded financial literacy in the United States is low. Researchers considering gender

discovered women score below men. Those pondering financial literacy's correlation to age found the young and older age groups scored lower than the mean. There are additional sources of financial literacy assessment research beyond the reoccurring survey administered by Jump\$tart, HRS, ALP, and NFCS.

Other financial literacy studies. Other studies exploring U.S. financial literacy were published using instruments different than the reoccurring survey's assessments. However, these studies concluded similar findings. Volpe, Chen, and Pavliko (1996) used the *What's your financial IQ?, a* 10-item questionnaire from the Money Forecast Issue of 1993 *Money* magazine as an assessment to test age 18 to 35 year old adults' investment knowledge. The results from 454 college students at a mid-sized metropolitan university in the U.S. showed an average financial IQ score of 44 out of 100, below the acceptable range of 70-90 (Volpe et al., 1996). Volpe et al. (1996) also noted male participants scored better than females.

Chen and Volpe (1998) examined responses to a 31-question multiple-choice survey provided by 924 college students from multiple universities across the United States. Reaffirming earlier evidence that young adults have low financial literacy, Chen and Volpe (1998) indicated, "college students need to improve their knowledge of personal finance" (p. 122).

Hilgert, Hogarth, and Beverly (2003) used the University of Michigan's monthly Survey of Consumers Finances (SCF) to administer 28 financial literacy questions in November and December of 2001. The research explored connections between financial knowledge and recommended behaviors. The survey covered cash-flow management, general credit management, savings, investment, mortgages, and

general financial-management. The research found consumers responded correctly to two-thirds (67%) of the questions, with mortgage knowledge responses having the highest scores (80%). Statistical significance was noted between specific financial knowledge scores and financial behaviors. Hilgert et al. (2003) found knowledge and experience can lead to improved financial practices, but cautioned causality could not be determined, suggesting an improvement in financial practices could flow from knowledge or vice versa.

Lusardi and Mitchell (2011b) included the basic financial literacy questions they created for HRS and APL in the 2008 National Longitudinal Survey of Youth. The National Longitudinal Survey of Youth research found young adults, age 23-28 years, were ill-equipped to make financial decisions due to severely lacking knowledge in financial literacy.

The Cognitive Economics survey (Delavande, Rohwedder, & Willis, 2008) included 24 financial knowledge and two self-assessment questions. The National Institute of Aging administered the survey to a national sample of age 51 years and older U.S. adults through the CogUSA study (Hsu, 2011). Findings by Delavande et al. (2008) and Hsu (2011) indicated the financial condition among older U.S. adults are less about the impacts of aging and more about a lack of knowledge, in other words, low financial literacy.

Lusardi and Tufano (2009) partnered with Taylor Nelson Sofres (TNS) Global, a commercial market research firm, to develop a financial literacy survey capturing information covering debt literacy, a component of financial literacy. Three debit literacy questions and one financial knowledge self-assessment questions were included in this

survey together with questions about participants' personal assessments of debt levels and financial experiences, rather than behavior. TNS Global administered the survey to 1,000 U.S. adults. Lusardi and Tufano's (2009) NBER working paper reported, "low levels of debt literacy are the norm, and understanding of the basic mechanics of debt is especially limited among the elderly, women, certain minorities, and people with lower incomes and wealth" (p. 24).

Alhenawi and Elkhal (2013) conducted research in southern Indiana using a survey containing seven financial knowledge questions developed for their research. Once developed, the instrument gathered demographic financial knowledge, planning, and financial behavior data from 350 Midwest household financial decision makers. The overall results of the research showed surveyed households scored 75.1% indicating financial knowledge at a much higher level than most other studies. The researchers determined there was a low positive correlation between financial knowledge scores and financial planning, noting, "financially knowledgeable people could make bad financial choices and those who score high on the planning test are not always the most knowledgeable ones" (Alhenawi & Elkhal, 2013, p. 17).

The Financial Literacy Assessment Test (FLAT) was developed based on research from the Financial Literacy Project by a Texas Tech University team (Finke & Huston, 2014). The research team worked with a panel of experts who included financial planners, personal-finance experts, and five Ph.Ds. to develop the 20-item instrument (16 tested knowledge and 4 self-assessed knowledge) designed to measure knowledge and skills required to make basic financial decisions (Texas Tech University, 2015). From December 2010 through 2013, the survey was included as a module of

the Consumer Finance Monthly Survey conducted by the Center for Human Resource Research at the Ohio State University (Finke et al., 2011). Finke et al. (2011) evaluated responses by participants aged 60 years and older to "show that is not so much the imbalance between confidence and knowledge that is causing poor financial decisions, but the low financial literacy itself" (p. 19).

Research studies highlighted within this literature review focused on adults of various age groups and geographic locations in the U.S. populations. All the studies have resulted in similar findings: financial literacy in the U.S. is low. Among women, younger adults, older adults, certain ethnic groups, and low-income and/or education levels, financial literacy is below the average mean. Moreover, the research correlated low financial literacy assessment scores and financial behaviors with financial outcomes. This phenomenon came to the attention of government officials at the highest levels of the U.S. government.

President's Advisory Council on Financial Capability. President Obama reiterated findings related to the United States' low financial literacy assessment scores when he created the 2010 President's Advisory Council on Financial Capability (Obama, 2010). Hearings by the Oversight of Government Management, the Federal Workforce, and the District of Columbia Subcommittee of the Committee on Homeland Security and Governmental Affairs United States Senate in 2011 and 2012 focused on financial literacy education of the American public (PACFC, 2013b). The hearings revealed at least 20 agencies spent billions of dollars on more than 50 different programs. The objective of these programs is to elevate personal financial literacy to

achieve the goal of improving the U.S. economy through implementing the PACFC's

recommendations to:

- 1. Incorporate critical personal financial competencies into the teaching of Common Core State Standards for English and mathematics.
- 2. Encourage federal government and all employers to embrace responsibility for financial well-being of employees and share best practices in employee education.
- 3. Encourage federal government to support and build upon the state, local, and tribal financial capability councils to promote financial well-being among communities.
- 4. Establish an online clearing for research in the field of financial education, behavioral economists, and regulators. (PACFC, 2013a, p. III-IV)

The PACFC's objectives defined the U. S. National Strategy for financial literacy (FLEC, 2011a; 2011b; 2014). Although the national strategy for the U.S. financial literacy was defined at the federal level, execution of the strategy falls into the arena of state governments (FLEC, 2010, 2014; Grifoni & Messy, 2012; Hillman, 2009). Acceptance and approval of state common core standards or similar educational laws, including a financial literacy initiative, was integral to facilitate the improvement of the youth's financial literacy (Bell, 2013). Because the U.S. financial literacy national strategy falls to the state level for execution, the next section focuses on the financial literacy of in the State of Florida.

Florida financial literacy. FINRA Investor Education Foundation (2013) NFCS

State-by-State survey findings showed less than 40% of the Americans correctly answered at least four out of five financial literacy questions related to everyday life (compound interest, inflation, risk, and diversification, bond prices vs. interest rates, and mortgage/interests). Examining the NFCS results for the State of Florida, only 36% could answer four out of five questions correctly, ranking Florida in the bottom 20% of the nation (FINRA Investor Education Foundation, 2013). The Florida Council for Economic Education, created in 1975, focuses on training K-12 in economics, entrepreneurship, personal finance, and the free-enterprise system. The goal of the Florida Council for Economic Education was training Florida teachers to implement the portions of the Florida's Next Generation Sunshine State Standards that contain financial literacy (Florida Council for Economic Education, 2014). In 2006, Florida House passed Bill 825 creating the Florida Financial Literacy Council for the expressed purpose of studying the

financial problems that affected consumers, particularly small business owners, young people, working adults, and seniors who arise from a lack of basic knowledge of financial issues and to provide recommendations to the Department of Financial Services, which will assist the department in developing financial literacy programs and resources and providing a single state resource for financial literacy for the general public in order to empower individuals and businesses to manage their financial matters in order to reduce debt, increase savings, and avoid bankruptcy. (Financial Literacy Council, FL, 2006, p. 1)

The Florida Financial Literacy Council's (2008) first report dated January 2008 indicated, "There is no single answer to the problem of financial illiteracy. A comprehensive campaign utilizing media, individual, and group outreach can create a *wave effect* that will teach Floridians how to make sound financial decisions" (p. 6).

August 2013, the Florida Council for Economic Education held the Florida Financial Literacy Summit at the University of South Florida, Tampa. The summit's overall goals were to focus on the financial education of the next generation of Floridians (Florida Financial Literacy Summit, 2013). Recommendations from the summit included "Encourage programs for teacher training on current financial topics and provide access to resources for training materials" (Florida Financial Literacy Council, 2008, p. 4). Roberts, Sorgman, and Parkison (2010) identified "teachers' knowledge, their comfort with that knowledge, and application of content and curriculum in the classroom as essential elements for successful student learning outcomes" (p. 9). Concluding their research commentary, Roberts et al. (2010) noted, "professional development needed to enhance K-12 teacher readiness may not be available in the current economic environment" (p. 15). Roberts et al. (2010) emphasized the importance of making professional development and adult education available to teachers in order to create successful economic and financial literate educators for the next generation.

A limited number of Florida educational programs directed towards adult financial literacy exist. Examples of Florida financial literacy programs available to adults included the Florida Council for Economic Education's *Financial Freedom* program and the *360 Degrees of Financial Literacy* program. The *Financial Freedom* program was designed for use in classrooms from middle-school to adult education classes, and the *360 Degrees of Financial Literacy* program provided by Florida Institute of Certified Public Accountants was created for all ages and stages of life (Florida Council for Economic Education, 2014; Florida Institute of Certified Public Accountants, n.d.). Additionally, the State of Florida developed an online tutorial, *Family Foundations: Building a Strong Financial Future,* directed towards Hispanic Floridians to address adults in this community (Atwater, n.d.). Prior to this research, there was no empirical research related to any Florida adult financial literacy programs. The adults selected for this research study were small business owners. The next section explores the financial literacy of this group.

Small business owner financial literacy. Financial literacy is among several competencies necessary to be a successful small business owner. Mojab, Zaefarian,

and Azizi (2011) conceptually defined entrepreneurship competencies to include three features: characteristics, skills, and knowledge emphasizing the importance of entrepreneurial competencies as factors for growth and success. Included within the entrepreneurship knowledge competency is the construct of financial literacy.

According to Fairlie and Krashinsky (2012), small business success was linked to improvement in the economy and was a global priority. The financial literacy aptitude of small business owners was considered a necessary attribute in the ability to run a successful business (Dahmen & Rodriguez, 2014); however, the financial literacy of small business owners has not been evaluated or analyzed in the United States. Examining 14 individual clients from Small Business Development Center (SBDC) at the University of South Florida, Dahmen and Rodriguez (2014) observed the West Central Florida region small business owner's performance compared to business owners' selfperceived of financial literacy and business financial habits. This case study noted 50% of business owners were experiencing financial difficulties. Of those experiencing difficulties, 86% did not review financial statements regularly, and 100% of those not reviewing financial statements expressed a perception of their financial literacy as inadequate. No financial literacy knowledge assessments were performed to determine financial literacy knowledge scores related to the subjects of this study. As a case study, Dahmen and Rodriguez's (2014) work could not be generalized to an entire population (Gall, Gall, & Borg, 2007); however, the high percentage of small business owners struggling with financial difficulties also expressing low self-perceived financial literacy raised cause for concern. Since small business success is an important ingredient in a prosperous U.S. economy, financially literacy of small business owners is

a significant factor (Dahmen & Rodriguez, 2014; Jones & Tullous, 2002; Mojab et al., 2011; Obama, 2010, para. 2).

Jones and Tullous (2002) conducted a study of 133 pre-venture clients of a regional SBDC to explore the self-perception of the need for financial literacy (financial and accounting) consulting. This study examined client perceptions vs. consultant perceptions for Anglo and Hispanic men and women. According to Jones and Tullous (2002), the study noted female small business owners perceived themselves to be ill-prepared to take on the financial aspects of creating and maintaining a small business.

Classifying barriers for small business owners starting businesses in Indiana, Marshall (2012), noted the lack of human capital skills, including financial literacy, gained through experience and education inhibited business startups. Economists use the term human capital to describe the sum of accumulated knowledge and skills people can draw upon when making decisions and/or performing various tasks (Finke & Huston, 2014; Huston, 2010, 2012). Delavande et al. (2008) consider financial knowledge from two perspectives: (a) form of human capital and (b) financial knowledge acquisition an investment.

Small businesses account for 54% of all the United States sales (SBA, 2013). Dahmen and Rodriguez (2014), Jones and Tullous (2002), Mojab et al. (2011), and Obama (2010) consider small business owner financial literacy an important factor in small business owners' ability to succeed and impact the United States economy. As such, exploring small business owner's financial literacy and its potential impact on small business success can provide value to the U.S. Government, governmental

agencies, and private industry. Empirical statistical analysis of the financial literacy assessment scores of small business owners has not been reported.

Research findings from around the world, whether government or privately conducted and funded, concurred financial literacy of all populations was low. This determination was based on an array of commonly used measurements (Atkinson & Messy, 2012; Hung et al., 2009; Huston, 2010; Nicolini et al., 2013; Remund, 2010). To understand what this actually means and how financial literacy assessment scores were derived, the next section of this literature review explores the financial literacy measurement.

Financial Literacy Measurement

The measurement of financial literacy, although the subject of numerous surveys, studies, working papers, and research articles, has been assessed from several perspectives without a consensus of a uniform measurement tool. According to Hung et al. (2009) and Huston (2010), it is difficult to identify a uniform measurement. This stemmed from two issues: (a) lack of a consensus in defining of the construct (core concept) and (b) an agreement of the content of the construct (Hung et al., 2009; Huston, 2010). After exploring the use of the positivist theoretical framework, this section looks at financial literacy measurement and explores the various definitions of financial literacy, followed by a model of the financial literacy construct used in this study. Finally, this section looks at methods and measurement instruments used to create a financial literacy assessment score.

Theoretical framework. The positivist perspective was the underlying theoretical framework of this study. Positivism relies heavily on scientific knowledge,

and that *confidence in science* is the result of the view that science is precise and accurate (Crotty, 1998). Positivism, a product of enlightenment and the *Age of Reason*, maintains two central tenets regarding the *truth* or knowledge about reality: (a) the supposition that only verifiable statements resulting from logical and mathematical treatments and (b) reports of experience have meaning and valid knowledge is the result (Crotty, 1998; Paul, 2005). Positivism is a philosophical tradition aligned with behaviorism, which measures competencies (knowledge, skills, and behaviors) in behavioral terms (Elias & Merriam, 2005).

The collection of financial literacy research from 1996 through 2015 evaluated in this literature review, used the scientific perspective prescribed by positivism. The research portrayed the positivist viewpoint of objectivity when assessing financial literacy around the world. Examples of the positivist perspective in current financial literacy research include: Atkinson and Messy (2012) who stressed analysis when reporting Organization of Economic Co-operation and Development (OECD)/International Network of Financial Education findings stating, "from analysis of data from each of these [14] countries focusing particularly on levels of financial literacy" (p. 6). Several researchers concentrated on the logical and mathematical or analytical nature of research in financial literacy. Knoll and Houts (2012) used modern psychometric techniques to analyze items from three national surveys. Research conducted by Lusardi and several associates simplified financial literacy measurement to a few questions to provide statistical interpretations and results (Lusardi, 2011, 2012a, 2012b; Lusardi & Beeler, 2007; Lusardi & Mitchell 2007a, 2007b, 2008, 2011a, 2011b, 2013, 2014; Lusardi & Scheresberg, 2013; Lusardi & Wallace, 2013). The

purpose of Huston's (2010) work was to "develop a more standardized measure of financial literacy" (p. 297). Nicolini et al. (2013) echoed the idea of positivist analysis stating,

most previous studies analyzed data about financial literacy from a single country or a subset of the population of a country. Additionally, the literature suggested the value of assessing financial knowledge lies in examining its relationship with financial behaviors. (p. 691)

Nicolini et al. (2013) brought the idea of behaviorism to financial literacy research. Finke and Huston (2014) also identified with behaviorism when asserting "most of the research focuses on the impact of financial literacy on behavior or financial outcomes rather than the construct itself" (p. 66). Hung et al. (2009) made the connection between financial literacy research and Behaviorism when they contended "few studies have been able to construct a sophisticated measure of financial literacy and definitively establish causal links between financial education, literacy and behavior in the U.S." (p. 1). Robb and Woodyard (2011) analyzed "a composite measure of financial literacy" (p. 62) associated with behavior to understand the relationship. Assessing behaviors as an indication of financial literacy and suggesting a change in financial literacy will impact behavior was the focal point in much of the financial literacy research (Lusardi, 2011, 2012a, 2012b; Lusardi & Beeler, 2007; Lusardi & Mitchell 2007a, 2007b, 2008, 2011a, 2011b, 2013, 2014; Lusardi & Scheresberg, 2013; Lusardi & Wallace, 2013, Mottola, 2013; Nicolini et al., 2013; Robb & Woodyard, 2011; Scheresberg, 2013; Schmeiser & Seligman, 2013; Woodyard & Robb, 2012). Following the precedent set by these researchers, the basis for the present research was the positivist perspective.

Definitions of financial literacy. In discussing financial literacy, several terms were used by researchers to describe similar concepts. These terms included:

financial awareness, financial knowledge, financial skills, and financial capability. Three scholarly works in 2009 and 2010 focused on the definition, or lack of definition, of financial literacy. All three, *Defining and Measuring Financial Literacy* (Hung et al., 2009); *Measuring Financial Literacy* (Huston, 2010); and *Financial Literacy Explicated: The Case for a Clearer Definition in an Increasingly Complex Economy* (Remund, 2010), individually or in combination, were cited in the majority of research when defining financial literacy. Table 1 shows, by author(s), the number of times each article has been cited in academic journal articles.

Table 1

Number of Times That Financial Literacy Definition was Cited in Journal Articles by Author(s) Based in Google Scholar Citations

Researchers	Research Article	Cited
		n
Hung et al. (2009)	Defining and Measuring Financial Literacy	158
Huston (2010)	Measuring Financial Literacy	336
Remund (2010)	Financial Literacy Explicated: T Case for a Clearer Definition in a Increasing Complex Economy	

Financial knowledge. Hung et al. (2009) prepared a Labor and Population working paper for the RAND Corporation providing "a review of theoretical and operational approaches to financial literacy, as well as a composite definition" (p. 22). Exploring the breadth of conceptual definitions, the Hung et al. (2009) research found knowledge (or understanding) was the most common basis for defining financial

literacy. According to Hung et al. (2009), definitions "focus primarily on the actual financial knowledge, rather than on skills, behavior or perceived knowledge" (p. 21), and proposed a composite definition of financial literacy: "Financial Literacy: knowledge of basic economic and financial concepts, as well as the ability to use that knowledge and other financial skills to manage financial resources effectively for a lifetime of financial wellbeing" (p. 20).

Huston (2010) examined 71 studies between 1996 and 2008 in a literature content analysis to "propose an approach, to develop a more standardized measure of financial literacy" (p. 297). While 72% of the studies did not include a definition of financial literacy, Huston (2010) identified eight definitions within the group of studies. Six of the eight definitions included knowledge, or understanding, and five referred to the ability to use knowledge or other critical thinking such as to make informed judgments, read, analyze, manage, and communicate. According to Huston (2010), the terms *financial literacy* and *financial knowledge* were used synonymously" (p. 303) in 47% of the studies examined. Huston (2010) indicated a financially literate person "has the knowledge and the ability to apply the knowledge" (p. 308).

Remund (2010) analyzed more than 100 resources evaluating the ways financial literacy has been interpreted and measured. Referring back to 2000, Remund (2010) identified "five categories: (a) knowledge of financial concepts, (b) ability to communicate about financial concepts, (c) aptitude in managing personal finances, (d) skills in making appropriate financial decisions, and (e) confidence in planning effectively for future financial needs" (p. 279). According to Remund (2010), the

common attribute of the financial literacy definitions is knowledge. Remund

consolidates the many definitions of financial literacy as:

A measure of the degree to which one understands key financial concepts and possesses the ability and confidence to manage personal finances through appropriate, short-term decision-making, and sound, long-range financial planning, while mindful of life events and changing economic conditions. (p. 284)

Remund (2010) went on to refer to financial capability as "the next possible iteration of

the financial literacy concept" posing the argument "knowledge is of little value without

ability or skill" (p. 283). Other researchers expanded on this idea and weighed in on the

relationship between financial literacy and financial capability.

Financial capability. When defining financial capability, Sherraden (2010)

referred to scholars and practitioners in the United Kingdom and Canadian. The first to

use the term financial capability, the United Kingdom's government adopted the

definition of financial capability as:

A broad concept, encompassing people's knowledge and skills to understand their own financial circumstances, along with the motivation to take action. Financially capable consumers plan ahead, find and use information, know when to seek advice and can understand and act on this advice, leading to greater participation in the financial services market. (HM Treasury, 2007, p. 19)

Sherraden asserted financial literacy to be a necessary component of financial capability with capability representing the ability (knowledge, skills, confidence, and motivation) and opportunity to act (access to institutions and beneficial financial products). Barr (2004), Cover, Spring, and Kleit (2011), and Finke and Huston (2014), as well as numerous other researchers, addressed financial capability in terms of individuals' abilities based on access. That is to say, an individual having financial knowledge, but no wealth to invest could have a high financial literacy assessment score, but no financial capability. According to Cover et al. (2011), *Minorities on the*

Margins? The Spatial Organization of Fringe Banking Services, the U.S. financial service sector functions on a two-tiered system: traditional banking and alternative banking services. Traditional banking services the middle- and upper-income households, while alternative banking services providers (payday lenders, check cashers, pawn brokers, and alike) accommodate low- and moderate-income households. There is a financial and physical divide between traditional and alternative banking service households with the traditional being more likely to be financially capable, while the alternative banking services households struggle for opportunities to become capable (Barr, 2004; Cover et al., 2011).

To Lachance (2014), financial capability represented a broader set of financial outcomes, citing the PACFC Executive Order 13520 (Executive Order 13520, 2010) to define financial capability:

Financial capability is the capacity, based on knowledge, skills, and access to manage financial resources effectively. In order to develop this capability, individuals must have appropriate access to and understanding of financial products, services and concepts. Financial capability empowers individuals to make informed choices, avoid pitfalls, know where to go for help, and take other actions to improve their present and long-term financial well-being. (Executive Order 13520, 2010, p. 1)

As a member of the group involved the with U.S. national financial literacy strategy development and implementation, the Financial Literacy and Education Commission (FLEC) felt consumers, and financial education providers needed defined competencies, stating, "the financial education field lacks a common understanding of what we [the FLEC] collectively are trying to achieve" (Financial Education Core Competencies, 2010, p. 52596). The lack of consistency or agreement in the definition of financial literacy was the grounds for this FLEC position. The concepts defined by the FLEC to represent the core competencies are earnings, spending, borrowing, saving and investing, and protecting. These competencies represent knowledge content areas of financial literacy. The next section contrasts definitions that have been used in the literature.

Divergent definitions of financial literacy. Financial literacy has been defined in a multitude of inconsistent ways. Nicolini et al. (2013) conducted a meta-analysis comparative study across four countries concurring with Hung et al. (2009), Huston (2010), and Remund (2010) that the definitions by numerous scholars are inconsistent. Prior to the prominent articles by Hung et al. (2009), Huston (2010), and Remund (2010), Noctor, Stoney, and Stradling research for the National Foundation for Education Research (as cited in Atkins, McKay, Collard, & Kempson, 2010) defined financial literacy as "the ability to make informed judgments and to make effective decisions regarding the use and management of money" (p. 29). The 2012 OECD Pilot Study of 14 countries defined a financially literate person as one considered to "have some basic knowledge of key financial concepts and the ability to apply numeracy skills in financial situations" (Atkinson & Messy, 2012, p. 16). Behavioral economics' standard definition is "having the knowledge, skills and confidence to make responsible financial decisions" (Altman, 2012, p. 677). Lusardi (2012b), and Lusardi and Mitchell (2007a. 2007b, 2007c, 2008, 2011a, 2011b, 2013, 2014) emphasized planning for retirement when defining financial literacy as knowledge of basic concepts to make savings and investment decisions. Lusardi and Tufano (2009) and Mottola (2013) expanded the definition to include the basic knowledge of compound interest in conjunction with making simple decisions regarding debt contracts (debt literacy).

Understanding financial market principles, instruments, organizations, and regulations or familiarity with basic economic principles, knowledge about U.S. economy, and key economic terms were considered financial literacy and tested in selected studies (FINRA Investment Education Foundation, 2013; National Council on Economic Education, 2005, 2014).

Mandell (2006, 2008) defined financial literacy for the Jump\$tart program as "the ability to use knowledge and skills to manage financial resources effectively for a lifetime financial security" (Jump\$tart Coalition, 2007, p. 1). Mandell's (2006, 2008) definition was cited by Knolls and Houts (2012); Nicolini et al. (2013); and PACFL (2009) in addition to other researchers. ANZ Bank (2008, 2011) used a simpler definition: the use and management of money.

With the lack of an empirical unified consensus on the conceptual or operational definition of financial literacy, several attributes were applied to financial literacy (Huston, 2010; Kunovskaya et al., 2014; Nicolini et al., 2013). According to Fernandes et al. (2014), conceptually financial literacy refers to a skill. Knolls and Houts (2012) explained financial literacy as possessing two attributes: financial knowledge and financial skills. Hung et al. (2009) and Huston (2010) suggested financial literacy is made up of knowledge (financial concepts) and ability (application). Remund (2010) summarized financial literacy as knowledge (of financial concepts), ability (to communicate about financial concepts), aptitude (to manage personal finances), skill (in making appropriate financial decisions), and confidence (in planning for financial future). Economic behaviorists contended behavior is also an attribute of financial literacy (Altman, 2012; Schuchardt, Hanna, Hira, Lyons, Palmer, & Xiao, 2009). Many

definitions imply positive outcome of financial decisions based on knowledge, such as financial well-being or lifetime financial security (Hung et al., 2009; Knolls & Houts, 2012; Nicolini et al., 2013; PACFL, 2009).

The cornucopia of definitions of financial literacy includes many attributes: knowledge, skills, ability, aptitude, attitude, confidence, behavior, positive outcomes, financial well-being, and financial security. Finke and Huston (2014) suggested, "definitions of financial literacy that include behavior and/or financial outcomes fall outside the purview of human capital and are not as useful from a research perspective" (p. 67). Huston (2012) explained financial literacy as a specific knowledge and set of skills representing a form of human capital. Finke and Huston (2014) characterized financial decision-making, financial behavior, financial outcome/well-being, and financial education as a distinct and different construct than personal finance. Finke and Huston (2014) also noted that if "the goal of research is to understand how human capital specific to personal finance is related to financial behavior and outcome, then the financial literacy concept needs to be limited to the knowledge and skill level of individuals" (p. 67).

Some researchers agreed financial literacy is a form of quantitative literacy, often called numeracy (Gilliland, Melfi, Sikorskii, Corcoran, & Melfi, 2011). Research supports the link between financial literacy and quantitative literacy. Examination of financial literacy assessment questions confirmed quantitative literacy as an underlying component based on the frequent requirement of mathematical calculation (Agnew et al., 2013). Quantitative reasoning entails understanding why a selected course of action is justified, while quantitative literacy is the ability to read and interpret

graphs and numerical calculations (Lusardi & Wallace, 2013). Numeracy, quantitative literacy, and quantitative reasoning are interrelated with financial literacy (Scheresberg, 2013). Lusardi and Mitchell (2013) considered numeracy a fundamental concept at the root of financial literacy related decisions. With this in mind, they developed five questions, also known as the *Big Five*, to assess financial literacy, numeracy, and capability to include calculations related to interest rates (Hastings, Madrian, & Skimmyhorn, 2012; Lachance, 2014; Lusardi & Mitchell, 2013). Researchers agreed numeracy, quantitative literacy, and quantitative reasoning cast a wider net of abilities than financial literacy (Agnew et al., 2013; Gilliland et al., 2011; Lusardi & Mitchell, 2013; Scheresberg, 2013). This research considered financial literacy a subset of numeracy, quantitative literacy, and quantitative reasoning and focused on the knowledge aspect of financial literacy alone.

Decomposing definition of financial literacy. With numerous interpretations of financial literacy by various researchers and scholars, breaking the phrase *financial literacy* into its components, the words *financial* and *literacy*, may shed light on the meaning. Shuttleworth (2009) indicated each word separately represents a myriad of issues that together easily lose their relevance. Defining financial, Shuttleworth (2009) used the words: economic, business, commercial, monetary, fiscal, and pecuniary. Mckenzie (2009) traced the origins of the use of literacy from 1432 Latin word *litterae* meaning letters or literature to the 1886 word *literate* and finally to the present-day meaning that is often defined as "the state of being educated, instructed or learned" (Mckenzie, 2009, p. 27). Simply stated, literacy is competence, in particular, type of

knowledge within a particular subject. Mckenzie (2009) also noted synonyms of literate, knowledge, and competence are proficiency, resourcefulness, and skilled.

Examining the conceptual definition of literacy, Remund (2010) looked outside the arena of financial services to the familiar use of literacy relates to reading. Inspecting the definition of literacy in the context of learning to manage money, Remund (2010) proposed the higher-order competencies (e.g., ability to calculate numbers and understand basic economic concepts). Remund (2010) claimed competencies of financial literacy are more complicated to achieve than simply to read, write, and speak. The obvious recurring theme throughout researchers' financial literacy definitions is knowledge or understanding (Hung et al., 2009; Huston, 2010; Remund, 2009). Summarizing the individual definitions of each word in *financial literacy* to derive an operational definition, financial literacy is knowledge and understanding of information related to money, enabling an individual the ability or skill to make effective decisions about money matters or situations.

Throughout the literature, the terms *financial literacy* and *financial capability* were used interchangeably (Kunovskaya et al., 2014; PACFC, 2013a; HM Treasury, 2007). Lachance (2014) and PACFC (2013a) defined financial capability as focused on a broader set of outcomes than financial literacy. They advocated financial capability addresses the contextual, cultural, and environmental aspects of the individuals' interactions with money (HM Treasury, 2007; Lachance, 2014; PACFC, 2013a). Reflecting on researchers' interpretation of financial literacy vs. financial capability, the two are not interchangeable. Furthermore, financial literacy is a necessary attribute to reach financial capability. But, financial literacy does not ensure financial capability.

Hung et al. (2009), Huston (2010), Lee and Hanna (2014), Mason and Watson (2000), Sheraden (2010), and Shuttleworth (2009) moved beyond defining financial literacy to develop models showing the attributes of financial literacy. The attributes of all of these researchers' models provided the foundation to conceptualize the model for this research. Next, this research's financial literacy model is defined.

Model of financial literacy. The financial literacy model for this study was based on synthesizing the research of Hung et al. (2009), Huston (2010), Lee and Hanna (2014), Mason and Watson (2000), Sheraden (2010), and Shuttleworth (2009). These researchers portrayed financial literacy in six different conceptual models. Each model provided a visual exploration of the attributes of financial literacy. The attributes of these models can be divided into three domains: cognitive, affective, and social.

According to Kolb (1984), "learning is the major process of human adaptation" (p. 32). Kolb (1984) viewed the learning process from an experiential perspective. He emphasized adaptation rather than content or outcomes. Kolb (1984) believed knowledge was continuously created and recreated in a transformative process. He also thought learning transformed both objective and subjective knowledge during the experience. Drawing on the works of Dewey, Lewin, Piaget, James, Jung, Freire, Roger, and others, Kolb and Kolb (2005), determined experiential learning is a holistic process of adaptation to the world that involves more than the cognition, also including feeling, perceiving, and behaving. Defining learning from this perspective sets the foundation for the cognitive, affective, and social dimensions of acquiring knowledge (James & Blank, 1993; Kolb & Kolb, 2005). Applying this to financial literacy knowledge, cognitive includes tested (objective) knowledge, skills (using financial

information), and self-assessed (subjective) knowledge. Affective is an individual's attitude and behavior related to money and financial activities. Social includes experiences with, and socialization around, money such as financial activities, inclusion, resources, and access to financial knowledge, information, and services. Table 2 summarizes the three domains by researcher(s) and journal article, report, or dissertation.

A useful working model providing a foundation for measuring financial literacy requires the ability to assess the components. Adapting and synthesizing all of the models generated by other researchers into one simplified perspective facilitates linking an assessment to a model component. The model developed for this research starts with the attribute domains: cognitive, affective, and social. These three domains exist within an individual to form their total financial literacy. Financial products, information, resources, and inclusion exist outside an individual and are necessary for financial capabilities.

The Financial Literacy/Capability model (Figure 1) considers financial literacy an integral part of financial capabilities. In order for financial capacity to exist, financial information and products must be available together with financial resources and inclusion in cooperation with access to financial markets and products. As noted by Lachance (2014) and Sherraden (2010), financial capability is a broader concept that takes in financial plaines outside of the individual. Economic circumstances and inclusion in the financial community are factors contributing to the individual's financial capability. Circumstances and community determine access to financial products (i.e., checking accounts, credit cards, loans, mortgages, retirement savings, annuities, IRAs,

Table 2

Financial Literacy Attributes by Domain, Researcher(s), and Journal Article, Report or Dissertation

		Cognitive		Affective		Social				
Researcher(s)	Article/Report/Dissertation	Knowledge	Skills/ Application/ Using Financial Information	Perceived Knowledge	Behavior	Attitude	Experiences	Socialization	Inclusion	Resources
	•									
Mason and Watson (2000)	Conceptualizing Financial Literacy	Х	Х							Х
Hung et al. (2009)	Defining and Measuring Financial Literacy	х	Х	Х	Х					
Shuttleworth (2009)	Towards a financial literacy model as a coordinating interface between information and decision makers	Х	Х				Х			
Huston (2010)	Measuring Financial Literacy	Х	Х		Х	Х	Х	Х		
Sherraden (2010)	Financial Capability: What is It, and How Can It be Created?	Х	х					Х	х	
Lee and Hanna (2014)	Gender Differences and Asian College Students' Financial Knowledge Pathways	х			Х	Х	х			



Figure 1. Financial Literacy/Capability model summarizing the financial literacy attributes relationships within financial capabilities

401Ks, stocks, and mutual funds) and related information (Barr, 2004; Cover et. al., 2011; Finke & Huston, 2014).

Shuttleworth (2009) saw financial literacy as a conduit for this information. Inspecting the financial literacy aspect of the Financial Literacy/Capability model, the three attribute domains (cognitive, affective, and social) are clear. Knowledge (cognitive) is the common attribute present in all the models and definitions (Carlson et al., 2009; Fernandes et al., 2014; Hsu, 2011; Hung et al., 2009; Hung et al., 2012; Huston, 2010; Lee & Hanna, 2014; Mason & Watson, 2000; Remund, 2010; Sherraden, 2010; Shuttleworth, 2009). Except Lee and Hanna (2014), all models also included skills. Financial knowledge and skills are the primary attribute of the cognitive domain of financial literacy. Perceived knowledge is the secondary attribute of cognitive. Only Hung et al. (2009) included perceived knowledge in their model. However, numerous researchers have evaluated perceived knowledge comparing it to tested knowledge or financial behaviors and financial well-being. Correlations between perceived knowledge and tested knowledge indicate perceived knowledge is a significant factor in the analysis of financial literacy (Allgood & Walstad, 2012, 2016; Carlson et al., 2009; Collins, 2013; Fernandes et al., 2014; Finke & Huston, 2014; Huston, 2012; Lee,. & Hanna, 2014; Miller et al., 2014; Nicolini et al., 2013; Shim, Serido, Bosch, & Tang, 2012; Robb & Woodyard, 2011; Tokar-Asaad, 2015; Van Rooij et al., 2011).

Looking at the affective domain, the models developed by Hung et al. (2009), Huston (2010), and Lee and Hanna (2014) included attitudes and/or behaviors. The Hung et al. (2009) model showed an interaction between financial knowledge and financial behavior with knowledge directly impacting behavior and behavior indirectly

impacting knowledge. Huston's (2010) model indicated knowledge and attitude separately impact behavior. Lee and Hanna's (2014) model suggests financial knowledge impacts attitude and behavior separately, and attitude also impacts behavior after the effects of financial knowledge. For Lee and Hanna (2014), knowledge (cognitive) was a fundamental part of behavior (affective). In previous models, financial knowledge was portrayed as impacting behavior or impacting attitude. Attitude was reflected as connected to behavior. This working model considers financial knowledge, whether tested or perceived, existing within behavior and attitude. Attitude and behavior can materialize no matter what the level of knowledge. Behavior may also be independent of knowledge. For this reason, knowledge resides within behavior and attitude.

The cognitive and affective domains of financial literacy exist within the social. Sherraden (2010) proposed economic socialization influences financial education. Huston (2010) also included a facet of socialization within the model calling it *cultural familial influence*. Shuttleworth (2009), Huston (2010), and Lee and Hanna (2014) included experiences, another trait of social, in their models. In this study's model, the experience and socialization aspects of financial literacy are included in the social domain, while cognitive and affective domains exist within social.

The Financial Literacy/Capabilities model places financial literacy within financial capabilities interacting with financial information and financial products. Total financial literacy is the combination of cognitive, affective, and social domains. The cognitive domain of financial literacy, knowledge including both tested and perceived, was the

focal point of this study. Next, the instruments used to assess financial knowledge are examined.

Financial literacy assessment instruments. In *Assessing Financial Literacy,* Huston (2012) posed the question "What to include in a financial literacy assessment?" (p. 111). In response to this question, Huston (2012) identified "the three *C*'s namely, assessment context, concept, and content" (p. 111). Acknowledging context and concept must align with the assessment goals and objectives; Huston (2012) focused on content.

All financial literacy researchers considered tested knowledge a primary focus of financial literacy. However, there was no single agreed-upon instrument to assess financial literacy (Huston, 2010; Remund, 2010). Several different instruments were used to assess tested and/or perceived financial knowledge. Next, the content of existing financial literacy assessment instruments is reviewed.

Financial literacy assessment content. Remund (2010) reviewed research studies between 2000 and 2006 identifying the four most common content topics within financial literacy as budgeting, saving, borrowing, and investing. Finke and Huston's (2014) Financial Literacy Assessment Test (FLAT) covered four content areas for tested knowledge: (a) money basics, (b) borrowing, (c) investing, and (d) protection (Finke & Huston, 2014). The FLAT content areas somewhat align with the FLEC core competencies defined by the U. S. Treasurer (Financial Education Core Competencies: Comment Request, 2010). Table 3 is summary and comparison of the Finke and Huston (2014) FLAT and the FLEC content categories together with the five to eight financial literacy concepts included in each.

Table 3

Summary and Comparison of Financial Literacy Content Categories Defined by FLEC and FLAT

FLEC (U.S. Treasury, 2010)	Finke & Huston (2014)	Concepts Included in Categories
Earning: Gross vs. net pay, benefits, & taxes ¹ , Sources of Income Spending: Prioritizing spending choices, Long-term vs. Short-term spending, purpose, & use of checking accounts	Money Basics: Compound interest ² , Time value of money ¹ , purchasing power, transaction (checking) accounts, budgeting, cash flow, & balance sheets	Compound interest, Time value of money, purchasing power/inflation, personal financial statements (balance sheet, cash flow, budget, & ratios)
Borrowing: cost of borrowing, credit scores	Borrowing: Use of credit & loan products (e.g., credit cards, consumer loans or mortgages)	APR, finance charges, loan terms, credit scores, type of borrowing (e.g., education, home, vehicles)
Saving & Investing: Compound interest ² , Time Value of money ² , Savings account/Certificates of Deposit, Investment products (bonds, stocks mutual funds)	Building: Certificates of deposit, stocks, bonds, mutual funds	Compound interest, Savings accounts/Certificates of deposit, stocks, bonds, mutual funds
Protecting: Risk management, emergency fund, insurance, fraud/scams/identity theft	Capital protection : Managing risk, insurance (e.g. property health, disability income, life), diversification, tax ¹ /estate planning	Investment product risk (stocks vs. bonds vs. mutual funds), insurance protection

Note. ¹FLEC (2010) includes taxes in earnings content area, while Finke & Huston (2014) include tax planning in capital protection. ² FLEC (2010) includes compound interest and time value of money in savings and investing content area, while Finke & Huston (2014) include it in money basics content area. APR=Annual percentage rate.

Since 2004, numerous researchers studying adult financial literacy have used data from six publically available data collections surveys: (a) Health and Retirement Study (HRS), (b) De Netherlandsche Bank Household Survey (DHS), (c) Rand American Panel Life (ALP), (d) National Financial Capability Study (NFCS), (e) TNS Global, and (f) World Bank Group. Four of the surveys (HRS, DHS, ALP, & NFCS) used either the *Big Three* or *Big Five* questions. Questions for the fifth survey, TNS Global, designed by Lusardi and Tufano (2009) deviate from the *Big Five* theme of savings and investment addressing debt literacy. The sixth survey, World Bank Global, was not administered in the United States; however, two of the four financial literacy questions used in World Bank Global were similar to *Big Five* questions. HRS, AP, and NFCS surveys strictly concentrate on the United States population.

These surveys contained 36 different questions in ten concept areas focusing on tested financial literacy. As a group, the questions cover short-term financial literacy topics (25%) and long-term financial topics (75%). Basic financial literacy knowledge was tested in 47% of the questions, while advanced knowledge was tested in the remaining 53%. Five of the questions were the *Big Five*.

Big Five financial literacy questions. Many of the academic articles discussing tested financial literacy scores cited in this literature review are based on five primary questions assessing tested financial knowledge. See Appendix A for a copy of the *Big Five* financial literacy survey questions. Lusardi and Mitchell (2011b) evolved the five standard questions from three questions created for the 2004 HRS study. Initially, four key principles were used to develop the questions: (a) simplicity, (b) relevance, (c) brevity, and (d) capability to differentiate.

According to Lusardi and Mitchell (2011b), the goal was to create a small number of questions that would be "relevant across ethnic/cultural groups rather than focus on any specific market" (p. 498) with the expectation of translation to an international context.

Economic models of saving and portfolio choice were the basis of the initial three questions concentrating on economic concepts needed to make financial decisions: (a) understanding compound interest, (b) understanding inflation, and (c) understanding risk diversification. Questions covering mortgage interest and interest rates vs. bond prices were added to complete the five questions. Keeping the number of questions to a minimum was to encourage widespread adoption. It was Lusardi and Mitchell's (2011b) intent to measure basic financial concepts using questions "pertinent to people's day-to-day financial decision over the life cycle" (p. 498).

The five questions designed by Lusardi and Mitchell focused on financial concepts most important to retirement planning, and long-term money management. Knolls and Houts (2012) noted the questions "are skewed toward investing and savings knowledge" content areas" (p. 406). FLEC core (i.e., earning, spending, saving, borrowing, and protecting against risk) represent balanced content areas competencies (Financial Education Core Competencies: Comment Request, 2010). Financial knowledge covering to day-to-day money management such as cash-flow, budgeting, and credit cards was not assessed by the *Big Five* (Knolls & Houts, 2012). Surveys with more questions existed, but were not as frequently used as the Lusardi and Mitchell questions. Ultimately, Lusardi and Mitchell (2011b) designed the survey
questions to differentiate financial knowledge levels of study participants by making scores on a common set of questions comparable.

Lusardi was credited with creating the standard in financial literacy assessments and has an international reputation for work in financial literacy. Publishing 25 National Bureau of Economic Research (NBER) working papers on the subject (NBER, n.d.), Lusardi was the founder and director of Global Financial Literacy Excellence Center (GFLEC, 2016a) as well as the director of RAND Labor and Population Financial Literacy Center (RAND Financial Literacy Center, 2016a, 2016b). Considered a pioneer in the study of financial literacy, Lusardi received the 2014 William A. Forbes Public Awareness Award bestowed by Council for Economic Education (PRWeb, 2015). Morrison, CEO and President of Council for Economic Education, comments on the award saying, "Through groundbreaking work as a researcher and advocate, Lusardi has been a leading voice in the national conversation on financial literacy, raising global awareness of its critical need" (PRWeb, 2015, para. 5).

The *Big Five* financial literacy questions Lusardi developed with Mitchell (Lusardi & Mitchell, 2011b) were not without shortcomings. Critics have suggested the five questions are narrow coverage and lack of breadth. Nicolini et al. (2013) in *Financial Literacy: A Comparative Study Across Four Countries*, states:

While the Lusardi and Mitchell questions are a start, they cover a relatively narrow range of content. Little is known about how well responses to these questions correlate with knowledge about other aspects of financial management. In addition, all the questions include numbers; it is unknown whether they assess financial literacy or numeracy or both (pp. 697-698).

Finke and Huston (2014) identified the main drawback of the Lusardi and Mitchell (2007a, 2007b, 2007c, 2008, 2011a, 2011b, 2014) *Big Five* questions as the lack of

items in the area of borrowing and protections of resources. According to Huston (2010), *Measuring Financial Literacy*, "initial instruments consisting of as few as three items would appear to be deficient to capture the breadth of human capital specifically related to personal finance" (p. 309).

Evaluation of assessments. Although a number of researchers have reviewed the literature and compared financial literacy assessment scores based on type and composition of participants, only three journal articles have evaluated the quality of financial literacy assessment test questions. Knoll and Houts (2012), The Financial Knowledge Scale: An Application of Item Response Theory to the Assessment of Financial Literacy, focused on psychometrically developing a more systematic and consistent approach to assessing financial literacy knowledge. In Evaluation of a Financial Literacy Test Using Classical Test Theory and Item Response Theory, Kunovskaya et al. (2014) evaluated the quality of a six-item financial literacy test used in World Bank Financial Capability and Consumer Protection Surveys and the De Netherlandsche Bank. The six questions were adapted from those used by Lusardi and Mitchell (2007a, 2007b, 2008, 2011a, 2011b), Van Rooij et al. (2011), and Cole, Sampson, and Zia (2009). Finally, Fernandes et al. (2014) developed a 13-items scale with psychometric properties in conjunction with a meta-analysis of 168 papers covering 201 prior studies. Each article looked at financial literacy assessment questions from a different perspective.

Knoll and Houts. Evaluating financial literacy questions from large-scale surveys (ALP, HRS and NFCS), Knoll and Houts (2012) used the Item Response Theory model to identify 20 items for inclusion in a general-purpose assessment tool. Knoll and Houts

select 20 items for the general-purpose assessment tool to ensure a broad range of practical topics with a wide range of difficulty parameters. Items were limited to 20 to reduce possibilities of respondent fatigue and allow the assessment to be acceptable for inclusion in large-scale surveys (Knoll & Houts, 2012). Item Response Theory described how each question performs in a population. It used a collection of statistical models to determine values for each question. The quantitative values were then used to develop a scale to analyze and inform which questions to include in the financial literacy assessment (Knoll & Houts, 2012).

According to Knoll and Houts (2012), the 20-question scale "provides *empirical* support for the 'goodness,' or suitability, of several of the most commonly used questions to measure financial knowledge" (p. 404). Knolls and Houts' (2012) goal was to provide an empirically tested comprehensive index of the tested financial knowledge while reducing variability in the measurement. To support the appropriateness and predictive nature of the scale, exercises confirming the validity of the questions were conducted.

Kunovskaya, Cude, and Alexeev. Using both the Classical Test Theory and the Rasch model, a form of Item Response Theory, Kunovskaya et al. (2014) administered a six-question assessment in Azerbaijan (situated at the crossroads of Eastern Europe and Western Asia), Romania, and Russia. The goal of Kunovskaya et al.'s (2014) research was to thoroughly evaluate the construct validity and reliability through rich information about individual and grouped items statistics.

Kunovskaya et al.'s (2014) findings included moderately high internal consistencies across countries, high reliability measured by an item separation and

reliability index, and sufficiently good overall fit. Rasch evaluation also specified problems with the money illusion and interest rate questions indicating a misfit with reliability indexes. The overall conclusion by Kunovskaya et al. (2014) was that a financial literacy assessment required more items with varied difficulties.

Fernandes, Lynch, and Netemeyer. In developing a financial literacy assessment instrument, Fernandes et al. (2014) reduced the items used in previous research from 26 items to 13 and conducted three separate studies. Fernandes et al. (2014) performed procedures "(a) to reflect widely adopted definitions of the concept; (b) to tap its key agreed-upon content domain areas; and (c) brief enough to encourage study participation in survey or experimental research" (p. 75). In the initial study, psychometric properties of one-factor confirmatory Item Response Theory, Rasch modeling, reliability, and construct validity testing were established. In the second and third studies, a two-factor model of the financial literacy and numeracy items was used to correlate both financial literacy and the financial behaviors to confidence in financial information search, planning for money long term, willingness to take investment risk.

Across all three studies, Fernandes et al. (2014) performed correlations with other factors (i.e., preference for numerical information, attitude toward/concern for money scale, spendthrift-tightwad scale, cognition scale, numeracy scale, confidence in financial information search scale, and planning for money in the short-run. Using correlations, construct validity was confirmed in all significant expected or predicted directions.

Summary of evaluations. The key finding identified in the Fernandes et al. (2014), Knoll and Houts (2012), and the Kunovskaya et al. (2014) articles applying to

this research study was the suggestion for the number of items in a financial literacy assessment instrument. Recommending the number of items be larger than five, these researchers also cautioned the number should not be too large to prevent fatiguing or causing unease to participants. Kunovskaya et al. (2014) determined three to five items per concept were more appropriate than only one as in the *Big Five*. Knoll and Houts (2012) and Finke and Huston (2014) also noted this finding. Kunovskaya et al. (2014) suggested a greater number of questions would support including questions with differing degrees of difficulty. Knoll and Houts (2012) believed a 20-question assessment would allow for basic and advanced (or more sophisticated questions) within a mix of financial literacy topics. The need for a more balanced mix between content areas, for example, the distribution between the FLEC core competencies rather than a heavy emphasis on investment and savings, was also suggested. Knoll and Houts (2012) selected 20 items for inclusion based on item response theory analysis. Knoll and Houts' goal was to identify a general-purpose assessment tool to cover a comprehensive range of substantive topics and include a reasonably wide range of difficulty factors sufficiently. Finally, the financial literacy assessment questions evaluated by these researchers were psychometrically tested for validity and reliability (Fernandes et al., 2014; Finke & Huston, 2014; Knoll & Houts, 2012; Kunovskaya et al. 2014). The suggestion for the number of assessment questions in an instrument was used as a target in developing the instrument for this study.

Summary

Financial literacy has been a subject of concern for over 200 years. In recent years, the financial turmoil around the world has escalated the concern. Researchers

evaluating the level of financial literacy within populations around the world have deemed it is below acceptable levels. The global financial crises in 2008 and 2009 renewed interest in financial literacy. An abundance of researchers, government agencies, and private organizations came forward to look for a cause and possible solution. There still is no agreed-upon definition of financial literacy or standard measurement assessment (Hung et al., 2009; Huston, 2010; Remund, 2010).

None-the-less, financial literacy has been researched around the world. In the United States, much of the research has been based on four reoccurring surveys, Jump\$tart, Health and Retirement Survey (HRS), American Life Panel (ALP), and National Financial Capabilities Study) NFCS. The findings associated with data from these surveys concurred with each other that financial literacy is low, with women, young adults, older adults and ethnic groups below average. The only financial literacy research for the state of Florida was based on Florida data extracted from in the NFCS State-by-State survey. No research related to small business owners in the U.S. or in the state of Florida has been published.

This study aimed to investigate the financial literacy continuing professional educational needs of the Florida small business owners with a focus on gender, age, education level, and small business classification differences. The financial literacy assessment instrument used in this study was developed from assessment questions used in prior research.

Chapter 3

Methods

The purpose of this study was to assess the financial literacy continuing professional education cognitive needs of Florida small business owners. This chapter defines and discusses the research questions, population and sample, instrumentation, data collection, data analysis, and summary.

Research Design

This research study was a nonexperimental quantitative survey design. This type of research investigates relationships between variables, but does not manipulate the variables (Ary, Jacobs, & Sorenson, 2010). Survey research represents studies primarily relying on questionnaires or interviews for data collection (Gall et al., 2007). A questionnaire instrument developed for this study was used to identify the financial literacy profile of Florida small business owners. The questionnaire included financial literacy assessment scores of tested knowledge and self-assessed financial knowledge as well as demographic information related to gender, age, education level, and small business classification.

Research questions. Based on prior research, this study investigated the following research questions:

- 1. What is the financial literacy profile of the Florida small business owners?
- 2. What are the direction and extent of differences in the financial literacy profile by gender?

- 3. What are the direction and extent of differences in the financial literacy profile by age?
- 4. What are the direction and extent of differences in the financial literacy profile by education level?
- 5. What are the direction and extent of differences in the financial literacy profile by small business classification?

Variables. There were six variables in this study. The variables were: tested

knowledge, self-assessed knowledge, gender, age, education level, and small business

classification. Each variable's format of response, type, and level of variable is detailed

in Table 4. Tested knowledge and self-assessed knowledge were response variables

(dependent variables).

Table 4

Level of Format of Variable Variable Response Type Response/ Tested knowledge Number correct Interval Criterion Self-assessed Response/ 1 to 7 Interval knowledge Criterion Gender Male=0 Explanatory/ Nominal Predictor Female=1 Age (adults, 18+) In years Explanatory/ Ratio Predictor (four age groups based on year of birth) Ordinal Education Level Level of schooling Explanatory/ Predictor (4 levels) Explanatory/ Small business Categories Nominal Predictor classification 1 through 3

Profile of Variables Related to the Conduct of the Study

The explanatory variables (independent variables) were gender, age, education level, and small business classification.

Population and Sample

Participants for this study were selected from small businesses registered with the West Central region of the Florida Small Business Development Center Network (FSBDCN). The FSBDCN serviced over 13,000 small business owners in 2012 (FSBDCN, 2016b). The West Central region is one of the 10 FSBDCN regions throughout the State of Florida. The sample of small business owners registered with the West Central region is representative of the Florida small business population. The director of West Central region of the FSBDCN, E. Rodriguez, and the FSBDCN state director, M. Myhre, were supportive of this research, as the researcher has been an FSBDCN Certified Business Analyst in the West Central region since 2011. E. Rodriguez agreed to provide access to the network clients for sampling (personal communication, December 10, 2014). In June 2015, M. Myhre provided the IRB (Institutional Review Board) with a letter of support. See Appendix B for a copy of the letter of support from the FSDBCN state director. West Central Florida region FSBDCN clients receive consulting services from the non-profit public outreach of the University of South Florida business department at no charge. As part of the consulting services agreement, clients agree to participate in FSBDCN surveys. Based on this agreement, it was reasonable to expect FSBDCN clients would participate in this research study.

The desired sample size, calculated using a confidence level of 95% with a confidence interval of 5%, was 160. This was based on power of .80, α at .05, and the effect size (ES - *f*²) of .25 effect size. For a one-way ANOVA with four groups, the

required sample size was n = 160 (Stevens, 2007). With three groups and an effect size of .25, the sample size was n = 120 (Stevens, 2007, p. 415).

The effect size of .25 was deemed reasonable based on three studies conducted by Fernandes et al. (2014) developing an instrument for a correlational study to measure financial literacy and financial behaviors. Included in the studies were correlations of financial literacy with gender, age, education, and self-efficacy (self-assessed knowledge). Table 5 shows results of the Fernandes et al. (2014) studies showing correlations of financial literacy with gender, age, education, and self-efficacy. According to Cohen (1992), Pearson Product-Moment Correlation (*r*) of the population is considered the effect size. The Fernandes et al. (2014) correlation results for the effect size ranged from .20 to .51. Using a .25 effect size was consistent with the research performed by Fernandes et al. (2014). The actual sample size of completed surveys was 156.

Table 5

	Study 1	Study 2	Study 3
Variable	r	r	r
Gender	0.26	0.26	0.21
Age	0.28	0.35	0.20
Education		0.29	0.51
Self-Efficacy			0.21

Fernandes, Lynch, and Netemeyer (2014) Correlations of Financial Literacy with Gender, Age, Education, and Self-Efficacy

Instrumentation

The instrument developed for this survey consists of three parts: (a) demographic data, (b) tested knowledge, and (c) self-assessed knowledge. Demographic data were collected to answer research questions related to gender, age, education level, and small business classification. This data included: gender, age group (based on age, determined by year of birth), education level, number of years in business, and number of employees. The form of the demographic data collected was as follows: gender was coded as either male (0) or female (1), while the age group was based on year of birth. The age groups were coded 1 through 4 and included (a) 18 to 40 years, (b) 41 to 50 years, (c) 51 to 60 years, and (d) 61 years and over. Education level was based on the following four categories (coded 1 through 4): (a) high school, (b) Associates degree, (c) Bachelor's degree, and (d) Master's degree or higher.

The tested knowledge portion of the instrument included 18 items. The content areas covered comprised money basics, borrowing, building assets (savings and investments), and protecting assets (diversity and risk management). According to Financial Literacy and Education Commission (FLEC) (2010) and Finke and Huston (2014), these four areas create a comprehensive view of basic financial literacy. Money basics covered the concepts of compound interest, time value of money, and personal net worth. Annual percentage rate (APR), finance charges, loan terms, credit scores, and type of borrowing (e.g., education, home, vehicles) were covered under the borrowing category. Building assets included concepts of compound interest, savings accounts/certificates of deposit, stocks, and mutual funds. Investment product risk and risk diversification were included in protecting assets. Tested knowledge scores were

reported based on the number of correct responses method, one of the two common assessment-scoring methods (Cude, Kunovskaya, Kabuci, & Henry, 2013). The second method is the percentage correct. Both methods yielded the similar statistical outcomes in this study. In addition to the possible true/false or multiple-choice questions, participants also had the ability to answer these questions with the responses *Don't know* and *Prefer not to say*. These additional responses were used in the Health Retirement Study (HRS), American Life Panel (ALP), and National Financial Capabilities Study (NFCS) surveys (Lusardi, 2011; Lusardi & Mitchell, 2007b, 2007c). Providing these options gave the participants alternative choices to guessing. *Don't know* and *Prefer not to say* answers were considered incorrect in determining tested knowledge.

Self-assessed knowledge was based on participants' responses to the questions reflecting self-confidence and beliefs about their financial knowledge using a 7-point Likert scale. The mean of the five self-assessed questions responses was used to determine the individual's self-assessed knowledge scores. The five self-assessed questions were modifications of the questions used in the NFCS survey (Lusardi & Mitchell, 2011), HRS (Lusardi & Mitchell, 2007a, 2007c, 2008), ALP (Lusardi & Mitchell, 2007b, De Nederlandsche Bank Household Survey (Van Rooij et al., 2009), Organization for Economic Co-operation and Development (OECD) surveys (Atkinson & Messy, 2012), the Cognitive Economics Survey (Hsu, 2011), and Financial Literacy Assessment Test (FLAT) (Finke, 2011; Finke & Huston, 2014; Huston, 2012).

Instrument development. The instrument for this study was developed from existing financial literacy tested and assessed knowledge questions. The sources used

for this process are listed in Table 6, which includes information about the sources used to develop the initial financial question pool. The primary purpose of the tested knowledge portion of the assessment instrument was to identify the level of financial literacy tested knowledge of participants. The self-assessed knowledge portion of the assessment instrument was to determine participants' level of confidence or selfperceived financial literacy. The mean scores from both sections were used to define the financial literacy profile of Florida small business owners.

A five-step process with three panels of experts and a field-test group detailed in Table 7 was used to develop the tested knowledge and self-assessed knowledge questions. The process involved the following steps:

- 1. Collect a pool of suitable financial literacy instrument questions. (Researcher)
- Refining the pool of instrument questions by excluding question not meeting prescribed criteria and categorizing financial literacy concepts by content area. (Initial Panel)
- Evaluating the appropriateness of question language, wording, assigned categories, rating difficulty, and ranking questions, level of appropriateness. This was undertaken for the tested knowledge. For the self-assessed knowledge, questions were evaluated for appropriateness of language and wording of the questions. (Validation and Verification Panels)
- 4. Evaluating questions for understandability, comprehension, and user difficulty. (Field Test Group)
- 5. Final review of the assessment instrument. (Final Review Panel).

Expert panels consisting of 32 individuals performed content validation procedures.

After evaluating questions for inclusion, the panel reviewed them for appropriateness of

language, understanding of questions, and elimination of word bias on both tested

knowledge and self-assessed knowledge questions. The panels also evaluated the

assignment of content subjects to categories and question difficulty. The demographic

profile of all panel members is compiled in Table 8, the expert panel demographics.

Table 6

Fourteen Sources Used to Develop Initial Financial Literacy Question Pool Including Criteria for Selection, Number of Questions (Initial, Eliminated, and Retained), and Participant Age and Originating Country

#	Survey	Researcher/ Author(s)	Year	Criteria	# of Questions	# Eliminated	Total # Retained	Participant age/country
1	Investing IQ	Volpe et al. (1996)	1996	2	10	1	9	j
2	Survey of Personal Financial Literacy	Chen & Volpe (1998)	1998	2	31	2	29	18+; U.S. 18-29; U.S
3	Survey of Consumers	Hilgert, Hogarth, & Beverly (2003)	2002	2	28	4	24	18+; U.S
4	HRS	Lusardi & Mitchell (2007c)	2004	1	3		3	> 50; U.S.
5	DHS	van Rooij et al. (2007, 2009)	2005	1	16	4	12	30 - 60; Non-
6	ALP	Lusardi & Mitchell (2007b)	2006	1, 2	23	17	6	0.0.
								18+; U.S.
7	TNS Global	Lusardi & Tufano (2009)	2007	1	3		3	10
8	Jump\$tart	Mandell (2008)	2008	2	31	7	24	10+, 0.3.
C C	• •			_	•	-		15 - 19; U.S.
9	CogEcon	Delavande et al. (2008) Hsu (2011)	2008	2, 3	24		24	> 50; U.S.
10	NFCS	Lusardi (2011)	2009	1	5	4	1	18+; U.S.

Continued on the next page

Table 6 (continued)

Fourteen Sources Used to Develop Initial Financial Literacy Question Pool including Criteria for Selection, Number of Questions (Initial, Eliminated, and Retained), and Participant Age and Originating Country

#	Survey	Researcher/ Author(s)	Year	Criteria	# of Questions	# Eliminated	Total # Retained	Participant age/country
11	WBG	World Bank (2009)	2009	1	4	3	1	18+; U.S.
12	FLAT	Finke & Huston (2014)	2011	2	16		16	18+; U.S.
13	OECD	Atkinson & Messy (2011)	2012	2	8	4	4	18+; Non-U.S.
14	USIS	Alhenawi & Elkhal (2013)	2013	3	7	3	4	18+; U.S.
		Total			209	49	160	

Note. Survey: HRS--Health & Retirement Study, DHS—De Nederlandsche Bank Household Survey, ALP--American Life Panel, TNS Global--Taylor, Nelson, Sofres Global, CogEcon--Cognitive Economic Survey, NFCS--National Financial Capability Study, WBG--World Bank Global, FLAT--Financial Literacy Assessment Test, OECD--Organization for Economic Co-operation and Development, and USIS--University of Southern Indiana Survey; Selection Criteria: 1=Instruments receiving content evaluation (Fernandes et al., 2014; Knoll & Houts, 2012; Kunovskaya et al., 2014), 2=Instruments heavily cited in research covered in the literature review, 3=Instrument questions reviewed and developed through comprehensive research or psychometric processes variables (gender, age, education level, and small business classification)

Table 7

Research Process Steps to Achieve Content Validation

STEP #1	STEP #1	STEP	#3	STEP #4	STEP #5
Researcher	Initial	Validation Panel	Verification Panel	Field Test Group	Final Review
1 Individual	12 Individuals	4 Individuals	4 Individuals	12 Individuals	3 Individuals
Select Questions	160 Questions	64 Questions	28 Questions	20 Questions	18 Questions
From 14 sources:					
-182 questions - 8 Self-assess questions	2. Evaluated concepts, assign categories	 Appropriateness -of Language -Word Bias 	 Verify Appropriateness -of Language 	7. Reflective cognitive process	10. Final Review
1. Eliminate questions	3. Identify questions for	Rate difficulty 5. Evaluate	-Word Bias -Categories -Difficulty rating	8. Initial test of Assessment	
	Inclusion and Exclusion	Inclusion, Identify Category	,	9. Feedback	

Note. Tasks description:

1. Detailed review and analysis of tested knowledge and self-assessed knowledge questions by Researcher to eliminate questions that are duplicates or similar wording, open-ended, dated content, contain currency/cultural bias, or are not a concept in the four categories.

- 2. Initial Panel reviews financial literacy concepts and assigns each to one of four financial literacy assessment categories.
- 3. Initial Panel reviews the tested knowledge questions for inclusion or exclusion to reduce the total number of questions.
- 4. Validation Panel's assesses the appropriateness of the questions: Language, word bias (tested knowledge and self-assessed knowledge) questions, and financial literacy category for each question (tested knowledge).
- 5. Verification Panel's assesses the appropriateness of the questions: Language and word bias. Then, agreement with questions categories.
- 6. Validation Panel rank order of question suitability within concept for inclusion in the final instrument.
- 7. Verification Panel rates each questions for difficulty (tested knowledge).
- 8. Two-three Field Test Group members perform reflective cognitive process with field test instrument.
- 9. Field Test Group completes the instrument and responds to online interview questions providing feedback of understandability and difficulty of questions.
- 10. Final Review Panel performs unstructured review of the instrument (18 questions).

Instrument development steps. A series of tasks was executed for the development and content validation process of this study's instrument. There were 10 separate tasks (reviews) within the five-step process aimed at reducing the pool of questions to the final instrument. There were defined goals for each review. The questionnaires and responses in the validation process of this study were also hosted on the Qualtrics' secured server. Qualtrics software enables researchers to collect data online through the use of survey questionnaires. Qualtrics.com was chosen because it provides features simplifying the process for the researcher and panel participants. The steps to validate the instrument were as follows:

Step #1. The researcher collected suitable financial literacy instruments based on the review of the literature. The pool of instruments contained 14 assessment instruments or groups of instrument questions, previously shown in Table 6. The selection of 14 instruments was based on the following criteria:

1. Instruments receiving content evaluation (Fernandes et al., 2014; Knoll & Houts, 2012; Kunovskaya et al., 2014).

2. Instruments heavily cited in research covered in the literature review, as shown in Table 9, Number of Times Researchers Were Cited.

3. Instrument questions reviewed and developed through comprehensive research or psychometric processes.

Table 8

Expert Panel Demographics

Category	n	%*
Gender		
Male	11	34.4
Female	21	65.6
Education Level		
Ph.D.	10	31.3
Ph.D. Candidate	5	15.6
MBA	8	25.0
Masters	6	18.8
Other	3	9.4
Credentials		
SBDC Certification	9	28.1
CPA Certification	7	21.9
Other Relevant Certification**	3	9.4
Total holding certifications	19	59.4
Area of Expertise		
Financial Literacy	12	37.5
Adult Education	10	31.3
Small Business	14	43.8
Banking, Financial Planning or Economics	12	37.5
Area of Employment		
Education	14	43.8
Business	18	56.3
Government/Other	3	9.4

* Percentage add to more than 100% due to panel members in more than one group; ** Other certification areas included: Valuation Analyst, Profit Mastery, and Economist & Policy Analyst In Task #1, a total of 209 questions were accumulated from the 14 sources. This researcher reviewed all the questions for duplications or similarities in wording. Twenty-seven questions were recognized as duplications and were removed. See Appendix C for the list of the pool of financial literacy questions containing 182 questions. These questions were evaluated for any of the following conditions: open-ended questions, currency or cultural bias (international), dated content, and questions covering concepts outside of the four content areas identified in the literature review.

Table 9

Researchers	Cited
Hilgert, Hogarth, & Beverly (2003)	777
Chen & Volpe (1998) Volpe et al. (1996)	710
Mandell (1998, 2008) Mandell & Klein (2007, 2009)	632
Hung et al. (2009)	158

Selection Criteria #2: Number of Times Researchers Were Cited

Note. Google Scholar Citations

During this evaluation, 22 questions were eliminated from the pool, leaving 160 for expert panel review. The summary of the disposition of the pool questions by source is included in Table 6, previously shown. With the first review of pool questions completed by this researcher, the Initial Panel content validation process began.

Step #2. The Initial Panel was asked to conduct two rounds of review:

(a) evaluating 28 financial concepts and assigning each concept to one of the four

content categories (Task #2) and (b) determination of inclusion or exclusion of the questions for the instrument (Task #3). This panel consisted of 20 individuals with backgrounds in education, accounting, financial planning, banking, or financial literacy. The panel was divided into five groups for the second round of review, and the question pool was systematically divided into sets of 32 questions. Each panel group received 32 questions to review. The panel members received explicit instructions with criteria for evaluating a portion of the total pool in Task #3. See Appendix D for the names of the initial panel members.

Initial Panel members were invited and confirmed via email in advance of receiving instructions. See Appendix E for the request for participation to potential panel members email. Once acceptance was confirmed, each panel member received an email with instructions and the link to an online tailored questionnaire. See Appendix F for the initial panel communication emails. See Appendix G for a copy of the initial panel round 1 online survey. Instructions were repeated on the opening screen of the online review survey. The panel members were first asked to evaluate financial literacy concepts for inclusion, followed by assigning included concepts to one of four financial literacy content categories (Task #2). Upon completion of Task #2, each panel member received another email with instructions to evaluate each question in a group of 32 questions for inclusion or exclusion in the knowledge test portion of the financial literacy instrument (Task #3). The email instructions included the link to another online tailored questionnaire with the instructions provided again on the opening screen (see Appendix F). See Appendix H for the copy of the initial panel round 2 online survey. Based on the Initial Panel's Round #2 review where each question was reviewed by four panel

members, the pool of questions was reduced from 160 to 64. A summary of the Initial Panel's disposition of the pool questions by source is also shown in Table 6.

Step #3. The Verification and Validation Panels completed a total of three rounds of review. The Validation Panel consisted of six individuals while the Verification Panel included three individuals (see Appendix D). Individuals for both panels had backgrounds in education, accounting, financial planning, banking, or financial literacy. Panel members were invited and confirmed via email in advance of receiving instructions. Each panel member received an email with instructions and the link to an online tailored questionnaire. Instructions were also on the opening screen of the online review survey. See Appendix I for a copy of the Validation Panel communication emails. See Appendix J for Verification Panel communication email.

In the Validation Panel first round, the remaining 64 questions were divided into two groups of 32. Each panel member reviewed 32 tested knowledge questions and the eight self-assessment questions. See Appendix K for a copy of the validation panel round 1 online survey. See Appendix L for a copy of the self-assessment financial literacy questions. During the review, the panel rated each tested knowledge question on a five-point scale for language, wording, and difficulty (Task #4). The scale for language was as follows: 1 meant *very unclear*, 2 *unclear*, 3 *neutral*, 4 *clear*, and 5 *very clear*. For wording or word bias, the rating scale was 1, *poor*, 2 *needs improvement*, 3 *neutral*, 4 *good*, and 5 *excellent*. Difficulty was rated as follows: 1 *too easy*, 2 *easy*, 3 *average*, 4 *difficult*, and 5 *too difficult*. When evaluating the eight self-assessment questions, the panel was asked to determine whether each should be included in or excluded from the instrument. During this round, the panel suggested exclusion of one self-assessed question. It was removed.

Once the Validation Panel had completed Task # 4 in the second round, the panel evaluated questions for inclusion in the instrument, and identified the categories each question addressed (Task #5). See Appendix M for a copy of the Validation Panel round 2 online survey. The self-assessment questions were evaluated for language and wording in this round using the same scale as was used for the tested knowledge questions.

After the Validation Panel completed both rounds of review and evaluation, the researcher analyzed the Qualtrics output to reduce the size of the question pool. First, all knowledge questions identified by the panel for exclusion were removed. Next, all questions rated *too easy* or *too difficult* were removed. All questions identified by the panel with language as very clear and wording as excellent or good were retained in the pool. Finally, any question the panel members suggested be retained with improved alternative wording was reworded and retained in the pool. The results of the Validation Panel review and researcher analysis reduced the pool of knowledge questions from 64 to 26.

Analyzing the self-assessment questions, the Validation panel suggested two questions be removed due to lack of relevancy. These questions were removed. Wording was adjusted on three questions to remove the possibility of misleading the respondents. During both rounds, the original eight self-assessment questions were reduced to six.

The remaining 26 knowledge questions and six self-assessment questions were presented to the Verification Panel for review. The panel was asked to evaluate each question and note agreement or disagreement for the following (Task #6): inclusion in the instrument, language, wording, difficulty rating, and categories questions addressed.

The panel was also asked to comment on any items of disagreement. See Appendix N for a copy of the verification panel online survey.

The results of the Verification Panel review were analyzed by the researcher. Six questions were removed from the tested knowledge question pool and wording adjustments were made to two questions. An additional self-assessment question was removed. The post-Validation Panel review instrument contained 20 tested knowledge and five self-assessed knowledge questions.

Step #4. Field-t evaluation of questions for understandability and difficulty was performed using two separate procedures: (a) reflective cognitive process evaluation and (b) field test. For the reflective cognitive process (Task #7), a brief reflective cognitive process evaluation was conducted with five individuals to provide response process evidence. The process was performed aloud and recorded. The individuals completing the process varied in ages (28, 31, 46, 60, and 73) and had business experience from two to over 20 years. Based on a review of the notes and recordings from the reflective cognitive process, the tested knowledge and self-assessed knowledge questions were appraised. Wording of six tested knowledge questions was slightly adjusted.

Prior to the cognitive interviews, this researcher identified the response pattern of the 20 questions finding the mix between true/false and multiple-choice was 60-40. Of the 12 true/false questions, 11 had true responses. Three of these questions identified for wording were changed to create false correct responses. During the cognitive process, each individual gave feedback comparing the true version to the false version. All individuals agreed. The false version properly tested the same content. A total of nine questions were slightly revised based on the reflective cognitive process.

Following the reflective cognitive process, the field test group completed the assessment (Task #8) and provided feedback about to the understandability and difficulty of the questions (Task #9).

Pilot studies, in this research referred to as the field-test study, are designed to trial test the instrument and administration process, as well as, evaluate the data analysis design to inform planning for the main study (Thabane, Ma, Chu, Cheng, Ismila, Rios, Robson, Thabane, Giangregoria, & Goldsmith, 2010). The purpose of a field study is to identify any weaknesses, including inadequate protocols and whether proposed methods or the items on the instrument are inappropriate or too complicated (De Vaus, 2002). The focus of this field test study was to check: (a) if the instructions were understandable, (b) whether the wording of the survey was clear, and (c) if planned statistical and analytical processes would provide the desired outcomes for examination. Baker (1998) suggested the number of participants to consider for a pilot study should be 10-20% of the sample size for the actual study. Thabane et al. (2010) suggest a rule of thumb to "be large enough to provide useful information about the aspects that being assessed for feasibility" (p. 5).

The field-test study was administered with the same procedures planned for the full study (Task #8). Follow-up interview questions at the end of the online questionnaire queried feedback about the clarity of the instructions, and understandability of the survey questions (Task #9). To conduct this field-test study, an email with a link to the survey was personally sent to a group of 44 SBDC registered business owners in the Florida West Central region. These recipients were also former consulting clients of the researcher. See Appendix O for a copy of the field-test online survey. The

field-test group consisted of 11 individuals. The group comprised 36.4% males and 63.6% females. The age demographics of the group were 72.1% between 51 and 60 years old, with the balance of the group between 31 and 50 years old. The years of business operation and the education were spread evenly among all field-test group members. The mean of the tested knowledge score was 13.6, and the mean of the self-assessed score was 5.2. The field test indicated planned procedures were adequate, and participants did not have difficulties understanding questions or comprehending the instructions.

Step #5: The Final Review Panel consisted of 12 members. An unstructured evaluation of the final 20 questions tested knowledge and five self-assessed knowledge, plus the demographic questions (Task #10) was conducted in this review. See Appendix Q for the final review panel communication email. See Appendix R for a copy of the initial review online survey. This panel evaluated the instrument based on their expertise and any previous experience as an expert panel member. Based on the Final Review Panel's recommendations, two additional questions were removed. Wording was further clarified on two questions, demographic question possible responses were refined, and a back button was added to the survey.

Final instrument. The final financial literacy instrument for this research contained 18 knowledge and 5 self-assessing questions. See Appendix S for a copy of the final instrument. Several of the knowledge questions were associated with more than one category. As a result, looking at the distribution of questions in each category yielded a coverage percentage greater than 100%. The breakdown of questions in each of the financial literacy content categories was: 67% money basics, 39% borrowing, 61% building assets, and 12% protecting. Nine of the original 14 sources

contributed questions to the final survey. The source contributing the most questions (n=7) was Survey of Consumers (Hilgert et al., 2003).

Data Collection

Field test. A field test was part of the instrument validation process of this study. The field test was administered through Qualtric.com and conducted in the same manner as the final survey. Eleven individuals completed the field-test survey. Table 10, Descriptive Statistics of Field Test, details the results of the field-test responses included in the final sample count for data analysis. The Shapiro-Wilk Test for this sample indicates normal distribution (tested knowledge *p*=.03446; self-assessed knowledge *p*=.8093).

Procedures. Participants were selected from active clients of West Central region the FSBDCN. The FSBDCN has 10 regions. The West Central region is one of the largest regions in the state representing Florida's small business population. FSBDCN technology personnel created a list of active clients over the past five years resulting in approximately 12,000 emails. An invitation to participate in this financial literacy research was emailed by the FSBDCN technology department to the entire list. See Appendix T for copies of the participant communications emails. The email contained a link to the online survey hosted on <u>www.qualtrics</u>. The first email was sent November 4, 2015. Two reminder emails were sent on November 18th and December 2nd. A final invitation, for a total of four requests, was sent on December 15th.

Surveys were administered anonymously and participants' identities remained anonymous, based on Institutional Research Board (IRB) guidelines. The following steps were taken to ensure all aspects of this study were planned and executed according to ethical standards:

Table 10

Descriptive Demographic Statistics of Field Test

			Tested Kno Corre (Total possi	Tested Knowledge Correct (Total possible=18)		Self-assessed Knowledge (Max Score=7)	
	n	%	\overline{X}	SD	\overline{X}	SD	
Gender:							
Male	4	36.4	13.5	1.3	5.8	0.7	
Female	7	63.6	13.7	1.9	4.8	1.0	
Age Group:*							
18-50 years	3	27.3	12.0	1.7	5.1	0.3	
51 years and over	6	54.5	14.5	1.5	4.9	1.2	
Years in Business:							
3-5 years	3	27.3	13.3	1.2	5.3	0.3	
6-10 years	3	27.3	13.0	2.6	5.3	1.2	
10 years and over	4	36.4	14.3	1.5	5.4	1.0	
Education:							
High School	1	9.1	13.0	0.0	4.8	0.0	
Associate's degree	2	18.2	13.5	2.1	5.9	1.0	
Bachelor's degree	5	45.5	13.0	1.7	4.8	0.9	
Master's degree or higher	3	27.3	15.0	1.0	5.2	1.0	
Business Classification:							
Pre-venture/start-up	0		0.0	0.0	0.0	0.0	
Small Business	11		13.6	1.6	5.2	1.0	
Small-Medium Enterprise	0		0.0	0.0	0.0	0.0	
Total	11		13.6	1.6	5.2	1.0	

Note. * Two participants did not provide year of birth

1. The researcher was current on all IRB educational requirements.

 The study plan was submitted to the University of South Florida IRB panel for review and approved prior to the onset of any, and all research activities. See Appendix U for a copy of the IRB approval letter.

3. Once approved, the plan was followed and executed by the researcher and any and all agents of the researcher.

Participants were informed of the anonymous nature of the survey in the original invitation email, as well as, in the online survey opening screen. All dealings with participants were governed by the American Educational Research Association's Code of Ethics, approved by the American Educational Research Association Council February 2011 (American Educational Research Association, 2015). The final survey results were downloaded from Qualtrics.com.

Data Analysis

The instrument was administered through Qualtrics.com where the instrument responses were hosted on a secure server. Qualtrics.com was used because it provided the following features aligned with guidelines for effective online survey design (Andrews, Nonnecke, & Preece, 2003; Shropshire, Hawdon, & Witte, 2009; Yan, Conrad, Tourangeau, & Couper, 2010): (a) question types include multiple-choice, true/false, or text entry; (b) ensuring the confidentiality of the participant; (c) design of individual screens to reduce question burden per page; and, (d) incorporating a progress bar allowing participants to see their movement through the survey.

The data collected from the Qualtrics.com survey were exported to EXCEL. SAS was used to generate statistical analysis. Three primary statistical methods were used for the analysis: (a) descriptive statistics, (b) *t* test, and (c) one-way ANOVAs.

Secondary statistical analysis was produced using the Shapiro-Wilk test, Levene's test, Satterhwaite test, and Tukey's test.

Content validity. Content validation was performed from two perspectives to substantiate the validity of the instrument: (a) evidence demonstrating validity and (b) process of validation (validation process procedures). Ary et al. (2010) suggested, "Historically, validity was concerned with the extent to which an instrument measured what it claims to measure" (p. 225). Current views of instrument validity focus on the interpretation and meaning of scores derived from the instrument. Gall et al. (2007) highlight the 1999 *Standards for Educational and Psychological Testing* definition of validity noting, "the fact that test scores themselves are neither valid nor invalid. Rather, it is interpretations of the scores that are either valid or invalid" (p. 195).

The instrument developed for this study was based on questions from empirically tested assessments from 14 sources (Table 6). The questions from these sources were examined, evaluated, revised, and tested in the development of the questionnaire to assessed tested knowledge and self-assessed knowledge. All questions selected for evaluation in this study were previously tested for validity and reliability in connection with the original empirical articles, working papers, and reports from which they were identified.

Evidence demonstrating validity. Gall et al. (2007) identified five main types of evidence for demonstrating the validity of test-score interpretations: "(a) evidence from test content, (b) response processes, (c) internal structure, (d) relationship to other variables, and (e) consequences of testing" (p 195).

Test content. Gall et al. (2007) suggested content-related evidence of test validity requires content experts to review systematically specific content to determine

how well test items depict a respective sample of the content domain. During the literature review, numerous studies evaluated the content of financial literacy questions (Alhenawi & Elkhal, 2013; Atkinson & Messy, 2011; Agnew et al., 2013; Beckmann, 2013; Bumcrot et al., 2013; Chen & Volpe, 1998; Fernandes et al., 2014; Finke, 2011; Finke & Huston, 2014; Gale & Levine, 2010; Hastings et al., 2012; Hilgert et al., 2003; Hsu, 2011; Huston, 2012; JumpStart Coalition, 2007; Knolls & Houts, 2012; Kunovskaya et al., 2014; Lachance, 2014; Lusardi, 2011, 2012a, 2012b; Lusardi & Mitchell, 2007, 2008, 2011a, 2011b; Lusardi, Mitchell, & Curto, 2010; Lusardi & Tufano, 2009; Mandell, 2006, 2008; Mandell & Klein, 2007, 2009; Mottola, 2013; Nicolini et al., 2013; Scheresberg, 2013; Simms, 2014; Van Rooij et al., 2007, 2009, 2011; Volpe et al., 1996; World Bank, 2009). Questions evaluated in these studies were tested for evidence of reliability in the original research. For this reason, the questions were included in an initial pool of financial literacy questions. This study's expert panels systematically evaluated the pool to develop the final instrument.

Response processes. Gall et al. (2007) indicated supporting validity of test-score interpretations can be accomplished by having a participant reflect aloud on the cognitive and evaluative processes being used while responding to test questions. Recording these verbal expressions of thought processes allows for analysis. Study of the consistency within the construct to be measured can provide response-process evidence (Gall et al., 2007). A reflective cognitive process was part of the field test for this study.

Internal structure. The relationship between items on tests can provide evidence of the validity of test-score interpretations because responses to individual questions are often related to each other (Gall et al., 2007). Gall et al. (2007) pointed out

correlations between individual's correct answers can demonstrate that a test does, in fact, measure the construct it was designed to measure. According to Hof (2012) and Osborne (2014), construct validity of a questionnaire can be tested with exploratory factor analysis. The results of the exploratory factor analysis group the items of the instrument. Identifying possible assignment of items to content areas (factors) was the goal of exploratory factor analysis in this study. The review of exploratory factor analysis results allowed exploration of whether the study data fits a model for the previously acknowledged content areas (i.e., money basics, borrowing, building assets, and protecting assets) based on the Finke and Huston (2014) and FLEC (2010) research.

Relationship to other variables. The relationship of other variables is often used to explore predictability (Gall et al., 2007). Since this study was nonexperimental, not seeking to predict, this aspect of evidence validation was not considered.

Consequences of testing. According to Gall et al. (2007), consequential validity

refers

to the fact that test scores, the theory and beliefs behind the construct, and the language used to label the construct also embody certain values and have valueladen consequences when used to make decisions about individuals. The values and consequences need to be checked to determine whether our interpretations of test scores and the way we use these scores to make decisions are valid for particular uses. (p. 199)

Gall et al. (2007) considered intelligence and personality testing are not neutral constructs. Financial literacy testing is similar and is becoming a measure of knowledge valued in society (HM Treasury, 2007; Lusardi & Mitchell, 2007a, 2007b, 2007c 2008, 2011a, 2011b, 2013, 2014; Obama, 2010; PACFC, 2013a, 2013b). Despite the lack of a standard financial literacy assessment (Hung et al., 2009; Huston, 2010; Remund,

2010), the *Big Five* was repetitively used of in large-scale studies. This wide-scale usage has been deemed as some evidence these questions can be considered an international standard (Lusardi & Mitchell, 2011b). The various assessments used around the world have resulted in interpretations of test scores similar to those of the *Big Five*. That is, financial literacy is low around the world with women, younger adults, older adults, and those with lower education and income score below the average. Researchers' recommendations advocate improvement of financial literacy leads to better financial behaviors, and in turn, will advance local, national, and global economies (Atkins et al., 2010; Atkinson & Messy, 2012; Beckmann, 2013; Buckland, 2010; Cude, 2010; Delavande et al., 2008; Hung et al., 2009; Hung et al., 2012; Huston, 2010; Lee & Hanna, 2014; Lusardi & Mitchell, 2011a, 2011b; Mottola, 2013; Nicolini et al., 2013; Remund, 2010; Schmeiser & Seligman, 2013; Wolfe-Hayes, 2010).

Gall et al. (2007) emphasized the American Education Research Standards stating, "Although information about the consequences of testing may influence decisions about test use, such consequences do not in and of themselves detract from the validity of intended interpretations" (p. 199). Consequential validity calls for careful consideration of the interpretation and use of a standard financial literacy assessment and its societal impact. Although not specifically addressing consequential validity, this study used recommendations for optimizing an assessment from previous researchers (e.g., varied degrees of difficulty and more than one question per concept) as a guide in developing the tested knowledge portion of the instrument (Finke & Huston, 2014; Knoll & Houts, 2012; Kunovskaya et al., 2014). The five main types of evidence described by Gall et al. (2007) were followed implementing the instrument validation procedures to support the validity of this study.

Validation process procedures. Crocker and Algina (2008) recommend four

steps in content validation:

- 1. Defining the performance domain of interest.
- 2. Selecting a panel of qualified experts in the content domain.
- 3. Providing a structured framework for the process of identifying items in the content domain.
- 4. Collecting and summarizing the data from the process. (p. 218)

The principles prescribed by Crocker and Algina (2008) were used as guidelines to

design the tested knowledge and self-assessed knowledge portions of the instrument in

this study. Cocker and Algina (2008) guidelines include:

- 1. Identify the primary purpose(s) for which the test scores will be used.
- 2. Identify behaviors that represent the construct or define the domain.
- 3. Prepare a set of test specifications, delineating the proportion of items that should focus on each type of behavior identified in step 2.
- 4. Construct an initial pool of items.
- 5. Have items reviewed (and revise as necessary).
- 6. Hold preliminary item tryouts (and revise as necessary).
- 7. Field-test the items on a large sample representative of the examinee population for whom the test is intended.
- 8. Determine statistical properties of item scores and when appropriate, eliminate items that do not meet pre-established criteria.
- 9. Design and conduct reliability and validity studies for the final form of the test.
- 10. Develop guidelines for administration, scoring, and interpretation of the test scores (e.g., prepare norm tables, suggest recommended cutting scores or standards for performance, etc.). (Crocker & Algina, 2008, p. 66)

The guidelines prescribed by Cocker and Algina (2008) guided the research process

steps listed in Table 7 as a foundation for the content validity of this study.

Construct validity. Construct validity tests the degree to which an instrument

measures what it was designed to measure. Factor analysis is a well-accepted method

of testing for construct validity. One of the objectives of exploratory factor analysis is to

evaluate the construct validity of a scale, test, or instrument (Williams, 2012). In this

study, the construct was financial literacy. Based on the literature review, the financial

literacy tested knowledge construct was divided into four content areas: money basics,

borrowing, building assets, and protecting assets. Exploratory factor analysis was applied to tested knowledge questions to identify possible groupings of the survey items.

According to Williams (2012), it is important to assess the suitability of the sample data. Sample size is a vital element of assessing suitability. Based on recommendations of Costello and Osborne (2005) a sample size that reflects a 10:1 ratio of subjects to items is appropriate for factor analysis. Beavers, Lounsbury, Richards, Huck, Skolits, and Esquivel (2013) also acknowledged the suggestion of 10 subjects per questionnaire item. Osborne (2014) notes a minimum of five subjects per item is acceptable with the rule of thumb being 10 subjects per item. For this study, there were 186 subjects in the tested knowledge portions of the survey fully completed. Based on the 10:1 ratios, the sample size was adequate.

In addition to sample size, there are other tests of suitability of the respondent data. Two such tests are Kaiser-Meyer-Olkin Measure of Sampling Adequacy and Bartlett's Test of Sphericity. The overall Kaiser-Meyer-Olkin (KMO) Measure of Sampling Adequacy (MSA) is .7454. When KMO-MSA is 0.50 or above, data are suitable for analysis. Bartlett's Test of Sphericity resulted in an approximate Chi-square of 602.31, *df*=153 and *p*>0.0001. For the data to be suitable, Barlett's Test of Sphericity should be significant (*p*<.05). Both tests indicate the data were suitable.

After evaluating the suitability of the sample data, SAS was used for the exploratory factor analysis statistical computations. The maximum likelihood extraction method was chosen as "it allows for the computation of a wide range of indexes of the goodness of fit of the model [and] permits statistical significance testing of factor loading and correlations among factors and the computations of confidence intervals" (Fabrigar,

Wegener, MacCallum, & Strahan, 1999, p. 277). Using this method, the number of factors to be extracted was determined by creating a correlation matrix and estimating the communalities using the squared multiple correlations (R²). Squaring correlations between the variable and all other variables was the method used to create a scree plot. The scree plot, shown in Figure 2, depicts the rate of change in magnitude of the eigenvalues for the factors. The number of factors is determined by the point where the curve in the scree plot bends. Three factors were indicated.

Next, the simplest rotation method, orthogonal rotation, was used to determine the final factor pattern and identify questions in each factor through significant loadings. Only factor loadings greater than or equal to .30 absolute value were assigned to a factor. This pattern is shown in Table 11. According to O'Rourke and Hatcher (2014), there are four criteria to address in the interpretation of exploratory factor analysis: (a) does each factor have at least three questions with significant loading identified? (b) do the questions that load on each factor share a common theme or meaning? (c) do the questions that load on each factor differ from loading on other factors? and (d) is a simple structure indicated?

Reviewing Table 11, Factor 1 contained eight of the tested knowledge questions. Factor 2 contains seven questions, and Factor 3 contains four questions. To meet O'Rourke and Hatcher's (2014) first criteria, there must be at least three items in each factor. These criteria were met. Looking at the remaining criteria prescribed by O'Rouke and Hatcher (2014), the second criterion is the ability to identify common themes or idea within each factor.



Figure 2. Scree plot of eigenvalues depicting decision point for number of factors to use Based on loading.
	Rotated Factor Pattern			Communality	Weight
Question	Money	Money	Managing		
Question	Basics	Concepts	Money		
Q1			0.504	0.3090	1.4473
Q2	0.743			0.6165	2.6074
Q3	0.459			0.2256	1.2913
Q4	0.547			0.3302	1.4930
Q5		0.461		0.2684	1.3669
Q6		0.322		0.1259	1.1441
Q7	0.454		0.730	0.7668	4.2877
Q8				0.0894	1.0982
Q9		0.385		0.1758	1.2133
Q10			0.305	0.1414	1.1647
Q11		0.434		0.2401	1.3161
Q12				0.1038	1.1158
Q13				0.1243	1.1420
Q14	0.390	0.382		0.3029	1.4345
Q15	0.411	0.417	0.365	0.4761	1.9088
Q16	0.345			0.1282	1.1471
Q17		0.511		0.3704	1.5883
Q18		0.318		0.1286	1.1475

Factor Pattern Coefficients and Communalities Based on Exploratory Factor Analysis

Note. Factor values less than 0.3 are not included.

The emerging theme for Factor 1 was money basics. Factor 2 was dominated by the items related to money concepts, and the theme of Factor 3 was managing money. The third criteria queries, whether the questions load on each factor differs from those loading on other factors. Based on the final loadings (Table 11), there were three questions loaded on more than one factor and three not loaded on any factors. The final criterion is looking for a simple structure. This exploratory factor analysis identified three factors with varied themes: basic money knowledge, money concepts (definitions), and managing money. This exploratory factor analysis was completed to

determine if factors, financial literacy categories, existed in the instrument. It was not performed to confirm predetermined factors.

The same exploratory factor analysis methods were applied to the five selfassessed knowledge questions to identify possible groupings. The results indicated only one factor for all five questions.

Reliability. When an instrument measures what it is intended to measure with a high degree of consistency, it is said to have a reliability of measurement. Reliability together with validity is necessary attributes for an instrument to provide dependable and usable data for interpretation in research (Ary et al., 2010). Validity in this study has been previously discussed.

Reliability of the instrument begins with the instrument's history of use (Ary et al., 2010). The instrument developed in this study was based on questions from empirically tested assessments from 14 sources. The questions from these sources were examined, evaluated, revised, and tested in the study through the content validation process. All questions were also previously tested for validity and reliability in connection with the original empirical articles, working papers, and reports they were identified. Prior testing, evaluation, and usage of the instrument questions provided some confirmation and assurance the instrument was testing and gathering responses reliably and is, in fact, measuring the construct validity of financial literacy (Ary et al., 2010). This study did not rely on the original testing, rather it was used as a basis for inclusion of the questions in the development pool of instrument questions. In addition to the original validity and reliability measures, three expert panels and a field-test group evaluated the instrument developed for this study. Cronbach's alpha (Cronbach, 1951) was used to assess the reliability of the survey scores and address measurement error.

When determining the reliability of responses, if Cronbach's alpha is .7 or above on a scale of 0 to 1 indicating the internal consistency of the data collected is considered good to excellent (Ary et al., 2010). Cronbach's alpha for tested knowledge questions in this study was .6743, and for self-assessed knowledge was .8967. Both were near the reasonable threshold of .70.

Descriptive statistics and *t* **test.** Descriptive statistics were used to organize and summarize the data to address research question #1 about the profile of the Florida small business owners. The participants' mean scores, both tested knowledge score and self-assessed knowledge score, were used to describe the financial literacy profile.

Independent *t* tests compare means of two groups to determine significant differences. Independent *t* tests were used to test differences in gender to answer research question #2 inquiring about the direction and extent of differences in the financial literacy profile by gender.

Analysis of variance (ANOVA). According to Gall et al. (2007), ANOVA is used to "compare the amount of between-groups variance individuals' scores with the amount of within-group variance" (p. 318). The *F* ratio generated in an ANOVA indicates whether the difference in scores is significant. ANOVAs were utilized in the research to determine differences and answer research questions #3 through #5 (since research question #1 and #2 have been discussed already). Research question #3 considered the direction and extent of differences in the financial literacy profile by age, while #4 contemplated the direction and extent of differences in the financial literacy profile by age profile by education level, and #5 pondered the direction and extent of differences in the financial literacy profile by small business classification.

According to Glass and Hopkins (1996), assumptions when using one-way ANOVAs are: (a) observations are independent, (b) variances are homogeneous, and (c) the dependent variable is approximately normally distributed for each category of the independent variable. This research was designed in a way that independence of observations existed. Levene's test was used to confirm homogeneity. The Shapiro-Wilk test was initially used to determine normality of distribution. The Shapiro-Wilk test for this study's data was p < 0.0001 indicating a non-normal distribution. Shapiro-Wilk test is a formal normality test for small to medium samples (e.g., n < 300), but may also show incompatible results (Kim, 2013). According to Kim (2013), assessing normality, using skewness and kurtosis of the distribution may be used in both small and large samples. Skewness, a measure of asymmetry, measures the distribution of the variable. Kurtosis measures the distribution's peakedness or height of the distribution. When evaluating skewness and kurtosis for normality, West, Finch, and Curran (1995) proposed a skew absolute value <2 and a kurtosis value <7 to be a departure from normality, in other words, values less than two for skewness and less than seven for kurtosis would be considered within the acceptable level of normality. Table 12 summarizes skewness and kurtosis of the variables in this study for tested knowledge and self-assessed knowledge. Both are within this range prescribed by West et al. (1995). Therefore, the sample for this study was considered to have a normal distribution for ANOVA statistical analysis purposes. Tukey's Honest Significant Difference (HSD) is a post-hoc test used to determine which groups differ from each other. Tukey's test is conducted when an ANOVA indicates a significant statistical difference. This test was used to identify pairwise comparisons for significant differences with ANOVA analyzes.

Analysis of Dependent Variables Skewness and Kurtosis

Dependent Variable							Shapiro-
(Knowledge)	Mean	SD	max	min	skewness	kurtosis	Wilk
Tested	14.93	2.51	5.0	18	-1.4917	2.2967	<0.0001
Self-assessed	5.56	1.29	1.2	7	-1.1268	1.4424	<0.0002
N=156							

Summary

The purpose of this study was to assess the financial literacy continuing professional education cognitive needs of Florida small business owners. This chapter covers the steps utilized in this study to design, validate, analyze the study, as well as develop, validate, and administer the instrument. The instrument used in this study was based on 14 sources of financial literacy questions and included three parts (demographic data, tested knowledge, and self-assessed knowledge).

The development of the instrument was based on expert panels and field-testing. Clients of the West Central Region of FSBDCN were sampled. Qualtrics surveys were used to gather responses from the expert panels' evaluations as well as field-test and study participants. Once data were collected, they were exported to EXCEL and analyzed using SAS.

Chapter 4

Findings

The purpose of this study was to assess the financial literacy continuing professional education cognitive needs of Florida small business owners. This chapter contains the research questions, study participants, research question one, research question two, research question three, research question four, research question five, and observations.

Research Questions

Based on prior research, this study attempted to investigate the following

research questions:

- 1. What is the financial literacy profile of the Florida small business owners?
- 2. What are the direction and extent of differences in the financial literacy profile by gender?
- 3. What are the direction and extent of differences in the financial literacy profile by

age?

- 4. What are the direction and extent of differences in the financial literacy profile by education level?
- 5. What are the direction and extent of differences in the financial literacy profile by small business classification?

Study Participants

Response rates of participants. Respondents for this study were from the

clients for the Florida West Central region of the Florida Small Business Development

Center Network. Approximately 12,000 emails with a link to the study survey were sent to active clients from over the past five years. There were four rounds of emails distributed to the list. The four emails resulted in 237 recipients clicking through the email to the survey. Sixty-seven recipients clicked through the first email sent November 4, 2015. The second email was sent November 18, 2015, and resulted in an additional 52 clicks to the survey. The third email, sent December 2, 2015, resulted in 59. The fourth, and final, email was sent December 15, 2015, contributed another 59 for a total of 237 recipients who clicked through to the survey. Although the recipient response level continued to be high after the fourth email, further emails were discontinued due to restrictions based on the agreement with the FSBDCN. Of the 237 individuals who clicked through the email to the survey, 35 did not start the survey. Of those, starting the survey (202), only 177 completed the survey. Of the 177 completing the survey, there were 34 who did not provide a year of birth and/or did not complete the self-assessed questions, leaving 147 complete responses. An additional nine responses were included from the field-test survey (two of the 11 field-test responses were incomplete), bringing the final sample to 156. The Listwise Deletion method was used to discard the cases with incomplete information (Cheema, 2014).

The actual response rate for complete surveys was 1.475%, which is low. There were several possible contributing factors. Emails were from clients dated back five years. It is possible many emails were no longer valid. FSBDCN did not provide the researcher with information on email bounce or fail rates. It is also possible previous clients included on the list were no longer in business, and therefore, did not respond to emails. Additionally, many FSBDCN clients may seek consulting services when considering a business, but never actually start a business. Finally, if clients only met 103

with the FSBDCN once several years ago, they may not recall or recognize the organization and would not have responded to the emails.

Demographic profile of respondents. Five questions provided demographic data related to the four independent variables of this study: gender, age, education level, and small business classification, derived from years in business and number of employees. Table 13, Demographic Characteristics of Respondents in the Survey, summarizes the demographics of the respondents within each variable. There were more female respondents in the group than males. Over 60% of respondent ages were between 41 years of age and 60 years of age. The mean age was 51, and the median age was 52. Level of education was predominately Bachelor's degree and higher (71.8%), and the minimum education level was high school. There were no participants who reported they did not complete high school. The number of years in business, a determinate of small business classification, disclosed more respondents in business less than five years (56.4%) while the classification small business (less than five years and less than five employees) contained 59% of the respondents.

Analysis of Research Questions

Research question 1. What is the financial literacy profile of the Florida small business owners? The financial literacy survey resulted in two measures of financial literacy: tested financial knowledge and self-assessed financial knowledge. Tested knowledge was based on 18 questions covering various concepts in financial literacy (Appendix S). Self-assessed knowledge was based on responses to five questions related to the respondents' agreement with statements rating their belief or confidence in their knowledge of financial literacy. A 7-point Likert scale with selections ranging from *strongly disagree* to *strongly agree* was used (Appendix S).

Demographic Characteristics of Respondents in the Survey

Variable	п	%*
Gender:		
Males	69	44.2
Females	87	55.8
Age Group:		
18-40 years	23	14.7
41-50 years	46	29.5
51-60 years	53	34.0
61 years and over	34	21.8
Education:		
High School	23	14.7
Associates Degree	21	13.5
Bachelor Degree	62	39.7
Master's Degree or Higher	50	32.1
Years in Business:		
0-2 years	57	36.5
3-5 years	31	19.9
6-10 years	19	12.2
>10 years	49	31.4
Small Business Classification:		
Pre-Venture/Start-up	35	22.4
Small Business (< 5 employees)	92	59.0
Small-Medium Enterprise >5 employees	29	18.6

N=156; *% may not = 100 due to rounding

Analyses of the respondents' tested knowledge and self-assessed knowledge are summarized in Table 14. The number of correct responses to the tested knowledge portion of the survey ranged from 5 to 18 (out of 18) with a total mean 14.9. Selfassessed knowledge ranged from 1.2 to 7.0 out of 7.0 with the mean of 5.6.

Descriptive Analysis Profiling the Financial Literacy of Florida Small Business Owners

Variable	\overline{X}	SD	95% Confidence Leve	
Tested knowledge	14.9	2.5	14.54	15.33
Self-assessed knowledge	5.6	1.2	5.37	5.71
N=156				

The correlation between the tested knowledge and self-assessed knowledge using *z*-scores and Pearson Correlation Coefficient was .5065. Histograms showing the distribution of tested knowledge scores and self-assessed knowledge scores are presented in Figures 3 and 4.



Tested Knowledge Scores

Figure 3. Histogram of tested knowledge score percentage distribution.





Figure 4. Histogram of self-assessed knowledge score percentage distribution.

Research question 2. What are the direction and extent of differences in the financial literacy profile by gender? This research question sought to identify the differences in financial literacy that were influenced by gender. The mean financial knowledge score for males was 15.4 with a standard deviation of 1.81 while the mean for females, was 14.6 with a standard deviation of 2.9. For self-assessed scores, the mean for males was 6.0 with a standard deviation of .84, and for females, the score was 5.2 with a standard deviation of 1.28. The descriptive statistics are summarized in Table 15. Because there are two groups within gender, an independent *t*-test statistic was used to evaluate statistical differences.

	(Tested Kno Corre Total possik	Self-Ass Financial (Max So	Self-Assessed Financial Literacy (Max Score 7)	
Variable	n	\overline{X}	SD	\overline{X}	SD
Males	69	15.4	1.81	6.0	0.84
Females	87	14.6	2.90	5.2	1.28
N=156					

Descriptive Analysis Profiling the Financial Literacy by Gender

In order to conduct usable t tests, the soundness of four assumptions needed to be evaluated: independence, normality, and homogeneity. For this study, the two groups, male and female, were independent of each other. Normality was tested using the Shapiro-Wilk test indicating a statistical difference, p<.0001. A review of the skewness and kurtosis (see Table 12) confirms normality was within an acceptable range. Heterogeneity was confirmed using Levene's Test indicating a statistically significant result of p < .05 (tested knowledge p = .0076; self-assessed knowledge p<.0001). From these results, equal variance could not be assumed, and the Satterthwaite p value was used instead of the pooled statistic in the SAS output. The t test conducted for the difference in means by gender disclosed a significant result for tested knowledge, t(1,154)=2.29, p=.0237 and self-assessed knowledge, t(16,154)=5.29, p=.0001. The effect size measured by Cohen's d for tested knowledge was .339, which was between small and medium. Cohen's d for self-assessed knowledge was .739, which was large. The mean tested knowledge and self-assessed knowledge score differences by gender were statistically significant. Male mean scores for both the tested and self-assessed knowledge were higher than female mean scores. **Research question 3.** What are the direction and extent of differences in the financial literacy profile by age? This research question sought to identify any differences in financial literacy that were influenced by age. The descriptive analysis by age group shows 18-40 years at a 14.9 mean for tested knowledge with a standard deviation of 2.42. The age group 41-50 years had a mean tested knowledge score of 14.7 with a standard deviation of 2.39. The 51-60 years age group mean tested knowledge score was 14.9, with a standard deviation of 2.92, and 61 years and older tested knowledge score was 15.4 with a standard deviation of 2.03. The mean self-assessed knowledge score for the 18-40 years age group was 5.6 with a standard deviation of 1.19; for 41-50 years, the mean self-assessed score was 5.5 with a standard deviation of 1.09. The mean self-assessed score for 51-60 years was 5.4, with a standard deviation of 1.32, and 61 years and over had a self-assessed score of 6.1 with a standard deviation of .79. The descriptive statistics by age group are summarized in Table 16.

Table 16

	(1	Tested Kno Corre Total possit	Self-Ass Financial (Max So	Self-Assessed Financial Literacy (Max Score 7)	
Variable	n	\overline{X}	SD	\overline{X}	SD
Age Group:					
18-40 years	23	14.9	2.42	5.2	1.19
41-50 years	46	14.7	2.39	5.5	1.09
51-60 years	53	14.9	2.92	5.4	1.32
61 years and Over	34	15.4	2.03	6.1	0.79
N-156					

Descriptive Analysis Profiling the Financial Literacy by Age Group

N=156

A one-way ANOVA test was conducted to assess the direction and extent of differences in the financial literacy profile by age. The result of the tested knowledge ANOVA test, summarized in Table 17, was F(3.152)=0.51, p=.6778 with an effect size, as measured by the eta squared of .01. Levene's test results for homogeneity of variance for tested knowledge (p=.44) indicated non-significance (p>.05). The Shapiro-Wilk test indicated the absence of normality. The results of a review of the skewness and kurtosis (Table 12) indicated normality was within an acceptable range. The results of the ANOVA suggested there was no statistical significant difference in the tested knowledge means across the four age groups. The self-assessed knowledge ANOVA test results, summarized in Table 18, were F(3.152)=3.02, p=0.0318 with an effect size of .06.

Table 17

ANOVA Summary Table of Tested Knowledge for Age Group						
	Sum of		Mean			
Source	Squares	df	Square	F	р	
Between	9.69	3	3.22	0.51	0.678	
Within	967.67	152	6.37			
Total	977.36	155				
<i>N</i> =156; <i>eta squared</i> = <i>.01</i> ; α = <i>.</i> 05						

Table 18

ANOVA Summary Table of Self-assessed Knowledge for Age Group

	Sum of		Mean		
Source	Squares	df	Square	F	р
Between	12.39	3	4.13	3.02	0.0318
Within	208.17	152	1.37		
Total	220.56	155			
N/ 450 /	1 00	05			

N=156; *eta* squared = .06; α = .05

The tested knowledge effect size is considered small while the self-assessed knowledge effects size is medium. Levene's test results for homogeneity of variance for self-assessed knowledge indicated non-significance self-assessed p=.06. Although the Shapiro-Wilk test indicated the absence of normality, skewness and kurtosis (Table 12) were within an acceptable range. The ANOVA test for self-assessed knowledge indicated there was a statistically significant mean difference. Tukey's test for self-assessed knowledge, summarized in Table 19, indicated statistical significant between the 18 to 40 and the 61 years and over age groups. The mean for 18 to 40 years (mean=5.6) was less than 61 years and over (mean=6.1).

Table 19

		Mean Differences $(\overline{X}_i - \overline{X}_k)$				
Group	Mean	18-40	41-50	51-60	61 and Over	
18-40 years	5.2	0.00				
41-50 years	5.5	0.33	0.00			
51-60 years	5.4	0.18	0.15	0.00		
61 years and Over	6.1	0.84**	0.51	0.66	0.00	

Tukey's Test Post Hoc Results of Self-assessed Knowledge by Age Group

N=156; ***p*<.0001

Research question 4. What are the direction and extent of differences in the financial literacy profile by education level? This research question sought to identify the differences in financial literacy that were influenced by education level. The descriptive analysis by education level indicated the mean tested knowledge score for high school level of education was 13.5 with a standard deviation of 3.88. The associates degree mean tested knowledge score was 14.7 with a standard deviation of

2.11. Bachelor's degree level of education mean score was 14.9, with a standard deviation of 2.40, and master's degree or higher mean score was 15.8, with a standard deviation of 1.5. The mean self-assessed score for high school education level was 5.2 with a standard deviation of 1.47. The associates degree education level mean self-assessed score was 5.5 with a standard deviation of 1.05, and bachelor's degree mean score was 5.4 with a standard deviation 1.32. For the master's degree or higher, the mean score was 6.0 with a standard deviation of .78. The descriptive analysis by education level is summarized in Table 20.

Table 20

		Tested Knowledge Correct (Total possible = 18)		Self-Assessed Financial Literacy (Max Score 7)	
Variable	n	\overline{X}	SD	\overline{X}	SD
Education:					
High School	23	13.5	3.88	5.2	1.47
Associates Degree	21	14.7	2.11	5.5	1.05
Bachelor Degree	62	14.9	2.40	5.4	1.32
Master's Degree or Higher	50	15.8	1.55	6.0	0.78

Descriptive Analysis Profiling the Financial Literacy by Education Level

N=156

A one-way ANOVA test was conducted to assess the direction and extent of differences in the financial literacy by education level. The results of the tested knowledge ANOVA test, summarized in Table 21, were F(3.152)=5.01, p=0.0024 with an effect size, as measured by eta squared, of .09 (medium). Levene's test for tested knowledge scores resulted in statistical significance (p<.0001) and, therefore, confirms heterogeneity.

	Sum of		Mean			
Source	Squares	df	Square	Welch F	р	
Between	87.98	3	29.33	4.60	0.0024	
Within	889.37	152	5.85			
Total	977.36	155				
N=156; eta squared = 00; $\alpha = 05$						

ANOVA Summary Table of Tested Knowledge for Education Level

N=156; eta squared = .09; α = .05

Because the heterogeneous and *n* were unequal, the Welch test was used to determine the Welch F value of 4.60. Again, the Shapiro-Wilk test indicated the absence of normality. The results of a review of the skewness and kurtosis (Table 12) indicated normality was within an acceptable range.

The ANOVA suggested statistically significant mean differences across the four education level groups for tested knowledge existed. Following up with Tukey's test for tested knowledge summarized in Table 22, statistical significance was indicated between high school and master's degree and higher. The high school mean (13.5) was lower than the master's degree and higher mean (15.8). The self-assessed knowledge ANOVA test results, summarized in Table 23, were F(3.152)=2.35, p=.0746with an effect size of .04 (small). Levene's test for tested knowledge scores resulted in statistical significance (p<.0001) and, therefore, confirms heterogeneity. Because the heterogeneous and n were unequal, the Welch test was used to determine the Welch F value of 2.40. As before, the Shapiro-Wilk test indicated the absence of normality, but the results of a review of the skewness and kurtosis (Table 12) were within an acceptable range. This suggested no statistically significant mean differences across the four education level groups for self-assessed knowledge.

		Mean Differences $(\overline{X}_i - \overline{X}_k)$				
	-				Master's	
		High	Associates	Bachelor's	Degree &	
Education Level Group	Mean	School	Degree	Degree	Higher	
High School	13.5	0.00				
Associates Degree	14.7	1.18	0.00			
Bachelor's Degree	14.9	1.39	0.20	0.00		
Master's Degree and Higher	15.8	2.32**	1.13	0.93	0.00	

N=156; ***p*< .0001

Table 23

ANOVA Summary Table of Self-assessed Knowledge for Education Level

	Sum of		Mean		
Source	Squares	df	Square	Welch F	p
Between	9.77	3	2.26	2.40	0.0746
Within	210.78	152	1.39		
Total	220.56	155			
	1	<u> </u>			

N=156; *eta* squared = .04; α = .05

Research question 5. What are the direction and extent of differences in the financial literacy profile by small business classifications? This research question sought to identify any differences in financial literacy that were influenced by small business classification. Descriptive statistics by small business classification for the mean tested knowledge score were 14.0 for pre-venture/start-up, 15.1 for small business, and 15.7 for small-medium enterprises, with the standard deviations of 2.91, 2.39, and 2.09 respectively. The mean self-assessed scores were 5.1 for pre-venture/start-up, 5.6 for small business and 6.0 for small-medium enterprise, with 1.45,

1.11, and .95 standard deviations respectively. Descriptive statistics by small business classification are summarized in Table 24. A one-way ANOVA test was conducted to assess the direction and extent of differences in the financial literacy profile by business classification. The results of the tested knowledge ANOVA test, summarized in Table 25, were F(2.153)=3.88, p=.0227 with an effect size, as measured by eta squared of .05 (medium). Levene's test results for homogeneity of variance for tested knowledge indicated no statistical significance, p=.3967. Once again, the Shapiro-Wilk test indicated the absence of normality. The results of a review of the skewness and kurtosis (Table 12) indicated normality was within an acceptable range.

Table 24

	Tested Knowledge Correct (Total possible = 18)		wledge ct ble = 18)	Self-As Financial (Max S	sessed Literacy core 7)
Variable	n	\overline{X}	SD	\overline{X}	SD
Small Business Classification:					
Pre-Venture/Start-up	35	14.0	2.91	5.1	1.45
Small Business	92	15.1	2.39	5.6	1.11
Small-Medium Enterprise	29	15.7	2.09	6.0	0.95
N=156					

Descriptive Analysis Profiling the Financial Literacy by Small Business Classification

Table 25

ANOVA Summary Table of Tested Knowledge for Small Business Classification

_	Sum of		Mean		
Source	Squares	df	Square	F	p
Between	47.20	2	23.60	3.88	0.0227
Within	930.16	153	6.08		
Total	977.36	155			
		05			

N=156; *eta* squared = .05; α = .05

Tukey's test, summarized in Table 26, pointed to statistically significant differences between pre-venture/start-up and small-medium enterprise and no statistically significant differences between small business and pre-venture/start-up or smallmedium enterprise.

Table 26

Tukey's Test Post Hoc Results of Tested Knowledge by Small Business Classification

		Mean Differences $(\overline{X}_i - \overline{X}_k)$				
Small Business		Pre-Venture/	Small	Small-Medium		
Classification Group	Mean	Start-up	Business	Enterprise		
Pre-Venture/Start-up	13.4	0.00				
Small Business	15.1	1.07	0.00			
Small-Medium Enterprise	15.7	1.66**	0.59	0.00		
N=156; **p< .0001						

The self-assessed knowledge ANOVA test results summarized in Table 27 was F(2.153)=4.29, p=.00154 with an effect size of .05 (medium). Levene's test results for homogeneity of variance for tested knowledge indicated no statistical significance, p=.0927. The Shapiro-Wilk test indicated the absence of normality, but the results of a review of the skewness and kurtosis (Table 12) were within an acceptable range. This suggested there was a significant mean difference across the three small business classifications for self-assessed knowledge. Performing Tukey's tests, summarized in Table 28, pointed to statistical significance between pre-venture/start-up and small-medium enterprise, and no statistical significance between small business and pre-venture/start-up or small-medium enterprise.

ANUVA SUMMARY TADIE OF SEMI-ASSESSED KNOWLEDDE TOF SMAIL BUSINESS CLASSIFICATION
--

	Sum of		Mean		
Source	Squares	df	Square	F	р
Between	11.71	2	5.85	4.29	0.0154
Within	208.85	153	1.37		
Total	220.56	155			
NI 4EO ULU ULU	05 05				

N=156; *eta* squared = .05; α = .05

Table 28

Tukey's Test Post Hoc Results of Self-assessed Knowledge by Small Business Classification

		Mean Differences $(\overline{X}_i - \overline{X}_k)$			
Small Business		Pre-Venture/	Small	Small-Medium	
Classification Group	Mean	Start-up	Business	Enterprise	
Pre-Venture/Start-up	5.0	0.00			
Small Business	5.6	0.49	0.00		
Small-Medium Enterprise	6.0	0.84**	0.36	0.00	
$M = 156 \cdot ** n < 0001$					

N=156; ***p*< .0001

Observations

An observation worth noting during this study was the interest by respondents in the results of the research. At the end of the survey, the respondents were given an opportunity to provide an email to receive information about the results of the research. Of the 177 respondents who completed the survey, 108 (61%) expressed an interest in the results. It appears from the expressed interest of the respondents, small business owners have an interest in their own financial literacy comparable to the researcher's aim to isolate small business owners' financial literacy from the general population in order to provide stakeholders of the FSBDC network evidence of a need for continuing professional education.

Approximately 15% of the surveys completed had incomplete data, which removed them from the statistical analysis. Excluding the year of birth represented the primary omission, 12.7%. In most cases, the actual year of birth was omitted, but in a few cases, a location was recorded instead of a year. In future research, consideration might be given to designing this demographic question in a manner that would not allow omitting the data. Whether a drop-down menu is used and/or a programmed mandatory response, omitting the birth year voided a substantial amount of useful data.

During the reflective cognitive interviews, commentary from some of the participants revealed life experience and/or stage of life as the reason they did not know correct answers to questions. For example, one participant had never owned a home. Therefore, any questions related to mortgages and mortgage interest were not useful information and were not part of the financial knowledge. Another participant was of retirement age and was no longer concerned about borrowing or credit reporting. For this individual, questions related to these concepts were not relevant, and they were not sure of the correct answer. These comments put forth the notion that financial literacy knowledge might require considering more than merely learning concepts. The importance of the concept in the individual's life might be crucial.

There were 32 expert panel members with backgrounds in education, accounting, financial planning, banking, or financial literacy. All varied in educational degrees and amount and type of experience with financial literacy (see Table 8). Instructions for panel review did not include a definition of financial literacy because the researcher wanted to develop the instrument based on a real-world perspective of

financial literacy, rather than confining experts to an academic definition of the construct. None of the experts requested a definition to complete the panel reviews. All the experts concentrated on money basics rather than a balance between all four financial literacy categories. The coverage of money basics in the final survey was 68%.

Of the 14 surveys selected for this study, questions were used from nine. The most interesting observation was that only one question was selected from the National Financial Capacity Survey (NFCS). The NFCS has been a primary source of data for research on U.S. financial literacy. The NFCS survey is limited to five financial literacy knowledge questions. It has been suggested that it is the benchmark for financial literacy assessment (Lusardi & Mitchell, 2011b); however, only one question relating to mortgages was selected by the experts for this study's survey. Other researchers have proposed NFCS survey is too narrow in focus to concentrate on investing (building assets) concepts. The expert panels' final financial literacy question choices appeared to support this critique of NCFS.

In the four rounds of email send to FSBDCN recipients, the number of respondents to click through to the survey was strong in every round. The first email received 67 clicking through; the second email was 52, and the third email was 59. Initially, the FSBDCN agreed to send three emails to gather data so as not to interfere with the FSBDCN annual survey conducted at the beginning of every year. After seeing the results of the first three emails, a request was made to send one more, the fourth. The request was approved, and an additional 59 recipients clicked through to the survey. It is not known whether this strong response was influenced by the nature of the recipients (small business owners) or the subject of the survey (financial literacy).

Financial literacy concepts in the literature were placed into four groups: money basics, borrowing, building assets, and protecting assets. Results of the exploratory factor analysis indicated only three groups for the items in the instrument created in this study. The group themes were: money basics, money concepts (understanding of definitions or concepts), and money managing. It appears that rather than viewing the financial literacy construct from the perspective of difference subject concept groups, financial literacy should be defined by the activity (i.e., knowledge and understanding, basic concepts of using money, managing money).

Chapter 5

Summary, Conclusions, Implications, and Recommendations

The purpose of this study was to assess the financial literacy continuing professional education cognitive needs of Florida small business owners. The parts of this chapter include a summary, conclusions, implications, and recommendations for further research.

The following research questions guided this study:

- 1. What is the financial literacy profile of the Florida small business owners?
- 2. What are the direction and extent of differences in the financial literacy profile by gender?
- 3. What are the direction and extent of differences in the financial literacy profile by age?
- 4. What are the direction and extent of differences in the financial literacy profile by education level?
- 5. What are the direction and extent of differences in the financial literacy profile by small business classification?

Summary

Financial literacy has been a topic of concern around the world (Lusardi & Mitchell, 2011a, 2011b, 2013, 2014). In the U.S., it has captured the attention of the President and numerous public and private agencies as well as the business world (PACFC, 2013a, 2013b; PACFL, 2009; Schwab, 2009). This research examined financial literacy as it related to establishing the continuing professional education cognitive needs of Florida small business owners.

The research consisted of two components: (a) development of a financial literacy assessment instrument and (b) survey of West Central Florida small business owners. The instrument was developed using a five-step process with an expert panel of 32 professionals whose backgrounds were in education, accounting, financial planning, banking, or financial literacy. After review and evaluation, the panel reduced a pool of 160 tested knowledge and 8 self-assessed knowledge financial literacy questions to the final instrument containing 18 tested knowledge and 5 self-assessed questions. After a reflective cognitive review and field testing, the instrument was administered via an online survey which was emailed four different times to clients of the Florida West Central region of the Florida Small Business Development Center Network at the University of South Florida. The resulting data provided a financial literacy profile of Florida small business owners, and any significant differences in financial literacy within the four variables considered.

Conclusions

Based on the results of this study, the conclusions that accrued from this study are as follows.

The small business owners' financial literacy profile reflects a high understanding of financial literacy. All scores for tested knowledge and self-assessed knowledge were high. Small business owners have knowledge as well as confidence in their understanding of financial literacy.

The mean financial literacy score of males was higher than females. Both tested knowledge and self-assessed knowledge scores were greater for males than females. The results of this research were similar to those of other studies conducted globally

(Allgood & Walstad, 2013, Atkinson & Messey, 2012; GFLEC, 2016b; Lusardi & Mitchell, 2008, 2011a, 2011b, 2014; Lusardi & Wallace, 2013; Mottola, 2013).

Age difference did not contribute to differences in tested knowledge financial literacy scores, except between the young (18 to 40 year olds) and elderly (61 years and older) age groups. Older small business owners, over the age of 61 years scored slightly higher than those between the ages of 41 years and 60 years old. Younger small business owners, 18 to 40 years of age, scored slightly lower than those between the ages of 41 years and 60 years old.

Age was a contributor to differences in confidence with financial literacy knowledge. Confidence, based on in the self-assessed scores, increased with age. There were notable differences in the age group self-assessed knowledge scores. Small business owners over the age of 61 years had the highest scores, indicating the most confidence in their financial literacy knowledge. Younger small business owners, aged 18 to 40 years, had the lowest level of confidence. Middle-aged small business owners, ages 41 to 60 years, were in the middle.

Education levels also impacted financial literacy knowledge. The greater the level of education obtained by a small business owner, the higher the scores of financial literacy knowledge in agreement with previous research (Atkinson & Messey, 2012; GFLEC, 2016b, Lachance, 2014). Self-assessed scores also increased with the education level, but confidence in financial literacy knowledge was not as strongly impacted by education levels as was tested knowledge.

Pre-venture/start-up business owners had the lowest tested and self-assessed knowledge scores in small business classifications. Knowledge scores improved with

business owners who had more time running a business. Business owners in the smallmedium enterprise business classification had the most confidence and highest tested knowledge.

Implications

The findings of this study may have continuing professional education implications for private businesses and government agencies providing educational programs, entrepreneurship programs at Florida universities, Florida SBDC stakeholders, Florida SBDC consultants, small business owners, and the Florida legislature.

Private businesses and government agencies. Implications of this research indicate the need for organizations providing educational programs for small business owners to consider addressing the lower levels of financial literacy (both tested and selfassessed) realized by the younger age group (18 to 40 years) and less experienced business owners (in business less than three years). Additionally, women of all ages and business experience might benefit from programs specifically targeted at improving their financial knowledge and confidence level. In addition to educational programs for women, these organizations might consider creating or supporting volunteer outreach programs to provide mentoring and consulting.

Entrepreneurship programs. University entrepreneurial programs could consider providing education (formal and informal, such as at the SBDC) addressing the need for financial literacy education for younger individuals. These programs might also be designed to provide varying levels of education, rather than a generic single program for all small business owners. With this in mind, younger start-up and newer business owners would be provided a basic level of education, while older experienced business owners could have access to a more challenging program of education.

Entrepreneurship programs could also consider supporting education at the high school level aiming to improve financial literacy prior to starting college or a small business.

Small Business Development Center (SBDC) stakeholders. In supporting the SBDC, stakeholders might encourage educational programs to be directed towards improvement of financial literacy for groups falling below the average level of small business owner financial literacy (e.g., younger women start-up business owners). Stakeholders, such as financial institutions, might also consider participating in the programs they support to create a greater source of knowledge for small business owners.

Small Business Development Center (SBDC) consultants. SBDC

consultants may consider a new client's profile (gender, age, education level, and business classification) when responding to the client's requests. This could help the consultants improve understanding of the client's potential financial literacy needs. With a better understanding, consultants' ability to direct new clients to the best possible continuing professional education programs in support of the small business owners' attempts to build and grow their businesses might be enhanced. SBDC consultants should consider evaluating their own level of financial literacy and improve any areas where their knowledge or confidence might be lacking.

Small business owners. With an awareness of their own financial literacy profile, small business owners can assess personal strengths and weakness and seek opportunities to increase their own financial literacy. Aided by guidance from

knowledgeable empathetic consultants, small business owners' confidence in their financial literacy can also be improved, resulting in more effective businesses in the state of Florida.

Florida Legislature. In 2013, the Florida Legislature passed House Bill No. 2007 allocating four million dollars to the FSBDCN to provide support for Florida small business owners (K-20 Education, 2013). As the legislature continues to distribute these dollars, it behooves lawmakers to understand the small business owners' financial literacy profile. With this knowledge, the legislature can provide dedicated resources to financial literacy initiatives within the budget to address small business owners' financial literacy needs.

Recommendations for Further Research

Based on the findings from this research, there are several recommendations for further study:

1. This study was limited to small business owners in the West Central region of Florida. Conducting research using the entire state of Florida for sample selection would allow evaluation of different regions to confirm whether the same financial literacy profile exists throughout the state.

2. This study focused on Florida small business owners' financial literacy. Conducting research with this study's instrument on a state-by-state basis would facilitate evaluating whether small business owners in other states have similar or different financial literacy profiles. Since Florida ranks third in the U.S. for the number of small business, behind California and Texas, particular attention could be given to these three states (SBA, 2012).

3. The literature review revealed the lack of a consensus of an instrument to assess financial literacy. The questions used in the National Financial Capabilities Study (NFCS) have been called a benchmark for financial literacy assessment (Lusardi & Mitchell, 2011b). These questions have also been criticized as too narrow in focus and too short (Finke & Huston, 2014; Huston, 2010); Nicolini et al., 2013). An exploration of correlations between responses to this study's assessment and the NFCS assessment from the same participants could provide new insights into past evaluations of financial literacy.

4. Significant gender differences were one of the primary findings in most of the literature (Allgood & Walstad, 2013, Atkinson & Messey, 2012; GFLEC, 2016b; Lusardi & Mitchell, 2008, 2011a, 2011b, 2014; Lusardi & Wallace, 2013; Mottola, 2013). Time and time again, females score lower than males. Exploring gender differences in more quantitative detail together with qualitative research for each variable (age, education level, and business classification) might give better insight reasons or causes for the differences. This could improve tailoring of continuing professional educational programs to women's needs.

5. Further research to investigate the relationship between the tested knowledge score and self-assessed score might assist in identifying correlations, including possible predictability, between the two.

6. The model developed for this study identified three financial literacy domains: cognitive, affective, and social. Further exploration of the affective and social domains of financial literacy, including possible correlations or interdependencies, might provide a view of the whole person's financial literacy, not just the intellect.

7. The instrument in this research focused on basic financial literacy (money basics, borrowing, building assets, and protecting assets). There were several concepts excluded from the instrument based on the expert panel's recommendation. Some of these concepts were identified: investing, investment analysis, and economics. Identifying the next level financial literacy might be helpful in developing continuing profession education for more experienced small business owners.

References

- Agnew, J., Bateman, H., & Thorp, S. (2013). Financial literacy and retirement planning in Australia. *Numeracy, 6*(2), Article 7. doi:10.5038/1936-4660.6.2.7
- Agnew, S., & Cameron-Agnew, T. (2015). The influence of consumer socialization in the home on gender differences in financial literacy. *International Journal of Consumer Studies, 39*(6), 630-639. doi:10.1111/jcs.12179
- Alhenawi, Y., & Elkal, K. (2013). Financial literacy of U. S. households: Knowledge vs. long-term financial planning. *Financial Services Review, 22*(3), 1-43. Retrieved from http://papers.ssrn.com/sol3/papers.cfm?abstract_id=2532068
- Allgood, S., & Walstad, W. B. (2012). The effects of perceived and actual financial literacy on financial behaviors. SSRN Electronic Journal. doi:10.2139/ssn.2191606
- Allgood, S., & Walstad, W. B. (2013). Financial literacy and credit card behaviors: A cross-sectional analysis by age. *Numeracy*, *6*(2), Article 3. doi:10.5038/1936-4660.6.2.3
- Allgood, S., & Walstad, W. B. (2016). The effects of perceived and actual financial literacy on financial behaviors. *Economic Inquiry*, 54(1), 675-697. doi:10.1111/ecin.12255
- Altman, M. (2012). Implications of behavioral economics for financial literacy and public policy. *Journal of Socio-Economics*, 41(5), 677-690. doi:10.1016/j.socec.2012.06.002
- American Educational Research Association. (2015). *Professional ethics*. Retrieved from http://www.aera.net/AboutAERA/AERARulesPolicies /ProfessionalEthics/tabid/10200/Default.aspx
- American Savings Education Council. (n.d.). *What is ASEC?* Retrieved from http://www.choosetosave.org/asec/index.cfm?fa=about
- American Small Business Development Center (ASBDC). (2016). *Mission and vision*. Retrieved from http://americassbdc.org/about-us/mission-vision/
- Andrews, D., Nonnecke, B., & Preece, J. (2003). Conducting research on the internet: Online survey design, development and implementation guidelines. *International Journal of Human-Computer Interaction*, 16(2), 185-210. Retrieved from http://auspace.athabascau.ca/bitstream/214

- Ansong, A. (2011). Level of knowledge in person finance by university freshman business students. *African Journal of Business Management, 5*(22), 8933-8940. doi:10.5897/AJBM11.483
- ANZ Bank, Department of Financial Inclusion & Capability. (2008). ANZ survey of adult financial literacy in Australia (Social Research Center Report). Retrieved from http://www.anz.com/resources/5/4/54a7b400413360d8 b5d8bda2fd298cdf/Survey-Adult-Financial-Literacy-2008.pdf
- ANZ Bank, Department of Financial Inclusion & Capability. (2011). Adult financial literacy in Australia (Social Research Center Report). Retrieved from http://www.anz.com/resources/2/8/284e81004f1d3e80a555b558b54e5b8d/Makin g-Difference-Financial-Literacy.pdf?MOD=AJPERES
- Applied Research & Consulting LLC. (2016). Retrieved from http://www.arcllc.com
- Arrondel, L., Debbich, M., & Savignac, F. (2013). Financial literacy and financial planning in France. *Numeracy*, *7*(2), Article 8. doi:10.5028/1036-4660.6.2.8
- Ary, D., Jacobs, L.C., & Sorensen, C. (2010). *Introduction to research in education.* Belmont, CA: Wadsworth.
- Atkins, A., McKay, S., Collard, S., & Kempson, E. (2010). Levels of financial capability in the UK. *Public Money & Management*, *27*(1), 37-41. doi:10.1111/j.1467-9302.2007.00552.x
- Atkinson, A., & Messy, F. A. (2012). *Measuring financial literacy* (Finance, Insurance and Private Pension Working Paper No. 15). Retrieved from Organisation for Economic Co-operation and Development (OECD) website, http://dx.doi.org/10.1787/ 5k9csfs90fr4-en
- Atwater, J. (n.d.). Your money matter\$: Small business. Retrieved from http://www.myfloridacfo.com/YMM/smallBusiness/Default.aspx
- Atwater, J. (2013). 2013 Small business survey (Florida Chamber Foundation and Florida Small Business Development Center Report). Retrieved from Florida's Chief Financial Officer website, http://www.myfloridacfo.com/ floridasbottomline/docs/FBL2013_special_report.pdf
- Baker, T. (1998). Doing social research (3rd ed.). Boston, MA: McGraw-Hill.
- Barr, M. (2004). *Banking the poor: Policies to bring low-income mainstream.* (Research Brief). Washington, DC: The Brookings Institution. Retrieved from http://repository.law.umich.edu/law_econ_archive/art48/

- Beavers, A. S., Lounsbury, J. W., Richards, J. K., Huck, S. W., Skolits, G. J., & Esquivel, S. L. (2013). Practical considerations for using exploratory factor analysis in educational research. *Practical Assessment, Research & Evaluation,* 18(6), 1-13. Retrieved from http://pareonline.net/getvn.asp?v=18&n=6
- Beckmann, E. (2013). Financial literacy and household savings in Romania. *Numeracy,* 6(2), Article 9. doi:10.5038/1936-4660.6.2.9
- Bell, M. L. (2013). Florida financial literacy summit: A call for education. Tampa, FL: Florida Council for Economic Education. Retrieved from http://www.requirethemoneycourse.com/downloads/FinancialLiteracySummit-WhitePaper.pdf
- Bonte, W., & Filipiak, U. (2012). Financial literacy, information flows, and caste affiliation: Empirical evidence from India. *Journal of Banking & Finance, 36*(12), 3399-3414. doi:10.1016/j.jbankfin.2012.07.028
- Braunstein, S., & Welch, C. (2002). Financial literacy: An overview of practice, research and policy. *Federal Reserve Bulletin, 88*(11), 445-457. Retrieved from http://eds.a.ebscohost.com.ezproxy.lib.usf.edu/ehost/pdfviewer/pdfviewer?vid=2 &sid=d8b5125b-36ef-4664-977b-0faa3bcd3489%40sessionmgr4001&hid=4208
- Brown, M., & Graf, R. (2013). Financial literacy and retirement planning in Switzerland. *Numeracy, 6*(2), Article 8. doi:10.5038/1936-4660.6.2
- Buckland, J. (2010). Are low-income Canadians financial literacy? Placing financial literacy in the context of personal and structural constraints. *Adult Education Quarterly, 60*(4), 357-376. doi:10.1177/0741713609358449
- Buckland, J., Task Force on Financial Literacy. (2011). *Money management on a shoestring: A critical literature review of financial literacy and low income people*. Retrieved from http://www.publications.gc.ca/collections/collection_2011/fin/F2-220-2011-eng.pdf
- Buckland, J. (2014). Structural barriers, financial exclusion, and the possibilities of situated learning for financial education. In K. S. Forte, E. W. Taylor, & J. Tisdell (Eds.), *Special issues: Financial literacy and adult education* (pp. 15-24). New Directions for Adult and Continuing Education, no. 141. San Francisco, CA: Jossey-Bass. doi:10.1002/ace/20081
- Bumcrot, C., Lin, J., & Lusardi, A. (2013). The geography of financial literacy. *Numeracy, 6*(2), Article 2. doi:10.5038/1936-4660.6.2.2
- Carlson, J. P., Vincent, L. H., Hardesty, D. M., & Bearden, W. O. (2009). Objective and subjective knowledge relationship: A quantitative analysis of consumer research findings. *Journal of Consumer Research*, *25*(2), 864-876. doi:10.1086/593688

- Cheema, J. R. (2014). A review of missing data handling methods in education research. *Review of Educational Research, 84*(4), 487-508. doi:10.3102/0034654314532697
- Chen, H., & Volpe, R. P. (1998). An analysis of personal financial literacy among college students. *Financial Services Review*, 7(2), 107-128. Retrieved from http://eds.a.ebscohost.com.ezproxy.lib.usf.edu/ehost/pdfviewer/pdfviewer?sid=0b 30c5fc-1d80-4f00-9953-c85656183ac8%40sessionmgr4001&vid=0&hid=4111
- Cohen, J. (1992). A power primer. *Psychological Bulletin*, *112*(1), 155-159. Retrieved from http://classes.deonandan.com/hss4303/2010/cohen%201992%20 sample%20size.pdf
- Cole, S., Sampson, T., & Zia, B. (2009). Financial literacy, financial decision, and the demand for financial services: Evidence from India and Indonesia (Working Paper 09-117). Boston, MA: Harvard Business School, Harvard University. Retrieved from http://www1.worldbank.org.ezproxy.lib.usf.edu/prem/ poverty/ie/dime_papers/1107.pdf
- Collins, J. (2013). The impacts of mandatory financial education: Evidence from a randomized field study. *Journal of Economic Behavior & Organization, 95*(11), 146-158. doi:10.1016/j.jebo.2012.08.011
- Consumer Financial Protections Bureau. (2013). *Financial literacy annual report* (CPFB Annual Report). Washington, DC: Consumer Financial Protections. Retrieved from http://files.consumerfinance.gov/f/201307_cfpb_report_financial-literacyannual.pdf
- Costello, A. B., & Osborne, J. W. (2005). Best practices in exploratory factor analysis: Four recommendations for getting the most from your analysis. *Practical Assessment, Research, & Evaluation, 10*(7), 1-9. Retrieved from http://pareonline.net/getvn.asp?v=10&n=7
- Cover, J., Spring, A. F., & Kleit, R. G. (2011). Minorities on the margins? The spatial organization of fringe banking services. *Journal of Urban Affairs*, *33*(3), 317-344. doi:10.1111/j.1467-9906.2011.00553.x
- Crocker, L., & Algina, J. (2008). *Introduction to classical and modern test theory*. Mason, OH: Cengage Learning.
- Cronbach, L. J. (1951). Coefficient alpha and the internal structure of tests. *Psychometrika, 16*(3), 296-334. doi:10.1007/BF02310555
- Crossan, D., Feslier, D., & Hurnard, R. (2011). Financial literacy and retirement planning in New Zealand. *Journal of Pension Economics and Finance*, *10*(4), 619-635. doi:10.1017/S14747211000473
Crotty, M. (1998). The foundations of social research. Thousand Oaks, CA: Sage.

- Cude, B. J. (2010). Financial literacy 501. *The Journal of Consumer Affairs*, *44*(2), 271-275. doi:10.1111/j.1745-6606.2010.01168.x
- Cude, B. J., Kunovskaya, I., Kabaci, M. J., & Henry, T. (2013). Assessing changes in the financial knowledge of college seniors. *Consumer Interest Annual, 59*, 1-17. Retrieved from https://acci.memberclicks.net/assets/docs/CIA/CIA2013/ OralSess2013/assessing%20changes%20in%20the%20financial%20knowledge %20of%20college%20seniors%20-%20cude%20kunovskaya%20kabaci%20henr y.pdf
- Dahmen, P., & Rodriguez, E. (2014). Financial literacy and the success of small businesses: An observation from a Small Business Development Center. *Numeracy*, *7*(1), Article 3. doi:10.5038/1936-4660.7.1.3
- Davies, D. (1795). *The case of labourers in husbandry states and considered.* (Hathi Trust Digital Library). Retrieved from http://babel.hathitrust.org/cgi/pt?id=mdp. 39015082457022; view=1up;seq=7
- Delavande, A., Rohwedder, S., & Willis, R. (2008). *Preparation for retirement, financial literacy, and cognitive resources* (Working Paper No. 2008-190). Retrieved from University of Michigan Retirement Research Center website, http://www.investmentnews.com/assets/docs/CI580281021.PDF
- De Vaus, D. A. (2002). Analyzing social science data. Thousand Oaks, CA: Sage.
- Donohue, M. (2011). *Financial literacy and women: A mixed method study of challenges and needs* (Doctoral dissertation, University of Massachusetts). Retrieved from http://search.proquest.com.ezproxy.lib.usf.edu/pqdtft/docview/883078179/8D935 788B29F4D8DPQ/8?accountid=14745
- Drexler, A., Fischer, G., & Schoar, A. (2010). Keeping it simple: Financial literacy and rules of thumb. *London School of Economics Online*. Retrieved from http://eprints.lse.ac.uk/33502/1/__libfile_REPOSITORY_Content_Fischer%2C%2 0G_Keeping%20it%20simple_Fischer_Keeping%20it%20simple_2013.pdf
- Elias, J. L., & Merriam, S. B. (2005). *Philosophical foundations of adult education* (Rev. ed.). Marlabar, FL: Krieger.
- Executive Order 13520. (2010). President's Advisory Council on Financial Capability. *Federal Register*, 75(22), 5481-5483. Retrieved from http://www.whitehouse.gov/sites/default/files/2010financial.eo_.rel_.pdf

- Fabrigar, L. R., Wegener, D. T., MacCallum, R. C., & Strahan, E. J. (1999). Evaluating the use of exploratory factor analysis in psychological research. *Psychological Methods, 4*(3), 272-299. Retrieved from http://psycnet.apa.org.ezproxy. lib.usf.edu/journals/met/4/3/272/
- Fairlie, R. W., & Holleran, W. (2012). Entrepreneurship training, risk aversion and other personality traits: Evidence from a random experiment. *Journal of Economic Psychology*, 33(2), 366-378. doi:10.1016/j.joep.2011.02.001
- Fairlie, R. W., & Krashinsky, H. A. (2012). Liquidity constraints, household wealth, and entrepreneurship revisited. *Review of Income and Wealth*, 58(2), 279-306. Retrieved from University of California website, http://escholarship.org /uc/item/6hv0m2q6
- Fernandes, D., Lynch, J. G., & Netemeyer, R. G. (2014). Financial literacy, financial education, and downstream financial behaviors. *Management Science*, 60(8), 1861-1883. doi:10.1287/mnsc.2013.1849
- *Financial education core competencies: Comment request, 75* Fed. Reg. 165, 52597 (August 26, 2010). FR Doc. 2010–21305 Filed 8–25–10. Retrieved from http://www.treasury.gov/about/organizational-structure/offices/Domestic-Finance/Documents/FLEC%20-%20Core%20Competencies%20-%20FRN.pdf
- Financial Literacy and Education Commission (FLEC). (2010, November). *Why and how: Background report developing the 2011 national strategy.* Retrieved from United States Department of the Treasury website, http://www.treasury.gov/resource-center/financial-education/Pages/commission-index.aspx
- Financial Literacy and Education Commission (FLEC). (2011a). *Promoting financial success in the United States: National strategy for financial literacy.* Retrieved from United States Department of the Treasury website, http://www.treasury.gov/resource-center/financial-education/Documents/NationalStrategyBook_12310%20(2).pdf
- Financial Literacy and Education Commission (FLEC). (2011b). *Why and how:* Background report developing the 2011 national strategy (FLEC November 2011 Report). Retrieved from http://www.treasury.gov/resource-center/financialeducation/Documents/National%20Startegy%20Background.pdf
- *Financial Literacy and Education Commission (FLEC).* (2014, February 25). Retrieved from United States Department of the Treasury website, http://www.treasury.gov/resource-center/financial-education/Pages/commission-index.aspx
- *Financial Literacy Council*, FL. HB 825 § 2006-140. (2006). (enacted). Retrieved from http://laws.flrules.org/files/Ch_2006-140.pdf

- Finke, M. S. (2011). Old age and the decline of financial literacy. *Behavior determinants of household financial choices: Three essays* (Doctoral dissertation, University of Missouri). Retrieved from https://mospace.umsystem.edu/xmlui/ bitstream/handle/10355/14204/research.pdf?sequence=2
- Finke, M. S., Howe, J. S., & Huston, S. (2011). Old age and decline in financial literacy. SSRN Electronic Journal. doi:10.2139/ssrn.1948627
- Finke, M. S., & Huston, S. (2014). Personal finance issues. In V. Baker & H. K. Riccardo (Eds.), *Investor behavior: The psychology of financial planning and investing* (pp. 65-82). New York, NY: Wiley.
- FINRA Investor Education Foundation. (2009). *Financial capability in the United States: National survey--Executive summary* (Applied Research & Consulting LLC Report). Retrieved from http://www.usfinancial capability.org/downloads/NFCS_2009_Natl_Exec_Sum.pdf
- FINRA Investor Education Foundation. (2013). *Financial capability in the United States: Report of findings from the 2012 National Financial Capability Study* (FINRA Investor Education Report 13_0135.1). Retrieved from www.usfinancialcapability.org /downloads/NFCS_2012_Report_Natl_ Findings.pdf
- FINRA Investor Education Foundation. (2016). *About the national financial capability study*. Retrieved from http://www.usfinancialcapability.org/about.php
- Florida Council for Economic Education. (2014). *About us: Our mission, history.* Retrieved from Florida Council for Economic Education website, http://www.fcee.org/about-us/
- *Florida Financial Literacy Council.* (2008). Financial literacy council: First report, January 1, 2008. Retrieve from http://edocs.dlis.state.fl.us/fldocs/dfs /financial_literacy/ar/FLC2008(1).pdf
- Florida Financial Literacy Summit. (2013). *A call for education* (Summit Report, 2014, July 31). Retrieved from http://www.requirethemoneycourse.com/ downloads/FinancialLiteracySummit-WhitePaper.pdf
- Florida Institute of Certified Public Accountants. (n.d.). *360 degrees of financial literacy*. Retrieved from http://www.ficpa.org/content/CPAResources /Professional/360.aspx
- Florida Small Business Development Center Network (FSBDCN). (2016a). *About us: Mission, vision*. Retrieved from http://floridasbdc.org/about-us/mission-vision/
- Florida Small Business Development Center Network (FSBDCN). (2016b). *Helping* businesses grow & succeed. Retrieved from http://floridasbdc.org/

- Fonseca, R., Mullen, K., Zamarro, G., & Zissimopoulos, J. (2012). What explains the gender gap in financial literacy? The role of household decision making. *Journal of Consumer Affairs*, *46*(1), 90-106. doi:10.1111/j.1745-6606.2011.01221.x
- Fornero, E., & Monticone, C. (2011). Financial literacy and pension plan participation in Italy. *Journal of Pension Economics and Finance*, *10*(4), 547-564. doi:10.1017/S1474747211000473
- Fox, J., Bartholomae, S., & Lee, J. (2005). Building the case for financial education. *Journal of Consumer Affairs, 39*(1), 195-214. doi:10.1111/j.1745-6606.2005.00009.x
- Gale, W. G., & Levine, R. (2010). Financial literacy: What works? How could it be more effective? (Financial Security Series Working Paper No. 2011-1). Retrieved from Boston College website, http://squaredawayblog.bc.edu/wpcontent/uploads/2011/03/FSP-2011-1.pdf
- Gall, M. D., Gall, J. P., & Borg, W. R. (2007). *Educational research: An introduction*. Boston, MA: Pearson Education.
- Gallery, G. T., & Gallery, N. (2010). Rethinking financial literacy in the aftermath of global financial crisis. *Griffith Law Review, 19*(1), 30-50. Retrieved from http://eprints.qut.efu.au/
- Glass, G. V., & Hobkins, K. D. (1996). *Statistical methods in education and psychology*. Needham Heights, MA: Simon & Schuster.
- Global Financial Literacy Excellence Center (GFLEC). (2016a). About GFLEC: Mission and vision. Retrieved from http://www.gflec.org/#!about/csgz
- Global Financial Literacy Excellence Center (GFLEC). (2016b). *Featured academic research.* Retrieved from http://www.gflec.org/#!research/cifp
- Gilliland, D., Melfi, V., Sikorskii, A., Corcoran, E., & Milfi, E. (2011). Quantitative literacy at Michigan State University, 2: Connection to financial literacy. *Numeracy*, *4*(2), Article 6. doi:10.5038/1936-4660.4.2.6
- *Google Scholar Citations*. (2016, February 8). Retrieved February 8, 2016 from Google Scholar website, https://scholar.google.com/citations
- Grifoni, A., & Messy, F. (2012). Current status of national strategies for financial education: A comparative analysis (OECD Working Paper on Finance, Insurance and private Pensions No. 16). Retrieved from OECD website, http://dx.doi.org/10.1787/5k9bcwct7xmn-en

- Hastings, J., Madrian, B., & Skimmyhorn, W. L. (2012). *Financial literacy, financial education and economic outcomes* (NBER Working Paper 18412). Retrieved from NBER website, http://www.nber.org/papers/w18412
- Hastings, J., & Mitchell, O. S. (2011). *How financial literacy and impatience shape retirement wealth and investment behaviors* (NBER Working Paper No. 16740). Retrieved from NBER website, http://www.nber.org.ezproxy.lib.usf.edu /papers/w16740
- Hilgert, M., Hogarth, J., & Beverly, S. (2003). Household financial management: The connection between knowledge and behavior. *Federal Reserve Bulletin*, 89, 309-322. Retrieved from http://eds.b.ebscohost.com.ezproxy.lib.usf.edu/ ehost/pdfviewer/pdfviewer?vid=1&sid=b5f36c2f-c6b1-4f28-bd27-9411bf6407bf%40sessionmgr111&hid=101
- Hillman, R. (2009). Progress made in fostering partnerships, but national strategy remains largely descriptive rather than strategic (Report GAO-09-638). Retrieved from http://www.gao.gov/products/GAO-09-638T
- HM Treasury. (2007). *Financial capability: The government's long-term approach* (Report HM Treasury UP080). Retrieved from http://webarchive.nationalarchives.gov.uk/20091123151331/http://www.hmtreasury.gov.uk/d/fincap_150107.pdf
- Hof, M. (2012). Questionnaire evaluation with factor analysis and Cronbach's alpha: An example. Retrieved from http://www.let.rug.nl/nerbonne/teach/remastatsmeth seminar/student-papers/MHof-QuestionnaireEvaluation-2012-Cronbach Factor Analysis
- Hopely, V. (2003). *Financial education: What is it and what makes it so important?* (Community Reinvestment Report). Cleveland, OH: Federal Reserve Bank of Cleveland. Retrieved from http://ideas.repec.org/a/fip/fedccr/y2003ispr.html
- Hsu, J. (2011). Aging and strategic learning: The impact of spousal incentives of financial literacy (Finance and Economics Discussion Series Working Paper No. 2011-53). Retrieved from Federal Reserve website, http://www.federalreserve.gov/pubs/feds/2011/201153/201153pap.pdf
- Hung, A., Parker, A., & Yoong, J. (2009). Defining and measuring financial literacy (Working Paper No. WR708). Retrieved from RAND Corporation website, http://www.prgs.edu/content/dam/rand/pubs/working_papers/2009/RAND_WR70 8.pdf

- Hung, A., Yoong, J., & Brown, E. (2012). Empowering women through financial awareness and education (Working Paper on Finance, Insurance, and Private Pensions, No. 14). Retrieved from OECD website, http://www.oecdilibrary.org.ezproxy.lib.usf.edu/docserver/download/5k9d5v6kh56g.pdf?expires=1 423494610&id=id&accname=guest&checksum=3D5B67801F79F73D37A01032F B74B2A9
- Huston, S. J. (2010). Measuring financial literacy. *Journal of Consumer Affairs, 44*(2), 296-316. doi:10.1111/j.1745-6606.2010.01170.x
- Huston, S. J. (2012). Assessing financial literacy. In D. Durband & S. Britt (Eds.), Student financial literacy: Campus-based program development (pp. 109-124). New York, NY: Springer.
- James, W. B., & Blank, W. E. (1993). Review and critique of available learning-style instruments for adults. In D. D. Flannery (Ed.), *Applying cognitive learning theory* to adult learning (pp. 47-57). New Directions for Adult and Continuing Education, no. 59, San Francisco, CA: Jossey-Bass. doi:10.1002/ace.36719935907
- Jones, K., & Tullous, R. (2002). Behaviors of pre-venture entrepreneurs and perceptions of their financial needs. *Journal of Small Business Management, 40*(3), 233-249. Retrieved from http://eds.b.ebscohost.com.ezproxy.lib.usf.edu/ eds/pdfviewer/pdfviewer?sid=0bfbad6d-cd1d-4f56-87ab-5c7f0c307bf5@sessionmgr115&vid=3&hid=110
- Jump\$tart Coalition. (2007). *National standards in K-12 personal finance education* (Report for Jump\$tart Coalition). Retrieved from http://jumpstart.org/assets /files/standard_book-ALL.pdf
- Jump\$tart Coalition. (2016). Jump\$tart Coalition mission statement. Retrieved from http://www.jumpstartcoalition.org/mission.html
- *K-20 Education*, Florida Senate Bill No. 1076 Chapter 2013-27 (approved April 22, 2013). Retrieved from http://laws.flrules.org/files/Ch_2013-027.pdf
- Kim, H. Y. (2013). Statistical notes for clinical researchers: assessing normal distribution using skewness and kurtosis. *Restorive Dentistry & Endodontics, 38*(1), 52-54. doi:10.5395/rde.2013.38.1.52
- Klapper, L., Lusardi, A., & Panos, G. A. (2013). Financial literacy and its consequences: Evidence from Russia during the financial crisis. *Journal of Banking & Finance,* 37(10), 3904-3923. doi:10.1016 /j.jbankfin.2013.07.014
- Knoll, M. A. Z., & Houts, C. R. (2012). The financial knowledge scale: An application of item response theory to the assessment of financial literacy. *Journal of Consumer Affairs, 46*(3), 381-410. doi:10.1080/00220485.2012.686385

- Kolb, D. (1984). The process of learning. In D. Kolb, *Experiential learning: Experience* as the source of learning and development. Englewood Cliffs, NJ: Prentice Hall. Retrieved from http://www.learningfromexperience./images/uploads/process-ofexperiential-learning.pdf
- Kolb, A., & Kolb, D. (2005). Learning styles and learning spaces: Enhancing experiential learning in higher education. Academy of Management Learning & Education, 4(2), 193-212. Retrieved from http://eds.b.ebscohost.com.ezproxy.lib.usf.edu /eds/pdfviewer/pdfviewer?sid=9c60d4ae-6ecf-46c9-a18b-47d29b0ca0b7%2540 sessionmgr115&vid=2&hid=126
- Kozup, J., & Hogarth, J. M. (2008). Financial literacy, public policy, and consumers' selfprotection—More questions, fewer answers. *Journal of Consumer Affairs*, 42(2), 127-136. doi:10.1111/j.1745-6606.2008.00101.x
- Kumar, N., Mathur, A., & Sal, S. (2014). Banking 101: Mobile-izing financial inclusion in an emerging India. *Bell Labs Technical Journal*, *17*(4), 37-42. doi:10.1002/bltj
- Kunovskaya, I. A., Cude, B. J., & Alexeev, N. (2014). Evaluation of a financial literacy test using classical test theory and item response theory. *Journal of Family and Economic Issues*, *34*(1), 1-16. doi:10.1080/00220485.2012.686385
- Lachance, M. (2014). Financial literacy and neighborhood effects. *Journal of Consumer Affairs, 48*(1), 1-23. doi:10.1111/joca.12024
- Lee, E., & Hanna, S. D. (2014). Gender differences of Asian college students' financial knowledge pathways. Asian Women, 30(2), 27-55. Retrieved from http://www.easianwomen.org/xml/02243/02243.pdf
- Liston, M. I., & Bivens, G. (1993). *History of family economics research: 1862-1962.* Ames, IA: Iowa State University Research Foundation.
- Lusardi, A. (2011). *Americans' financial capability* (NBER Working Paper No. 17103) Retrieved from National Bureau of Economic Research website, http://www.nber.org/papers/w17103
- Lusardi, A. (2012a). Financial literacy and financial decision-making in older adults. *Generations, 36*(2), 23-33. Retrieved from http://eds.a.ebscohost.com. ezproxy.lib.usf.edu/eds/pdfviewer/pdfviewer?sid=77397045-f5ca-4cea-b102-374d141c9082%2540sessionmgr4005&vid=3&hid=4203
- Lusardi, A. (2012b). Numeracy, financial literacy, and financial decision-making. *Numeracy*, *5*(1), Article 2. doi:10.5038/1936-4660.5.1.2
- Lusardi, A. (2015). Financial literacy skills for the 21st century: Evidence from PISA. *Journal of Consumer Affairs, 49*(3), 639-659. doi:10.1111/joca.12099

- Lusardi, A., & Beeler, J. (2007). Saving between cohorts: The role of planning. In B. Madrian, O. Mitchell, & B. Soldo (Eds.), *Redefining retirement: How will boomers fare?* (pp.1-21). Cary, NC: Oxford University Press.
- Lusardi, A., & Mitchell, O. S. (2007a). Baby boomer retirement security: The roles of planning, financial literacy, and housing wealth. *Journal of Monetary Economics,* 54(1), 205-224. doi:10.1016/j.jmoneco.2006.12.001
- Lusardi, A., & Mitchell, O. S. (2007b). *Financial literacy and retirement planning: New evidence from the Rand American Life Panel* (CFS Working Paper No. 2007/33). Retrieved from Econstor website, http://hdl.handle.net/10419/25534
- Lusardi, A., & Mitchell, O. S. (2007c). Financial literacy and retirement preparedness: Evidence and implications for financial education. *Business Economics, 4*2(1), 35–44. Retrieved from http://eds.b.ebscohost.com. ezproxy.lib.usf.edu/eds/pdfviewer/pdfviewer?sid=169a5de1-c5dc-4aee-ad21-24b997b210b0%40sessionmgr110&vid=3&hid=115
- Lusardi, A., & Mitchell, O. S. (2008). Planning and financial literacy: How do women fare? *American Economic Review*, *98*(2), 413-417. doi:10.1257/aer.98.2.413
- Lusardi, A., & Mitchell, O. S. (2011a). Financial literacy around the world: An overview. *Journal of Pension Economics and Finance, 10*(4), 197-508. doi:10.1017/S1474747211000448
- Lusardi, A., & Mitchell, O. S. (2011b). *The outlook for financial literacy* (Working Paper No. 17077). Retrieved from National Bureau of Economic Research website, http://www.nber.org/papers/w17077
- Lusardi, A., & Mitchell, O. S. (2013). *The economic importance of financial literacy: Theory and evidence* (Working Paper No. 18952). Retrieved from National Bureau of Economic Research website, http://www.nber.org.ezproxy.lib. usf.edu/papers/w18952
- Lusardi, A., & Mitchell, O. S. (2014). The economic importance of financial literacy: Theory and evidence. *Journal of Economic Literature, 52*(1), 5-44. doi: 10.1257/jel.52.1.5
- Lusardi, A., Mitchell, O. S., & Curto, V. (2010). Financial literacy among the young. *Journal of Consumer Affairs, 44*(2), 358-380. doi: 10.1111/j.1745-6606.2010.01173.x
- Lusardi, A., & Scheresberg, C. (2013). *Financial literacy and high-cost borrowing in the United States* (NBER Working Paper 18969). Retrieved from National Bureau of Economic Research website, http://www.nber.org/papers/w18969

- Lusardi, A., & Tufano, P. (2009). *Debt literacy, financial experiences, and over indebtedness* (NBER Working Paper 14808). Retrieved from National Bureau of Economic Research website, http://www.nber.org/papers/w14808
- Lusardi, A., & Wallace, D. (2013). Financial literacy and quantitative reasoning in the high school and college classroom. *Numeracy, 6*(2), Article 1. doi:10.5038/1936-4660.6.2.1
- Mandell, L. (2006). *Financial literacy: If it's so important, why isn't it improving?* (Working Paper No. 2006-PB-8). Retrieved from Network Financial Institute at Indiana State University website, http://indstate.edu/business/NFI/ leadership/briefs/2006-PB-08_Mandell.pdf
- Mandell, L. (2008). The financial literacy of young American adults: Results of the 2008 national Jump\$tart Coalition survey of high school seniors and college students (Report for Jump\$tart Coalition). Retrieved from http://deca.smgww.org/assets /pdf/2008%20JumpStart%20Financial%20Literacy%20Survey.pdf
- Mandell, L., & Klien, L. S. (2007). Motivation and financial literacy. *Financial Services Review*, *16*(2), 105-116. Retrieved from http://eds.a.ebscohost.com. ezproxy.lib.usf. edu/ehost/pdfviewer/pdfviewer?sid=1f04f9f9-31c6-4ef4-946cfed1217b323b%40sessionmgr4002&vid=1&hid=4203
- Mandell, L., & Klien, L. (2009). The impact of financial literacy education on subsequent financial behavior. *Journal of Financial Counseling and Planning, 20*(10), 15-25. Retrieved from http://www.afcpe.org/assets/pdf/lewis_mandell_linda_schmid_klein.pdf
- Marshall, M. (2012). Outreach and education boost entrepreneurs in Indiana. *Choices,* 27(2). Retrieved from http://eds.b.ebscohost.com.ezproxy.lib.usf.edu/eds/ pdfviewer/pdfviewer?sid=0bfbad6d-cd1d-4f56-87ab-5c7f0c307bf5@sessionmgr115&vid=3&hid=110
- Mason, C. L. J., & Watson, R. M. S. (2000). *Conceptualising financial literacy* (Working Paper Business School Research Series 2000:7). Retrieved from Loughborough University website, https://dspace.lboro.ac.uk/2134/6977
- McKenzie, V. (2009). The financial literacy of university students: A comparison of graduating seniors' financial literacy and debt level (Doctoral dissertation, University of South Florida). Retrieved from http://search.proquest.com .ezproxy.lib.usf.edu/pqdtft/docview/749943909/C95AA8E8E6E44144PQ/4?accountid=14745
- Mendes-da-Silva, W., Nakamura, W. T., & Moraes, D. C. (2012). Credit card risk behavior on college campuses: Evidence from Brazil. *BAR Brazilian Administration Review*, 9(3), 351-373. Retrieved from http://www.scielo.br/pdf/bar/v9n3/a07v9n3.pdf

- Miller, M., Reichelstein, J., Salas, C., & Zia, B. (2014). Can you help someone become financially capable? A meta-analysis of the literature (Policy Research Working Paper No. 6745). Retrieved from World Bank website, http://siteresources. worldbank.org.ezproxy.lib.usf.edu/EXTGLOBALFINREPORT/Resources/881609 6-1361888425203/9062080-1383688365382/Meta_Analysis_Paper.pdf
- Mojab, F., Zaefarian, R., & Azizi, A. (2011). Applying competency based approach for entrepreneurship education. *Procedia--Social and Behavioral Sciences*, 12(3), 436-447. doi:10.1016/j.sbspro.2011.02.054
- Mottola, G. R. (2013). In our best interest: Women, financial literacy, and credit card behavior. *Numeracy*, *6*(2), Article 4. doi:10.5038/1936-4660.6.2.4
- National Bureau of Economic Research (NBER). (n.d.). *NBER Publications by Annamaria Lusardi.* Retrieved from NBER website, http://www.nber.org/authors/annamaria_lusardi
- National Council on Economic Education. (2005). Survey of the states: Economic and personal finance education in our national's schools in 2004. Retrieved from National Council on Economic Education website, http://www.councilforeconed .org/wp/wp-content/uploads/2011/11/2004-Survey-of-the-States.pdf
- National Council on Economic Education. (2014). Survey of the states: economic and personal finance education in our national's schools in 2014. Retrieved from National Council on Economic Education website, http://www.councilforeconed .org/wp/wp-content/uploads/2014/02/2014-Survey-of-the-States.pdf
- National Endowment for Financial Education. (2014, May 18). Retrieved from National Endowment for Financial Education website, http://www.nefe.org/what-weprovide/research-funding/grant-guidelines.aspx
- National Financial Capability Study (NFCS). (2016a). 2009 state-by-state survey (Data File Information). Retrieved from http://www.usfinancialcapability.org/ downloads.php
- National Financial Capability Study (NFCS). (2016b). 2012 state-by-state survey (Data File Information). Retrieved from http://www.usfinancialcapability.org/ downloads.php
- Newman, I., Ridenour, C. S., Newman, C., & DeMarco, G. M. P., Jr. (2003). A typology of research purposes and its relationship to mixed methods. In A. Tashakkori & C. Teddlie (Eds.), *Handbook of mixed methods in social and behavioral research* (pp. 167-188). Thousand Oaks, CA: Sage.

Nicolini, G., Cude, B. J., & Chatterjee, S. (2013). Financial literacy: A comparative study across four countries. *International Journal of Consumer Studies*, *37*(6), 689-705. doi:10.1111/ijcs.12050

Norvilitis, J. M., & Mendes-da-Silva, W. (2013). Attitudes toward credit and finances among college students in Brazil and the United States. *Journal of Business theory and Practice, 1*(1), 132-135. Retrieved from http://papers.ssrn.com/sol3/papers.cfm?abstract_id=2283901

- Novak, J. (2014). The six living generations in America. *Category Archives*. Retrieved from http://www.marketingteacher.com/the-six-living-generations-in-america/
- Nunoo, J., & Andoh, F. (2012). Sustaining small and medium enterprises through financial service utilization. Does financial literacy matter? Seattle, WA. Agricultural & Applied Economics Association's 2012 AAEA Annual Meeting. Retrieved from http://ageconsearch.umn.edu/bitstream/12341
- Obama, B. (2010, April 2). *Presidential proclamation--Financial literacy month*. The White House. Retrieved from http://www.whitehouse.gov/the-press-office/presidential-proclamation-financial-literacy-month
- O'Connor, A. (2013). A conceptual framework for entrepreneurship education policy: Meeting government and economic purposes. *Journal of Business Venturing*, *28*(4), 546-563. doi:10.1016/j.jbusvent.2012.07.003
- O'Rourke, N., & Hatcher, L. (2014). A step-by-step approach to using the SAS system for factor analysis and structural equation modeling. Cary, NC: SAS Institute.
- Osborne, J. W. (2014). *Best practices in exploratory factor analysis*. Retrieved from http://www.researchgate.net/publication/265248967_Best_Practices_in_Explorat ory_Factor_Analysis
- Pahnke, L., & Honekamp, I. (2010). *Different effects of financial literacy and financial education in Germany* (MPRA Working Paper No. 22900). Retrieved from University of Bamber website, http://mpra.ub.uni-muenchen.de/22900/
- Paul, J. L. (2005). Introduction to the philosophies of research and criticism in education and the social sciences. Upper Saddle River, NJ: Pearson.
- Potrich, A. C., Vieira, K. M., Coronel, D. A., & Filho, R. B. (2015). Financial literacy in Southern Brazil: Modeling and invariance between genders. *Journal of Behavioral and Experimental Finance, 3*(2). doi:10.1016/j/jbef.2015.03002
- President's Advisory Committee on Financial Capability (PACFC). (2013a). Annual report to the President: Executive summary (Report No. 22513). Retrieved from http://www.treasury.gov/connect/blog/Documents/PACFC%20Final%20Report% 202013.PDF

- President's Advisory Committee on Financial Capability (PACFC). (2013b). *Final report: President's Advisory Council on Financial Capability* (Report No. 1-29-13). Retrieved from http://www.treasury.gov/resource-center/financialeducation/Documents/PACFC %20final%20report%20revised%2022513%20(8)_ R.pdf
- President's Advisory Committee on Financial Literacy (PACFL). (2009). 2008 annual report to the President (Report No. 1-16-09). Retrieved from http://www.treasury.gov/resource-center/financialeducation/Documents/PACFL_ANNUAL_REPORT_1-16-09.pdf
- PRWeb. (2015). Council for Economic Education honors Dr. Lusardi with 2014 W. A. Forbes Public Awareness Award. Retrieved from http://www.prweb.com/releases/2015/01/prweb12442619.htm
- RAND Financial Literacy Center. (2016a). RAND Labor and Population: Financial literacy center. Retrieved http://www.rand.org /labor/centers/financial-literacy.html
- RAND Financial Literacy Center. (2016b). *Financial literacy center staff.* Retrieved from http://www.rand.org/labor/centers/financial-literacy/staff.html
- Remund, D. (2010). Financial literacy explicated: The case for a clearer definition in an increasingly complex economy. *Journal of Consumer Affairs, 44*(2), 276-295. doi:10.1111/j.1745-6606.2010.01169.x
- Robles, J. (2014). Economic inclusion and financial education in culturally divers communities: Leveraging cultural capital and whole-family learning. In K. Spro-Forte, E. Taylor, & E. Tisdell (Eds.), *Financial literacy and adult education* (pp. 57-66). New Directions for Adult and Continuing Education, no. 141. San Francisco, CA: Jossey-Bass. doi:10.1002/ace
- Robb, C., & Woodyard, A. (2011). Financial knowledge and best practice behavior. Association for Financial Counseling and Planning Education, 22(1), 60-70. Retrieved from http://eds.b.ebscohost.com.ezproxy.lib.usf.edu/ eds/pdfviewer/pdfviewer?sid=e1928a18-3502-44f0-9efcb3fdea42c7ea%40sessionmgr111&vid=3&hid=109
- Roberts, H., Sorgman, M., & Parkison, K. (2010). Surveying the importance of economics and financial literacy descriptors. *Journal of Applied Business and Management Studies*, 1(2), 1-18. Retrieved from http://www.jabms.net/Volume1No2/Surveying%20the%20Importance%20of%20E conomics%20and%20Financial%20Literacy%20Descriptors.pdf
- Scheresberg, D. B. (2013). Financial literacy and financial behavior among young adults: Evidence and implications. *Numeracy*, *6*(2), Article 5. doi:10.5038/1936-4660.6.2.5

- Schmeiser, M. D., & Seligman, J. S. (2013). Using the right yardstick: Assessing financial literacy measures by way of financial well-being. *Journal of Consumer Affairs*, 47(2), 243-262. doi:1011.111/joca.12010
- Schuchardt, J., Hanna, S. D., Hira, T. K., Lyons, A. C., Palmer, L., & Xiao, J. J. (2009). Financial literacy and education research priorities. *Journal of Financial Counseling & Planning, 20*(1), 84-96. Retrieved from http://eds.b.ebscohost.com .ezproxy.lib.usf.edu/ehost/pdfviewer/pdfviewer?sid=87e9ba3a-a510-47de-808a9c8b894b0749%2540sessionmgr198&vid=1&hid=101
- Sekita, S. (2011). Financial literacy and retirement planning in Japan. *Journal of Pension Economics and Finance, 10*(4), 637-656. doi:10.1017/S1474747211000527
- Shambare, R., & Rugimbana, R. (2012). Financial literacy among the educated: An exploratory study of selected university students in South Africa. *Thunderbird International business Review*, *54*(4), 581-590. doi:10.1002/tie.21485:
- Sherraden, M. (2010). *Financial capability: What is it, and how can it be created*? (CSD Working Paper 10-17). Retrieved from University of Missouri website, http://csd.wustl.edu/Publications/Pages/displayresultitem.aspx?ID1=1034
- Sherraden, M. S., & Grinstein-Weiss, M. (2015). Creating financial capability in the next generation: An introduction to the special issue. *Journal of Consumer Affairs, 49*(1), 1-12. doi: 10.1111/joca.12067
- Shim, S., Serido, J., & Tang, C. (2012). The ant and the grasshopper revisited: The present psychological benefits of saving and future oriented financial behaviors. *Journal of Economic Psychology, 33*(1), 155-165. doi:10.1016/joep.2011.08.005
- Shropshire, K. O., Hawdon, J. E., & Witte, J. C. (2009). Web survey design: Balancing measurement, response, and topical interest. *Sociological Methods & Research*, 37(3), 344-370. doi:10.1177/0049124108327130
- Shuttleworth, C. (2009). Towards a financial literacy model as a coordinating interface between information and decision makers (Doctoral dissertation, University of Pretoria). Retrieved from http://www.repository.up.ac.za/bitstream/ handle/2263/28205/ Complete.pdf?sequence=8
- Simms, K. (2014). Is universal financial education putting the cart before the horse? *Journal of Economics and Behavioral Studies, 6*(4), 318-332. Retrieved from http://search.proquest.com/docview/1530407979?accountid=14745
- Small Business Administration (SBA). (2012). Annual report of the office of economic research FY 2012. Retrieved from http://www.sba.gov/content/annual-report-office-economic-research-fy-2012

- Small Business Administration (SBA). (2013). U.S. Small Business Administration: FY2014 Congressional budget justification and FY 2012 annual performance report. Retrieved from http://www.sba.gov/sites/default/files/files/1-508-Compliant-FY-2014-CBJ%20FY%202012%20APR.pdf
- Small Business Administration (SBA). (2015). U.S. Small Business Administration: FY2015 Congressional budget justification and FY2013 annual performance report. Retrieved from https://www.sba.gov/sites/default/files/files/FY%202015% 20CBJ%20FY%202013%20APR%20FINAL%20508(1).pdf
- Small Business Administration (SBA) Office of Advocacy. (2016a). Small business profile: Florida. Retrieved from SBA Office of Advocacy website, www.sba.gov/advocacy
- Small Business Administration (SBA) Office of Advocacy. (2016b). *Small business profile: Florida.* Retrieved from SBA Office of Advocacy website, https://www.sba.gov/sites/default/files/advocacy/FL_0.pdf
- Small Business Administration (SBA) Office of Advocacy. (2016c). Small business profile for the states and territories. Retrieved from SBA Office of Advocacy website, https://www.sba.gov/sites/default/files/advocacy/SB%20Profiles% 202024-15_0.pdf
- Stevens, J. P. (2007). A modern approach: Intermediate statistics. New York, NY: Routledge.
- Texas Tech University. (2015). *Personal financial planning: Financial literacy assessment project at Texas Tech University*. Retrieved from http://www.depts.ttu.edu/pfp/ documents/Literacy_Survey_ Schwab_Overview.pdf
- Thabane, L., Ma, J., Chu, R., Cheng, J., Ismila, A., Rios, L. P., Robson, R., Thabane, M., Giangregoria, L., & Goldsmith, C. H. (2010). A tutorial on pilot studies: The what, why and how. *BMC Medical Research Methodology*, *10*(1), 1-10. doi: 10.1186/1471-2288-10-1
- Theodos, B., Kalish, E., McKernan, S. M., & Ratcliffe, C. (2014). Do financial knowledge, behavior, and well-being differ by gender? Retrieved from FINRA Investor Education foundation website, ftp://timecard.urban.org/pubs_prod/ 2014/pdf/batch1/413077-Do-Financial-Knowledge-Behavior-and-Well-Being-Differ-by-Gender.pdf
- Tippet, J., & Kluvers, R. (2007). A practical evaluation of an adult financial literacy programme. *International Review of Business Research Papers, 3*(2), 442-455. Retrieved from http://www.bizresearchpapers.com/Tippet.pdf

- Tokar-Assaad, C. (2015). Financial literacy and financial behavior: Assessing knowledge and confidence. *Financial Services Review, 24*(2), 101-117. Retrieved from http://eds.a.ebscohost.com.ezproxy.lib.usf.edu/eds/ pdfviewer/pdfviewer?sid=a7dda83a-5355-4d77-b29e-a4489de08abd% 40sessionmgr4002&vid=2&hid=4102
- Tschache, C. A. (2009). *Importance of financial literacy and financial literacy content in curriculum* (Doctoral dissertation, Montana State University). Retrieved from http://scholarworks.montana.edu/xmlui/handle/1/2449
- U.S. Census. (2010). *Statistics of U.S. Businesses*. Retrieved from http://www.census.gov/econ/susb/
- Van Rooij, M., Lusardi, A., & Alessie, R. (2007). *Financial literacy and stock market participation* (NBER Working Paper 18412). Retrieved from NBER website, http://www.nber.org.ezproxy.lib.usf.edu/papers/w13565.pdf
- Van Rooij, M., Lusardi, A., & Alessie, R. (2009). Financial literacy and retirement planning in the Netherlands (DNB Working Paper 231/2009). Retrieved from DNB website, http://www.dnb.nl/binaries/231%20Financial%20 Literacy%20and%20Retirement%20Planning%20in%20the%20Netherlands_tcm 46-225547.pdf
- Van Rooij, M., Lusardi, A., & Alessie, R. (2011). Financial literacy and stock market participation. *Journal of Financial Economics*, *101*(2), 449-472. doi:10.1016/jfinceo.2011.03.006
- Vieira, E. F. S. (2012). What do we know about financial literacy? A literature review. *Marmara Journal of European Studies, 20*(2), 23-38. Retrieved from http://ria.ua.pt/bitstream/10773/10055/1/Paper_Fin_Lit_Rev.pdf
- Vitt, L. A., Anderson, C., Kent, J., Lyter, D. M., Siegenthaler, J. K., & Ward, J. (2000). Personal finance and the rush to competence: Financial literacy education in the U.S. (Fannie Mae Foundation Report). Middleburg, VA: Institute for Social-Financial Studies. Retrieved from http://www.isfs.org/ documents-pdfs/repfinliteracy.pdf
- Volpe, R. P., Chen, H., & Pavliko, J. J. (1996). Personal investment literacy among college students: A survey. *Financial Practice and Education*, 6(2), 86-94. Retrieved from http://maagblog.ysu.edu/financialliteracy/files/2009/03/volpe4.pdf
- West, S. G., Finch J. F., & Curran P. J. (1995). Structural equation models with nonnormal variables. In R. H. Hoyle, *Structural equation modeling, concepts, issues, and applications* (pp. 56-75). Thousand Oaks, CA: Sage.

- Williams, B. (2012). Exploratory factor analysis: A five-step guide for novices. *Australiasian Journal of Paramedicine, 8*(3), Article 1. Retrieved from http://ro.ecu.edu.au/jephc/vol8/iss3/1
- Wilson, F., Kickul, J., & Marlino, D. (2007). Gender, entrepreneurial self-efficacy, and entrepreneurial career intentions: Implications or entrepreneurship education. *Entrepreneurship Theory and Practice*, 31(3), 387–406. doi:10.1111/j.1540-6520.2007.00179.x
- Wolfe-Hayes, M. A. (2010). Financial literacy and education: An environmental scan. International Information and Library Review, 42(2), 105-110. doi:10.1016/j.iilr.2010.04.006
- Woodyard, A., & Robb, C. (2012). Financial knowledge and the gender gap. *Journal of Financial Therapy*, *3*(1), 1-16. doi:10.4148/jft.v3i1.1453
- World Bank Group. (2009). World Bank Russian financial literacy and financial education program: 2009 diagnostic financial literacy survey questionnaire. Retrieved from World Bank website, http://responsiblefinance.worldbank.org/~/ media/GIAWB/FL/Documents/eLibrary/FCCP-Survey-questionnaire-Russia-2009.pdf
- Yan, T., Conrad, F. G., Tourangeau, R., & Couper, M. P. (2010). Should I stay or should I go: The effects of progress feedback, promised task duration, and length of questionnaire on completing web surveys. *International Journal of Public Opinion Research*, 23(2), 131-137. doi:10.1093/jjpor/edq046

Appendices

Big Five Financial Literacy Survey Questions

Questions 1 – 3 were used by Health and Retirement Study (2004); 1-4 were used by International Market Research Company (2011), Euro Survey of Austrian Central Bank (2007), Pureprofile Web Panel (2013); and, questions 1-5 were used by NFCS (2009, 2012).

- a) Suppose you had \$100 in a savings account and the interest rate was 2% per year. After 5 years, how much do you think you would have in the account if you left the money to grow?
 - More than \$102
 - Exactly \$102
 - Less than \$102?

- Do not know
- Refuse to answer
- b) Imagine that the interest rate on your savings account was 1% per year and inflation was 2% per year. After 1 year, would you be able to buy
 - More than

- Do not know
- Exactly the same as
- Refuse to answer
- Or less than today with the money in this account
- c) True or false? "Buying a single company stock usually provides a safer return than a stock mutual fund."
 - True
 - False

- Do not know
- Refuse to answer
- d) If interest rates rise, what will typically happen to bond prices?
 - They will rise
 - They will fall
 - They will stay the same
- There is no relationship between bond prices and interest rates
- Do not know
- Prefer not to say
- e) True or false: A 15-year mortgage typically requires higher monthly payments than a 30-year mortgage, but the total interest paid over the life of the loan will be less.
 - True

Don't know

False

• Prefer not to say

Sources: Agnew, 2013; Allgood, 2013; Beckmann, 2013; Bumcrot et al., 2013: Lusardi, 2012; Scheresberg, 2013

Appendix B

Institutional Letter of Support

	neiping businesses Grow & Succeed
SBDC	FSBDC Network State Offic UWF, Center for Research and Economic Opportunity Suntrust Tower, Suite 30 220 West Garden Stree Pensacola, FL 3250
TEORIDA	P 850.898.347
7/7/2015	www.FloridaSBDC.org
University of South Election	
Research and Innovation	
Institutional Review Board (IRB)	
4202 E. Fowler Avenue,	
Tampa, FL 33620	
Dear Members of the Committee:	
On behalf of the Florida Small Business Development Center Networl support of the proposal, "Financial Literacy Professional Continuing E Small Business Owners", being submitted to the University of South I Review Board by Pearl Dahmen Para of the College of Education Dep and Higher Education at the University of South Florida.	(FSBDCN), it is my pleasure to write a letter in ducation Cognitive Needs Assessment for Florida lorida Research and Innovation, Institutional artment of Leadership, Counseling, Adult, Career
As CEU and State Director of the FSBDCN, I have been associated with Consultant with the FSBDC at the University of South Florida. As a Ce Bay area in developing and delivering financial education to our clien	n Ms. Para during her tenure as a Business rtified Public Accountant, she has led the Tampa ts. This experience inspired her research, and I am
delighted that she has chosen our clients for the study's population.	
l am aware that Ms. Para intends to conduct her research by surveyin research that can help us to better understand our clients' needs and providing education will benefit our organization, our clients, and the	ng our clients. I fully support her research as any enhance our decision-making processes in State of Florida.
delighted that she has chosen our clients for the study's population. I am aware that Ms. Para intends to conduct her research by surveyir research that can help us to better understand our clients' needs and providing education will benefit our organization, our clients, and the Should you have any questions or concerns, please feel free to conta	ng our clients. I fully support her research as any I enhance our decision-making processes in I State of Florida. Ct my office at (850) 898-3479.
delighted that she has chosen our clients for the study's population. I am aware that Ms. Para intends to conduct her research by surveyir research that can help us to better understand our clients' needs and providing education will benefit our organization, our clients, and the Should you have any questions or concerns, please feel free to conta Sincerely,	ng our clients. I fully support her research as any I enhance our decision-making processes in I State of Florida. Ct my office at (850) 898-3479.
delighted that she has chosen our clients for the study's population. I am aware that Ms. Para intends to conduct her research by surveyin research that can help us to better understand our clients' needs and providing education will benefit our organization, our clients, and the Should you have any questions or concerns, please feel free to conta Sincerely, Michael W. Mydua	ng our clients. I fully support her research as any l enhance our decision-making processes in l State of Florida. ct my office at (850) 898-3479.
delighted that she has chosen our clients for the study's population. I am aware that Ms. Para intends to conduct her research by surveyin research that can help us to better understand our clients' needs and providing education will benefit our organization, our clients, and the Should you have any questions or concerns, please feel free to conta Sincerely, Michael W. Myfue CEO and Network State Director	ng our clients. I fully support her research as any l enhance our decision-making processes in l State of Florida. Ct my office at (850) 898-3479.
delighted that she has chosen our clients for the study's population. I am aware that Ms. Para intends to conduct her research by surveyin research that can help us to better understand our clients' needs and providing education will benefit our organization, our clients, and the Should you have any questions or concerns, please feel free to conta Sincerely, Michael W. MyJuan Michael Myhre CEO and Network State Director	ng our clients. I fully support her research as any l enhance our decision-making processes in State of Florida. ct my office at (850) 898-3479.
delighted that she has chosen our clients for the study's population. I am aware that Ms. Para intends to conduct her research by surveyin research that can help us to better understand our clients' needs and providing education will benefit our organization, our clients, and the Should you have any questions or concerns, please feel free to conta Sincerely, Michael W. Myfue CEO and Network State Director	ng our clients. I fully support her research as any l enhance our decision-making processes in l State of Florida. Ct my office at (850) 898-3479.
delighted that she has chosen our clients for the study's population. I am aware that Ms. Para intends to conduct her research by surveyin research that can help us to better understand our clients' needs and providing education will benefit our organization, our clients, and the Should you have any questions or concerns, please feel free to conta Sincerely, Michael W. MyJu Michael Myhre CEO and Network State Director	ng our clients. I fully support her research as any l enhance our decision-making processes in 2 State of Florida. ct my office at (850) 898-3479. <i>Hosted by</i>
I am aware that Ms. Para intends to conduct her research by surveyin research that can help us to better understand our clients' needs and providing education will benefit our organization, our clients, and the Should you have any questions or concerns, please feel free to conta Sincerely, Michael W. Mydua Michael Myhre CEO and Network State Director	ig our clients. I fully support her research as any enhance our decision-making processes in State of Florida. tr my office at (850) 898-3479. <i>Hosted by</i>

Appendix C

Pool of Financial Literacy Questions

Table C1

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT/FR
1	Suppose you had \$100 in a savings account and the interest rate was 2% per year. After 5 years how much do you think you would have in the account, if you left the money to grow? • More than \$102 • Don't know • Exactly \$102 • Less than \$102	K1 K2	ALP DHS HRS NFCS			Chg	EX	
2	 Less than \$102 If interest rates rise, what will typically happen to bond prices? They will rise They will fall They will stay the same Don't know Prefer not to say ALTERNATIVE WORDING If interest rates fall, what should happen to bond prices? They will fall They will stay the same They mill fall They will stay the same They mill stay the same They mill stay the same Prefer not to say 		ALP (2) NFCS DHS		EX			
3	Imagine that the interest rate on your savings account was 1% per year and inflation was 2% per year. After 1 year, would you be able to buy: • More than today • Exactly the same • Less than today	K1 K2 FLN	APL DHS HRS NFCS USIS				EX	

Note: ICR=IRT/CIT Review, Rshr=Researcher, IP=Initial Panel, Val=Validation Panel, Ver=Verification Panel, FT/FR=Field Test/Final Review; * See Table C2 for key to abbreviations, source and reference;

**Included in final instrument; Dated=question eliminated due to not being current subject matter; Cur/Cul=question eliminated due to currency/culture not applicable to this study's participants; Inc/Econ=questions outside this study's content areas; EX=excluded by expert panel; Chg=wording changed by expert panel suggestions; Correct answers are highlighted in bold.

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

			100	^ *	D 1	10			
#	Question		ICR	Source*	Rshr	IP	Val	Ver	FI/FR
4	Suppose that in the year 2010, y goods have doubled too. In 2010 income? • More than today • The same • Less than today	our income has doubled and prices of all), how much will you be able to buy with your • Don't know • Refuse to answer	K2	ALP DHS			Chg	EX	
5	Assume a friend inherits \$10,000 years from now. Who is richer be • My friend • His sibling • They are equally rich	 today and his sibling inherits \$10,000 three ecause of the inheritance? Don't know Refuse to answer 	K1	ALP DHS		EX			
6	Normally, which asset displays th • Savings accounts • Bonds • Stocks	ne highest fluctuations over time?Don't knowRefuse to answer	K1 FLN	APL DHS		EX			
7	There are annual contribution limplan or IRA that depend on your True False	nits on the amount you can save in a 401(k) income. • Don't know • Refuse to answer	K1	ALP		EX			
8	After age 70 1/2, you have to wit plan or IRA. • True • False	hdraw at least some money from your 401(k) Don't know Refuse to answer 	K1 FLN	ALP		EX			

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT/FR
9	Consider a long time period (e.g., 10 or 20 years), which asset normally gives the highest return? • Savings accounts • Bonds • Stocks	K1 FLN	ALP DHS		EX			
10	An annuity is a financial product that pays a lump sum when you die. • True • False • Refuse to answer	K1	ALP				EX	
11	 Which statement describes the main function of the stock market? The stock market helps to predict stock earnings The stock market results in an increase in the price of stocks The stock market brings people who want to buy stock together with those who want to sell stocks None of the above Don't know Refuse to answer 		ALP DHS		EX			
12	If you buy a company's stock • You own a part of the company • You have lent money to the company • You are liable for the company's debt • You can vote on shareholder resolutions • Don't know • Refuse to answer	K1	ALP DHS		EX			

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT/FR
13	If you buy a company's bond • You own a part of the company • You have lent money to the company • You are liable for the company's debt • You can vote on shareholder resolutions • Don't know. • Refuse to answer	FLN	ALP DHS		EX			
14**	If you were to invest \$1000 in a stock mutual fund, it would be possible to have less than \$1000 when you withdraw your money. • True • False • Don't know • Refuse to answer	K1 FLN	ALP					
15	 Which statement about mutual funds is correct? Once one invests in a mutual fund, one cannot withdraw the money in the 1st year Mutual funds can invest in several assets, for example, invest in both stocks & bonds Mutual funds pay a guaranteed rate of return which depends on their past performance None of the above Don't know Refuse to answer 	FLN	ALP DHS		EX			

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question		ICR	Source*	Rshr	IP	Val	Ver	FT/FR
16	A stock mutual fund combines th of stocks, not a single stock. • True • False	 e money of many investors to buy a variety Don't know Refuse to answer 	K1 FLN	ALP		EX			
17	"Whole life" insurance has a savi • True • False	ngs feature while "term" insurance does not. • Don't know • Refuse to answer	K1	ALP HHB		EX			
18	The cash value of a life insurance surrender your life insurance poli • True • False	e policy is the amount available if you cy while you're still alive. • Don't know • Refuse to answer	K1	ALP HHB		EX			
19	Is the following statement true or Housing prices in the US can new • True • False	false? /er go down. • Don't know • Refuse to answer	K1	ALP			EX		
20	Do you think that the following sta "Bonds are normally riskier than a • True • False ALTER Do you think that the following sta "Stocks are normally riskier than	atement is true or false? stocks." • Don't know • Refuse to answer NATIVE WORDING atement is true or false? Bonds."	K1 FLN	APL (2) DHS			EX		

• True • Don't know

• False • Refuse to answer

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question		ICR	Source*	Rshr	IP	Val	Ver	FT/.FR
21	Buying a company stock usually fund. • True	 provides a safer return than a stock mutual Don't know Refuse to answer 	K1	APL DHS HRS		EX			
22	Suppose you had \$100 in a savin per year and you never withdraw years, how much would you have • More than \$200 • Exactly \$200 • Less than \$200	ngs account and the interest rate is 20% money of interest payments. After five in the account in total? • Don't know • Refuse to answer	FLN	DHS	Cur/Cul	EX			
23	If you buy a 10-year bond, it mea without incurring a major penalty • True • False	ans you cannot sell it after five years • Don't know • Refuse to answer		DHS		EX			
24	When an investor spreads his mo of losing money: • Increase • Decrease • Stay the same	oney among different assets, does the risk • Don't know • Refuse to answer	K1 FLN	DHS			EX		
25**	A 15-year mortgage typically req year mortgage, but the total inter less. • True • False	uires higher monthly payments than a 30- est paid over the life of the loan will be • Don't know • Refuse to answer	K1 FLN	NFCS					

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT/FR
26	Suppose you owe \$3,000 on your credit card. You pay a minimum paymentof \$30 each month. At an Annual Percentage Rate of 12% (or 1% per month), how many years would it take to eliminate your credit card debt if you made no additional new charges?• Less than 5 years • Between 5 and 10 years • Between 10 and 20 years• Never, you will continue to be in debt • Don't know • Prefer not to say	K1 FLN	TNS		EX			
27	Suppose you owe \$1,000 on your credit card and the interest rate you are charged is 20% per year compounded annually. If you didn't pay anything off, at this interest rate, how many years would it take for the amount you owe to double? a. 2 years b. Less than 5 years c. 5 to 10 years d. More than 20 years c. Don't know d. Prefer not to answer		TNS			Chg		EX
28	You purchase an appliance that costs \$1,000. To pay for this appliance, you are given the following two options: (a) Pay 12 monthly installments of \$100 each; (b) Borrow at a 20% annual interest rate and pay back \$1,200 a year from now. Which is the more advantageous offer? • Option (a) • Option (b) • Prefer not to answer • They are the same		TNS		EX			
29	You lent X to a friend on evening and he gives you X back the next day. How much interest has he paid on this loan? [open response: 0]		TNS		EX			

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT/FR
30	Imagine that five brothers are given a gift of \$1,000. If the brothers have to share the money equally, how much does each one get? [open response: \$200] Note: Follow up question-#22		OECD				EX	
	Now imagine that the brothers have to wait for one year to get their share of the X. In one year's time will they be able to buy:[Multiple choice] a) More b) the same amount c) less than they could buy today							
31	High inflation means that the cost of living is increasing rapidly. [True /False]		OECD			EX		
32	An investment with a high return is likely to be high risk. [True/False]		OECD			EX		
33**	It is usually possible to reduce the risk of investing in the stock market by buying a wide range of stocks and shares. [True /False]		OECD			Chg		
34	Financially, investing in the stock market is better than buying lottery tickets. [True /False] ALTERNATIVE WORDING		OECD		EX			
	Financially, investing in the stock market is not better than buying lottery tickets. [True/ False]							
35	Suppose you put \$100 into a savings account with a guaranteed interest rate of 2% per year. You don't make any further payments into this account and you don't withdraw any money. How much would be in the account at the end of the first year, once the interest payment is made? [Open response: \$102] Continued with #36		OECD	Open- ended	EX			

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question		ICR	Source*	Rshr	IP	Val	Ver	FT/FR
36	and how much would be in the account at a) More than \$110 b) Exactly \$110 c) Less than \$110 d) Or is it impossible to tell from the info	the end of five years? Would it be: ormation given		OECD	Open- ended	EX			
37	Let's assume that you took a bank credit of during a year in equal monthly payments. rubles. Give rough estimate of the annual interest rate is about: • 3% • 6% • 9%	of 10,000 rubles to be paid back The credit charge [interest] is 600 I interest rate on your credit. The 13% I cannot estimate it even roughly	K2	WBG	Curren cy/Cult ure	EX			
38	 Which is the riskier asset to invest in? Shares in a single company stock Shares in a unit fund 	Risks are identical in both cases Don't Know		WBG		EX			
39	Assume that you deposit 100,000 rubles in interest rate. How much money will you had do not withdraw from or add to this account • More than \$100,000 rubles • Exactly \$100,000 rubles	n a bank account for 2 years at 8% ave in your account in 2 years if you nt any money? Less than \$100,000 rubles I cannot come up with even a rough number	K2	WBG	Cur/Cul	EX			
40	Assume that you saw a TV set of the sam shops. The initial retail price of it was 10, discount of 1,500 rubles, while the other of one is the better bargain—a discount of 1, • A discount of 1,500 rubles • A 10% discount	e model on sale in two different 000 rubles. One shop offered a on offered a 10% discount. Which 500 rubles or 10%? I cannot estimate it even roughly	K2	WBG	Cur/Cul	EX			

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT/FR
41	You invested \$1,000 in a stock two years ago. The stock's trading price declined 40% the first year and rose 40% the next. As a result, you've a) lost money b) made money		VCP		EX			
42	Let's say the price/earnings ratio on Standard & Poor's 500-stock index is 23 and its dividend yield is 2.5%. This means that the stock market is relatively a) undervalued by historical standards b) over valued c) fairly value		VCP			EX		
43	If a financial planner's business card says that he or she Is a Registered Investment Adviser, the planner a) meets rigorous standards set by the SEC b) is recommended by the SEC c) has simply paid a \$150 registration fee to the SEC		VCP		EX			
44	 You own shares in the Germany Fund. The value of your fund's investment in U.S. dollars would be higher if a) the dollar weakens against the Deutsche mark b) the dollar strengthens against in either; c) a change in the dollar's value doesn't matter in this case 		VCP	Cur/Cul	EX			
45	A broker recommends a municipal bond that matures in 1999 but is likely to be called, or redeemed, as early as 1994. The best gauge of your expected return is its a) current yield b) yield to maturity c) yield to call		VCP		E>	X		
46	You're considering investing in a mutual fund expected to distribute \$1 a share in dividends. You should a) buy now so you'll get the distribution b) buy after the distribution is paid c) buy either way, because it doesn't matter		VCP		EX	ntinus	d on the	2014 2022
						minueo	a on the	next page

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

# Q	uestion	ICR	Source* Rshr	IP Va	l Ver	FT/FR
47	The figure that best reflects a mutual fund's performance over a period of years is its current yield the total of dividends and capital gains it has paid its total return		VCP		EX	
48	If interest rates climb one percentage point, which of these securities would be hurt the least? 1. a 20-year zero-coupon bond 2. a 20-year bond selling at its face value 3. a 20-year bond selling at a premium above its face value		VCP	EX		
49	Investing in a mutual fund that holds a diversified portfolio of stocks protects your investment against market declines. True or false ?		VCP	EX		
50	You can't lose money investing in a U.S. Treasury bond, because it is backed by the United States Government. True or false ?		VCP	EX		`
51	 Which bank account is likely to pay the highest interest rate on money saved? 1. Savings account 2. Six month CD or certificate of deposit 3. Three year CD 		HF	I	ΞX	
52	Net worth is equal to: 1. Total assets 2. Total assets plus liabilities 3. Total assets minus liabilities		HF		EX	< compared with the second sec
53**	If your assets increase by \$5,000 and your liabilities decrease by \$3,000, your net worth would 1. Increase by \$2,000 2. Increase by \$8,000 3. Increase by \$3,000		HF			

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FF/FT
54	Savings accounts and money market accounts are most appropriate for 1. Long-term investments like retirement 2. Emergency funds and short-term goals 3. Earning a high rate of return		HF		EX			
55	On which type of loan is interest never tax deductible? 1. A home equity loan 2. An adjustable rate mortgage 3. A personal vehicle loan		HF		EX			
56	 The main advantage of a 401(k) plan is that it: 1. Provides a high rate of return with little risk 2. Allows you to shelter retirement savings from taxation 3. Provides a well-diversified mix of investment assets 		HF			EX		
57**	 To ensure that some of your retirement savings will not be subject to income tax upon withdrawal, you would contribute to: 1. A Traditional IRA or Individual Retirement Account 2. A Roth IRA 3. A 401(k) plan 		HF			Chg		
58	To ensure that some of your retirement savings will not be subject to income tax upon withdrawal, you would contribute to: 1. Treasury bills 2. Money market mutual funds 3. Balanced stock funds		HF		EX			
59	Over the long term, stocks have the highest rate of return on money invested. [True /False]		HF			EX		

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT/FR
60**	To reduce the total finance costs paid over the life of an auto loan, you should choose a loan with the 1. Lowest monthly payment 2. Longest repayment term 3. Shortest repayment term		HF					Chg
61**	If you always pay the full balance on your credit card, which of the following is least important? 1. Annual interest rate 2. Annual fees 3. Line of credit		HF					
62	 Which type of mortgage would allow a first-time homebuyer to qualify for the highest loan amount? 1. Fixed-rate mortgage 2. Adjustable-rate mortgage 3. Reverse mortgage 		HF			EX		
63.	If you have an insurance policy with a higher deductible, the premiums will be? 1. Higher 2. Lower 3. The same 4. Don't know 5. Prefer not to answer		HF		EX			
64	 Which household would typically have the greatest life insurance needs? 1. A middle-class retired couple 2. A middle-aged working couple with children in college 3. A single-earner family with two young children in preschool 4. Don't know e. Prefer not to answer 		HF		EX			

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source* Rshr	IP Val	Ver	FT/FR
65	Which of the following types of insurance is most important for single workers without children?	5	HF	EX		
	 a. Life insurance b. Disability income insurance c. Dental insurance d. Don't know e. Prefer not to answer 					
66	Which policy provides the most coverage at the lowest cost for a young family?		HF	EX		
	a. Renewable term life b. Whole life c. Universal life					
67	Checking account reconciliation involves		CV	EX		
	a. balancing bank statements with your checkbook records to determine if there are errors.					
	b. reconciling current bank statement with the previous month's statement to determine if there are errors					
	c. subtracting outstanding checks to your checkbook balance to					
	 determine if your checks have been properly processed. adding outstanding checks to your checkbook balance to improve your creditstanding. 	e				
	e. none of the above.					

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT/FR
68	 Personal finance literacy can help you A. avoid being victimized by financial scams. B. buy the right kind of insurance to protect you from catastrophic risk. C. learn the right approach to invest for your future needs. D. lead a financially secure life through forming healthy spending habits. E. do all of the above. 		CV		EX			
69	If you invest \$1,000 today at 4% for a year, your balance in a year will be A. higher if interest is compounded daily rather than monthly. B. higher if the interest is compounded quarterly rather than weekly. C. higher if the interest is compounded yearly rather than quarterly. D. \$1,040 no matter how the interest is computed. E. \$1,000 no matter how the interest is computed.		CV		EX			
70	If interest rates rise, the price of a Treasury bond will A. increase. B. decrease. C. remain the same. D. trade at a premium. E. be impossible to predict.		CV		EX			
71	 You are not overspending if A. you write checks for more than what you have in your checking account. B. your monthly wages are \$500 and credit charges \$1,000. C. you frequently receive calls from collection agencies. D. your monthly debt payment is: 30% of your take-home pay. E. you meet your minimum monthly credit card payments. 		CV		EX			

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP Va	l Ver	FT/FR
72	 An overdraft A. occurs when you write a \$1,000 dollar check when you have \$500 in account. B. is a stop-payment order written by the payee. C. will result in fines. D. All of the above. E. Both A and C. 	your	CV			EX	
73	 If you sign a twelve-month lease for \$300 per month, but never occupied the apartment, you legally owe the landlord. A. your security deposit B. your first month's rent of \$300 C. your twelve month's rent of \$3,600 D. nothing E. whatever the landlord requires. 	e	CV			EX	
74	 is not a cost of leasing an apartment. A. Security deposit B. Monthly rental payment C. Expenses incurred for non-compliance of lease terms D. Medical expenses of your friend who fell and broke his arm on the icy pavement E. Security deposit retained by the landlord for damages to property bey normal wear and tear 	h e vond	CV		EX		
75*	 Your net worth is A. the difference between your expenditures and income. B. the difference between your liabilities and assets. C. the difference between your cash inflow and outflow. D. the difference between your bank borrowing and savings E. none of the above. 		CV				Chg

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT/FR
76	The most liquid asset is A. money in a certificate of deposit account. B. money in a checking account. C. a car. D. an investment account. E. a house (real estate).		CV USIS			EX		
77	Assume you have dependent children, is a \$500 tax credit per child or a \$500 tax deduction per child more valuable to you? A. A \$500 tax credit B. A \$500 tax deduction C. They are the same D. Depends on your tax bracket E. Depends on the number of children you have		CV	Dated	EX			
78	 Which of the following is FALSE? A. As shareholders of a mutual fund, you have a right to tell fund managers what securities to buy. B. A mutual fund is a diversified collection of securities used as an investment vehicle. C. A mutual fund is an investment corporation that raises funds from investors and purchases securities. D. Your ownership in a mutual fund is proportional to the number of shares you own in the fund. E. None of the above. 		CV		EX			
79	The returns from a balanced mutual fund include A. interest earned on cash in the fund. B. dividends from common stock in the fund. C. interest earned on bonds in the fund. D. capital gains from stocks and bonds in the fund. E. all of the above.		CV		EX			
Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT/FR
79	 Which of the following investments requires that you keep your money invested for a specified period or face an early withdrawal penalty? A. Certificate of deposit B. Checking account that pays interest C. Government savings bond D. Money market mutual fund E. Passbook savings account 		CV		EX			
80	 Which of the following investments requires that you keep your money invested for a specified period or face an early withdrawal penalty? A. Certificate of deposit B. Checking account that pays interest C. Government savings bond D. Money market mutual fund E. Passbook savings account 		CV			EX		
81	 Personal financial planning involves A. establishing an adequate financial record keeping system. B. developing a sound yearly budget of expenses and income. C. minimizing taxes and insurance expenses. D. preparing plans for future financial needs and goals. E. examining your investment portfolios to maximize returns. 		CV		EX			
82	A dollar-cost-averaging approach to investing involves A. buying low and selling high. B. complex calculations of risk and return. C. selling securities to minimize capital.		CV		EX			
83	 No-load mutual funds are recommended over load funds because investors A. do not pay for 12b-l fees. B. can reduce their tax liability. C. are not charged with sales commissions. D. can avoid the funds' administrative expenses. E. believe that the funds have no management charges. 	5	CV		EX			

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT/FR
84	 No-load mutual funds are recommended over load funds because investors A. do not pay for 12b-l fees. B. can reduce their tax liability. C. are not charged with sales commissions. D. can avoid the funds' administrative expenses, E believe that the funds have no management charges. F. Don't know G. Prefer not to answer 		CV		EX			
85	 Which of the following statements is TRUE about the annual percentage rate (APR)? A. APR is the actual rate of interest paid over the life of the loan. B. APR is expressed as a percentage on an annual basis. C. APR is a good measure of comparing loan costs. D. APR takes into account all loan fees. E. All of the above 		CV			EX		
86	 You will improve your credit worthiness by A. visiting your local commercial bank. B. showing no record of personal bankruptcies in recent years. C. paying cash for all goods and services. D. borrowing large amounts of money from your friends. E. donating money to charity. 		CV		EX			
87**	You can receive your credit report from A. a credit union. B. a commercial bank. C. the Better Business Bureau. D. a credit bureau. E. a professor.		CV			Chg	Chg	Chg

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Prcess

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT/FR
88	 Which is FALSE concerning credit cards? A. You can use your credit card to receive a cash advance. B. If your credit card balance is \$1,000 and you pay \$300, interest is charged on the unpaid balance of \$700. C. The rate of interest on your credit card is normally higher than you can earn on a certificate of deposit. D. A credit card company will not charge you interest if you payoff the entire balance by the due date. E. you cannot spend more than your line of credit. 		CV		EX			
89	 If you co-sign a loan for a friend, then A. you become responsible: for the loan payments if your friend defaults. B. it means that your friend cannot receive the loan by himself. C. you are entitled to receive part of the loan. D. Both A and B. E. Both A and C. 		CV			EX		
90	 The main reason to purchase insurance is to A. protect you from a loss recently incurred. B. provide you with excellent investment returns. C. protect you from sustaining a catastrophic loss. D. protect you from small incidental losses. E. improve your standard of living by filing fraudulent claims. 		CV		EX			
91	 Which of the following statements is FALSE? A. Term insurance is an excellent investment vehicle. B. You receive no benefits when your term insurance policy expires. C. A term insurance policy is the least expensive form of life insurance. D. A decreasing-term policy reduces coverage over time. E. A level-term policy guarantees a fixed premium over the life of the contract. 		CV		EX			

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT/FR
92	 The main reason to purchase health insurance is: A. After buying health insurance, you are normally covered for pre- existing conditions. B. You have a better chance to choose doctors with a health maintenance organization rather than with a traditional health care insurance company. C. Most policies contain deductible and coinsurance clauses. D. A policy purchased by the individual is cheaper than one purchased through a group. E. None of the above. 		CV		EX			
93	 Auto insurance companies determine your premium based on A. age of insured. B. record of accidents. C. type and age of vehicle. D. completion of a driver education course. E. all of the above. 		CV		EX			
94	 A high-risk and high-return investment strategy would be most suitable for A. an elderly retired couple living on a fixed income. B. a couple needing funds for their children's education in two years. C. a young married couple without children. D. all of the above because they all need high return. E. none of the above because they are equally risk adverse. 		CV USIS		EX			
95	 Your savings accounts in a federally insured commercial bank are insured by A. SIPC to the maximum amount of \$10,000 per account. B. FDIC to the maximum amount of \$100,000. C. FDIC to the maximum amount of \$50,000 per account. D. SLIC to the maximum amount of \$100,000. E. FNMA to the maximum amount of \$100,000 per account. 		CV		EX			

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT/FR
96	would not ordinarily be covered under a home owner's policy. A. War B. Earthquake C. Flood D. Your being sued by someone for slander E. All of the above		CV		EX			
97	 You have a better chance of resolving a complaint against an insurance company by bringing the issue to a government agency at the A. federal level. B. state level. C. county level. D. township level. E. None of the above. 		CV		EX			
98	Over the long term, stocks have the highest rate of return on money invested. [True/ False]		HHB		EX			
99	Employers are responsible for providing the minority of funds that you will need for retirement. [True/ False]		HHB	Dated				
100**	With compound interest, you earn interest on your interest, as well as on your principal. [True /False]		HHB			Chg	Chg	Chg
101**	The earlier you start saving for retirement, the more money you will have because the effects of compounding interest increase over time. [True /False]		HHB			Chg		
102	Your bank will usually call to warn you if you write a check that would overdraw your account. [True/ False]		HHB	Dated				
103**	You should have an emergency fund that covers two to six months of your expenses. [True /False]		HHB			Chg		

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT/FR
104	If you have a savings account at a bank, you may have to pay taxes on the interest you earn. [True/False]		HHB				Chg	EX
105	Mutual funds pay a guaranteed rate of return. [True/False]		HHB		ΕX			
106	A stock mutual fund combines the money of many investors to buy a variety of stocks. [True /False]		HHB			EX		
107**	If the interest rate on an adjustable-rate mortgage loan goes up, your monthly mortgage payments will also go up. [True /False]		HHB				Chg	
108**	Repeatedly refinancing your home mortgage over a short period of time results in added fee and points that further increase your debt. [True /False]		HHB					
109	The finance charge on your credit card statement is what you pay to use credit. [True /False]		HHB			EX		
110	If your credit card is stolen and someone uses it before you report it missing, you are only responsible for \$50 no matter how much they charge on it. [True /False]		ННВ	Dated				
111	Using extra money in a bank savings account to pay off high interest rate credit card debt is a good idea. [True /False]		HHB			EX		
112**	Your credit report includes employment data, your payments history, any inquiries made by creditors, and any public record information. [True/ False]		ННВ					Chg
113	If you are behind on debt payments and go to a credit counseling service, they can get the federal government to apply your income tax refund to pay off your debts. [True/ False]		HHB		EX			

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT/FR
114	If you have any negative information on your credit report, a credit repair agency can help you remove that information. [True/ False]		HHB		EX			
115	Creditors are required to tell you the APR that you will pay when you get a loan. [True /False]		HHB			EX		
116	Your credit rating is not affected by how much you charge on your credit cards. [True/ False]		HHB			EX		
117	If you expect to carry a balance on your credit card, the APR is the most important thing to look at when comparing credit card offers. [True/ False]		HHB			EX		
118	After signing a contract buying a new car, you have three days to change your mind. [True/ False]		HHB			EX		
119	When you use your home as collateral for a loan, there is no chance of losing your home. [True/ False]		HHB	Dated				
120	Making payments late on your bills can make it more difficult to take out a loan. [True/ False]		HHB			EX		
121	You could save thousands of dollars in interest costs by choosing a 15-year rather than a 30-year mortgage. [True/False]		HHB		ΕX			
122**	If you buy certificates of deposit, savings bonds, or Treasury bills, you can earn higher returns than on a savings account, with little or no added risk. [True /False]		HHB					Chg
123	All investment products bought at your bank are covered by FDIC insurance [True/ False]		HHB		EX			
124	If you start out with \$1,000 and earn an average return of 10% per year for 30 years, after compounding, the initial \$1,000 will have grown to more than \$6,000. [True /False] ALTERNATIVE WORDING		CogE		EX			
	If you start out with \$1,000 and earn an average return of 10% per year for 30 years, even after compounding, the initial \$1,000 will have grown to less than \$6,000. [True/ False]							

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source* Rshr	IP	Val	Ver	FT/FR
125	If the interest rate falls, bond prices will rise. [True /False] ALTERNATIVE WORDING		CogE	EX			
	If the interest rate falls, bond prices will fall. [I rue/False]						
126	It is better for young people saving for retirement to combine stocks with long-term (inflation protected) bonds than with short-term (inflation protected) bonds. [True /False] ALTERNATIVE WORDING		CogE	EX			
	It is better for young people saving for retirement to combine stocks with short-term (inflation protected) bonds than with long-term (inflation protected) bonds. [True/ False]						
127	Taxes affect how you should invest your money. [True/False] ALTERNATIVE WORDING		CogE		EX		
	Taxes do not affect how you should invest your money. [True/False]						
128	If you invest for the long run, the annual fees of mutual funds are important. [True /False]		CogE	EX			
	ALTERNATIVE WORDING						
	If you invest for the long run, the annual fees of mutual funds are unimportant. [True/ False]						
129	It is possible to invest in the stock market in a way that makes it hard for people to take unfair advantage of you. [True /False] ALTERNATIVE WORDING		CogE	EX			
	There is no way to avoid people taking advantage of you if you try to invest in the stock market. [True/ False						

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source* R	shr	IP	Val	Ver	FT/FR
130	It is a good idea to own stocks of foreign companies. [True/ False] ALTERNATIVE WORDING		CogE		EX			
	It is best to avoid owning stocks of foreign companies. [True/False]							
131	Even older retired people should hold some stocks. [True/ False] ALTERNATIVE WORDING		CogE		EX			
	Older retired people should not hold any stocks. [True/False]							
132	When an investor spreads money between 20 stocks, rather than 2, the risk of losing a lot of money decreases. [True /False] ALTERNATIVE WORDING		CogE		EX			
	When an investor spreads money between 20 stocks, rather than 2, the risk of losing a lot of money increases. [True/ False]							
133	To make money in the stock market, you should not buy and sell stocks too often. [True/ False] ALTERNATIVE WORDING		CogE		EX			
	To make money in the stock market, you have to buy and sell stocks often. [True/ False]							
134	Mutual funds do not pay a guaranteed rate of return. [True /False] ALTERNATIVE WORDING		CogE		EX			
	Mutual funds pay a guaranteed rate of return. [True/False]							

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source* Rshr	IP	Val	Ver	FT
135	It is easy to find mutual funds that have annual fees of less than one percent of assets. [True/ False] ALTERNATIVE WORDING		CogE	EX			
	It is hard to find mutual funds that have annual fees of less than one percent of assets. [True/ False]						
136	Buying a stock mutual fund usually provides a safer return than a single company stock. [True/ False] ALTERNATIVE WORDING		CogE	EX			
	Buying a single company stock usually provides a safer return than a stock mutual fund. [True/ False]						
137	You should invest in either mutual funds or a large number of different stocks instead of just a few stocks. [True/ False] ALTERNATIVE WORDING		CogE	EX			
	You should invest most of your money in a few good stocks that you select rather than in lots of stocks or in mutual funds. [True/ False]						
138	Even if you are smart, it is hard to pick individual company stocks that will have better than average returns. [True /False] ALTERNATIVE WORDING		CogE	EX			
	If you are smart, it is easy to pick individual company stocks that will have better than average returns. [True/ False]						

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT/FR
139	An employee of a company with publicly traded stock should have little or none of his or her retirement savings in the company's stock. [True /False] ALTERNATIVE WORDING		CogE		EX			
	An employee of a company with publicly traded stock should have a lot of his or her retirement savings in the company's stock. [True/ False]							
140	An investment advisor tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate would double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would be worth \$4,000 after 40 years (by the time they are 70). [True/False] ALTERNATIVE WORDING		CogE		EX			
	An investment advisor tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate would double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would not be worth \$4,000 for at least 45 years (until they are at least 75). [True/ False]							
141	It is important to take a look at your investments periodically to see if you need to make changes. [True /False] ALTERNATIVE WORDING		CogE		EX			
	Once you have made an initial decision about the investment mix for your portfolio, you should avoid making changes to your portfolio until you are close to retirement. [True/ False]							
142	You could save money in interest costs by choosing a 15-year rather than a 30-year mortgage. [True /False] ALTERNATIVE WORDING		CogE		EX			
	You could save money in interest costs by choosing a 30-year rather than a 15-year mortgage.[True/ False]				С	ontinue	d on the	next page

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

# Qu	iestion	ICR	Source* Rshr	IP	Val	Ver	FT/FR
143	Using money in a bank savings account to pay off credit card debt is usually a good idea. [True /False] ALTERNATIVE WORDING		CogE	EX			
	Using money in a bank savings account to pay off credit card debt is usually a bad idea. [True/ False]						
144	For a family with a working husband and a wife staying home to take care of their young children, life insurance that will replace three years of income is not enough life insurance. [True /False] ALTERNATIVE WORDING		CogE	EX			
	For a family with a working husband and a wife staying home to take care of their young children, life insurance that will replace three years of income is more than enough. [True/ False]						
145	Buying a single company stock usually provides a safer return than a stock mutual fund. [True /False] ALTERNATIVE WORDING		CogE	EX			
	Buying a stock mutual fund usually provides a safer return than a single company stock. [True/ False]						
146	Young people should hold somewhat riskier financial investments than older people. [True /False] ALTERNATIVE WORDING		CogE	EX			
	Older people should hold somewhat riskier financial investments than young people. [True/ False]						

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source* Rshr	IP	Val	Ver	FT/FR
147	The more you diversify among stocks, the more of your money you can invest in stocks. [True /False] ALTERNATIVE WORDING		CogE	EX			
	The more you diversify among stocks, the less of your money you should invest in stocks. [True/ False]						
148	 Inflation can cause difficulty in many ways. Which group would have the greatest problem during periods of high inflation that last several years? a) Older, working couples saving for retirement. b) Older people living on fixed retirement income. c) Young couples with no children who both work. d) Young working couples with children. 		J\$	EX			
149	 Which of the following types of investment would best protect the purchasing power of a family's savings in the event of a sudden increase in inflation? a) A 10-year bond issued by a corporation. b) A certificate of deposit at a bank. c) A 25-year corporate bond. d) A house financed with a fixed-rate mortgage. 		J\$		EX		
150	 Which of the following statements is NOT correct about most ATM (Automated Teller Machine) cards? a) You can generally get cash 24 hours-a-day. b) You can generally obtain information concerning your bank balance at an ATM machine. c) You can get cash anywhere in the world with no fee. d) You must have a bank account to have an ATM Card. 		J\$			EX	

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP Val	Ver	FT
151	 Which of the following instruments is NOT typically associated with spending? a) Debit card. b) Certificate of deposit. c) Cash. d) Credit card 		J\$		EX		
152	David just found a job with a take-home pay of \$2,000 per month. He must pay \$900 for rent and \$150 for groceries each month. He also spends \$250 per month on transportation. If he budgets \$100 each month for clothing, \$200 for restaurants and \$250 for everything else, how long will it take him to accumulate savings of \$600. a) 3 months. b) 4 months. c) 1 month. d) 2 months.		\$L		EX		
153	 Many people put aside money to take care of unexpected expenses. If Juan and Elva have money put aside for emergencies, in which of the following forms would it be of LEAST benefit to them if they needed it right away? a) Invested in a down payment on the house. b) Checking account. c) Stocks. d) Savings account 		J\$		EX		
154	 Which of the following best describes the primary sources of income for most people age 20-35? a) Dividends and interest. b) Salaries, wages, tips. c) Profits from business. d) Rents. 		J\$	Inc			

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT
155	Don and Bill work together in the finance department of the same company and earn the same pay. Bill spends his free time taking work-related classes to improve his computer skills; while Don spends his free time socializing with friends and working out at a fitness center. After five years, what is likely to be true? a) Don will make more because he is more social. b) Don will make more because Bill is likely to be laid off. c) Bill will make more money because he is more valuable to his company. d) Don and Bill will continue to make the same money.		J\$	Inc				
156	 Matt has a good job on the production line of a factory in his hometown. During the past year or two, the state in which Matt lives has been raising taxes on its businesses to the point where they are much higher than in neighboring states. What effect is this likely to have on Matt's job? a) Higher business taxes will cause more businesses to move into Matt's state, raising wages. b) Higher business taxes can't have any effect on Matt's job. c) Matt's company may consider moving to a lower-tax state, threatening Matt's job. d) He is likely to get a large raise to offset the effect of higher taxes. 		J\$	Econ				
157	 If you went to college and earned a four-year degree, how much more money could you expect to earn than if you only had a high school diploma? a) About 10 times as much. b) No more; I would make about the same either way. c) A little more; about 20% more. d) A lot more; about 70% more. 		J\$	Inc				

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

# Qu	estion	ICR	Source*	Rshr	IP	Val	Ver	FT/FR
158	 If you had a savings account at a bank, which of the following would be correct concerning the interest that you would earn on this account? a) Earnings from savings account interest may not be taxed. b) Income tax may be charged on the interest if your income is high enough. c) Sales tax may be charged on the interest that you earn. d) You cannot earn interest until you pass your 18th birthday 		J\$			EX		
159	 Your take home pay from your job is less than the total amount you earn. Which of the following best describes what is taken out of your total pay? a) Social security and Medicare contributions. b) Federal income tax, property tax, and Medicare and social security contributions. c) Federal income tax, social security and Medicare contributions. d) Federal income tax, sales tax, and social security contribution. 		\$L		EX			
160	 Which of the following is true about sales taxes? a) The national sales tax percentage rate is 6%. b) The federal government will deduct it from your paycheck. c) You don't have to pay the tax if your income is very low. d) It makes things more expensive for you to buy. 		J\$		EX			
161	 Chelsea worked her way through college earning \$15,000 per year. After graduation, her first job pays \$30,000. The total dollar amount Chelsea will have to pay in Federal Income taxes in her new job will: a) Double, at least, from when she was in college. b) Go up a little from when she was in college. c) Stay the same as when she was in college. d) Be lower than when she was in college. 		J\$		EX			

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT
162	Rebecca has saved \$12,000 for her college expenses by working part-time. Her plan is to start college next year and she needs all of the money she saved. Which of the following is the safest place for her college money? a) Locked in her closet at home. b) Stocks. c) Corporate bonds. d) A bank savings account.		J\$			EX		
163	 Sara and Joshua just had a baby. They received money as baby gifts and want to put it away for the baby's education. Which of the following tends to have the highest growth over periods of time as long as 18 years? a) A checking account. b) Stocks. c) A U.S. Govt. savings bond. d) A savings account. 		J\$			EX		
164	 Rob and Mary are the same age. At age 25, Mary began saving \$2,000 a year while Rob saved nothing. At age 50, Rob realized that he needed money for retirement and started saving \$4,000 per year while Mary kept saving her \$2,000. Now they are both 75 years old. a) They would each have the same amount because they put away exactly the same. b) Rob, because he saved more each year. c) Mary, because she has put away more money. d) Mary, because her money has grown for a longer time at compound interest. 		J\$			EX		

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FTFR
165	Retirement income paid by a company is called: a) 401 (k). b) Pension. c) Rents and profits. d) Social Security.		J\$		EX			
166	 Scott and Eric are young men. Each has a good credit history. They work at the same company and make approximately the same salary. Scott has borrowed \$6,000 to take a foreign vacation. Eric has borrowed \$6,000 to buy a car. Who is likely to pay the lowest finance charge? a) Eric will pay less because the car is collateral for the loan. b) They will both pay the same because the rate is set by law. c) Scott will pay less because people who travel overseas are better risks. d) They will both pay the same because they have almost identical financial backgrounds. 		\$L			EX		
167	 Under which of the following circumstances would it be financially beneficial to you to borrow money to buy something now and repay it with future income? a) When you need to buy a car to get a much better paying job b) When you really need a week vacation. c) When some clothes you like go on sale. d) When the interest on the loan is greater than the interest you get on your savings. 		\$L			EX		
168	 Dan must borrow \$12,000 to complete his college education. Which of the following would NOT be likely to reduce the finance charge rate? a) If he went to a state college rather than a private college. b) If his parents cosigned the loan. c) If his parents took out an additional mortgage on their house for the loan. d) If the loan was insured by the Federal Government 		J\$		ΕX			

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT/FR
169 170	 Which of the following statements is true? a) Banks and other lenders share the credit history of their borrowers with each other and are likely to know of any loan payments that you have missed. b) People have so many loans it is very unlikely that one bank will know your history with another bank c) Your bad loan payment record with one bank will not be considered if you apply to another bank for a loan. d) If you missed a payment more than 2 years ago, it cannot be considered in a loan decision. Which of the following statements best describes your right to check your credit history for accuracy? a) Your credit record can be checked once a year for free. 		J\$ J\$			EX	EX	
	 b) You cannot see your credit record. c) All credit records are the property of the U.S. Government and access is only available to the FBI and Lenders. e) You can only check your record for free if you are turned down for credit based on a credit report. 							
171	 If you are behind on your debt payments and go to a responsible credit counseling service such as the Consumer Credit Counseling Services, what help can they give you? a) They can cancel and cut up all of your credit cards without your permission. b) They can get the federal government to apply your income taxes to pay off your debts. c) They can work with those who loaned you money to set up a payment schedule that you can meet. d) They can force those who loaned you money to forgive all your debts. 	ıt r o	J\$		EX			

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT/FR
172	 Which of the following credit card users is likely to pay the GREATEST dollar amount in finance charges per year, if they all charge the same amount per year on their cards? a) Jessica, who pays at least the minimum amount each month and more, when she has the money. b) Vera, who generally pays off her credit card in full but, occasionally, will pay the minimum when she is short of cash c) Megan, who always pays off her credit card bill in full shortly after she receives it d) Erin, who only pays the minimum amount each month. 		J\$		EX			
173	If your credit card is stolen and the thief runs up a total debt of \$1,000, but you notify the issuer of the card as soon as you discover it is missing, what is the maximum amount that you can be forced to pay according to Federal law? a) \$500 b) \$1000 c) Nothing d) \$50		J\$	Dated				
174	 Barbara has just applied for a credit card. She is an 18-year-old high school graduate with few valuable possessions and no credit history. If Barbara is granted a credit card, which of the following is the most likely way that the credit card company will reduce ITS risk? a) It will make Barbara's parents pledge their home to repay her credit card debt. b) It will require Barbara to have both parents co-sign for the card. c) It will charge Barbara twice the finance charge rate it charges older cardholders. d) It will start Barbara out with a small line of credit to see how she handles the account. 		J\$				EX	

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT/FR
175	 If each of the following persons had the same amount of take home pay, who would need the greatest amount of life insurance? a) An elderly retired man, with a wife who is also retired. b) A young married man without children. c) A young single woman with two young children. d) A young single woman without children. 		J\$		EX			
176	If you have caused an accident, which type of automobile insurance would cover damage to your own car? a) Comprehensive. b) Liability. c) Term. d) Collision.		J\$		EX			
177	 Many young people receive health insurance benefits through their parents. Which of the following statements is true about health insurance coverage a) You are covered by your parents' insurance until you marry, regardles of your age. b) If your parents become unemployed, your insurance coverage may stop, regardless of your age. c) Young people don't need health insurance because they are so healthy. d) You continue to be covered by your parents' insurance as long as you live at home, regardless of your age. 	s. ? ss	J\$	Dated				
178	 Many savings programs are protected by the Federal government against loss. Which of the following is not? a) A U. S. Savings Bond. b) A certificate of deposit at the bank. c) A bond issued by one of the 50 States. d) A U. S. Treasury Bond. 		J\$		E	x		

Table C1 (continued)

Initial Pool of Financial Literacy Questions for Content Validation Process

#	Question	ICR	Source*	Rshr	IP	Val	Ver	FT/FR
179**	Bank A offers monthly compounding and Bank B offers yearly compounding. Both banks offer the same interest rate. In your opinion, which bank would you choose if you wanted a higher return? a) Bank A b) Bank B c) Both banks offer the same return d) I do not know		USIS					
180	If interest rates are currently low but are expected to rise, which one of the following is the most appropriate action? a) Use long-term loans and short-term savings. b) Use short-term loans and long-term savings. c) Do nothing d) I do not know		USIS		EX			
181	You are likely to obtain the lowest interest on loans with? a) Credit card companies b) Banks/credit unions c) Check-cashing firms d) I do not know.		USIS			EX		
182**	Do you think buying stock in a single company is safer than buying stock in several different companies? a) Yes b) No		USIS					

Note: ICR=IRT/CIT Review, Rshr=Researcher, IP=Initial Panel, Val=Validation Panel, Ver=Verification Panel, FT/FR=Field Test/Final Review; * See Table C2 for key to abbreviations, source and reference;

**Included in final instrument; Dated=question eliminated due to not being current subject matter; Cur/Cul=question eliminated due to currency/culture not applicable to this study's participants; Inc/Econ=questions outside this study's content areas; EX=excluded by expert panel; Chg=wording changed by expert panel suggestions; Correct answers are highlighted in bold.

Table C2

Key to Pool of Financial Literacy Questions Source Codes: Source, Number (n), and References

Code	Source	n	Reference
ALP	SurveyAmerican Life Panel	23	Lusardi & Mitchell, 2007; RAND American Life Panel, 2014
CogE	Cognitive Economics Journal Article	24	Hsu, 2011
CV	Journal Article	31	Chen & Volpe, 1998
DHS	Survey—De Nederlandsche Bank Household Survey	16	van Rooij et al., 2007, 2009, 2011
FH	Journal Article	16	Finke & Huston, 2014
FLN	Journal Article	13	Fernandes et al., 2014
HHB	Journal Article	28	Hilgert, Hogarth, & Beverly, 2003
HRS	SurveyHealth and Retirement Study	3	Agnew, 2013; Beckmann, 2013; Gale & Levine, 2010; Lusardi, 2012; Lusardi & Mitchell, 2008, 2011; Lusardi, Mitchell, & Curto, 2010
J\$	Jump\$tart	31	JumpStart Coalition, 2007; Mandell, 2006, 2008; Mandell & Klein, 2007, 2009
K1	Journal Article	20	Knolls & Houts, 2012
K2	Journal Article	6	Kunovskayaet al., 2014
NFCS	SurveyNational Financial Capabilities Survey	5	Bumcrot et al., 2013; Fernandes et al., 2014; Hastings, Madrian, & Skimmyhorn, 2012; Lachance, 2014; Lusardi, 2011; Mottola, 2013; Nicolini et al., 2013; Scheresberg, 2013; Simms, 2014

Table C2 (continued)

Key to Pool of Financial Literacy Questions Source Codes: Source, n, and References

Code	Source	n	Reference
OEDC	SurveyOrganization for Economic Co-operation and Development	8	Atkinson & Massey, 2011
TNS	SurveyTaylor, Nelson, Sofres Global Survey	3	Lusardi & Tufano, 2009
USIS	University of Southern Indiana Study	7	Alhenawi & Elkhal, 2013
VCP	Journal Article	10	Volpe et al., 1996
WBG	SurveyWorld Bank Group	4	World Bank, 2009

Appendix D

Panel Members

Table D1

Initial Panel Members and Their Areas of Expertise

Panel Member		Expertise	Organization	
Heba	Abuzayyad, MBA, PhD Candidate	Adult Education Economics Research & Measurement	USF Graduate Student	
Levon	Blue, PhD	Financial Literacy Adult Education	Griffith University, Australia	
Charlene	Bostic, MBA	Small Business Economics	SBDC at USF	
Wayne	Brass	Small Business	SBDC at USF	
Elizabeth	Brietcbach, PhD	Financial Literacy Economics	University of Arkansas	
Dean	Douglas, CPA, MA	Small Business Banking Accounting	Vertex, Inc.	
Mario	lezzoni, CPA, MBA	Small Business Economics Accounting	City of New Port Richey	
Carol	Johns, MBA	Small Business Banking	SBDC at USF	
Deborah	Kozdras, PhD	Financial Literacy Small Business	Stravos Center	
Kimberly	Martin, CPA	Small Business Accounting	Vertex, Inc.	

Table D1 (continued)

Initial Panel Members and Their Areas of Expertise

Panel Member		Expertise	Organization
Bill	McKown, MBA	Financial Literacy Small Business	SBDC at USF
David	Noel, MA	Adult Education Small Business	SBDC at USF
Holly	Rick, PhD	Financial Literacy Economics	University of Phoenix
Eileen	Rodriguez, MBA*	Financial Literacy Small Business	SBDC at USF Regional Director
Rhonda	Rogers, CPA, MA	Financial Literacy Small Business Accounting	Vertex, Inc.
David	Scott	Financial Literacy	Stravos Center
Dmitry	Shevchneko, PhD	Adult Education Economics	Southern Federal University, Russia
Tammy	Talerico, MBA	Financial Literacy	Vertex, Inc.
James	Taylor, MBA	Adult Education Small Business Financial Planning	SBDC at USF
Ashley	Tharayil, PhD	Financial Literacy Economics	University of Nebraska

Note: *Participated in 1st round only; SBDC—Small Business Development Center, USF--University of South Florida

Table D2

Validation Panel Members and Their Areas of Expertise

Panel Member		Expertise	Organization
Andrea	Bennett, MA, PhD Candidate	Adult Education	USF Graduate Student
Rebecca	Hodges, CPA, MA	Accounting	Vertex, Inc.
Monique	Joanette. MA	Small Business Financial Literacy Accounting	Vertex, Inc.
Natasha	McKenna, MA	Adult Education	University of Toronto, Canada
Hasan	Nuseibeh, MBA, PhD Candidate	Economics Finance	USF Graduate Student USF Adjunct Instructor
Wendy	Plant, MBA	Small Business	Florida State University

Table D3

Verification Panel Members and Their Areas of Expertise

Panel Member		Expertise	Organization	
Pele	Bagwell, CPA, MBA, PhD Student	Economics Finance Accounting	Coca-Cola Company	
Eileen	Rodriguez, MBA	Financial Literacy Small Business	SBDC at USF	
George	Salis, PhD, JD	Economics Finance Accounting	Vertex, Inc.	

Note: SBDC—Small Business Development Center, USF--University of South Florida

Table D4

Final Review Panel Members and Their Areas of Expertise

Panel Member		Expertise	Organization
Pele	Bagwell, CPA, MBA PhD Student	Economics Finance Accounting	Coca-Cola
Levon	Blue, PhD	Financial Literacy Adult Education	Griffith University
Elizabeth	Brietcbach, PhD	Financial Literacy Economics	Moore School of Business
Mario	lezzoni, CPA, MBA	Small Business Economics Accounting	Economic Development Director New Port Richey USF Adjunct Professor
Lynne	Key, PhD	Adult Education	Citigroup
Cheryl	Kirby, PhD	Small Business	FSBDCN Director
Wendy	Plant, MBA	Small Business	FSU Professor
Holly	Rick, PhD	Financial Literacy Economics	University of Phoenix
Yanina	Rosario, MBA	Small Business	SBDC Associate Director
Tanya	Scotese, PhD	Adult Education	Farley Funeral Home
James	Taylor, MBA PhD Candidate	Adult Education Small Business Financial Planning	SBDC Consultant USF Graduate Student
Ashley	Tharayil, PhD	Financial Literacy Economics	University of Nebraska

Note: FSBDCN—Small Business Development Center Network, SBDC—Small Business Development Center, USF--University of South Florida

Appendix E

Request for Participation to Potential Panel Members Email

From: Pearl Para pjdahmen@mail.usf.edu Date: Tue, 9 Jun 2015 11:33:10 -0400 Subject: Small Business Financial Literacy Research To: <Potential Expert Panel Members> I am a PhD Candidate at the University of South Florida completing my dissertation (Financial Literacy Continuing Professional Education Needs Assessment for Florida Small Business Owners). I have also been an SBDC consultant in Florida since 2011. As part of my research, I am developing a financial literacy assessment instrument. Your background would be an excellent addition to my expert panel. I suspect you are wondering what the expert panel members' responsibilities will entail. Let me give you a brief overview. The total anticipated time required is estimated to be less than an hour. All evaluation work will be in the form of surveys via the Internet, which will be also mobile device accessible. The tasks panel members will be asked to complete include: · Reviewing existing financial literacy questions for inclusion · Review question for appropriateness of language · Ranking questions order of suitability · Rating question difficulty Each task is approximately 20 to 30 minutes, and you will only be asked to complete two. Explicit instructions will accompany the surveys. Please click on the link below to respond to this email. Feel free to respond using a smartphone or tablet. http://usf.qualtrics.com/SE/?SID=SV_50hLHZXZm2RhfeJ If you have any questions, or would like to discuss my research, please feel free to contact me at pjdahmen@mail.usf.edu. Thank you. Pearl Dahmen Para PhD Candidate University of South Florida

Appendix F

Initial Panel Communications

Initial Panel Confirmation & Round 1 Email

Subje To: Pe	ct: Financial Literacy Expert Panel earl Dahmen < <u>pjdahmen@mail.usf.edu</u> >
Hello	Panel Expert,
Thank	you, again, for agreeing to serve as a member of the Expert Panel for my dissertation research.
This e	mail is to provide you with instructions and a link to the online review survey.
The re which profes	esults of your involvement will contribute to the development of the basic financial literacy assessment instrument, will be used in my research. The purpose of my research study is to identify the financial literacy continuing sional education needs of Florida small business owners.
There should so des	are 25 items in the Round 1 survey. You are asked to deliberate on each financial concept to determine whether it I be included in a financial literacy assessment. You will also be given an opportunity to provide comments if you ire.
Please finaliz assista	complete this first round promptly as there are several more rounds of review before the assessment can be ted. The estimated time to complete this review is less than 15 minutes. If you have any questions or need ince, please feel free to contact me at <u>pjdahmen@mail.usf.com</u> or <u>941-962-1211</u> .
To sta brows	rt the survey, please click on this link, or copy or paste it to your er, <u>http://usf.qualtrics.com/SE/?SID=SV_bE0dEmJawtysiCV</u> . Then, follow the instructions are below.
NOTI	E: This survey is mobile friendly and can be accessed with a smartphone or tablet as well as a computer or laptop.
INST	RUCTIONS
The fo refer t	ollowing instructions will be on the opening screen of the survey. Please feel free to print this email so you may o them as you complete the survey.
Based impor includ the co	on your knowledge of financial concepts, please evaluate each concept listed below to determine whether it is tant in assessing the financial literacy of adults age 18 and over. Then, respond YES if you feel it should be led, and NO if not. If you selected YES, please identify which of the four content categories defined below you fee ncept belongs.
The co	ontent category definitions are:
Mone	y Basics—concepts necessary to understand and manage money or financial resources.
Borro fee fo	wing—concepts relating to the debt, using money from a financial institution under an agreement to repay it plus a r the privilege of using it.
Build	ing assets—concepts relating to growing and increasing financial resources.
Prote	cting assets—concepts relating to avoiding the possibility of loss of money and/or financial resources.
Please	feel free to comment on any of the concepts in the list.
Best r	egards,
Pearl I Unive	Dahmen Para rsity of South Florida, PhD Candidate

Initial Panel Round 2 Email

------ Forwarded message ------From: Pearl Para <pjdahmen@mail.usf.edu> Date: Sun, Jul 12, 2015 at 8:30 AM Subject: Financial Literacy Expert Panel Round 2 To: <email.address>

Dr. <Name>,

Thank you for completing expert panel round 1 and participating in this valuable financial literacy research. In this second round of review, you will be evaluating 32 questions in the pool for the basic financial literacy instrument.

The estimated time to complete the review is 30 minutes or less. Please complete your review as soon as possible. Your input will determine the questions included in future rounds of review by other members of the expert panel. For your convenience, survey allows you to stop at any point. It will automatically save and later return to the place you stopped. If you have any questions or need assistance, please feel free to contact me at <u>pjdahmen@mail.usf.com</u> or <u>941-962-1211</u>.

To start the survey, please click on this link, or copy and paste it to your browser: http://usf.gualtrics.com/SE/?SID=SV_bfQnMVIYcbzuMId Then, follow the instructions below.

NOTE: This survey is mobile friendly and can be accessed with a smart phone or tablet as well as a computer or laptop.

INSTRUCTIONS

The following instructions will be on the opening screen of the survey. Please feel free to print this email so you may refer to them as you complete the survey.

As you read each question and possible answers (correct answer is bold), identify whether you believe the question should be included or excluded from an instrument used to determine a basic financial literacy score. The goal of this round is to drastically reduce the number of questions in the pool. You are reviewing less than 20% of the pool, therefore, please feel free to be critical in your selection of the questions to include. Only included question you strongly feel should be included; all other questions should be excluded. Questions included will stay in the pool for the next round of review. If you determine a question should be excluded, please identify the reason you believe it should be excluded, please select one of the listed reasons for exclusion:

- Age/life cycle bias Question content is directed toward a specific age group or life cycle. A participant not in the age
 group or life cycle would probably respond with an incorrect answer.
- Judgment or belief bias--The question's response could be influenced by the participant's personal judgments or beliefs. Participants with different beliefs would respond differently to this question.
- Too specific--The content of this question is too specific and would not necessarily be financial knowledge useful to the population majority.
- Too vague--The content of this question is too general or vague and would not be a good assessment of financial literacy knowledge. Participants could easily guess the answer without having financial knowledge.
- Other-Any questions you feel would not be representative of a participant's financial literacy. If you choose this
 response, please note your reason and/or use the comment section to explain.

Please feel free to comment on any of the question you identify to include for the next round.

Best regards,

Pearl Dahmen Para University of South Florida, PhD Candidate

Appendix G

Initial Panel Round 1 Online Survey



Based on your financial knowledge, please consider each of the concepts listed below to determine whether it is important in assessing the basic financial literacy of adults age 18 and over. Then, respond YES if you feel it should be included, and NO if not. If you selected YES, please identify the content category the concepts belongs. The categories are defined below.

Money Basics—concepts necessary to understand and manage money or financial resources.

Borrowing-concepts relating to the debt or using money from a financial institution under an agreement to repay plus a fee for the privilege of using it.

Building assets-concepts relating to growing and increasing financial resources.

Protecting assets-concepts relating to avoiding the possibility of loss of money and/or financial resources.

Please be sure to complete the entire survey, and register your names to confirm your participation.

Initial Panel Round 1 Online Survey (continued)

	Concept be inc	t should luded		Appropriat	e Catego	У
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets
Bonds	0	0				
Borrowing Interest & APR (Annual Percentage Rate)	0	0				
Compound Interest	\bigcirc	\bigcirc				
Credit & borrorwing	\bigcirc	\bigcirc				
Credit Cards	\bigcirc	\bigcirc				
nflation	\bigcirc	\bigcirc				
Insurance (Life)	\bigcirc	\bigcirc				
nsurance (Personal. i.e. disability or long-term care)	\bigcirc	\bigcirc				
Insurance (Property, i.e. home or auto)	\bigcirc	\bigcirc				
Interest Rates	\bigcirc	0				
Investing	\bigcirc	\bigcirc				
Investment returns	\bigcirc	0				
Loans	\bigcirc	0				
Mortgage	\bigcirc	\bigcirc				
Mutual Funds	\bigcirc	\bigcirc				
Net Worth	\bigcirc	\bigcirc				
Retirement	\bigcirc	\bigcirc				
Risk	\bigcirc	\bigcirc				
Investment Diversification	\bigcirc	\bigcirc				
Short-Term Savings	\bigcirc	\bigcirc				
Spending	\bigcirc	\bigcirc				
Stocks	\bigcirc	\bigcirc				
Stocks vs. Mutual Funds	\bigcirc	\bigcirc				
Taxes	\bigcirc	\bigcirc				
Time Value of Money	\bigcirc	0				

Appendix H

Initial Panel Round 2 Online Survey

Initial Panel Round 2, Group 1 Online Survey

Initial Panel - Round 2 Instructions
Dear Expert Panel Member,
Thank you for participating in this valuable financial literacy research.
In Round 2 of the Expert Panel review, you will be evaluating questions in the pool for financial literacy assessment instrument. The final instrument will be administered to U.S. adults age 18 and over using an online survey.
As you read each question and possible answers (correct answer is bold), identify whether you believe the question should be included or excluded from an instrument used to determine a basic financial literacy score. The goal of this round is to drastically reduce the number of questions in the pool. You are only reviewing 20% of the pool, therefore, please feel free to be critical in your selection of the questions to include. Only include questions you strongly feel should be included; all other questions should be excluded.
Questions that are marked included will stay in the pool for the next round of review. If you determine a question should be excluded, please identify the reason you believe it should be excluded. Also, identify the reasons for exclusion from the list below:
 Age/life cycle bias-Question content is directed toward a specific age group or life cycle. A participant not in the age group or life cycle would probably respond with an incorrect answer. Judgment or belief bias-The question's response could be influenced by the participant's personal judgments or beliefs. Participants with different beliefs would respond differently to this question. Too specific-The content of this question is too specific and would not necessarily be financial knowledge useful to the population majority. Too vague-The content of this question is too general or vague and would not be a good assessment of financial literacy knowledge. Participants could easily guess the answer without having financial knowledge. Other-Any questions you feel would not be representative of a participant's financial literacy. If you choose this response, please note your reason and/or use the comment section to explain.
Please feel free to comment on any of the questions you identify to include for the next round.
Suppose you had \$100 in a savings account and the interest rate was 2% per year. After 5 years how much do you think you would have in the account if you left the money to grow?
a. More than \$102 b. Exactly \$102 c. Less than \$102 d. Don't know e. Prefer not to answer
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Include
Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other
Commonte
Comments

Initial Panel Round 2, Group 1 Online Survey Online Survey (continued)

If interest rates rise, what will typically happen to bond prices?

- a. They will rise
- b. They will fall
- c. They will stay the same
- d. There is no relationship between bond prices and the interest
- e. Don't know
- f. Prefer not to answer

ALTERNATIVE WORDING

If interest rates fall, what will typically happen to bond prices?

- a. They will rise
- b. They will fall
- c. They will stay the same
- d. There is no relationship between bond prices and the interest
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

- Include
- Exclude-Age/life bias
- Exclude-Judgment or belief bias

1

- Exclude-Too specific
- Exclude-To vague
- Exclude-Other
- Comments

Imagine that the interest rate on your savings account was 1% per year and inflation was 2% per year. After 1 year, would you be able to buy:

- a. More than today
- b. Exactly the same
- c. Less than today
- d. Don't know
- e. Prefer not not answer

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

- Include
- Exclude-Age/life bias
- Exclude-Judgment or belief bias
- Exclude-Too specific
- Exclude-To vague
- Exclude-Other

Initial Panel Round 2, Group 1 Online Survey Online Survey (continued)

Suppose that in the year 2015, your income has doubled and prices of all goods have doubled too. In 2015, how much will you be able to buy with your income?

a. More than today

- b. The same
- c. Less than today
- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

Include
molado

- Exclude-Age/life bias
- Exclude-Judgment or belief bias
- Exclude-Too specific
- Exclude-To vague
- Exclude-Other

Comments

Normally, which asset displays the highest fluctuations over time?

1

1

- a. Savings accounts
- b. Bonds
- c. Stocks
- d. Don't know
- c. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

Include	
Exclude-Age/life bias	
Exclude-Judgment or b	eliefbias
Exclude-Too specific	

- Exclude-To vague
- Exclude-Other
- Comments

There are annual contribution limits on the amount you can save in a 401(k) plan or IRA that depend on your income.
Initial Panel Round 2, Group 1 Online Survey Online Survey (continued)

a. True b. False c. Don't know d. Prefer not to say	
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?	
Include	
Exclude-Age/life bias	
Exclude-Judgment or belief bias	
Exclude-Too specific	
Exclude-To vague	
Exclude-Other	
Comments	
 a. Savings accounts b. Bonds c. Stocks d. Don't know c. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias 	
Exclude-10 vague	
Exclude-Other	
Comments	
Which statement describes the main function of the stock market? a. The stock market helps to predict stock earnings b. The stock market results in an increase in the price of stocks c. The stock market brings people who want to buy stock together with those who want to sell stocks d. None of the above e. Don't know f. Prefer not to answer	
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?	

Initial Panel Round 2, Group 1 Online Survey Online Survey (continued)

	Include
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
_	
	Comments
	h.
lf y mo	ou were to invest \$1,000 in a stock mutual fund, it would be possible to have less than \$1,000 when you withdraw your ney.
a. T	True
b. F c. L	ēalse Don't know
d. I	Prefer not to answer
Upo que	on reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of estions for the final basic financial literacy assessment test?
	Include
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
	Comments
Wh	ich statement about mutual funds is correct?
а. b.	Once one invests in a mutual fund, one cannot withdraw the money in the 1° year. Mutual funds can invest in several assets, for example, invest in both stocks & bonds.
c. d	Mutual funds pay a guaranteed rate of return which depends on their past performance. None of the above
е.	Don't know
<i>T.</i> 1	Prefer hot to answer
Upo que	on reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of stions for the final basic financial literacy assessment test?
	Include
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague

Initial Panel Round 2, Group 1 Online Survey Online Survey (continued)

	ude-Other
Con	iments
Whole I	fe" insurance has a savings feature while "term" insurance does not.
a. True 5. False 5. Don't d. Prefe	know r not to answer
Jpon rev question	viewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of s for the final basic financial literacy assessment test?
🗌 Inclu	de
Excl	ude-Age/life bias
Excl	ude-Judgment or belief bias
Excl	ude-Too specific
Excl	ude-To vague
Excl	ude-Other
Con	iments
Buying a a. True 5. False 5. Don't 1. Prefei	a company stock usually provides a safer return than a stock mutual fund. , , , , not to answer
Buying a a. True b. False c. Don't d. Prefei Jpon rev juestion	a company stock usually provides a safer return than a stock mutual fund. know r not to answer riewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool o s for the final basic financial literacy assessment test?
Buying a a. True b. False b. Don't d. Prefei Upon rev uestion	a company stock usually provides a safer return than a stock mutual fund. know r not to answer viewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of s for the final basic financial literacy assessment test?
Buying a . True . False . Don't . Prefea Jpon revuestion Incluestion Excl	a company stock usually provides a safer return than a stock mutual fund. know not to answer riewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of s for the final basic financial literacy assessment test? Inde ude-Age/life bias
Buying a b. True b. False c. Don't Joon rev uestion Inclu Excl Excl	a company stock usually provides a safer return than a stock mutual fund. know r not to answer viewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool o s for the final basic financial literacy assessment test? ide ude-Age/life bias ude-Judgment or belief bias
Buying a a. True b. False c. Don't d. Prefei Jpon rev juestion Inclu Excl Excl Excl	a company stock usually provides a safer return than a stock mutual fund. know not to answer riewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool o s for the final basic financial literacy assessment test? Inde ude-Age/life bias ude-Judgment or belief bias ude-Too specific
Buying a a. True Jan False b. False c. Don't d. Prefer Jpon rev uestion Inclu Excl Excl Excl Excl	a company stock usually provides a safer return than a stock mutual fund. know not to answer viewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool o is for the final basic financial literacy assessment test? Inde ude-Age/life bias ude-Judgment or belief bias ude-Too specific ude-To vague
Buying a . True . False . Don't . Prefei Jpon rev uestion Inclu Excl Excl Excl Excl Excl	a company stock usually provides a safer return than a stock mutual fund. know r not to answer riewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of s for the final basic financial literacy assessment test? ide ude-Age/life bias ude-Judgment or belief bias ude-Too specific ude-To vague ude-Other
Buying & a. True b. False c. Don't d. Prefei Jpon rev question Inclu Excl Excl Excl Excl	a company stock usually provides a safer return than a stock mutual fund. know r not to answer riewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of s for the final basic financial literacy assessment test? ide ude-Age/life bias ude-Judgment or belief bias ude-Too specific ude-To vague ude-Other
Buying a a. True D. False 5. Don't 1. Prefea Jpon rev juestion Inclu Excl Excl Excl Excl Excl Excl Com	a company stock usually provides a safer return than a stock mutual fund. know root to answer riewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool or s for the final basic financial literacy assessment test? Ide ude-Age/life bias ude-Judgment or belief bias ude-Too specific ude-To vague ude-Other iments

Initial Panel Round 2, Group 1 Online Survey Online Survey (continued)

a. True	
b. False c. Don't know	
d. Prefer not to answer	
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?	
Exclude-Age/life bias	
Exclude-Judgment or belief bias	
Exclude-Too specific	
Exclude-To vague	
Exclude-Other	
Comments	
When an investor spreads his money among different assets, does the risk of losing money:	
a. Increase b. Decrease	
c. Stay the same d. Don't know	
e. Prefer not to answer	
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?	
Exclude-Age/life bias	
Exclude-Judgment or belief bias	
Exclude-Too specific	
Exclude-To vague	
Exclude-Other	
Comments	
h.	
A 15-year mortgage typically requires higher monthly payments than a 30-year mortgage, but the total interest paid over th life of the loan will be less.	е
a. True	
b. False c. Don't know	
d. Prefer not to answer	
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?	

Initial Panel Round 2, Group 1 Online Survey Online Survey (continued)

Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other
Comments
h
Suppose you owe \$3,000 on your credit card. You pay a minimum payment of \$30 each month. At an Annual Percentage Rate of 12% (or 1% per month), how many years would it take to eliminate your credit card debt if you made no additional new charges? a. Less than 5 years b. Between 5 and 10 years c. Between 10 and 20 years d. Never, you will continue to be in debt c. Don't know d. Prefer not to answer
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Exclude Agenie blas
Exclude-Judgment of belief blas
Exclude-to vague
Exclude-Other
Comments
You purchase an appliance that costs \$1,000. To pay for this appliance, you are given the following two options: (a) Pay 12 monthly installments of \$100 each; (b) Borrow at a 20% annual interest rate and pay back \$1,200 a year from now. Which is the more advantageous offer?
a. Option (a) b. Option (b) c. They are the same d. Don't know e. Prefer not to answer
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific

Initial Panel Round 2, Group 1 Online Survey Online Survey (continued)

E	Exclude-To vague
E	Exclude-Other
	Comments
let v	vorth is equal to
. T	otal assets
. To T	otal asets plus liabilities
. D	on't know
. P	refer not to answer
lpon ues	reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of tions for thefinal basic financial literacy assessment test?
	nclude
E	Exclude-Age/life bias
E	Exclude-Judgment or belief bias
E	Exclude-Too specific
E	Exclude-To vague
E	Exclude-Other
	Comments
Savii	ngs accounts and money market accounts are most appropriate for
. Lo	ong-term investments like retirement
). E	mergency funds and short-term goals Increase by \$3,000
. D	on't know
. P	refer not to answer
lpon ues	reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of tions for the final basic financial literacy assessment test?
	nclude
E	Exclude-Age/life bias
E	Exclude-Judgment or belief bias
E	Exclude-Too specific
E	Exclude-To vague
E	Exclude-Other
	Comments

Initial Panel Round 2, Group 1 Online Survey Online Survey (continued)

On which type of loan is interest never tax deductible?

1

- a. Home equity loan
- b. An adjustable rate mortgage
- c. A personal vehicle loan
- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

- Include
- Exclude-Age/life bias
- Exclude-Judgment or belief bias
- Exclude-Too specific
- Exclude-To vague
- Exclude-Other
- Comments

If you have an insurance policy with a higher deductible, the premiums will be?

- a. Higher
- b. Lower
- c. The same
- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

- Include
- Exclude-Age/life bias
- Exclude-Judgment or belief bias
- Exclude-Too specific
- Exclude-To vague
- Exclude-Other
- Comments

Which of the following types of insurance is most important for single workers without children?

- a. Life insurance
- b. Disability income insurance

1

- c. Dental insurance
- d. Don't know
- e. Prefer not to answer

Initial Panel Round 2, Group 1 Online Survey Online Survey (continued)

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments Checking account reconciliation involves: a. balancing bank statement with your checkbook records to determine if there are errors. b. reconciling current bank statements with the. previous month's statements to determin if there are errors. c. subtracting outstanding checks to your checkbook to determine if your checks have been properly processed. d. adding outstanding checks to your checkbook blances to improve your credit standing. e. none of the above f. Don't know g. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments You will improve your credit worthiness by a. visiting your local commercial bank b. showing no record of personal bankruptcies in recent years. c. paying cash for all goods and services. d. borrowing large amounts of money from your friends. e donating money to charity. f. Don't know g. Prefer not to answer Upon reviewing the guestion above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

Initial Panel Round 2, Group 1 Online Survey Online Survey (continued)

	Include
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
	Comments
Vh	ich is FALSE concerning credit cards
	The rate of interest on your credit card is normally higher than you can earn on a certificate of deposit. A credit card company will not charge you interest if you payoff the entire balance by the due date. you cannot spend more than your line of credit. Don't know Prefer not to answer
ue	istions for the final basic financial literacy assessment lest?
	Include
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Judgment or belief bias Exclude-Too specific
	Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague
	Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other
	Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments
	Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments
	Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments
	Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments
	Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments igh-risk and high-return investment strategy would be most suitable for
	Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments igh-risk and high-return investment strategy would be most suitable for an eldedy retired couple living on a fixed income
A h	Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments igh-risk and high-return investment strategy would be most suitable for an elderly retired couple living on a fixed income. a couple needing funds for their children's education in two years.
A h 	Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments igh-risk and high-return investment strategy would be most suitable for an elderly retired couple living on a fixed income. a couple needing funds for their children's education in two years. a young married couple without children. all of the above because they all need high return.
A h a. b. c.	Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments igh-risk and high-return investment strategy would be most suitable for an elderly retired couple living on a fixed income. a couple needing funds for their children's education in two years. a young married couple without children. all of the above because they all need high return. none of the above because they are equally risk averse.
	Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments igh-risk and high-return investment strategy would be most suitable for an elderly retired couple living on a fixed income. a couple needing funds for their children's education in two years. a young married couple without children. all of the above because they all need high return. none of the above because they are equally risk averse. Don't know Prefer not to answer

Initial Panel Round 2, Group 1 Online Survey Online Survey (continued)

	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
	Comments
lf yo your a. Tri b. Fa c. Do d. Pr	u are behind on debt payments and go to a credit counseling service, they can get the federal government to apply income tax refund to payoff your debts. ue Ise on't know efer not to answer
Upor ques	n reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of tions for the final basic financial literacy assessment test?
	Include
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
	Comments
	h
lf yo a. Tri b. Fa c. Do d. Pr	u have any negative information on your credit report, a credit repair agency can help you remove that information. ue Ise n't know efer not to answer
Upor ques	n reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of tions for the final basic financial literacy assessment test?
	Exclude-Age/ine bias
	Exclude-10 vague
	Exclude-Other

Initial Panel Round 2, Group 1 Online Survey Online Survey (continued)

b.	False	

c. Don't know d. Prefer not to answer

ALTERNATIVE WORDING

If you are smart, it is easy to pick individual company stocks that will have better than average returns.

- a. True
- b. False
- c. Don't know d. Prefer not to answer

Upon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

Include

- Exclude-Age/life bias
- Exclude-Judgment or belief bias
- Exclude-Too specific
- Exclude-To vague
- Exclude-Other

Comments

An investment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate would double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would be worth \$4,000 after 40 years (by the time they are 70).

a. True

- b. False
- c. Don't know d. Prefer not to answer

ALTERNATIVE WORDING

An investment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate would double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would not be worth \$4,000 for at least 45 years (until they are at least 75).

- a. True
- b. False
- c. Don't know
- d. Prefer not to answer

Upon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

- Include
- Exclude-Age/life bias
- Exclude-Judgment or belief bias
- Exclude-Too specific
- Exclude-To vague
- Exclude-Other

tial I	anel Round 2, Group 1 Online Survey Online Survey (continued)	
b. Fal: c. Dor d. Pre	e t know er not to answer	
ALTERNATIVE WORDING		
lf you	are smart, it is easy to pick individual company stocks that will have better than average returns.	
a. Tru b. Fal : c. Dor d. Pre	e se 't know fer not to answer	
Upon (ident	reviewing the question and its alternative wording format above, do you feel it should be included in or excluded fying reason) from the pool of questions for the final basic financial literacy assessment test?	
📄 Ir	clude	
E	xclude-Age/life bias	
E	xclude-Judgment or belief bias	
E	xclude-Too specific	
E	xclude-To vague	
Ē	xclude-Other	
🔲 C	omments	
An in would	// restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would be worth \$4,000 after	
An in would 40 ye a. Tru b. Fai	restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would be worth \$4,000 after ars (by the time they are 70). e se	
An in would 40 ye a. Tru b. Fai c. Do d. Pro	restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would be worth \$4,000 after ars (by the time they are 70). Te se of know for not to answer.	
An in would 40 ye a. Tru b. Fai c. Do d. Pre	restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would be worth \$4,000 after ars (by the time they are 70). e se if know fer not to answer	
An in would 40 ye a. Tru b. Fai c. Do d. Pre ALTE	restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would be worth \$4,000 after ars (by the time they are 70). e se of know fer not to answer RNATIVE WORDING	
An in would 40 ye a. Tru b. Fai c. Do d. Pre ALTE An in would for at	restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would be worth \$4,000 after ars (by the time they are 70). e se of know fer not to answer RNATIVE WORDING restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would not be worth \$4,000 least 45 years (until they are at least 75).	
An inv would 40 ye a. Tru b. Fai c. Do d. Pre ALTE An inv would for at a. Tru	restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would be worth \$4,000 after ars (by the time they are 70). e se of know fer not to answer RNATIVE WORDING restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would not be worth \$4,000 least 45 years (until they are at least 75).	
An ini would 40 ye b. Fai c. Do d. Pre ALTE An ini would for at a. Trn b. Fa c. Do	restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would be worth \$4,000 after ars (by the time they are 70). re se it know fer not to answer RNATIVE WORDING restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would not be worth \$4,000 least 45 years (until they are at least 75). e se it know	
An ini would 40 ye b. Fai c. Do d. Pre An ini would for at a. Trn b. Fa c. Do d. Pre	restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would be worth \$4,000 after ars (by the time they are 70). e se of know fer not to answer RNATIVE WORDING restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would not be worth \$4,000 least 45 years (until they are at least 75). e se of know fer not to answer	
An in would 40 ye b. Fai c. Do d. Pre An in would for at a. Tr b. Fa c. Do d. Pre Upon (ident	<pre>restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would be worth \$4,000 after ars (by the time they are 70). re se se or to answer RNATIVE WORDING restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would be worth \$4,000 after of to answer RNATIVE WORDING restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would not be worth \$4,000 least 45 years (until they are at least 75). e se if know fer not to answer reviewing the question and its alternative wording format above, do you feel it should be included in or excluded fying reason) from the pool of questions for the final basic financial literacy assessment test?</pre>	
An in would 40 ye a. Tru b. Fai c. Do d. Pre ALTE An im would for at a. Tru b. Fai c. Do d. Pre Upon (ident	<pre>restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would be worth \$4,000 after ars (by the time they are 70). re se tk now fer not to answer RNATIVE WORDING restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would be worth \$4,000 after se tk now fer not to answer RNATIVE WORDING restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would not be worth \$4,000 least 45 years (until they are at least 75). reviewing the question and its alternative wording format above, do you feel it should be included in or excluded fying reason) from the pool of questions for the final basic financial literacy assessment test? clude</pre>	
An ini would 40 ye b. Fai c. Do d. Pre ALTE An ini would for at a. Trn would for at a. Trn b. Fai c. Do d. Pre Upon (ident	<pre>restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would be worth \$4,000 after ars (by the time they are 70). re se th know fer not to answer RNATIVE WORDING restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would not be worth \$4,000 least 45 years (until they are at least 75). re se se th know fer not to answer reviewing the question and its alternative wording format above, do you feel it should be included in or excluded fying reason) from the pool of questions for the final basic financial literacy assessment test? clude kclude-Age/life bias</pre>	
An ini would 40 ye b. Fai c. Do d. Pre ALTE An ini would for at a. Tr. b. Fai c. Do d. Pre Upon (ident Irre Upon (ident Irre E E E E	<pre>restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would be worth \$4,000 after ars (by the time they are 70). e se se tk now fer not to answer RNATIVE WORDING e se se th they are at least 75). e se th they are to answer reviewing the question and its alternative wording format above, do you feel it should be included in or excluded fying reason) from the pool of questions for the final basic financial literacy assessment test? clude kclude-Age/life bias kclude-Judgment or belief bias kclude-Judgment o</pre>	
An in would 40 ye b. Fai c. Do d. Pre ALTE An in would for at a. Tri b. Fai for at c. Do d. Pre Upon (ident ir i E E E E E	restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would be worth \$4,000 after ars (by the time they are 70). e se ot know fer not to answer RNATIVE WORDING restment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would not be worth \$4,000 least 45 years (until they are at least 75). e se of know fer not to answer reviewing the question and its alternative wording format above, do you feel it should be included in or excluded fying reason) from the pool of questions for the final basic financial literacy assessment test? clude kclude-Age/life bias kclude-Judgment or belief bias	
An in would 40 ye b. Fai c. Do d. Pre An in would for at a. Trn b. Fa c. Do d. Pre Upon (ident life E E E E E E	<pre>estment adviser tells a 30-year old couple that \$1,000 in an investment that pays a certain, constant interest rate double in value to \$2,000 after 20 years (by the time they are 50). If so, that investment would be worth \$4,000 after ars (by the time they are 70). e se se th know fer not to answer RNATIVE WORDING e se th know fer not to answer reviewing the question and its alternative wording format above, do you feel it should be included in or excluded for and to answer reviewing the question and its alternative wording format above, do you feel it should be included in or excluded for not to answer reviewing the question and its alternative wording format above, do you feel it should be included in or excluded final characteristic financial literacy assessment test? clude kclude-Age/life bias kclude-Judgment or belief bias kclude-Too specific kclude-To vague</pre>	

Initial Panel Round 2, Group 2 Online Survey Online Survey

Initial Panel - Round 2 Instructions

Dear Expert Panel Member,

Thank you for participating in this valuable financial literacy research.

In Round 2 of the Expert Panel review, you will be evaluating questions in the pool for financial literacy assessment instrument. The final instrument will be administered to U.S. adults age 18 and over using an online survey.

As you read each question and possible answers (correct answer is bold), identify whether you believe the question should be included or excluded from an instrument used to determine a basic financial literacy score. The goal of this round is to drastically reduce the number of questions in the pool. You are only reviewing 20% of the pool, therefore, please feel free to be critical in your selection of the questions to include. Only include questions you strongly feel should be included; all other questions should be excluded.

Questions that are marked included will stay in the pool for the next round of review. If you determine a question should be excluded, please identify the reason you believe it should be excluded. Also, identify the reasons for exclusion from the list below:

Age/life cycle bias-Question content is directed toward a specific age group or life cycle. A participant not in the age group or life cycle would probably respond with an incorrect answer.

Judgment or belief bias--The question's response could be influenced by the participant's personal judgments or beliefs. Participants with different beliefs would respond differently to this question.

Too specific-The content of this question is too specific and would not necessarily be financial knowledge useful to the population majority.

Too vague—The content of this question is too general or vague and would not be a good assessment of financial literacy knowledge. Participants could easily guess the answer without having financial knowledge.

Other--Any questions you feel would not be representative of a participant's financial literacy. If you choose this response, please note your reason and/or use the comment section to explain.

Please feel free to comment on any of the questions you identify to include for the next round.

Initial Panel - Round 2

Assume a friend inherits \$10,000 today and his sibling inherits \$10,000 three years from now. Who is richer because of the inheritance?

a. My friend

- b. His sibling
- c. They are equally rich
- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

Include

- Exclude-Age/life bias
- Exclude-Judgment or belief bias
- Exclude-Too specific
- Exclude-To vague
- Exclude-Other

Initial Panel Round 2, Group 2 Online Survey Online Survey (continued)

c	omments
πer	age 70 1/2, you have to withdraw at least some money from your 401(k) plan or IRA.
n. Tru Fal	10 199
. Doi	n't know
d. Pre	fer not to answer
Jpon questi	reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of ions for the final basic financial literacy assessment test?
🗌 In	Iclude
E	xclude-Age/life bias
E	xclude-Judgment or belief bias
E	xclude-Too specific
E	xclude-To vague
E	xclude-Other
	omments
	h
n. Tro 5. Fal 5. Doi 1. Pre	ue Ise n't know afer not to answer
Jpon	reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of
In In	iclude
E	xclude-Age/life bias
E	xclude-Judgment or belief bias
E	xclude-Too specific
E	xclude-To vague
E	xclude-Other
	comments
lf you	buy a company's stock
я. Үс b. You	או own a part of the company u have lent money to the company
c. You	u are liable for the company's debt

d. You can vote on shareholder resolutionse.

Initial Panel Round 2, Group 2 Online Survey Online Survey (continued)

	, 1 , 5 , 5 , 6 , 7 , 7
e. Don't know f. Prefer not to say	
Upon reviewing the questio questions for the final basic	n above, do you feel it should be included in or excluded (identifying reason) from the pool of financial literacy assessment test?
Exclude-Age/life bias	
Exclude-Judgment or be	liefbias
Exclude-Too specific	
Exclude-To vague	
Exclude-Other	
Comments	
A stock mutual fund combi	nes the money of many investors to buy a variety of stocks, not a single stock.
a. True	
. False . Don't know	
I. Prefer not to say	
uestions for the final basic	financial literacy assessment test?
Exclude-Age/life bias	
Exclude-Judgment or be	liefbias
Exclude-Too specific	
Exclude-To vague	
Exclude-Other	
Commonto	
Comments	
The cash value of a life ins	urance policy is the amount available if you surrender your life insurance policy while you're still
alive.	
a. True	
c. Don't know	
d. Prefer not to answer	
Jpon reviewing the questio questions for the final basic	n above, do you feel it should be included in or excluded (identifying reason) from the pool of financial literacy assessment test?
Exclude-Age/life bias	
Exclude-Judament or be	liefbias

Initial Panel Round 2, Group 2 Online Survey Online Survey (continued)

	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
	Comments
Ηοι	ising prices in the US can never go down.
a. 1	
b. I	False
c.L d.F	Don't know Prefer not to say
Upc que	on reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of stions for the final basic financial literacy assessment test?
	Include
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
	Comments
Sup	pose you owe \$1,000 on your credit card and the interest rate you are charged is 20% per year compounded annually.
If y	ou didn't pay anything off, at this interest rate, how many years would it take for the amount you owe to double?
a. 2	' years
b . L	.ess than 5 years i to 10 years
d. N	<i>fore than 20 years</i>
c. E d F	Don't know Prefer not to answer
Upo	on reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of stions for the final basic financial literacy assessment test?
\bigcirc	Include
\bigcirc	Exclude-Age/life bias
\bigcirc	Exclude-Judgment or belief bias
\bigcirc	Exclude-Too specific
\bigcirc	Exclude-To vague
\bigcirc	Exclude-Other
_	
\bigcirc	Comments

Initial Panel Round 2, Group 2 Online Survey Online Survey (continued)

ш _в	Exclude-Too specific
E	Exclude-To vague
E	Exclude-Other
	Comments
Hous	sing prices in the US can never go down.
a. Tr	ue
b. F a c. Do	alse on't know
d. Pr	refer not to say
Upon ques	n reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of tions for the final basic financial literacy assessment test?
	Include
E	Exclude-Age/life bias
E	Exclude-Judgment or belief bias
E	Exclude-Too specific
E	Exclude-To vague
E	Exclude-Other
	Comments
	h.
Supp If you	pose you owe \$1,000 on your credit card and the interest rate you are charged is 20% per year compounded annually. u didn't pay anything off, at this interest rate, how many years would it take for the amount you owe to double?
a. 2	vears
b. Le	ess than 5 years to 10 years
с. 51 d. M	ore than 20 years
c. Do d. Pr	on't know refer not to answer
Upon	n reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of tions for the final basic financial literacy assessment test?
0 1	include
E	Exclude-Age/life bias
() E	Exclude-Judgment or belief bias
() E	Exclude-Too specific
() E	Exclude-To vague
() E	Exclude-Other
\bigcirc	Comments

Initial Panel Round 2, Group 2 Online Survey Online Survey (continued)

High inflation means that the cost of living is increasing rapidly
a. True
b. False
c. Don't know d. Prefer not to answer
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other
Comments
d. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Exclude- Judament or belief bias
Exclude-onler
Comments
Which is the risker asset to invest in?
a. Shares in a single company stock
c. Risks are identical in both assets
a. Don't know e. Prefer not to answer
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

Initial Panel Round 2, Group 2 Online Survey Online Survey (continued)

Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other
Comments
h. h.
A broker recommends a municipal bond that matures in 2020 but is likely to be called, or redeemed, as early as 2015. The best gauge of your expected return is its a. current yield. b. yield to maturity. c. yield to call. d. Don't know e. Prefer not to answer
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other
Comments
The figure that best reflects a mutual fund's performance over a period of years is
a. its current yield. b. the total of dividends and capital gains it has paid.
c. its total return. d. Don't know
e. Prefer not to answer
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other

Initial Panel Round 2, Group 2 Online Survey Online Survey (continued)

	Comments		
	,		
W	hich bank account is likely to	pay the highest interest rate on money saved?	
a.	Savings account		
b.	Six month CD or certificate		
с. d.	Don't know		
е.	Prefer not to answer		
Up qu	oon reviewing the question al lestions for the final basic fina	bove, do you feel it should be included in or excluded (identifying reason) from the ncial literacy assessment test?	pool
	Include		
	Exclude-Age/life bias		
	Exclude-ludament or balief	hias	
	Exclude-100 specific		
	Exclude-To vague		
	Exclude-Other		
	Comments		
	1		
lf y a. b. c. d. e.	your assets increase by \$5,0 Increase by \$2,000 Increase by \$8,000 Increase by \$3,000 Don't know Prefer not to answer	100 and your liabilities decrease by \$3,000, your net worth would	
lf) a. b. c. d. e. Up qu	your assets increase by \$5,0 Increase by \$2,000 Increase by \$8,000 Increase by \$3,000 Don't know Prefer not to answer poon reviewing the question at testions for the final basic fina	000 and your liabilities decrease by \$3,000, your net worth would nove, do you feel it should be included in or excluded (identifying reason) from the ncial literacy assessment test?	pool
lf) a. b. c. d. e. Up	your assets increase by \$5,0 Increase by \$2,000 Increase by \$8,000 Increase by \$3,000 Don't know Prefer not to answer bon reviewing the question at lestions for the final basic fina	000 and your liabilities decrease by \$3,000, your net worth would bove, do you feel it should be included in or excluded (identifying reason) from the ncial literacy assessment test?	pool
lf) a. b. c. d. e. Up qu	your assets increase by \$5,0 Increase by \$2,000 Increase by \$8,000 Increase by \$3,000 Don't know Prefer not to answer boon reviewing the question at testions for the final basic final Include Exclude-Age/life bias	000 and your liabilities decrease by \$3,000, your net worth would bove, do you feel it should be included in or excluded (identifying reason) from the ncial literacy assessment test?	pool
If y a. b. c. d. e. Up qu	your assets increase by \$5,0 Increase by \$2,000 Increase by \$8,000 Increase by \$3,000 Don't know Prefer not to answer bon reviewing the question at estions for the final basic fina Include Exclude-Age/life bias Exclude-Judgment or belief	000 and your liabilities decrease by \$3,000, your net worth would bove, do you feel it should be included in or excluded (identifying reason) from the ncial literacy assessment test?	pool
If) a. b. c. d. e. Up	your assets increase by \$5,0 Increase by \$2,000 Increase by \$8,000 Increase by \$3,000 Don't know Prefer not to answer oon reviewing the question at testions for the final basic finat Include Exclude-Age/life bias Exclude-Judgment or belief Exclude-Too specific	000 and your liabilities decrease by \$3,000, your net worth would bove, do you feel it should be included in or excluded (identifying reason) from the ncial literacy assessment test?	pool
If) a. b. c. d. e. Up qu	your assets increase by \$5,0 Increase by \$2,000 Increase by \$8,000 Increase by \$3,000 Don't know Prefer not to answer bon reviewing the question at restions for the final basic fina Include Exclude-Age/life bias Exclude-Judgment or belief Exclude-Too specific Exclude-To vague	000 and your liabilities decrease by \$3,000, your net worth would bove, do you feel it should be included in or excluded (identifying reason) from the ncial literacy assessment test?	pool
If y a. b. c. d. e. Up qu	your assets increase by \$5,0 Increase by \$2,000 Increase by \$8,000 Increase by \$3,000 Don't know Prefer not to answer bon reviewing the question at estions for the final basic fina Include Exclude-Age/life bias Exclude-Judgment or belief Exclude-Too specific Exclude-To vague Exclude-Other	000 and your liabilities decrease by \$3,000, your net worth would bove, do you feel it should be included in or excluded (identifying reason) from the ncial literacy assessment test?	pool
If y a. b. c. d. e. Up	your assets increase by \$5,0 Increase by \$2,000 Increase by \$8,000 Increase by \$3,000 Don't know Prefer not to answer bon reviewing the question at lestions for the final basic fina Include Exclude-Age/life bias Exclude-Judgment or belief Exclude-Too specific Exclude-To vague Exclude-Other	000 and your liabilities decrease by \$3,000, your net worth would bove, do you feel it should be included in or excluded (identifying reason) from the ncial literacy assessment test?	pool
If) a. b. c. d. e. Up qu	your assets increase by \$5,0 Increase by \$2,000 Increase by \$8,000 Increase by \$3,000 Don't know Prefer not to answer con reviewing the question at testions for the final basic final linclude Exclude-Age/life bias Exclude-Judgment or belief Exclude-Too specific Exclude-To vague Exclude-To vague Comments	000 and your liabilities decrease by \$3,000, your net worth would hove, do you feel it should be included in or excluded (identifying reason) from the ncial literacy assessment test?	pool
If y a. b. c. d. e. Upque	your assets increase by \$5,0 Increase by \$2,000 Increase by \$8,000 Increase by \$3,000 Don't know Prefer not to answer bon reviewing the question at testions for the final basic fina Include Exclude-Age/life bias Exclude-Judgment or belief Exclude-Too specific Exclude-To vague Exclude-Other Comments	000 and your liabilities decrease by \$3,000, your net worth would pove, do you feel it should be included in or excluded (identifying reason) from the ncial literacy assessment test?	pool

Initial Panel Round 2, Group 2 Online Survey Online Survey (continued)

a. Lowest me	onthly payment
b. Longest re	payment term
l. Don't know	repayment V
. Prefer not	to answer
lpon reviewir uestions for	ng the question above, do you feel it should be included in or excluded (identifying reason) from the pool the final basic financial literacy assessment test?
Include	
Exclude-A	\ge/life bias
Exclude-J	udgment or belief bias
Exclude-1	oo specific
Exclude-1	o vague
Exclude-0	Dther
Comment	S
:. Reverse n 1. Don't knov	iortgage v
. Prefer not	to answer
lpon reviewir uestions for	ing the question above, do you feel it should be included in or excluded (identifying reason) from the pool the final basic financial literacy assessment test?
Include	
Exclude-A	\ge/life bias
Exclude-J	udgment or belief bias
Exclude-1	oo specific
Exclude-1	
Exclude-0	o vague
	o vague Dther
	o vague Dther
Comment	s vague
Comment	o vague Other s
Comment	o vague Dther s
Comment	o vague Dther s \$1.000 today at 4% for a year, your balance in a year will be:
Comment	o vague Dther s \$1,000 today at 4% for a year, your balance in a year will be:
Comment f you invest a. higher if t b. higher if t	o vague Dther s \$1,000 today at 4% for a year, your balance in a year will be: the interest is compounded daily rather than monthly. He interest is compounded quarterly rather than weekly.
Comment f you invest a. higher if t b. higher if tt c. higher if tt d. \$100 cc	s \$1,000 today at 4% for a year, your balance in a year will be: \$1,000 today at 4% for a year, your balance in a year will be: the interest is compounded daily rather than monthly. the interest is compounded quarterly rather than quarterly. The interest is compounded yearly rather than quarterly. The interest is compounded yearly rather than quarterly. The interest is compounded to the theorem of theorem of the theorem of

- f. Don't know g. Prefer not to answer

Initial Panel Round 2, Group 2 Online Survey Online Survey (continued)

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other
Comments
<i>n</i>
 Which of the following investments requires that you keep your money invested for a specified period or face an early withdrawal penalty? a. Certificate of deposit. b. Checking account that pays interest. c. Government savings bond. d. Money market mutual fund. e Passbook savings account. f. Don't know g. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other
Comments
You can receive your credit report from
a. a credit union.
b. a commercial bank. c. the Better Business Bureau
d. a credit bureau.
f. Don't know
g. Preter not to answer

Initial Panel Round 2, Group 2 Online Survey Online Survey (continued)

-1200101010	or the final basic financial literacy assessment test?
Include	
Exclude	-Age/life bias
Exclude	-Judgment or belief bias
Exclude	-Too specific
Exclude	-To vague
Exclude	-Other
Comme	nts
	h
The main r	ason to nurchase insurance is to
b. provide c. protect d. protect y e improve f. Don't kn g. Prefer n	you with excellent investment returns. you from sustaining a catastrophic loss. you from small incidental losse. your standard of living by filing fraudulent claims. ow ot to answer
Upon review questions fo	ving the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test?
Upon review questions fo	ving the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test?
Upon review questions fo Include Exclude	ving the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? -Age/life bias
Upon review questions fr Include Exclude Exclude	ving the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? -Age/life bias
Upon review questions fo Include Exclude Exclude	ving the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? Age/life bias Judgment or belief bias Too specific
Upon review questions fo Include Exclude Exclude Exclude Exclude	An event of the second
Upon review questions fr Include Exclude Exclude Exclude Exclude Exclude	Age/life biasAge/life biasJudgment or belief biasToo specificTo vagueOther
Upon review questions fo Exclude Exclude Exclude Exclude Exclude	ving the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? Age/life bias Judgment or belief bias Too specific To vague Other
Upon review questions fr Exclude Exclude Exclude Exclude Exclude Comme	 Age/life bias -Age/life bias -Judgment or belief bias -Too specific -To vague -Other Ints
Upon review questions for Exclude Exclude Exclude Exclude Exclude Comme	-Age/life biasJudgment or belief biasToo specificTo vagueOtherTo vagueOther
Upon review questions fo Exclude Exclude Exclude Exclude Exclude Comme	 -Age/life bias -Judgment or belief bias -Too specific -To vague -Other
Upon review questions fr Exclude Exclude Exclude Exclude Exclude Comme	 ving the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? -Age/life bias -Judgment or belief bias -Too specific -To vague -Other Ints
Upon review questions for Exclude Exclude Exclude Exclude Exclude Comme	ving the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? -Age/life bias -Judgment or belief bias -Too specific -To vague -Other
Upon review questions fo Exclude Exclude Exclude Exclude Exclude Comme Auto insura a. age of ir b. record o	ving the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? Age/life bias Judgment or belief bias -Too specific -To vague -Other
Upon review questions fo Exclude Exclude Exclude Exclude Exclude Comme	ving the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? -Age/life bias -Judgment or belief bias -Too specific -To vague -Other Ints // // nts //
Upon review questions for Exclude Exclude Exclude Exclude Exclude Comme Auto insura a. age of ir b. record o c. type and d. completi e all of th	ving the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? Age/life bias Judgment or belief bias Too specific To vague Other Ints
Upon review questions for Exclude Exclude Exclude Exclude Exclude Exclude Comme Auto insura a. age of ir b. record o c. type and d. complet e all of th f. Don't kn g. Prefer n	Age/life bias -Age/life bias -Too specific -To vague -Other -Other Ints In

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of

Initial Panel Round 2, Group 2 Online Survey Online Survey (continued)

questions for the final basic financial literacy assessment test?	
Exclude-Age/life bias	
Exclude-Judgment or belief bias	
Exclude-Too specific	
Exclude-To vague	
Exclude-Other	
Comments	
You should have an emergency fund that covers two to six months of your expenses.	
a True	
). False	
). Don't know J. Prefer not to answer	
Jpon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of	
Exclude-loo specific	
Exclude-To vague	
Exclude-Other	
Commente	
Comments	
h.	
f you have a savings account at a bank, you may have to pay taxes on the interest you earn.	
a. True	
). False c. Don't know	
1. Prefer not to answer	
Jpon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?	
Exclude-Age/life bias	
Exclude-Judgment or belief bias	
Exclude-Too specific	
Continued on the nex	/t n

Initial Panel Round 2, Group 2 Online Survey Online Survey (continued)

Comments	
	6
Mutual funds pay a guarar	iteed rate of return.
a. True	
c. Don't know	
d. Prefer not to answer	
Jpon reviewing the question questions for the final basi	on above, do you feel it should be included in or excluded (identifying reason) from the pool of c financial literacy assessment test?
Include	
Exclude-Age/life bias	
Exclude-Judgment or b	elief bias
Exclude-Too specific	
Exclude-To vague	
Exclude-Other	
Comments	
	6
The finance charge on you a. True b. False b. Don't know d. Prefer not to answer	ir credit card statement is what you pay to use credit.
The finance charge on you a. True b. False c. Don't know d. Prefer not to answer Upon reviewing the questio questions for the final basic	וי credit card statement is what you pay to use credit. אין above, do you feel it should be included in or excluded (identifying reason) from the pool of financial literacy assessment test?
The finance charge on you a. True b. False c. Don't know d. Prefer not to answer Upon reviewing the question questions for the final basic Include	ur credit card statement is what you pay to use credit. on above, do you feel it should be included in or excluded (identifying reason) from the pool of c financial literacy assessment test?
The finance charge on you a. True b. False c. Don't know d. Prefer not to answer Jpon reviewing the question questions for the final basic Include Exclude-Age/life bias	ir credit card statement is what you pay to use credit. on above, do you feel it should be included in or excluded (identifying reason) from the pool of c financial literacy assessment test?
The finance charge on you a. True b. False c. Don't know d. Prefer not to answer Upon reviewing the questing questions for the final basis Include Exclude-Age/life bias Exclude-Judgment or b	ir credit card statement is what you pay to use credit. on above, do you feel it should be included in or excluded (identifying reason) from the pool of c financial literacy assessment test?
The finance charge on you a. True b. False c. Don't know d. Prefer not to answer Upon reviewing the question questions for the final basis Include Exclude-Age/life bias Exclude-Judgment or bo Exclude-Too specific	ir credit card statement is what you pay to use credit. on above, do you feel it should be included in or excluded (identifying reason) from the pool o c financial literacy assessment test?
The finance charge on you a. True b. False c. Don't know d. Prefer not to answer Upon reviewing the questin questions for the final basis Include Exclude-Age/life bias Exclude-Judgment or bis Exclude-Too specific Exclude-To vague	ir credit card statement is what you pay to use credit. on above, do you feel it should be included in or excluded (identifying reason) from the pool o c financial literacy assessment test? elief bias
The finance charge on you a. True b. False c. Don't know d. Prefer not to answer Upon reviewing the question questions for the final basis Include Exclude-Age/life bias Exclude-Judgment or bo Exclude-Too specific Exclude-To vague Exclude-Other	ir credit card statement is what you pay to use credit. on above, do you feel it should be included in or excluded (identifying reason) from the pool o c financial literacy assessment test?
The finance charge on you a. True b. False c. Don't know d. Prefer not to answer Upon reviewing the questin questions for the final basis Include Exclude-Age/life bias Exclude-Judgment or bis Exclude-Too specific Exclude-To vague Exclude-Other	rr credit card statement is what you pay to use credit. on above, do you feel it should be included in or excluded (identifying reason) from the pool o c financial literacy assessment test? elief bias
The finance charge on you a. True b. False c. Don't know d. Prefer not to answer Upon reviewing the questing questions for the final basis Include Exclude-Age/life bias Exclude-Judgment or bis Exclude-Too specific Exclude-To vague Exclude-Other	Ir credit card statement is what you pay to use credit.
The finance charge on you a. True b. False c. Don't know d. Prefer not to answer Upon reviewing the questing questions for the final basis include Exclude-Age/life bias Exclude-Judgment or bis Exclude-Too specific Exclude-To vague Exclude-To vague Comments	Ir credit card statement is what you pay to use credit.
The finance charge on you a. True b. False c. Don't know d. Prefer not to answer Upon reviewing the questing questions for the final basin Include Exclude-Age/life bias Exclude-Judgment or br Exclude-Too specific Exclude-To vague Exclude-Other Comments	Ir credit card statement is what you pay to use credit.

Initial Panel Round 2, Group 2 Online Survey Online Survey (continued)

a. True
D. raise C. Don't know
d. Prefer not to answer
Upon reviewing the question shows, do you feel it should be included in an evaluated (identifying reason) from the pool of
opon reviewing the question above, oo you teen it should be included in or excluded (identifying reason) from the poor of questions for the final basic financial literacy assessment test?
Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Evaluate Other
Comments
If you invest for the long run, the annual fees of mutual funds are important.
a. True
b. False c. Don't know
d. Prefer not to answer
ALTERNATIVE WORDING
If you invest for the long run, the annual fees of mutual funds are unimportant.
a. True
D. raise c. Don't know
d. Prefer not to answer
Upon reviewing the question and its alternative warding format above, do you feel it should be included in an evaluated
(identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Exclude-Age/life bias
Exclude_ludgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other
Comments
It is a good idea to own stocks of foreign companies
It is a good idea to own stocks of foreign companies.
a. True
b. False
C. LONTKROW d. Profer not to approver

Initial Panel Round 2, Group 2 Online Survey Online Survey (continued)

ALTERNATIVE WORDING

It is best to avoid owning stocks of foreign companies.

a. True

- b. False
- c. Don't know d. Prefer not to answer

Upon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

Include

- Exclude-Age/life bias
- Exclude-Judgment or belief bias
- Exclude-Too specific
- Exclude-To vague
- Exclude-Other

Comments

Many people put aside money to take care of unexpected expenses. If Juan and Elva have money put aside for emergencies, in which of the following forms would it be of LEAST benefit to them if they needed it right away?

a. Invested in a down payment on the house

1

- b. Checking account
- c. Stocks
- d. Savings account
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

- Include
- Exclude-Age/life bias
- Exclude-Judgment or belief bias
- Exclude-Too specific
- Exclude-To vague
- Exclude-Other
- Comments

			1

Barbara has just applied for a credit card. She is an 18-year-old high school graduate with few valuable possessions and no credit history. If Barbara is granted a credit card, which of the following is the most likely way that the credit card company will reduce ITS risk?

a. It will make Barbara's parents pledge their home to repay Karen's credit card debt.

b. It will require Barbara to have both parents co-sign for the card.

Initial Panel Round 2, Group 2 Online Survey Online Survey (continued)

- c. It will charge Barbara twice the finance charge rate it charges older cardholders.
- d. It will start Barbara out with a small line of credit to see how she handles the account.
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

- Include
- Exclude-Age/life bias
- Exclude-Judgment or belief bias
- Exclude-Too specific
- Exclude-To vague
- Exclude-Other
- Comments

If each of the following persons had the same amount of take home pay, who would need the greatest amount of life insurance?

- a. An elderly retired man, with a wife who is also retired.
- b. A young married man without children.
- c. A young single woman with two young children.
- d. A young single woman without children.
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

Include

- Exclude-Age/life bias
- Exclude-Judgment or belief bias
- Exclude-Too specific
- Exclude-To vague
- Exclude-Other
- Comments

Appendix H

Initial Panel Round 2, Group 3 Online Survey Online Survey

Dear Expert Panel Member,

Thank you for participating in this valuable financial literacy research.

In Round 2 of the Initial Panel review, you will be evaluating questions in the pool for financial literacy assessment instrument. The final instrument will be administered to U.S. adults age 18 and over using an online survey.

As you read each question and possible answers (correct answer is bold), identify whether you believe the question should be included or excluded from an instrument used to determine a basic financial literacy score. The goal of this round is to drastically reduce the number of questions in the pool. You are only reviewing 20% of the pool, therefore, please feel free to be critical in your selection of the questions to include. Only include questions you strongly feel should be included; all other questions should be excluded.

Questions that are marked included will stay in the pool for the next round of review. If you determine a question should be excluded, please identify the reason you believe it should be excluded, please select one of the listed reasons for exclusion:

• Age/life cycle bias--Question content is directed toward a specific age group or life cycle. A participant not in the age group or life cycle would probably respond with an incorrect answer.

Judgment or belief bias--The question's response could be influenced by the participant's personal judgments or beliefs. Participants with different beliefs would respond differently to this question.

• **Too specific**--The content of this question is too specific and would not necessarily be financial knowledge useful to the population majority.

• **Too vague**—The content of this question is too general or vague and would not be a good assessment of financial literacy knowledge. Participants could easily guess the answer without having financial knowledge.

Other-Any questions you feel would not be representative of a participant's financial literacy. If you choose this response, please note your reason and/or use the comment section to explain.

Please feel free to comment on any of the questions you identify to include for the next round.

Initial Panel - Round 2

If you buy a company's bond

- a. You own a part of the company
- b. You have lent money to the company
- c. You are liable for the company's debt
- d. You can vote on shareholder resolutionse.
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

- Include
- Exclude-Age/life bias
- Exclude-Judgment or belief bias
- Exclude-Too specific
- Exclude-To vague
- Exclude-Other
- Comments

Initial Panel Round 2, Group 3 Online Survey Online Survey (continued)

Bonds are normally riskier than stocks. a. True **b. False**

c. Don't know

d. Prefer not to answer

ALTERNATIVE WORDING

Stocks are normally riskier than bonds.

a. True

- b. False
- c. Don't know

d. Prefer not to answer

Upon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessmenttest?

Include

Exclude-Age/life bias

- Exclude-Judgment or belief bias
- Exclude-Too specific
- Exclude-To vague
- Exclude-Other

Comments

Financially, investing in the stock market is better than buying lottery tickets.

- a. True
- b. False
- c. Don't know d. Prefer not to answer

ALTERNATIVE WORDING

Financially, investing in the stock market is not better than buying lottery tickets.

1

a. True
b. False
c. Don't know
d. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

Include

Exclude-Age/life bias

Exclude-Judgment or belief bias

Exclude-Too specific

Initial Panel Round 2, Group 3 Online Survey Online Survey (continued)

	Exclude-To vaque
	Exclude-Other
	Comments
Υοι	are considering investing in a mutual fund expected to distribute \$1 a share in dividends. You should
a. b	uy now so you will get the distribution. uv after the distribution is paid
c. b	uy either way, because it doesn't matter.
a. D e. P	ont know refer not to answer
Upo que	on reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of stions for the final basic financial literacy assessment test?
	Include
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
	Comments
	li li
If in	terest rates climb one percentage point, which of these securities would be hurt the least?
a	a 20-vear zero-coupon bond
b.	a 20-year bond selling at its face value
c. d.	a 20-year bond seiling at a premium above its face value Don't know
е.	Prefer not to answer
Upo que	n reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool o stions for the final basic financial literacy assessment test?
	Include
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
_	
	Comments

Initial Panel Round 2, Group 3 Online Survey Online Survey (continued)

Investing in a mutual fund that holds a diversified portfolio of stocks protects your investment against market declines.				
a. True b. False c. Don't know d. Prefer not to answer				
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?				
Exclude-Age/life bias				
Exclude-Judgment or belief bias				
Exclude-Too specific				
Exclude-To vague				
Exclude-Other				
Commonte				
Comments				
h.				
 a. Provides a high rate of return with little risk b. Allows you to shelfter retirment savings from taxation c. Provides a well diversified mix of investment assets d. Don't know e. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other 				
Over the long term, stocks have the highest rate of return on money invested. a. True b. False				
c. Don't know d. Prefer not to answer				
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?				
Include				
Exclude-Age/life bias				

Initial Panel Round 2, Group 3 Online Survey Online Survey (continued)

	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
	Comments
lf yo	u always pay the full balance on your credit card, which of the following is least important?
a. A	Innual interest rate
b. А с. L	nnual fee ine of credit
d. D	ion't know
e. P	refer not to answer
Upor ques	n reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the poo tions for the final basic financial literacy assessment test?
	Include
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
	Comments
	· · · · · · · · · · · · · · · · · · ·
Whic	h household would typically have the greatest life insurance needs?
a. A	middle-class retired couple
с. А	Nindale-aged working couple with children in college
d. D	on't know
е. <i>Р</i>	rerer not to answer
Upor ques	reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the poo tions for the final basic financial literacy assessment test?
	Include
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other

Initial Panel Round 2, Group 3 Online Survey Online Survey (continued)

You are not overspending if a. you write checks for more than what you have in your checking account. b. your monthly wages are \$500 and credit charges \$1,000. c. you frequently receive calls from collection agencies. d. your monthly debt payments is: 30% of your take-home pay. e you meet your minimum monthly credit card payments. f. Don't know g. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments 1 An overdraft a. occurs when you write a \$1,000 dollar check when you have \$500 in your account. b. is a stop-payment order written by the payee. c. will result in fines. d. all of the above. e both A and C. f. Don't know g. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other

Comments

Initial Panel Round 2, Group 3 Online Survey Online Survey (continued)

Your net worth is
 a. the difference between your expenditures and income. b. the difference between your liabilities and assets. c. the difference between your cash inflow and outflow. d. the difference between your bank borrowing and savings. e none of the above. f. Don't know g. Prefer not to answer
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other
Comments
If you co-sign a loan for a friend, then
 a. you become responsible: for the loan payments if your friend defaults. b. it means that your friend cannot receive the loan by himself. c. you are entitled to receive part of the loan. d. both A and B. e Both A and C. f. Don't know g. Prefer not to answer
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other

Comments

Initial Panel Round 2, Group 3 Online Survey Online Survey (continued)

The main reason to purchase health insurance is	
 a. After buying health insurance, you are normally covered for pre-existing conditions. b. You have a better chance to choose doctors with a health maintenance organization rather than with a traditional health care insurance company. c. Most policies contain deductible and coinsurance clauses. d. A policy purchased by the individual is cheaper than one purchased through a group. e None of the above. f. Don't know g. Prefer not to answer 	1
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?	
Exclude-Age/life bias	
Exclude-Judgment or belief bias	
Exclude-Too specific	
Exclude-To vague	
Exclude-Other	
Comments	
Comments	
You have a better chance of resolving a complaint against an insurance company by bringing the issue to a government agency at the a. federal level. b. state level. c. county level. d. township level e None of the above. f. Don't know g. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?	
Exclude-Age/life bias	
Exclude-Judgment or belief bias	
Exclude-Too specific	
Exclude-To vague	
Exclude-Other	
Comments	
a. Tru b. Fal	the long term, stocks have the highest rate of return on money invested.
------------------------------	--
c. Doi d. Pre	ue Ise on't know efer not to answer
Upon ques	n reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of tions for the final basic financial literacy assessment test?
	Include
🔲 E	Exclude-Age/life bias
E	Exclude-Judgment or belief bias
E	Exclude-Too specific
E	Exclude-To vague
E	Exclude-Other
	Comments
Upon quest	n reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of tions for the final basic financial literacy assessment test?
F	Include
	Include Exclude-Age/life bias
. E	include Exclude-Age/life bias Exclude-Judgment or belief bias
	include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific
	include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague
	Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other
	Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other
	Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments
	Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments
	Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments
If the	Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments a interest rate on an adjustable-rate mortgage loan goes up, your monthly mortgage payments will also go up.
If the	Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments e interest rate on an adjustable-rate mortgage loan goes up, your monthly mortgage payments will also go up. ue
If the a. Trut b. Fail	Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments e interest rate on an adjustable-rate mortgage loan goes up, your monthly mortgage payments will also go up. ue Ise mathematical additional additiona

Initial Panel Round 2, Group 3 Online Survey Online Survey (continued)

	Include
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
	Comments
	le le
nfo a. T b. F c. D d. P	refear report includes employment data, your payments history, any inquines made by creditors, and any public record rue alse on't know refer not to answer
Upc que	n reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of stions for the final basic financial literacy assessment test?
	Include
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
_	
	Comments
	1.
A// /	nvestment products bought at your bank are covered by FDIC insurance.
a. T b. F	ue alse
). D	on't know prifer not to answer
Jpc	n reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of stions for the final basic financial literacy assessment test?
	Include
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Judgment or belief bias Exclude-Too specific
	Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague

Initial Panel Round 2, Group 3 Online Survey Online Survey (continued)

Comments
If the interest rate falls, bond prices will rise.
a. True b. False c. Don't know d. Prefer not to answer
ALTERNATIVE WORDING
If the interest rate falls, bond prices will fall.
a. True b. False c. Don't know d. Prefer not to answer
Upon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other
Comments
h
It is better for young people saving for retirement to combine stocks with long-term (inflation protected) bonds than with short-term (inflation protected) bonds. a. True b. False c. Don't know d. Prefer not to answer
ALTERNATIVE WORDING
It is better for young people saving for retirement to combine stocks with short-term (inflation protected) bonds than with long-term (inflation protected) bonds.
a. True b. False c. Don't know d. Prefer not to answer
Upon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Exclude-Age/life bias
Exclude-Judgment or belief bias

Initial Panel Round 2, Group 3 Online Survey Online Survey (continued)

Exclude-Too specific	•	-	
Exclude-To vague			
Exclude-Other			
Comments			
1.			
Taxes affect how you should	invest your money.		
a. True			
b. False			
d. Prefer not to answer			
ALTERNATIVE WORDING			
Taxes do not affect how you	should invest your money		
-	should invest your money.		
a. True b. False			
c. Don't know d. Profer not to anower			
a. Freier not to answer			
Upon reviewing the question (identifying reason) from the	and its alternative wording format ab pool of questions for the final basic fir	ove, do you feel it should nancial literacy assessment	be included in or excluded test?
Exclude-Age/life bias			
Exclude-Judgment or bel	efbias		
Exclude-Too specific			
Exclude-To vague			
Exclude-Other			
Comments			
Even older retired people sh	uld hold some stocks.		
a. True			
b. False c. Don't know			
d. Prefer not to answer			
ALTERNATIVE WORDING			
Older retired people should i	ot hold any stocks.		
a. True			
b. False			
c. Don't know d. Prefer not to answer			
Upon reviewing the question (identifying reason) from the	and its alternative wording format ab pool of questions for the final basic fir	ove, do you feel it should nancial literacy assessment	be included in or excluded test?

Include

Initial Panel Round 2, Group 3 Online Survey Online Survey (continued)



Initial Panel Round 2, Group 3 Online Survey Online Survey (continued)

Upoi (ider	reviewing the question and its alternative wording format above, do you feel it should be included in or excluded tifying reason) from the pool of questions for the final basic financial literacy assessment test?
	nclude
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
	Johnnents
	h
Whie	h of the following instruments is NOT typically associated with spending?
~ r	labit cord
а. L b. (Certificate of deposit.
c. (ash.
а.С ≏Г	realt cara. Ion't know
f. P	refer not to answer
ques	tions for the final basic financial literacy assessment test?
	nclude
	Exclude-Age/life bias
	Exclude-Judament or belief bias
	Exclude-Too specific
	Exclude-Too specific Exclude-To vague
	Exclude-Too specific Exclude-To vague Exclude-Other
	Exclude-Too specific Exclude-To vague Exclude-Other
	Exclude-To vague Exclude-Other Comments
	Exclude-Too specific Exclude-To vague Exclude-Other Comments
	Exclude-Too specific Exclude-To vague Exclude-Other Comments
Scol	Exclude-Too specific Exclude-To vague Exclude-Other Comments
Scol	Exclude-Too specific Exclude-To vague Exclude-Other Comments t and Eric are young men. Each has a good credit history. They work at the same company and make approximated rame salary. Scott has borrowed \$6,000 to take a foreign vacation. Eric has borrowed \$6,000 to buy a car. Who is
Scot	Exclude-Too specific Exclude-To vague Exclude-Other Comments t and Eric are young men. Each has a good credit history. They work at the same company and make approximatel rame salary. Scott has borrowed \$6,000 to take a foreign vacation. Eric has borrowed \$6,000 to buy a car. Who is t to pay the lowest finance charge?
Scotthe stikely	Exclude-Too specific Exclude-To vague Exclude-Other Comments t and Eric are young men. Each has a good credit history. They work at the same company and make approximatel came salary. Scott has borrowed \$6,000 to take a foreign vacation. Eric has borrowed \$6,000 to buy a car. Who is to pay the lowest finance charge?
Scolution Scolut	Exclude-Too specific Exclude-To vague Exclude-Other Comments t and Eric are young men. Each has a good credit history. They work at the same company and make approximatel arme salary. Scott has borrowed \$6,000 to take a foreign vacation. Eric has borrowed \$6,000 to buy a car. Who is t to pay the lowest finance charge? Fric will pay less because the car is collateral for the loan. hey will both pay the same because the rate is set by law.
Scol the s likely a. E b. T c. S	Exclude-Too specific Exclude-To vague Exclude-Other Comments t and Eric are young men. Each has a good credit history. They work at the same company and make approximatel ame salary. Scott has borrowed \$6,000 to take a foreign vacation. Eric has borrowed \$6,000 to buy a car. Who is t to pay the lowest finance charge? Tric will pay less because the car is collateral for the loan. they will both pay the same because the rate is set by law. cott will pay less because people who travel overseas are better risks.
Scol the s likely a. E b. 7 c. S	Exclude-Too specific Exclude-To vague Exclude-Other Comments t and Eric are young men. Each has a good credit history. They work at the same company and make approximatel arme salary. Scott has borrowed \$6,000 to take a foreign vacation. Eric has borrowed \$6,000 to buy a car. Who is t to pay the lowest finance charge? Fric will pay less because the car is collateral for the loan. they will both pay the same because the rate is set by law. cott will pay less because people who travel overseas are better risks. they will both pay the same because they have almost identical financiak backgrounds. too the same because they have almost identical financiak backgrounds.
Scol the s likely a. E b. 7 c. S d. 7 e. C	Exclude-Too specific Exclude-To vague Exclude-Other Comments t and Eric are young men. Each has a good credit history. They work at the same company and make approximately arme salary. Scott has borrowed \$6,000 to take a foreign vacation. Eric has borrowed \$6,000 to buy a car. Who is t to pay the lowest finance charge? Tric will pay less because the car is collateral for the loan. hey will both pay the same because the rate is set by law. cott will pay less because people who travel overseas are better risks. hey will both pay the same because they have almost identical financiak backgrounds. hey will both pay the same because they have almost identical financiak backgrounds. hey will both pay the same because they have almost identical financiak backgrounds.
Scol the s likely a. E b. 7 c. S d. 7 e. E	Exclude-Too specific Exclude-To vague Exclude-To vague Exclude-Other Comments Commen
Scolution Scolut	Exclude-Too specific Exclude-To vague Exclude-Other Comments Comme
Scol the s likely a. E d. 7 f. F Upol ques	Exclude-Too specific Exclude-To vague Exclude-Other Comments Comments I and Eric are young men. Each has a good credit history. They work at the same company and make approximatel areame salary. Scott has borrowed \$6,000 to take a foreign vacation. Eric has borrowed \$6,000 to buy a car. Who is t to pay the lowest finance charge? I ric will pay less because the car is collateral for the loan. They will both pay the same because the rate is set by law. Toott will pay less because people who travel overseas are better risks. They will both pay the same because they have almost identical financiak backgrounds. They refer not to answer The equation above, do you feel it should be included in or excluded (identifying reason) from the pool of tions for the final basic financial literacy assessment test? The provide the final basic financial literacy assessment test? The provide the final basic financial literacy assessment test? The provide the final basic financial literacy assessment test? The provide the final basic financial literacy assessment test? The provide the final basic financial literacy assessment test? The provide the final basic financial literacy assessment test? The provide the final basic financial literacy assessment test? The provide the final basic financial literacy assessment test? The provide the final basic financial literacy assessment test? The provide the final basic financial literacy assessment test? The provide the final basic financial literacy assessment test? The provide the final basic financial literacy assessment test? The provide the provide the provide the provide the provide the provide the final basic financial literacy assessment test? The provide the provid

nitial	Panel Round 2, Group 3 Online Survey Online Survey (continued)
	Exclude-Other
	Comments
Wh	ich of the following statements is true?
a. Ioa b. c. d. e. f. Upo	Banks and other lenders share the credit history of their borrowers with each other and are likely to know of any n payments that you have missed. People have so many loans it is very unlikely that one bank will know your history with another bank. Your bad loan payment record with one bank will not be considered if you apply to another bank for a loan. If you missed a payment more than 2 years ago, it cannot be considered in a loan decision. Don't know Prefer not to answer
que	istions for the final basic financial literacy assessment test?
	Include
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
	Comments
	Comments
Υοι	are likely to obtain the lowest interest on loans with?
а. С b.	Sredit Card companies. Banks/credit unions.
c. d	Check-cashing firms. Don't know
е.	Prefer not to answer
Upo que	on reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of estions for the final basic financial literacy assessment test?
\bigcirc	Include
\bigcirc	Exclude-Age/life bias
\bigcirc	Exclude-Judgment or belief bias
\bigcirc	Exclude-Too specific
\bigcirc	Exclude-To vague
\bigcirc	Exclude-Other

Initial Panel Round 2, Group 3 Online Survey Online Survey (continued)

k in several different companies? or excluded (identifying reason) from the pool c
k in several different companies? or excluded (identifying reason) from the pool c
k in several different companies? or excluded (identifying reason) from the pool c
k in several different companies? or excluded (identifying reason) from the pool c
k in several different companies? or excluded (identifying reason) from the pool c
or excluded (identifying reason) from the pool o
or excluded (identifying reason) from the pool c
or excluded (identifying reason) from the pool c
or excluded (identifying reason) from the pool of
or excluded (identifying reason) from the pool c
or excluded (identifying reason) from the pool c

Initial Panel Round 2, Group 4 Online Survey Online Survey

Dear Expert Panel Member,

Thank you for participating in this valuable financial literacy research.

In Round 2 of the Initial Panel review, you will be evaluating questions in the pool for financial literacy assessment instrument. The final instrument will be administered to U.S. adults age 18 and over using an online survey.

As you read each question and possible answers (correct answer is bold), identify whether you believe the question should be included or excluded from an instrument used to determine a basic financial literacy score. The goal of this round is to drastically reduce the number of questions in the pool. You are only reviewing 20% of the pool, therefore, please feel free to be critical in your selection of the questions to include. Only include questions you strongly feel should be included; all other questions should be excluded.

Questions that are marked included will stay in the pool for the next round of review. If you determine a question should be excluded, please identify the reason you believe it should be excluded, please select one of the listed reasons for exclusion:

• **Age/life cycle bias**-Question content is directed toward a specific age group or life cycle. A participant not in the age group or life cycle would probably respond with an incorrect answer.

Judgment or belief bias--The question's response could be influenced by the participant's personal judgments or beliefs. Participants with different beliefs would respond differently to this question.

Too specific--The content of this question is too specific and would not necessarily be financial knowledge useful to the population majority.

Too vague--The content of this question is too general or vague and would not be a good assessment of financial literacy knowledge. Participants could easily guess the answer without having financial knowledge.

Other-Any questions you feel would not be representative of a participant's financial literacy. If you choose this response, please note your reason and/or use the comment section to explain.

Please feel free to comment on any of the questions you identify to include for the next round.

Initial Panel - Round 2

An investment with a high return is likely to be high risk.

a. True

- b. False
- c. Don't know
- d. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

- Include
- Exclude-Age/life bias
- Exclude-Judgment or belief bias
- Exclude-Too specific
- Exclude-To vague
- Exclude-Other
- Comments

Initial Panel Round 2, Group 4 Online Survey Online Survey (continued)

You invested \$1,000 in a stock two years ago. The stock's trading price declined 40% the first year and rose 40% the next. As a result, you've
a. lost money. b. made money. c. broke even. d. Don't know e. Prefer not to answer
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other
Comments
a. meets rigorous standards set by the SEC. b. is recommded by the SEC. c. has simply paid the \$150 registration fee to the SEC. d. Don't know e. Prefer not to answer
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other
Commonte
Comments
h.
You can't lose money investing in a U.S. Treasury bond, because it is backed by the United States Government.
a. True b. False c. Don't know d. Prefer not to answer

Initial Panel Round 2, Group 4 Online Survey Online Survey (continued)

Include	
Exclude-Age/life bias	
Exclude-Judgment or belief bias	
Exclude-Too specific	
Exclude-To vague	
Exclude-Other	
Comments	
a. A Traditional IRS or Individual Retirment Account b. A Roth IRA c. A 401(k) plan d. Don't know e. Prefer not to answer	
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) fro questions for the final basic financial literacy assessment test?	m the pool of
Include	
Exclude-Age/life bias	
Exclude-Judgment or belief bias	
Exclude-Too specific	
Exclude-To vague	
Exclude-Other	
Comments	
Which policy provides the most coverage at the lowest cost for a young family? a. Renewable term life b. Whole life c. Universal life d. Don't know e. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) fro questions for the final basic financial literacy assessment test?	m the pool of
Include	

Initial Panel Round 2, Group 4 Online Survey Online Survey (continued)



Exclude-Age/life bias

Initial Panel Round 2, Group 4 Online Survey Online Survey (continued)



Initial Panel Round 2, Group 4 Online Survey Online Survey (continued)

E E	xclude-To vaque
	volude_Other
C	omments
Perso	nal financial planning involves
a. esi	tablishing an adequate financial record keeping system.
b. de c. mi	veloping a sound yearly budget of expenses and income. nimizing taxes and insurance expenses
d. pr	eparing plans for future financial needs and goals.
e exa	amining your investment portfolios to maximize returns. on't know
g. Pre	efer not to answer
Upon questi	reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of ions for the final basic financial literacy assessment test?
la la	
	xclude-Age/life blas
E:	xclude-Judgment or belief blas
E:	xclude-1 oo specific
E:	xclude-To vague
E:	xclude-Other
	omments
	h
Which	of the following statements is TRUE about the annual percentage rate (APR)?
a AF	PR is the actual rate of interest paid over the life of the loan
b. AF	PR is expressed as a percentage on an annual basis.
c. AF d. AF	?R is a good measure of comparing loan costs. ?R takes into account all loan fees.
e All	of the above.
g. Pre	efer not to answer
Upon questi	reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of ions for the final basic financial literacy assessment test?
🗌 In	clude
E:	xclude-Age/life bias
E:	xclude-Judgment or belief bias
E:	xclude-Too specific

Initial Panel Round 2, Group 4 Online Survey Online Survey (continued)

Exclude-To vaque	
Exclude-Other	
Comments	
h	
would not ordina	arily be covered under a home owners policy.
a War	
b. Earthquake.	
c. Flood. d. Your being sund by se	maana far slander
e All of the above.	meone for stander.
f. Don't know	
g. Prefer not to answer	
Upon reviewing the questior questions for the final basic	above, do you feel it should be included in or excluded (identifying reason) from the pool of inancial literacy assessment test?
Include	
Exclude-Age/life bias	
Exclude-Judgment or bel	iefbias
Exclude-Too specific	
Exclude-To vague	
Exclude-Other	
Comments	
,	
The earlier you start saving increase over time. a. True b. False c. Don't know d. Prefer not to answer	for retirement, the more money you will have because the effects of compounding interest
Upon reviewing the questior questions for the final basic	above, do you feel it should be included in or excluded (identifying reason) from the pool of financial literacy assessment test?
Include	
Exclude-Age/life bias	
Exclude-Judgment or bel	iefbias
Exclude-Too specific	
Exclude-To vague	
Exclude-Other	
Comments	

Initial Panel Round 2, Group 4 Online Survey Online Survey (continued)

Repeatedly refinancing your home mortgage over a short period of time results in added fee and points that further increase your debt.
a. True b. False c. Don't know d. Prefer not to answer
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other
Comments
 c. Don't know d. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Evaluate the status have
Exclude-Age/Inte blas
Exclude-Judgment or belief blas
Exclude-To vague
Exclude-Other
Comments
Your credit rating is not affected by how much you charge on your credit cards. a. True b. False c. Don't know d. Prefer not to answer
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of

Initial Panel Round 2, Group 4 Online Survey Online Survey (continued)



Initial Panel Round 2, Group 4 Online Survey Online Survey (continued)

a. True b. False c. Don't kn d. Prefer n	ow ot to answer
Upon revi (identifyin	ewing the question and its alternative wording format above, do you feel it should be included in or excluded g reason) from the pool of questions for the final basic financial literacy assessment test?
	e
Exclue	de-Age/life bias
Exclusion	de-Judgment or belief bias
Exclusion	de-Too specific
Exclue	de-To vaque
Exclue	de-Other
~ ~	
Comn	nents
It is easy	to find mutual funds that have annual fees of less than one percent of assets.
a. True	
b. False c. Don't kn	ow
d. Prefer n	ot to answer
ALTERNA	TIVE WORDING
It is hard	to find mutual funds that have annual fees of less than one percent of assets.
a. True b. False c. Don't kn d. Prefer n	ow ot to answer
Upon revi (identifyin	ewing the question and its alternative wording format above, do you feel it should be included in or excluded g reason) from the pool of questions for the final basic financial literacy assessment test?
Includ	e
Exclusion	de-Age/life bias
Exclue	de-Judgment or belief bias
Exclus	de-Too specific
 Exclusion 	
U Exclu	de-Other
Comn	nents
0 001111	
	h
You shou	ld invest in either mutual funds or a large number of different stocks instead of just a few stocks.
a. True b. False c. Don't kn	ow
a. Preter n	
ALIERNA	TIVE WORDING

Initial Panel Round 2, Group 4 Online Survey Online Survey (continued)

Proce Procession	 True Prove Prove the service of the ser	You should invest most of you	r money in a few good stocks that you select rather than in lots of stocks or in mutual funds
A rate 2 point now 4 Preferencies answer Joon reviewing the question and its alternative wording format above, do you feel it should be included identifying reasons from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-To specific Exclude-To vague Exclude-Other Comments A rate A rate Induction answer Stag money in a bank savings account to pay off credit card debt is usually a good idea. 1 True Preference Dom'now 1 True Preference 1 True Preference 1 Preference Dom'now 1 Preference Dom'now 1 Preference Dom'now in a bank savings account to pay off credit card debt is usually a bad idea. 1 True Preference Preference Dom'now Preference Dom'now Preference	A Price Don't now # Prefer not to answer Jon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Judgment or belief bias Exclude-Too specific Exclude-Other Comments Influence Prefer not to answer Jaing money in a bank savings account to pay off credit card debt is usually a good idea. Influence Prefer not to answer LIEENATIVE WORDING Jaing money in a bank savings account to pay off credit card debt is usually a bad idea. 1. True Prefer not to answer LIEENATIVE WORDING Jaing money in a bank savings account to pay off credit card debt is usually a bad idea. 1. True Prefer not to answer Jpoint reviewing the question and its alternative wording format above, do you feel it should be included in or excluded identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Judgment or belief bias Exclude-Agelife bias Exclude-Judgment or belief bias Exclude	a. True	
	Prevention	b. False	
Include Include Exclude-Age/life bias Exclude-To specific Comments Include Include Include Exclude-To specific Exclude-Other Include Include Include Include Include Include Exclude-Other Include Include Include Include Include Exclude-To specific Exclude-Other Include Include-	ppon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Too specific Exclude-Too ague Exclude-Too ague Comments Comments Prefer not to answer Include Include Prefer not to answer Include Prefer not to answer Prefer not to answer Include Exclude-Jougnetor to belief bias Exclude-Jougnetor to belief bias Exclude-Too specific Comments Include Prefer not to answer Atternative wording format above, do you feel it should be included in or excluded dientifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Include Exclude-Jougnet to belief bias Exclude-Jougnet to belief bias Exclude-Judgment or belief bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-Too specific </td <td>d. Prefer not to answer</td> <td></td>	d. Prefer not to answer	
Include Exclude-Age/life bias Exclude-To specific Exclude-To squue Exclude-Other Comments Include Palse Particle To answer Exclude-To specific Ex	Include Exclude-Agelife bias Exclude-To specific Exclude-To vague Exclude-Other Comments Comments Palse Particle and the provide the provi	Jpon reviewing the question ar identifying reason) from the po	nd its alternative wording format above, do you feel it should be included in or excluded ool of questions for the final basic financial literacy assessment test?
Exclude-Age/life bias Exclude-Too specific Exclude-Too specific Exclude-To vague Exclude-Other Comments	Exclude-Age/life bias Exclude-Too specific Exclude-Too specific Exclude-Other Comments Palse Particle Paritele Particle Particle Par	Include	
Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments Comments Intervet False Don't now I. Prefer not to answer Harden on the pool of questions for the final basic financial literacy assessment test? I. Frude-Prefer not be answer Exclude-Agelifie bias Exclude-Agelifie bias Exclude-Agelifie bias Exclude-Agelifie bias Exclude-Other Exclude-Other I. Fuelow Comments Don reviewing the question and its alternative wording format above, do you feel it should be included in or excluded dientifying reason) from the pool of questions for the final basic financial literacy assessment test? Exclude-Agelifie bias Exclude-Other Exclude-Other Comments Direction Comments Direction Exclude-Other	Exclude-Judgment or belief bias Exclude-Too specific Exclude-Other Comments Or a family with a working husband and a wife staying home to take care of their young children, life insurance that will server a family with a working husband and a wife staying home to take care of their young children, life insurance that will server a family with a working husband and a wife staying home to take care of their young children, life insurance that will server a family with a working husband and a wife staying home to take care of their young children, life insurance that will server a family with a working husband and a wife staying home to take care of their young children, life insurance that will server a server a family with a working husband and a wife staying home to take care of their young children, life insurance that will server a server of incorane is not an any thin insurance.	Exclude-Age/life bias	
Exclude-To vague Exclude-Other Comments Junctify and the problem of the	Exclude-Too specific Exclude-To vague Exclude-Other Comments Sing money in a bank savings account to pay off credit card debt is usually a good idea. True True Parse Contract or answer LICENATIVE WORDING Jarse Jon throw Parse Parse Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Jor ague Comments To a family with a working husband and a wife staying home to take care of their young children, life insurance that will parse merce.	Exclude-Judgment or belief	bias
Exclude-Other Comments Comments Single control of a bank savings account to pay off credit card debt is usually a good idea. Trans False So Part Now Prefer not to answer Answer Arree Prefer Not to answer Jon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-To vague Exclude-To vague Exclude-To vague Exclude-To vague Comments	Exclude-To vague Exclude-Other Comments Comments Comments Parale Comments Parale Comments Parale Parale Comments Parale	Exclude-Too specific	
Exclude-Other Comments Comment	Exclude-Other Comments	Exclude-To vague	
Comments Using money in a bank savings account to pay off credit card debt is usually a good idea. True Sease Control False Comments Comm	Comments Co	Exclude-Other	
Comments Using money in a bank savings account to pay off credit card debt is usually a good idea. True False Contraction True Comments Cuterents Comments	Comments Comments Cusing money in a bank savings account to pay off credit card debt is usually a good idea. True Solar know Comments Comments Comments Cusing money in a bank savings account to pay off credit card debt is usually a bad idea. The Comments Cusing money in a bank savings account to pay off credit card debt is usually a bad idea. The Cusing money in a bank savings account to pay off credit card debt is usually a bad idea. The Cusing money in a bank savings account to pay off credit card debt is usually a bad idea. The Cusing money in a bank savings account to pay off credit card debt is usually a bad idea. The Cusing money in a bank savings account to pay off credit card debt is usually a bad idea. The Cusing money in a bank savings account to pay off credit card debt is usually a bad idea. The Cusing money in a bank savings account to pay off credit card debt is usually a bad idea. The Cusing money in a bank savings account to pay off credit card debt is usually a bad idea. The Cusing money in a bank savings account to pay off credit card debt is usually a bad idea. The Cusing money in a bank savings account to pay off credit card debt is usually a bad idea. The Cusing money in a bank savings account to pay off credit card debt is usually a bad idea. The Cusing money in a bank savings account to pay off credit card debt is usually a bad idea. The Cusing money in a bank savings account to pay off credit card debt is usually a bad idea. The Cusing money in a bank savings account to pay off credit card debt is usually a bad idea. The Cusing money in a bank savings account to pay off credit card debt is usually a bad idea. The Cusing money in a bank savings account to pay off credit card debt is usually a bad idea. The Cusing Cusing the Questions for the final basic financial literacy assessment test? The Cusing Cu		
Using money in a bank savings account to pay off credit card debt is usually a good idea.	Solution A Using money in a bank savings account to pay off credit card debt is usually a good idea. Price 2 False 2 Ont know 3. Prefer not to answer ALTERNATIVE WORDING 2. Salse 2. Ont know 3. Prefer not to answer 3. False 2. Don't know 3. Prefer not a bank savings account to pay off credit card debt is usually a bad idea. 1. True 2. False 2. Don't know 3. Prefer not to answer Jon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Judgment or belief bias Exclude-To vague Exclude-To vague Exclude-To vague Exclude-Other Image: Comments Image: So a family with a working husband and a wife staying home to take care of their young children, life insurance that will prefere the wars of income is not enough life insurance	Comments	
Using money in a bank savings account to pay off credit card debt is usually a good idea.	Using money in a bank savings account to pay off credit card debt is usually a good idea.	1.	
Using money in a bank savings account to pay off credit card debt is usually a good idea.	Using money in a bank savings account to pay off credit card debt is usually a good idea. a. True 0. Failse 2. Don't know ALTERNATIVE WORDING Using money in a bank savings account to pay off credit card debt is usually a bad idea. a. True b. Faise 2. Don't know b. Faise 2. Don't know b. Faise 2. Don't know 1. Prefer not to answer Jpon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Too specific Exclude-To vague Exclude-Other Comments		
 a. True b. False b. Don't know d. Prefer not to answer Upon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other 	a. True b. False c. Don't know c. Prefer not to answer Upon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-To vague Comments Comm	ALTERNATIVE WORDING Jsina monev in a bank savinas	s account to pay off credit card debt is usually a bad idea.
 b. False b. False b. Don't know d. Prefer not to answer Jpon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other 	 Faise Don't know Prefer not to answer Jpon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-To vague Exclude-Other Comments 	a True	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
2. Don't know 3. Prefer not to answer Joon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments	2. Don't know 3. Prefer not to answer 3. Jopon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-To vague Exclude-To vague Comments Comments	b. False	
Jpon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-Other Comments	Joon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-Other Comments	c. Don't know d. Profor not to answor	
Jpon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments	Jpon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded itdentifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments	a. Freier not to answer	
 Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments 	 Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments 	Upon reviewing the question ar (identifying reason) from the po	nd its alternative wording format above, do you feel it should be included in or excluded ool of questions for the final basic financial literacy assessment test?
 Exclude-Age/life bias Exclude-Too specific Exclude-To vague Exclude-Other Comments 	 Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments 	Include	
 Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments 	 Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments 	Exclude-Age/life bias	
 Exclude-Too specific Exclude-To vague Exclude-Other Comments 	 Exclude-Too specific Exclude-To vague Exclude-Other Comments For a family with a working husband and a wife staying home to take care of their young children, life insurance that will pplace three years of income is not enough life insurance. 	Exclude-Judgment or belief	bias
Exclude-To vague Exclude-Other Comments	Exclude-To vague Exclude-Other Comments For a family with a working husband and a wife staying home to take care of their young children, life insurance that will replace three years of income is not enough life insurance.	Exclude-Too specific	
Exclude-Other Comments	Exclude-Other Comments For a family with a working husband and a wife staying home to take care of their young children, life insurance that will replace three years of income is not enough life insurance.	Exclude-To vague	
Comments	Comments For a family with a working husband and a wife staying home to take care of their young children, life insurance that will replace three years of income is not enough life insurance.	Exclude-Other	
Comments	For a family with a working husband and a wife staying home to take care of their young children, life insurance that will replace three years of income is not enough life insurance.		
	For a family with a working husband and a wife staying home to take care of their young children, life insurance that will	Comments	
	For a family with a working husband and a wife staying home to take care of their young children, life insurance that will	1.	
	For a family with a working husband and a wife staying home to take care of their young children, life insurance that will polace three years of income is not enough life insurance.		
	For a family with a working husband and a wife staying home to take care of their young children, life insurance that will replace three years of income is not enough life insurance.		

Initial Panel Round 2, Group 4 Online Survey Online Survey (continued)

a. True
b. False
c. Don't know d. Prefer not to answer
AL TERNATIVE WORDING
For a family with a working husband and a wife staying home to take care of their young children, life insurance that will replace three years of income is more than enough.
a. True
c. Don't know
d. Prefer not to answer
Upon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other
Comments
Young people should hold somewhat riskier financial investments than older people.
a. True
c. Don't know
d. Prefer not to answer
ALTERNATIVE WORDING
Older people should hold somewhat riskier financial investments than young people.
a True
b. False
c. Don't know d. Prefer not to answer
Upon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
(
Include Exclude-Age/life bias
 Include Exclude-Age/life bias Exclude-Judgment or belief bias
 Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific
 Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague
 Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other
 Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other
 Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments
 Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments

Initial Panel Round 2, Group 4 Online Survey Online Survey (continued)

The more you diversify among stocks, the more of your money you can invest in stocks.

- a. True
- b. False
- c. Don't know
- d. Prefer not to answer
- ALTERNATIVE WORDING

The more you diversify among stocks, the less of your money you should invest in stocks.

- a. True
- b. False
- c. Don't know
- d. Prefer not to answer

Upon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

- Include
- Exclude-Age/life bias
- Exclude-Judgment or belief bias
- Exclude-Too specific
- Exclude-To vague
- Exclude-Other
- Comments

If you had a savings account at a bank, which of the following would be correct concerning the interest that you would earn on this account?

a. Earnings from savings account interest may not be taxed.

1

- b. Income tax may be charged on the interest if your income is high enough
- c. Sales tax may be charged on the interest that you earn.
- d. You cannot earn interest until you pass your 18th birthday.
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

- Include
- Exclude-Age/life bias
- Exclude-Judgment or belief bias
- Exclude-Too specific
- Exclude-To vague
- Exclude-Other
- Comments

Initial Panel Round 2, Group 4 Online Survey Online Survey (continued)

Chelsea worked her way through college earning \$15,000 per year. After graduation, her first job pays \$30,000. The total dollar amount Chelsea will have to pay in Federal Income taxes in her new job will:
a. Double, at least, from when she was in college.
b. Go up a little from when she was in college.
c. Stay the same as when she was in college.
d. Be lower than when she was in college.
e. Don't know
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Include
Exclude-Age/life bias
Exclude Judgment or belief bias
Exclude-To vague
Exclude-Other
Comments
 b. Stocks. c. A U.S. Government savings bond. d. A savings account. e. Don't know f. Prefer not to answer
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Include
Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other
Comments
h
Deb and Man and the same and Advance Of Man beau and in 20,000 a second different day the Advance Of Deb
Rob and wary are the same age. At age 25, wary began saving \$2,000 a year winte Rob saved nothing. At age 50, Rob realized that he needed money for retirement and stated saving \$4,000 ner year while Man, kent saving her \$2,000 Now
they are both 75
a. They would each have the same amount because they put away exactly the same.
h Roh herause he saved more each vear

Initial Panel Round 2, Group 4 Online Survey Online Survey (continued)

ving the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? Age/life bias
-Age/life bias
e-Age/life bias
. Underson the shall of this sector is a sector of the sec
-Judgment of benefibias
a-Too specific
a-To vague
o-Other
ints
OW
ow ot to answer ving the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test?
ow ot to answer ving the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test?
ow ot to answer ving the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? Age/life bias
ow of to answer ving the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? Age/life bias Judgment or belief bias
ow ot to answer ving the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? Age/life bias Judgment or belief bias
ow of to answer wing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? e-Age/life bias e-Judgment or belief bias e-Too specific e-To vague
ow of to answer wing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? Age/life bias Judgment or belief bias Too specific To vague
ow of to answer wing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? Age/life bias Judgment or belief bias Too specific To vague
ow of to answer wing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? Age/life bias Judgment or belief bias Too specific To vague Other
ow of to answer wing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? -Age/life bias Age/life bias Too specific To vague Other
ow of to answer wing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? Age/life bias Joudgment or belief bias To vague To vague Other
ow of to answer wing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? Age/life bias Judgment or belief bias Too specific To vague Other
ow of to answer wing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? Age/life bias Judgment or belief bias Too specific To vague Other
ow of to answer wing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? Age/life bias Judgment or belief bias Too specific To vague Other
ow of to answer wing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of or the final basic financial literacy assessment test? >-Age/life bias >-Judgment or belief bias >-Too specific >-To vague >-Other =

Initial Panel Round 2, Group 4 Online Survey Online Survey (continued)

Include	
Exclude-Age/life bias	
Exclude-Judgment or belief bias	
Exclude-Too specific	
Exclude-To vague	
Exclude-Other	
Comments	
If interest rates are currently low but are expected to rise, which one of the following is the most appropriate action? a. Use long-term loans and short-term savings. b. Use short-term loans and long-term savings. c. Do nothing. d. Don't know e. Prefer not to answer	
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of question for the final basic financial literacy assessment test?	ons
Exclude-Age/life bias	
Exclude-Judgment or belief bias	
Exclude-Too specific	
Exclude-To vague	
Exclude-Other	
Comments	

Appendix H

Initial Panel Round 2, Group 5 Online Survey Online Survey

Dear Expert Panel Member,

Thank you for participating in this valuable financial literacy research.

In Round 2 of the Expert Panel review, you will be evaluating questions in the pool for financial literacy assessment instrument. The final instrument will be administered to U.S. adults age 18 and over using an online survey.

As you read each question and possible answers (correct answer is bold), identify whether you believe the question should be included or excluded from an instrument used to determine a basic financial literacy score. The goal of this round is to drastically reduce the number of questions in the pool. You are only reviewing 20% of the pool, therefore, please feel free to be critical in your selection of the questions to include. Only include questions you strongly feel should be included; all other questions should be excluded.

Questions that are marked included will stay in the pool for the next round of review. If you determine a question should be excluded, please identify the reason you believe it should be excluded. Also, identify the reasons for exclusion from the list below:

• Age/life cycle bias-Question content is directed toward a specific age group or life cycle. A participant not in the age group or life cycle would probably respond with an incorrect answer.

Judgment or belief bias--The question's response could be influenced by the participant's personal judgments or beliefs. Participants with different beliefs would respond differently to this question.

Too specific--The content of this question is too specific and would not necessarily be financial knowledge useful to the population majority.

• **Too vague**—The content of this question is too general or vague and would not be a good assessment of financial literacy knowledge. Participants could easily guess the answer without having financial knowledge.

• **Other--**Any questions you feel would not be representative of a participant's financial literacy. If you choose this response, please note your reason and/or use the comment section to explain.

Please feel free to comment on any of the questions you identify to include for the next round.

Initial Panel - Round 2

Let's say the price/earnings ratio on Standard & Poor's 500-stock index is 23 and its dividend yield is 2.5%. This means that the stock market is relatively

a. undervalued by historical standards.

b. over valued.

c. fairly valued.

d. Don't know

e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

Include

Exclude-Age/life bias

- Exclude-Judgment or belief bias
- Exclude-Too specific
- Exclude-To vague
- Exclude-Other

Comments

Initial Panel Round 2, Group 5 Online Survey Online Survey (continued)

A young investor willing to take moderate risk for above-average growth would be most interested in:
a. Treasury bills b. Money market mutual funds c. Balanced stock funds d. Don't know e. Prefer not to answer
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other
Comments
h
Personal finance literacy can help you:
 a. avoid being victimized by financial scams. b. buy the right kind of insurance to protect you from catastrophic risk. c. learn the right approach to invest for your future needs. d. lead a financially secure life through forming healthy spending habits. e. do all of the above. f. Don't know
g. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other

Comments

Initial Panel Round 2, Group 5 Online Survey Online Survey (continued)

_	is not a cost of leasing an apartment.
а.	Security deposit.
) .	Monthly rental payments
1	Expenses incurred for non-compliance of lease terms. Medical expenses of your friend who fell and broke his arm on the icy pavement.
	Security deposit retained by the landlord for damages to property beyond normal wear and tear.
	Don't know
1.	Prefer not to answer
Jp	on reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool o estions for the final basic financial literacy assessment test?
	Include
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
	Comments
	h.
1/F	nich of the following is EALSE?
1.	As shareholders of a mutual fund, you have a right to tell fund managers what securities to buy.
	A mutual fund is a diversified collection of securities used as an investment vehicle.
	A mutual fund is an investment corporation that raises funds from investors and purchases securities.
	Your ownership in a mutual rund is proportional to the number of shares you own in the rund. None of the above
	Don't know
	Prefer not to answer
Jp	on reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool o estions for the final basic financial literacy assessment test?
	Include
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
	Exclude-Other

Initial Panel Round 2, Group 5 Online Survey Online Survey (continued)

 a. buying low and selling high. b. complex calculations of risk and return. c. selling securities to minimize capital. d. Don't know e. Prefer not to answer
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Include
Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other
Comments
Comments
h
 f. Don't know g. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool or questions for the final basic financial literacy assessment test?
Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other
Comments
"
Which of the following statements is FALSE?
a. Term insurance is an excellent investment vehicle. b. You receive no benefits when your term insurance policy expires.

- c. A term insurance policy is the least expensive form of life insurance.
 d. A decreasing-term policy reduces coverage over time.

Initial Panel Round 2, Group 5 Online Survey Online Survey (continued)

e A level-term policy guarant f. Don't know	ees a fixed premium over the life of the contract.
g. Prefer not to answer	
Upon reviewing the question a questions for the final basic fin	bove, do you feel it should be included in or excluded (identifying reason) from the pool of ancial literacy assessment test?
Include	
Exclude-Age/life bias	
Exclude-Judgment or belief	bias
Exclude-Too specific	
Exclude-To vague	
Exclude-Other	
Comments	
Your savings accounts in a fe	derally insured commercial bank are insured by
a SIPC to the maximum am	ount of\$10,000 per account
b. FDIC to the maximum an	nount of \$100,000.
c. FDIC to the maximum amo	ount of\$50,000 per account.
a. SLIC to the maximum among FNMA to the maximum and the second secon	unt of\$100,000. Jount of\$100,000 per account
f. Don't know	
g. Prefer not to answer	
Upon reviewing the question a questions for the final basic fin	bove, do you feel it should be included in or excluded (identifying reason) from the pool of ancial literacy assessment test?
lnclude	
Exclude-Age/life bias	
Exclude-Judgment or belief	bias
Exclude-Too specific	
Exclude-To vague	
Exclude-Other	
Commente	
Comments	
1.	
A stock mutual fund combines	the money of many investors to buy a variety of stocks.
a. True	
b. False	
c. Don't know d. Prefer not to answer	
Upon reviewing the question a	bove, do you feel it should be included in or excluded (identifying reason) from the pool of
Juestions for the final basic fin	anciar meracy assessment lest?
	Continued on the next

Initial Panel Round 2, Group 5 Online Survey Online Survey (continued)

_	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
	Comments
	h
Cre	ditors are required to tell you the APR (Annual Percentage Rate) that you will pay when you get a loan.
a.T	rue
c. D	ion" know
d. F	refer not to answer
Upd	on reviewing the guestion above, do you feel it should be included in or excluded (identifying reason) from the pool of
que	istions for the final basic financial literacy assessment test?
	Include
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
	Commente
	Comments
	6
You	ir credit rating is not affected by how much you charge on your credit cards.
a. T	rue
b.F	alse Ion" know
d. F	refer not to answer
Upo	on reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of
que	stions for the final basic financial literacy assessment test?
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
	Comments

Initial Panel Round 2, Group 5 Online Survey Online Survey (continued)

Making payments late on your bills can make it more difficult to take out a loan.
a. True b. False c. Don't know d. Prefer not to answer
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Include
Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other
Comments
1
It is possible to invest in the stock market in a way that makes it hard for people to take unfair advantage of you.
a. True b. False c. Don't know d. Prefer not to answer
ALTERNATIVE WORDING
There is no way to avoid people taking advantage of you if you try to invest in the stock market.
a. True b. False c. Don't know d. Prefer not to answer
Upon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?
Include
Exclude-Age/life bias
Exclude-Judgment or belief bias
Exclude-Too specific
Exclude-To vague
Exclude-Other
Comments

Initial Panel Round 2, Group 5 Online Survey Online Survey (continued)

To make money in the stock market, you should not buy and sell stocks too often. a. True b. False c. Don't know d. Prefer not to answer ALTERNATIVE WORDING To make money in the stock market, you have to buy and sell stocks often. a. True b. False c. Don't know d. Prefer not to answer Upon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments 1. An employee of a company with publicly traded stock should have little or none of his or her retirement savings in the company's stock. **a. True** b. False c. Don't know d. Prefer not to answer ALTERNATIVE WORDING An employee of a company with publicly traded stock should have a lot of his or her retirement savings in the company's stock. a. True b. False c. Don't know d. Prefer not to answer Upon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague

Exclude-Other

Initial Panel Round 2, Group 5 Online Survey Online Survey (continued)

It is important to take a look at your investments periodically to see if you need to make changes.

a. True

b. False

c. Don't know

d. Prefer not to answer

ALTERNATIVE WORDING

Once you have made an initial decision about the investment mix for your portfolio, you should avoid making changes to your portfolio until you are close to retirement.

a. True

- b. False
- c. Don't know
- d. Prefer not to answer

Upon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

Include

Exclude-Age/life bias

Exclude-Judgment or belief bias

1,

- Exclude-Too specific
- Exclude-To vague
- Exclude-Other

Comments

You could save money in interest costs by choosing a 15-year rather than a 30-year mortgage.

- a. True
- b. False
- c. Don't know
- d. Prefer not to answer

ALTERNATIVE WORDING

You could save money in interest costs by choosing a 30-year rather than a 15-year mortgage.

- a. True
- b. False
- c. Don't know

d. Prefer not to answer

Upon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

Include

Exclude-Age/life bias

Exclude-Judgment or belief bias

Initial Panel Round 2, Group 5 Online Survey Online Survey (continued)

_	
Exc	ude-Too specific
Exc	lude-To vague
Exc	lude-Other
Con	aments
	h
Junting	a single company stack yough, provides a safer rature than a stack mutual fund
suying	a single company stock usually provides a saler return than a stock mutual fund.
. True). False	
. Don't l	(now
i. Preier	not to answer
LTERN	IATIVE WORDING
Buying a	a stock mutual fund usually provides a safer retum than a single company stock.
a. True	
. False	
l. Prefer	now not to answer
Joon re	viewing the guestion and its alternative wording format above, do you feel it should be included in or excluded
identify	ing reason) from the pool of questions for the final basic financial literacy assessment test?
Incl	ude
Exc	lude-Age/life bias
Exc	lude-Judgment or belief bias
Exc	lude-Too specific
Exc	lude-To vague
Exc	lude-Other
0	
Con	nments
nflation	can cause difficulty in many ways. Which group would have the greatest problem during periods of high inflation
hat last	several years?
a. Olde	r, working couples saving for retirement.
o. Olde	er people living on fixed retirement income.
I. Your	ng working couples with children.
e. Don'i	
. Prefe	r not to answer
Jpon re	viewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of
Exc	ude-Age/life bias
Exc	ude-Judgment or belief bias
Exc	lude-Too specific

Initial Panel Round 2, Group 5 Online Survey Online Survey (continued)

~	Exclude-To vague
0	Exclude-Other
0	Comments
	1.
<i>Wh</i> sud	ich of the following types of investment would best protect the purchasing power of a family's savings in the event Iden increase in inflation?
a	A 10-year bond issued by a corporation. A certificate of deposit at a bank
	A twenty-five year corporate bond.
 9. i	A nouse infanced with a fixed-rate mortgage. Don't know
f. F	Prefer not to answer
Jpc que	on reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool o stions for the final basic financial literacy assessment test?
	Include
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
-	
	Comments
	h
Nh	ich of the following statements is NOT correct about most ATM (Automated Teller Machine) cards?
a.	You can generally get cash 24 hours-a-day.
b.	You can generally obtain information concerning your bank balance at an ATM machine.
d.	You must have a bank account to have an ATM card.
9. I	Don't know Prefer not to answer
Jpc	on reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool
que	stions for the final basic financial literacy assessment test?
	Include
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Judgment or belief bias Exclude-Too specific
	Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague

Initial Panel Round 2, Group 5 Online Survey Online Survey (continued)

David just found a job with a take-home pay of \$2,000 per month. He must pay \$900 for rent and \$150 for groceries each month. He also spends \$250 per month on it transportation. If he budgets \$100 each month for clothing, \$200 for restaurants and \$250 nerverything else, how long will it take him to accumulate savings of \$000. a. 3 months b. 4 months c. 1 months c. 1 months c. 2 months d. 3 months d. 4 months d. 5 months d. 5 molt equation above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? d. Include d. Exclude-Too specific d. Exclude-Too specific d. Exclude-Too specific d. Comments d. 4 months d.		Comments
David just found a job with a take-home pay of \$2,000 per month. He must pay \$900 for rent and \$150 for groceries each month. He also spends \$250 per month on transportation. If he budgets \$100 each month for clothing, \$200 for restaurants and \$250 for everything else, how long will it take him to accumulate savings of \$600. a. 3 months b. 4 months c. 1 month, d. 2 months, e. Don't know f. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-AgeIIfe bias Exclude-Too specific Exclude-Too specific Exclude-Too specific Comments The ferlor government will deduct if from your paycheck. You for the following is true about sales taxes? The final government will deduct if from your paycheck. You for the final basic financial literacy assessment test? Include Exclude-Too specific Exclude-Too specific Comments The ferlor government will deduct if from your paycheck. You don't have to pay the tax if your income is very low. The ferlor government will deduct if from your paycheck. You for the final basic financial literacy assessment test? Include Exclude-Too specific Exclude-Too specific Comments The ferlor government will deduct if from your paycheck. You don't have to pay the tax if your income is very low. The ferlor government will deduct if from your paycheck. You don't have to pay the tax if your income is very low. The ferlor government will be uncluded in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Too specific Exclude-Too specific Exclude-Too specific Exclude-Too to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Too specific Exclude-Too specific Exclude-Too specific Exclude-Too specific Exclu		
Devid just found a job with a take-home pay of \$2,000 per month. He must pay \$900 for rent and \$150 for groceries each month. He also spends \$250 per month on transportation. If he budgets \$100 each month for clothing, \$200 for restaurants and \$250 for everything else, how long will it take him to accumulate savings of \$600. a. 3 months b. 4 months c. 1 month, d. 2 months b. 5 months c. Don't know f. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment tes? budget base base for a second base base bases taxes? Comments c. Exclude-Too specific comments f. The following is true about sales taxes? a. The national sales tax percentage parts is 6%. b. The following government will deduct if from your psycheck. C. You don't have to pay the tax if your income is very low. d. If makes things more expensive for you to buy. Don't know f. Prefer not to answer both the dillowing is true about sales taxes? base base to answer but have to pay the tax if your income is very low. d. If makes things more expensive for you to buy. base base from the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Comments Comm		
 a. 3 months b. 4 months c. 1 months d. 2 months d. 3 months d. 4 months <lid< td=""><td>Da mo and</td><td>d just found a job with a take-home pay of \$2,000 per month. He must pay \$900 for rent and \$150 for groceries each h. He also spends \$250 per month on transportation. If he budgets \$100 each month for clothing, \$200 for restaurants \$250 for everything else, how long will it take him to accumulate savings of \$600.</td></lid<>	Da mo and	d just found a job with a take-home pay of \$2,000 per month. He must pay \$900 for rent and \$150 for groceries each h. He also spends \$250 per month on transportation. If he budgets \$100 each month for clothing, \$200 for restaurants \$250 for everything else, how long will it take him to accumulate savings of \$600.
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-To specific Exclude-Other Comments Image: The national sales tax percentage rate is 6%. b. The federal government will deduct it from your paycheck. c. You don't have to pay the tax if your income is very low. d. Imakes things more expensive for you to buy. e. Don't know f. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? include Exclude-Age/life bias Exclude-Coher Exclude-Other include	a. b. c. d. e. f.	months months month. months. on't know efer not to answer
 Include Exclude-Age/life bias Exclude-To specific Exclude-Other Comments Which of the following is true about sales taxes? a. The national sales tax percentage rate is 6%. b. The federal government will deduct if from your paycheck. c. You don't have to pay the tax if your income is very low. d. It makes things more expensive for you to buy. e. Don't know The refar to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Judgment or belief bias Exclude-Judgment or belief bias Exclude-Judgment or belief bias Exclude-Judgment or belief bias Exclude-Cother Exclude-To vague Exclude-Cother Exclude-Cother Exclude-Cother Exclude-To vague Exclude-Cother Exclude-To vague 	Upo que	reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of tions for the final basic financial literacy assessment test?
 Exclude-Age/life bias Exclude-Too specific Exclude-To vague Exclude-Other Comments Which of the following is true about sales taxes? a. The national sales tax percentage rate is 6%. b. The foderal government will deduct if from your paycheck. c. You don't have to pay the tax if your income is very low. d. It makes things more expensive for you to buy. e. Don't know f. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Judgment or belief bias Exclude-To vague Exclude-To vague Exclude-To vague Exclude-To vague Exclude-To vague Exclude-Other 	\bigcirc	nclude
 Exclude-Judgment or belief bias Exclude-Too specific Exclude-Other Comments Comments Which of the following is true about sales taxes? a. The national sales tax percentage rate is 6%. b. The federal government will deduct it from your paycheck. c. You don't have to pay the tax if your income is very low. d. It makes things more expensive for you to buy. e. Don't know f. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Judgment or belief bias Exclude-To vague Exclude-To vague Exclude-Other Comments 	\bigcirc	Exclude-Age/life bias
 Exclude-Too specific Exclude-Other Comments Which of the following is true about sales taxes? a. The national sales tax percentage rate is 6%. b. The federal government will deduct it from your paycheck. c. You don't have to pay the tax if your income is very low. d. It makes things more expensive for you to buy. e. Don't know f. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments 	\bigcirc	Exclude-Judgment or belief bias
 Exclude-To vague Exclude-Other Comments Which of the following is true about sales taxes? a. The national sales tax percentage rate is 6%. b. The federal government will deduct it from your paycheck. c. You don't have to pay the tax if your income is very low. d. It makes things more expensive for you to buy. e. Don't know f. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-To specific Exclude-To vague Exclude-Other Comments 	\bigcirc	Exclude-Too specific
Exclude-Other Comments Which of the following is true about sales taxes? a. The national sales tax percentage rate is 6%. b. The federal government will deduct it from your paycheck. c. You don't have to pay the tax if your income is very low. d. It makes things more expensive for you to buy. e. Don't know f. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Too specific Exclude-To vague Exclude-Other Comments	\bigcirc	Exclude-To vague
Comments Which of the following is true about sales taxes? a. The national sales tax percentage rate is 6%. b. The federal government will deduct it from your paycheck. c. You don't have to pay the tax if your income is very low. d. It makes things more expensive for you to buy. e. Don't know f. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Comments	\bigcirc	Exclude-Other
 Which of the following is true about sales taxes? a. The national sales tax percentage rate is 6%. b. The federal government will deduct it from your paycheck. c. You don't have to pay the tax if your income is very low. d. It makes things more expensive for you to buy. e. Don't know f. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments 	\bigcirc	Comments
Which of the following is true about sales taxes? a. The national sales tax percentage rate is 6%. b. The federal government will deduct it from your paycheck. c. You don't have to pay the tax if your income is very low. d. It makes things more expensive for you to buy. e. Don't know f. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Too specific Exclude-To vague Exclude-Other Comments		
Which of the following is true about sales taxes? a. The national sales tax percentage rate is 6%. b. The federal government will deduct it from your paycheck. c. You don't have to pay the tax if your income is very low. d. It makes things more expensive for you to buy. e. Don't know f. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Too specific Exclude-To vague Exclude-To vague Exclude-Other Comments		· · · · · · · · · · · · · · · · · · ·
 b. The federal government will deduct it from your paycheck. c. You don't have to pay the tax if your income is very low. d. It makes things more expensive for you to buy. e. Don't know f. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments 	Wh a.	h of the following is true about sales taxes? he national sales tax percentage rate is 6%.
d. It makes things more expensive for you to buy. e. Don't know f. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Too specific Exclude-To vague Exclude-Other Comments	b. с.	he federal government will deduct it from your paycheck. ou don't have to pay the tax if your income is very low.
 born know <i>Prefer not to answer</i> Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments 	d.	makes things more expensive for you to buy.
Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments	f. 1	efer not to answer
 Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments 	Upo que	reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of tions for the final basic financial literacy assessment test?
 Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments 		nclude
 Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments 		Exclude-Age/life bias
 Exclude-Too specific Exclude-To vague Exclude-Other Comments 		Exclude-Judgment or belief bias
 Exclude-To vague Exclude-Other Comments 		Exclude-Too specific
Comments		Exclude-To vague
Comments		Exclude-Other
		Comments
Initial Panel Round 2, Group 5 Online Survey Online Survey (continued)

 a. Locked in her closet at home. b. Stocks. c. Corporate bonds. d. A bank savings account. Don't know f. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-Too specific Exclude-Other Comments <i>x</i> Retirement income paid by a company is called: a. 401(k). b. Pension. c. Rents and profits. d. Social security. e. Don't know Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?	of
 b. Stocks. c. Corporate bonds. d. A bank savings account. e. Don't know f. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool or questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-To vague Exclude-To vague Exclude-Other Comments Comments <i>Retirement income paid by a company is called</i>: a. 401(k). b. Persion. Scoial security. Don't know Trefer not to answer 	of
A bark savings account A bark savings account Don't know Prefer not to answer Joon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool or questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-Too specific Exclude-Other Comments Comments Prefer not A park saving A park saving A park saving A park saving account. A park saving account is called: A park saving account of the final basic financial literacy assessment test? Include Exclude-Other Comments Comments A park saving account of the final basic financial literacy assessment test? Don't know A park saving account of the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Other Don't know Comments Don't know Don't know Don't know Exclude-Other Include	of
a. Don't know Yerfer not to answer Joon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-Too specific Exclude-Other Comments Comments Retirement income paid by a company is called: A 401(k). Persion. Retris and profits. Don't know Prefer not to answer Comments Don reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of the final basic financial literacy assessment test? Include Exclude-Other Comments Comments Comments Don't know Prefer not to answer Comments Commen	of
 Prefer not to answer Jpon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments Comments A 401(k). Pension. Retirement income paid by a company is called: A 401(k). Pension. Retrem to to answer Jpon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? 	of
Jpon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of uestions for the final basic financial literacy assessment test? Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments Retirement income paid by a company is called: Add(k). Pension. Retis and profits. Social security. Don't know Prefer not to answer Jpon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of uestions for the final basic financial literacy assessment test?	of
 Include Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments Comments Comments Retirement income paid by a company is called: a. 401(k). Pension. Serial security. Don't know Prefer not to answer Joon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool or puestions for the final basic financial literacy assessment test? Include Exclude-Age/life bias 	
 Exclude-Age/life bias Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments Comments Retirement income paid by a company is called: 401(k). Pension. Rents and profits. Social security. Don't know Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool or puestions for the final basic financial literacy assessment test? Include Exclude-Age/life bias 	
 Exclude-Judgment or belief bias Exclude-Too specific Exclude-To vague Exclude-Other Comments Comments Retirement income paid by a company is called: 401(k). Pension. Rents and profits. Social security. Don't know Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool or puestions for the final basic financial literacy assessment test? Include Exclude-Age/life bias 	
 Exclude-Too specific Exclude-To vague Exclude-Other Comments Comments Retirement income paid by a company is called: 401(k). Pension. Rents and profits. Social security. Don't know Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias	
 Exclude-To vague Exclude-Other Comments Comments Retirement income paid by a company is called: 401(k). Pension. Rents and profits. Social security. Don't know Prefer not to answer Jpon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias 	
Exclude-Other Comments Comments Retirement income paid by a company is called: A 401(k). Pension. Rets and profits. Social security. Don't know Prefer not to answer Joon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias	
Comments Retirement income paid by a company is called: 4. 401(k). 5. Pension. 7. Rents and profits. 7. Social security. 8. Don't know 7. Prefer not to answer Joon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias	
Comments Retirement income paid by a company is called: A 401(k), Pension. Rents and profits. Social security. Don't know Prefer not to answer Joon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias	
Retirement income paid by a company is called: a. 401(k). b. Pension. c. Rents and profits. d. Social security. e. Don't know f. Prefer not to answer Joon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias	
Retirement income paid by a company is called: a. 401(k). b. Pension. c. Rents and profits. d. Social security. e. Don't know f. Prefer not to answer Joon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias	
Retirement income paid by a company is called: a. 401(k). b. Pension. c. Rents and profits. d. Social security. a. Don't know c. Prefer not to answer Joon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? Include Exclude-Age/life bias	
Exclude-Age/life bias	of
Exclude-Age/life bias	
Exclude-Judgment or belief bias	
Exclude-Too specific	
Exclude-To vague	
Exclude-Other	
Comments	
h	
Dan must borrow \$12,000 to complete his college education. Which of the following would NOT be likely to reduce the inance charge rate?	
a If he want to a state college rather than a private college	
b. If his parents costaned the loan.	

d. If the loan was insured by the Federal Government.

Initial Panel Round 2, Group 5 Online Survey Online Survey (continued)

e. Don't know f. Prefer not to answer	
Upon reviewing the question a questions for the final basic fin	bove, do you feel it should be included in or excluded (identifying reason) from the pool of ancial literacy assessment test?
Include	
Exclude-Age/life bias	
Exclude-Judgment or belie	fbias
Exclude-Too specific	
Exclude-To vague	
Exclude-Other	
Comments	
If you are behind on your debu Counseling Services, what he	payments and go to a responsible credit counseling service such as the Consumer Credit Ip can they give you? p all of your credit cards without your permission.
b. They can get the federal g c. They can work with thos d. They can force those who e. Don't know f. Prefer not to answer	overnment to apply your income taxes to pay off your debts. e who loaned you money to set up a payment schedule that you can meet. loaned you money to forgive all your debts.
Upon reviewing the question a questions for the final basic fin	bove, do you feel it should be included in or excluded (identifying reason) from the pool of ancial literacy assessment test?
Include	
Exclude-Age/life bias	
Exclude-Judgment or belie	fbias
Exclude-Too specific	
Exclude-To vague	
Exclude-Other	
Comments	
1.	
Which of the following credit of all charge the same amount p	ard users is likely to pay the GREATEST dollar amount in finance charges per year, if they er year on their cards?
 a. Jessica, who pays at least b. Vera, who generally pays c. Megan, who always pays d. Erin, who pays the minir 	the minimum amount each month and more, when she has the money. off her credit card in full but, occasionally, will pay the minimum when she is short of cash. off her credit card bill in full shortly after she receives it. num amount each month.
e. Don't know	

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

Initial Panel Round 2, Group 5 Online Survey Online Survey (continued)

	Include
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
	Comments
	1.
lf yc a. C b. l	ou have caused an accident, which type of automobile insurance would cover damage to your own car? Comprehensive. Liability.
d.	Collision.
e. l f. l	Don't know Prefer not to answer
Upo for ti	n reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions ne final basic financial literacy assessment test? Include Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
	Comments
Mar a. A b. 7 c. A d. 7 e. 1 f. 1	ny savings programs are protected by the Federal government against loss. Which of the following is not? A. U. S. Savings Bond. A certificate of deposit at the bank. A bond issued by one of the 50 States. A. U. S. Treasury Bond. Don't know Prefer not to answer
Upo que:	n reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of stions for the final basic financial literacy assessment test?
	Include
	Exclude-Age/life bias

Exclude-Judgment or belief bias

Initial Panel Round 2, Group 5 Online Survey Online Survey (continued)

	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
	Comments
	.h.
Bar you	nk A offers monthly compounding and Bank B offers yearly compounding. Both banks offer the same interest rate. In Ir opinion, which bank would you choose if you wanted a higher return?
a. E	Bank A.
b. 1	Bank B.
c. d.	Both banks offer the same return Don't know
е.	Prefer not to answer
Upc que	on reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of stions for the final basic financial literacy assessment test? Include
	Exclude-Age/life bias
	Exclude-Judgment or belief bias
	Exclude-Too specific
	Exclude-To vague
	Exclude-Other
	Comments
	Comments
	Comments //

Appendix I

Validation Panel Communications

Validation Panel Round 1 Email

Da Su To	te: Fri, Jul 31, 2015 at 8:43 PM bject: Financial Literacy Expert Panel - Validation Round 1 : <panel member=""></panel>
H	ello <name>,</name>
Tł	ank you, again, for agreeing to serve as a member of the Expert Panel for my dissertation research.
Tł	is email is to provide you with instructions and a link to the online review survey.
Th ins lit	e results of your involvement will contribute to the development of the basic financial literacy assessment strument, which will be used in my research. The purpose of my research study is to identify the financial eracy continuing professional education needs of Florida small business owners.
Tł ap op	ere are 40 items in the Round 1 survey. You are asked to read each question and evaluated the propriateness of language and wording, as well as the difficulty of each question. You will also be given an portunity to provide comments if you so desire.
Pl be ne	ease complete this first round promptly as there are several more rounds of review before the assessment can finalized. The estimated time to complete this review is less than 25 minutes. If you have any questions or ed assistance, please feel free to contact me at pjdahmen@mail.usf.com or 941-962-1211.
To br	start the survey, please click on this link, or copy or paste it to your owser, <u>http://usf.qualtrics.com/SE/?SID=SV_06tsn3RgfDf7YIB</u> . Then, follow the instructions are below.
N(or	DTE: This survey is mobile friendly and can be accessed with a smartphone or tablet as well as a computer laptop.
IN	STRUCTIONS
Tł yo	e following instructions will be on the opening screen of the survey. Please feel free to print this email so u may refer to them as you complete the survey.
Ba lit su plo ad	used on your knowledge and experience, you will be evaluating 32 questions in the pool for the basic financial eracy assessment instrument and eight self-assessed financial literacy questions. The goal of this round is to bstantially reduce the number of questions in the pool. You are only reviewing 50% of the pool, therefore, ease feel free to critical and note questions for exclusion. The final instrument will be administered to U.S. ults age 18 and over using an online survey.
In 1. 2. wo 3.	this round, you are asked to read each question and possible answers (correct answer is bold) to evaluate if it has appropriate for language (whether the question is simple, clear and easy to understand), the wording is adequate (whether the question contains wording that is relevant and free from bias, in other ords, it portrays a neutral viewpoint), and rate the question difficulty based on the descriptions below.

Validation Panel Round 1 Email (continued)

Please evaluate each question on the following scale: Very Unclear Unclear Neutral Clear Very Clear 3 4 Language 2 5 Needs Poor Neutral Good Excellent Improvement Wording 2 3 4 5 Too Easy Easy Average Difficult Too Difficult Question Difficulty 1 2 3 4 5 Difficulty ratings should be based on the following descriptions: Rating Description Common Knowledge, most adults would know this. Very Easy Adult would have acquired this financial literacy knowledge in the course normal K-12 Easy education. Average, requires some understanding of financial literacy and learning over and Neutral above normal K-12 concepts. Requires a level of education specifically directed at financial literacy concepts learned Difficult through formal or informal study. Very Advance financial literacy concepts learned through formal financial courses in higher Difficult learning programs (i.e. financing or investing courses). Please feel free to comment on any of the questions as you move through this round. Also, note any question that you feel should be excluded or require changes in wording to assess more affectively. Following the basic financial literacy assessment questions, are eight questions for self-assessing financial literacy. Please read each question and determine whether you believe the question should be included or excluded from a self-assessment financial literacy instrument. If you feel it should be excluded, please comment. Best regards, Pearl Dahmen Para University of South Florida, PhD Candidate

Validation Panel Round 2 Email

From: **Pearl Para** <<u>pjdahmen@mail.usf.edu</u>> Date: Tue, Aug 11, 2015 at 7:09 AM Subject: Fwd: Financial Literacy Expert Panel - Validation Round 2 To: <u><panelmember.address></u>

Hello <Name>

Thank you for completing expert panel round 1 and participating in this valuable financial literacy research. In this second round of review, you will be evaluating 32 financial literacy knowledge questions in the pool for the basic financial literacy instrument and seven financial literacy self-assessment questions.

The estimated time to complete the review is 30 to 45 minutes. Please complete your review as soon as possible. Your input will determine the questions included in future rounds of review by other members of the expert panel. For your convenience, survey allows you to stop at any point. It will automatically save and later return to the place you stopped. If you have any questions or need assistance, please feel free to contact me at pjdahmen@mail.usf.com or 941-962-1211.

To start the survey, please click on this link, or copy and paste it to your browser: <u>http://usf.qualtrics.com/SE/?SID=SV_cGP1idQLl2URLGI</u> Then, follow the instructions below.

NOTE: This survey is mobile friendly and can be accessed with a smart phone or tablet as well as a computer or laptop.

The following instructions will be on the opening screen of the survey. Please feel free to print this email so you may refer to them as you complete the survey.

INSTRUCTIONS

As you read each financial literacy question and possible answers (correct answer is bold), identify whether you believe the question should be included or excluded from an instrument used to determine a basic financial literacy score. The goal of this round is to reduce the number of questions in the pool. You are reviewing 50% of the pool, therefore, please feel free to be critical in your selection of the questions to include. Only included question you strongly feel should be included; all other questions should be excluded. If you determine the question should be included, please identify the financial literacy content category or categories you feel it addresses. The categories for content areas are:

Financial Literacy Category	Topics Included in Category
Money basics	Compound interest, Time value of money, purchasing power/inflation, spending, short-term savings
Borrowing	APR, finance charges, loan terms, credit scores; type of borrowing (e.g., education, home, vehicles)
Building assets	Long-term-savings accounts/Certificates of deposit, stocks, bonds, mutual funds, annuities
Protecting assets	Investment product risk (stocks vs. bonds vs. mutual funds), investment diversity, insurance protection

Validation Panel Round 2 Email (continued)

Following the financial literacy knowledge questions are seven financial literacy self-assessment questions. Please read each question and evaluate if:

1. it has appropriate for language (whether the question is simple, clear and easy to understand),

2. the wording is adequate (whether the question contains wording that is relevant and free from bias, in other words, it portrays a neutral viewpoint). Evaluation is on the following scale:

	Very Unclear	Unclear	Neutral	Clear	Very Clear
Language	1	2	3	4	5
	Poor	Needs Improvement	Neutral	Good	Excellent
Wording	1	2	3	4	5

Please feel free to comment on any of the question included in this round.

Best regards,

Pearl Dahmen Para University of South Florida, PhD Candidate

Appendix J

Verification Panel Communication

From: **Pearl Para** <<u>pjdahmen@mail.usf.edu</u>> Date: Tue, Sep 15, 2015 at 7:23 PM Subject: Financial Literacy Expert Panel - Verification Round To: <u><panelmember.address></u>

Greetings,

My dissertation research has reached the final round of expert panel review. Thank you for agreeing to serve as a member of the expert panel. Hopefully, the timing of this panel will work with your schedule. If not, please let me know.

Your input will determine the final assessment instrument. In the verification round of review, you will be evaluating 26 knowledge questions for the basic financial literacy assessment instrument and five self-assessed financial literacy questions through an online survey. The final instrument will be administered to U.S. adults age 18 and over using an online survey.

The estimated time to complete the review is approximately 30 minute. Please complete your review as soon as possible. The survey allows for you to stop and come back, so please feel free to break when necessary. You will be able to save the survey and return where you left off. If you have any questions or need assistance, please feel free to contact me at <u>pjdahmen@mail.usf.com</u>or <u>941-962-</u><u>1211</u>.

To start the survey, please click on this link, or copy and paste it to your browser: <u>http://usf.qualtrics.com/SE/?SID=SV_4Hslwdg234VbnPD</u>

Please note that this survey is mobile friendly and can be access with a smartphone or tablet as well as a computer or laptop.

INSTRUCTIONS

The following instructions will be on the opening screen of the survey. Please feel free to print this email so you may refer to them as you complete the survey.

Please read each knowledge questions and possible answers (correct answer is bold). Then, respond to the following statements:

- This question should be included in a basic financial literacy assessment instrument. (Agree or disagree)
- The language used in this question is . . . (from very unclear to very clear, evaluate whether the question is simple, clear and easy to understand),
- The wording of this question is . . . (poor to excellent, evaluate whether the question contains wording that is relevant and free from bias, in other words, it portrays a neutral viewpoint)
- The difficulty rating of this question is . . . (Difficulty definitions are below)
- The financial literacy content category (categories) is (are) appropriate. (Definitions of categories are below.)

Varification F	Danal Co	ommunica	tion (continued	١
VEIIICAUOII F		JIIIIIuIIICa		continueu	J

	1			1	1 1
	1	2	3	4	5
Include question	Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree
Language	Very Unclear	Unclear	Neutral	Clear	Very Clear
Wording	Poor	Needs Improvement	Neutral	Good	Excellent
Difficulty Rating	Too Easy	Easy	Average	Difficult	Too Difficult
Content Category	Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree
Difficulty ratings are Rating	Description				
Very Easy	Common Knowled	lge, most adult	s would kn	ow this.	
Easy	Adult would have acquired this financial literacy knowledge in the course normal K-12 education.				
Neutral	Average, requires	some understa	anding of fi	nancial lite	racy and learning
Neutral Difficult	Average, requires beyond K-12. Requires a level of concepts.	some understa f education spe	anding of fi	nancial lite ected at fir	racy and learning nancial literacy
Neutral Difficult Very Difficult	Average, requires beyond K-12. Requires a level of concepts. Advance financial courses (i.e. financial	some understa f education spe literacy concep cing or investin	anding of fine ecifically dir ots learned og courses).	nancial lite ected at fir through fo	racy and learning nancial literacy ocused financial
Neutral Difficult Very Difficult The categories for c	Average, requires beyond K-12. Requires a level of concepts. Advance financial courses (i.e. financial ontent areas are:	some understa f education spe literacy concep cing or investin	anding of fine ecifically dir ots learned og courses).	nancial lite ected at fir through fo	racy and learning nancial literacy ocused financial
Neutral Difficult Very Difficult <u>The categories for c</u> Financial Literacy Category	Average, requires beyond K-12. Requires a level of concepts. Advance financial courses (i.e. financial ontent areas are: Topics Included in	some understa f education spe literacy concep cing or investin Category	anding of fine ecifically direction of the second s	nancial lite ected at fir through fo	racy and learning nancial literacy ocused financial
Neutral Difficult Very Difficult <u>The categories for c</u> Financial Literacy <u>Category</u> Money basics	Average, requires beyond K-12. Requires a level of concepts. Advance financial courses (i.e. financial courses (i.e. financial content areas are: Topics Included in Compound interes power/inflation, s	some understa f education spe literacy concep cing or investin Category st, Time value c pending, short-	ecifically dir ecifically dir ots learned og courses). of money, p	nancial lite ected at fir through fo urchasing gs	racy and learning nancial literacy ocused financial
Neutral Difficult Very Difficult <u>The categories for c</u> Financial Literacy <u>Category</u> Money basics Borrowing	Average, requires beyond K-12. Requires a level of concepts. Advance financial courses (i.e. financial courses (i.e. financial courses (i.e. financial courses (i.e. financial courses (i.e. financial content areas are: Topics Included in Compound interes power/inflation, s APR, finance charge borrowing (e.g., et	some understa f education spe literacy concep cing or investin Category st, Time value c pending, short- ges, loan terms ducation, home	anding of fine ecifically dir ots learned og courses). of money, p term savin , credit sco e, vehicles)	nancial lite ected at fir through fo urchasing gs res; type of	racy and learning nancial literacy ocused financial

Verification Panel Communication (continued)

Protecting assets Investment product risk (stocks vs. bonds vs. mutual funds), investment diversity, insurance protection
Please feel free to comment in the areas provided. When rating 1 or 2, providing reasons and/or suggestion for possible improvement are welcomed.
Following the knowledge financial literacy assessment questions, there are five questions for self-assessing financial literacy. Please read each knowledge questions and possible answers. Then, respond to the following statements:
This question should be included in a basic financial literacy assessment instrument. (Agree or disagree)
The language used in this question is ... (from very unclear to very clear, evaluate whether the question is simple, clear and easy to understand),
The wording of this question is ... (poor to excellent, evaluate whether the question contains wording that is relevant and free from bias, in other words, it portrays a neutral viewpoint)
Please evaluate each question on the following scale:

	1	2	3	4	5
Include question	Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree
Language	Very Unclear	Unclear	Neutral	Clear	Very Clear
Wording	Poor	Needs	Neutral	Good	Excellent
_		Improvement			

Please feel free to comment on any of the question as you move through this round.

Best regards,

Pearl Dahmen Para University of South Florida, PhD Candidate

Appendix K

Validation Panel Round 1 Online Survey

Validation Panel Round 1, Group 1 Online Survey

Dear Expert Panel member,

Thank you for participating in this valuable financial literacy research. The results of this survey will contribute to the development of the basic financial literacy assessment instrument that will be used to assess the participants of this research. Because of your background and expertise, your participation and input is invaluable.

In Round 1 of the Validation Panel review, you will be evaluating 32 questions in the pool for the basic financial literacy assessment instrument and eight self-assessed financial literacy questions. The goal of this round is to substantially reduce the number of questions in the pool. You are only reviewing 50% of the pool, therefore, please feel free to critical and note questions for exclusion. The final instrument will be administered to U.S. adults age 18 and over using an online survey.

In this round, you are asked to read each question and possible answers (correct answer is bold) to evaluate if

1. it has appropriate for language (whether the question is simple, clear and easy to understand),

2. the wording is adequate (whether the question contains wording that is relevant and free from bias, in other words, it portrays a neutral viewpoint), and

3. rate the question difficulty based on the descriptions below.

Please evaluate each question on the following scale:

	Very Unclear	Unclear	Neutral	Clear	Very Clear
Language	1	2	3	4	5
	Poor	Needs Improvement	Neutral	Good	Excellent
Wording	1	2	3	4	5
	Too Easy	Easy	Average	Difficult	Too Difficult
Question Difficulty	1	2	3	4	5

Difficulty ratings should be based on the following descriptions:

Rating	Description
Very Easy	Common Knowledge, most adults would know this.
Easy	Adult would have acquired this financial literacy knowledge in the course normal K-12 education.
Neutral	Average, requires some understanding of financial literacy and learning over and above normal K-12 concepts.
Difficult	Requires a level of education specifically directed at financial literacy concepts learned through formal or informal study.
Very Difficult	Advance financial literacy concepts learned through formal financial courses in higher learning programs (i.e. financing or investing courses).

Please feel free to comment on any of the questions as you move through this round. Also, note any question that you feel should be excluded or require changes in wording to assess more affectively.

Following the basic financial literacy assessment questions, are eight questions for self-assessing financial literacy. Please read each question and determine whether you believe the question should be included or excluded from a self-assessment financial

Validation Panel Round 1, Group 1 Online Survey (continued)

literacy instrument. If you feel it should be excluded,	please comment.	
inancial Literacy Questions		
Financial Literacy Assessment Question	IS:	
Suppose you had \$100 in a source account and	the internet rate was 2%	an usar Affar Fusara haw much da yau thisk
you would have in the account if you left the mor	ney to grow?	ber year. After 5 years now much do you think
a. More than \$102 b. Exactly \$102 c. Less than \$102 d. Don't know e. Prefer not to answer		
Upon reviewing the question above, do you feel in questions for the final basic financial literacy ass	t should be included in or e sessment test?	excluded (identifying reason) from the pool of
	1 2 Neutral 4 5	
Language: Very Unclear		Very Clear
Wording: Poor	0 0 0 0 0	Excellent
Difficulty of Question: Too Easy	$ \circ \circ \circ \circ \circ $	Too Difficult
Comments:		
		h
Imagine that the interest rate on your savings ac you be able to buy:	count was 1% per year and	d inflation was 2% per year. After 1 year, would
a. More than today b. Exactly the same c. Less than today d. Don't know e. Prefer not answer		
Upon reviewing the question above, do you feel in	t should be included in or e	excluded (identifying reason) from the pool of
	1 2 Neutral 4 E	
Language: Very Unclear		very clear
Woraing: Poor		
Difficulty of Question: 100 Easy		100 Dillicuit

Validation Panel Round 1, Group 1 Online Survey (continued)

igh inflation mean	s that the cost of living is inc	reasin	g rap	idly.			
. True							
Don't know							
Prefer not to ans	wer						
pon reviewing the uestions for the fi	question above, do you feel nal basic financial literacy asses	t shou sment	uld be t test'	inclu?	ided i	n or e	excluded (identifying reason) from the pool of
		1	2 1	leutra	4	5	
	Language: Very Unclear	\circ	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Very Clear
	Wording: Poor	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Excellent
D	ifficulty of Question: Too Easy	\odot	\odot	\bigcirc	\bigcirc	\bigcirc	Too Difficult
is usually possib	le to reduce the risk of invest	ing in	the s	tock i	narke	ət by	buying a wide range of stocks and shares.
True							
False Don't know							
Prefer not to answ	er						
pon reviewing the uestions for the fir	question above, do you feel nal basic financial literacy asses	t shou sment	uld be t test?	inclu	ded i	n or e	excluded (identifying reason) from the pool of
		1	2 1	leutra	4	5	
			\bigcirc	\bigcirc	\bigcirc	\bigcirc	Very Clear
	Language: Very Unclear						very clear
	Language: Very Unclear Wording: Poor	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Excellent
C	Language: Very Unclear Wording: Poor ifficulty of Question: Too Easy	0	0	0	0 0	0	Excellent Too Difficult

Validation Panel Round 1, Group 1 Online Survey (continued)

The figure that best reflects a mutual fund's performance over a period of years is

a. its current yield.

- b. the total of dividends and capital gains it has paid.
- c. its total return.
- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

	1	2 1	Veutra	al 4	5	
Language: Very Unclear		\bigcirc	\bigcirc	\bigcirc	\odot	Very Clear
Wording: Poor	0	\bigcirc	\bigcirc	\bigcirc	\odot	Excellent
Difficulty of Question: Too Easy	0	\bigcirc	0	0	0	Too Difficult

Comments:

If your assets increase by \$5,000 and your liabilities decrease by \$3,000, your net worth would

- a. Increase by \$2,000
- b. Increase by \$8,000
- c. Increase by \$3,000
- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

	1	2 N	leutra	al 4	5	
Language: Very Unclear	\circ	\bigcirc	\bigcirc	\bigcirc	\odot	Very Clear
Wording: Poor	\circ	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Excellent
Difficulty of Question: Too Easy	\circ	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Too Difficult

Comments:

Validation Panel Round 1, Group 1 Online Survey (continued)

To ensure that some of your retirement savings will not be subject to income tax upon withdrawal, you would contribute to:

a. A Traditional IRS or Individual Retirment Account

- b. A Roth IRA
- c. A 401(k) plan
- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

	1	2 1	Neutra	al 4	5	
Language: Very Unclear	0	\bigcirc	\bigcirc	\bigcirc	\odot	Very Clear
Wording: Poor	0	\bigcirc	\bigcirc	\bigcirc	0	Excellent
Difficulty of Question: Too Easy	0	0	0	0	0	Too Difficult

Comments:

To reduce the total finance costs paid over the life of an auto loan, you should choose a loan with the

- a. Lowest monthly payment
- b. Longest repayment term
- c. Shortest repayment
- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

	1	2	Neutra	al 3	5	
Language: Very Unclear	\bigcirc	C	\circ	\bigcirc	\bigcirc	Very Clear
Wording: Poor	\bigcirc	C	\bigcirc	\bigcirc	\bigcirc	Excellent
Difficulty of Question: Too Easy	\bigcirc	C	\bigcirc	\bigcirc	\bigcirc	Too Difficult

Comments:

Validation Panel Round 1, Group 1 Online Survey (continued)

If you always pay the full balance on your credit card, which of the following is least important?

a. Annual interest rate b. Annual fee c. Line of credit d. Don't know e. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? 1 2 Neutral 4 5 Language: Very Unclear Very Clear 0 0 0 0 0 Wording: Poor 0 0 \bigcirc Excellent 0 0 Difficulty of Question: Too Easy Too Difficult 0 0 0 0 0 Comments: Which type of mortgage would allow a first-time homebuyer to qualify for the highest loan amount? a. Fixed-rate mortgage b. Adjustable-rate mortgage c. Reverse mortgage d. Don't know e. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? 2 Neutral 4 5 1 Language: Very Unclear Very Clear \bigcirc \bigcirc \bigcirc \bigcirc Wording: Poor \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc Excellent Difficulty of Question: Too Easy \bigcirc \bigcirc \bigcirc \bigcirc Too Difficult Comments:

Validation Panel Round 1, Group 1 Online Survey (continued)

If you signed a twelve-month lease for \$300 month but never occupied the apartment, you legally owe the landlord

- a. your security deposit.
- b. your first month's rent of \$300.
- c. your twelve month's rent of \$3.600.
- d. nothing.
- e whatever the landlord requires.
- f. Don't know
- g. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

	1	2	Neutra	al 4	5	
Language: Very Unclear		0	0	0	\bigcirc	Very Clear
Wording: Poor	0	0	0	0	\odot	Excellent
Difficulty of Question: Too Easy	\odot	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Too Difficult
Comments:						
The most liquid accet is						
The most iquid asset is						
 a. money in a certificate of deposit. b. money in a checking account. 						
c. a car.						
e a house (real estate).						
f. Don't know a. Prefer not to answer						
 Upon reviewing the question above, do you feel it 	it sho	uld b	e incli	Ided	in or 4	excluded (identifying reason) from the pool of
questions for the final basic financial literacy asses	ssmen	nt tesť	?	ucu		excluded (identifying reason) norm the poor of
	1	2	Neutra	al 4	5	

Continued on the next page

 $| \circ \circ \circ \circ \circ$

Wording: Poor

Difficulty of Question: Too Easy

Excellent

Too Difficult

Validation Panel Round 1, Group 1 Online Survey (continued)

hich of the following investments requires that	vou k	eep	vour r	none	v inve	ested for a specified period or face an early
thdrawal penalty?						,
Certificate of deposit. Checking account that pays interest. Government savings bond. Money market mutual fund. Passbook savings account. Don't know Prefer not to answer						
on reviewing the question above do you feel it	t sho	uld b	e inclu	ided	in or 4	excluded (identifying reason) from the pool of
estions for the final basic financial literacy assess	smen	t tesť	?	lueu		excluded (identifying reason) from the poor of
	1	2	Neutra	al 4	5	
Language: Very Unclear		0	\bigcirc	\bigcirc		Very Clear
Wording: Poor			0	0	0	Excellent
Difficulty of Question: Too Easy	0	0	\bigcirc	0	0	Too Difficult
mments:						
mments:						
mments:						
mments:						
mments:						
mments:						
mments: th compound interest, you earn interest on you	ur inte	ərest,	as w	ell as	on y	our principal.
mments: ith compound interest, you earn interest on you True	ur inte	ərest,	as w	ell as	on y	our principal.
mments: ith compound interest, you earn interest on you True False Don't know Prefer not to answer	ur inte	erest,	as w	ell as	on y	our principal.
mments: ith compound interest, you earn interest on you True False Don't know Prefer not to answer yon reviewing the question above, do you feel it estions for the final basic financial literacy assess	t shoi	uld b	as w	ell as	on y	<i>our principal.</i> excluded (identifying reason) from the pool o
mments: ith compound interest, you earn interest on you True False Don't know Prefer not to answer pon reviewing the question above, do you feel it estions for the final basic financial literacy assess	ur inte t shou smen 1	uld b t test	as we	ell as uded	on y in or o	<i>our principal.</i> excluded (identifying reason) from the pool o
mments: ith compound interest, you earn interest on you True False Don't know Prefer not to answer yon reviewing the question above, do you feel it estions for the final basic financial literacy assess Language: Very Unclear	t shou smen 1	uld b t test ²	as we e inclu ?	ellas uded 113	on y	our principal. excluded (identifying reason) from the pool o
ith compound interest, you earn interest on you True False Don't know Prefer not to answer roon reviewing the question above, do you feel it estions for the final basic financial literacy assess Language: Very Unclear Wording: Poor	t shot smen 1	uld b t test	as we	ellas uded nl 3	on y	<i>our principal.</i> excluded (identifying reason) from the pool of Very Clear Excellent

Validation Panel Round 1, Group 1 Online Survey (continued)

Nutual funds pay a guaranteed rate of return.						
True						
False						
Prefer not to answer						
pon reviewing the question above, do you feel i	t shou	uld be	e inclu	ded i	in or e	excluded (identifying reason) from the pool of
uestions for the final basic financial literacy asses	smen	t test	2			
	1	2	Neutra	4	5	1
Language: Very Unclear	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Very Clear
Wording: Poor	0	\bigcirc	0	\bigcirc	0	Excellent
Difficulty of Question: Too Easy	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Too Difficult
Repeatedly refinancing your home mortgage ove our debt. . True . False . Don't know . Prefer not to answer	r a sh	ort p	əriod c	of tim	ne res	ults in added fee and points that further incre
Repeatedly refinancing your home mortgage over our debt. . True . False . Don't know . Prefer not to answer lpon reviewing the question above, do you feel i uestions for the final basic financial literacy asses	r a sh t shou smen	ort p	eriod c	of tim	in or e	sults in added fee and points that further incre excluded (identifying reason) from the pool of
Pepeatedly refinancing your home mortgage over our debt. False Don't know Prefer not to answer pon reviewing the question above, do you feel i uestions for the final basic financial literacy asses	r a sh t shou smen 1	ort po uld be t test	eriod c e inclu Neutra	of tim	ne res	sults in added fee and points that further incre excluded (identifying reason) from the pool of
Repeatedly refinancing your home mortgage over our debt. . True . False . Don't know . Prefer not to answer loon reviewing the question above, do you feel i uestions for the final basic financial literacy asses Language: Very Unclear	r a sh t shou smen 1	uld be t test?	eriod c e inclu Neutra	ded i	in or o	cults in added fee and points that further incre excluded (identifying reason) from the pool of Very Clear
Repeatedly refinancing your home mortgage over our debt. False Don't know Prefer not to answer pon reviewing the question above, do you feel i uestions for the final basic financial literacy asses Language: Very Unclear Wording: Poor	t shou smen 1	uld be t test?	eriod c eriod c eriod c	of time ded i	in or e	eults in added fee and points that further incre excluded (identifying reason) from the pool of Very Clear Excellent

Validation Panel Round 1, Group 1 Online Survey (continued)

Using extra money in a bank savings account to pay off high interest rate credit card debt is a good idea.

a. True b. False c. Don't know d. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? 1 2 Neutral 4 5 Language: Very Unclear Very Clear \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc Wording: Poor Excellent \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc Difficulty of Question: Too Easy \bigcirc \bigcirc \bigcirc \bigcirc Too Difficult Comments: Creditors are required to tell you the APR that you will pay when you get a loan. a. True b. False c. Don't know d. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the ffinal basic financial literacy assessment test? 1 2 Neutral 4 5 Language: Very Unclear \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc Very Clear Wording: Poor Excellent \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc Difficulty of Question: Too Easy \bigcirc \bigcirc Too Difficult Comments:

Validation Panel Round 1, Group 1 Online Survey (continued) Your credit rating is not affected by how much you charge on your credit cards. a. True b. False c. Don't know d. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? 1 2 Neutral 4 5 Language: Very Unclear 0 0 Very Clear 0 0 0 Wording: Poor Excellent \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc Too Difficult Difficulty of Question: Too Easy \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc Comments: Your credit rating is not affected by how much you charge on your credit cards. a. True b. False c. Don't know d. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test? 1 2 Neutral 4 5 Language: Very Unclear Very Clear \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc Wording: Poor \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc Excellent Difficulty of Question: Too Easy Too Difficult \bigcirc \bigcirc \bigcirc \bigcirc Comments:

Validation Panel Round 1, Group 1 Online Survey (continued)

Taxes affect how you should invest your money.

- a. True
- b. False

c. Don't know d. Prefer not to answer

ALTERNATIVE WORDING

Taxes do not affect how you should invest your money.

- a. True
- b. False

c. Don't know

d. Prefer not to answer

Upon reviewing the question and its alternative wording format above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

	1	2	Neutra	4	5	
Language: Very Unclear	0	0	\bigcirc	\bigcirc	0	Very Clear
Wording: Poor	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Excellent
Difficulty of Question: Too Easy	0	0	\bigcirc	0	\odot	Too Difficult

Comments:

Which of the following instruments is NOT typically associated with	h spendina?
---	-------------

- a. Debit card.
- b. Certificate of deposit.
- c. Cash.
- d. Credit card.
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

	1	2	Neutra	4	5	
Language: Very Unclear	0	0	\bigcirc	\bigcirc	\bigcirc	Very Clear
Wording: Poor	\odot	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Excellent
Difficulty of Question: Too Easy	0	0	\bigcirc	\bigcirc	\bigcirc	Too Difficult

Validation Panel Round 1, Group 1 Online Survey (continued)

Many people put aside money to take care of un emergencies, in which of the following forms would	expected e Id it be of	expenses. LEAST be	If Jua enefit	an and Elva have money put aside for to them if they needed it right away?
a. Invested in a down payment on the house				
a. Checking account a. Stocks				
I. Savings account				
. Prefer not to answer				
least reviewing the supplier should be used for the		المعادما معا	In a-	evoluted (identifying reason) from the sect of
Jpon reviewing the question above, do you feel i juestions for the final basic financial literacy asses	t should be sment test?	e included	in or	excluded (identifying reason) from the pool of
	121	Neutral 4	5	
Language: Very Unclear	0 0	0 0	\bigcirc	Very Clear
Wording: Poor	0 0	0 0	\bigcirc	Excellent
Difficulty of Question: Too Easy	0 0			Too Difficult
		0 0	\bigcirc	
		0 0	0	
		0 0		
Comments:		0 0		
Comments:		0 0		
Comments:				
comments:		0 0	0	
'omments:			0	
'omments:			0	
'omments:			0	
Comments:			0	
f you had a savings account at a bank, which of	the follow	ing would	be co	prrect concerning the interest that you would e
Comments: f you had a savings account at a bank, which of on this account?	the follow	ing would	be co	prrect concerning the interest that you would e
Comments: f you had a savings account at a bank, which of on this account? a. Eamings from savings account interest may b	the follow	ing would	be co	prrect concerning the interest that you would e
Comments: f you had a savings account at a bank, which of in this account? Earnings from savings account interest may be charged on the interest Seles tax may be charged on the interest that Seles tax may be charged on the interest that Seles tax may be charged on the interest that	the follow	ing would ed. come is	be cc	prrect concerning the interest that you would e enough
Comments: f you had a savings account at a bank, which of on this account? b. Earnings from savings account interest may b c. Income tax may be charged on the interest tha i. Sales tax may be charged on the interest tha i. You cannot earn interest until you pass your	the follow not be taxe t if your in t you eam. 8th birthda	ing would ed. acome is a	be cc	prrect concerning the interest that you would e enough
Comments: f you had a savings account at a bank, which of on this account? a. Earnings from savings account interest may b. Income tax may be charged on the interest c. Sales tax may be charged on the interest that f. You cannot earn interest until you pass your b. Don't know	the follow not be taxe t if your in t you eam. 18th birthda	ing would ed. acome is a	be co	prrect concerning the interest that you would e enough
Comments: f you had a savings account at a bank, which of on this account? a. Earnings from savings account interest may b c. Income tax may be charged on the interest 5. Sales tax may be charged on the interest tha 1. You cannot earn interest until you pass your b. Don't know i. Prefer not to answer	the follow not be taxe t if your in t you eam. 18th birthda	ing would ed. acome is a	be co	prrect concerning the interest that you would e enough
Comments: f you had a savings account at a bank, which of in this account? a. Earnings from savings account interest may b b. Income tax may be charged on the interest c. Sales tax may be charged on the interest that f. You cannot earn interest until you pass your b. Don't know c. Prefer not to answer Upon reviewing the question above, do you feel in upestions for the final basic financial literacy asses	the follow not be taxe t if your in t you eam. 18th birthde t should be sment test?	ing would ad. acome is a ay. e included	be cc high in or	prrect concerning the interest that you would e enough excluded (identifying reason) from the pool of
Comments: f you had a savings account at a bank, which of on this account? a. Earnings from savings account interest may in b. Income tax may be charged on the interest c. Sales tax may be charged on the interest the interest that 1. You cannot earn interest until you pass your 2. Don't know 3. Don't know 4. Prefer not to answer Joon reviewing the question above, do you feel in questions for the final basic financial literacy assesting the save save save save save save save sav	the follown tot be taxe t if you a m. tyou am. 18th birthda t should be sment test? 1 2 1	ing would ed. acome is ay. e included Neutral 4	be co high in or 5	prrect concerning the interest that you would e enough excluded (identifying reason) from the pool of
Comments: f you had a savings account at a bank, which of on this account? a. Earnings from savings account interest may be b. Income tax may be charged on the interest c. Sales tax may be charged on the interest that d. You cannot earn interest until you pass your a. Don't know c. Prefer not to answer Joon reviewing the question above, do you feel in questions for the final basic financial literacy asses Language: Very Unclear	the follown to be taxed tif your in tyou earn. 18th birthdo t should be sment test? 1 2 1	ing would ed. acome is ay. e included Neutral 4	be co high in or 5	prrect concerning the interest that you would en enough excluded (identifying reason) from the pool of Very Clear
Comments: If you had a savings account at a bank, which of on this account? a. Earnings from savings account interest may in b. Income tax may be charged on the interest c. Sales tax may be charged on the interest that d. You cannot earn interest until you pass your a. Don't know c. Prefer not to answer Joon reviewing the question above, do you feel in questions for the final basic financial literacy asses Language: Very Unclear Wording: Poor	the follow to be taxe t if your in t you eam. 18th birthda t should be sment test? 1 2 1	ing would ed. acome is ay. e included Neutral 4	be co high in or 5	errect concerning the interest that you would ea enough excluded (identifying reason) from the pool of Very Clear Excellent
Comments: If you had a savings account at a bank, which of on this account? a. Earnings from savings account interest may in b. Income tax may be charged on the interest c. Sales tax may be charged on the interest that d. You cannot earn interest until you pass your a. Don't know f. Prefer not to answer Jpon reviewing the question above, do you feel in questions for the final basic financial literacy asses Language: Very Unclear Wording: Poor Difficult of Question: Tay Four	the follow to be taxe tif your in you earn. 18th birthda t should be sment test? 1 2 1	ing would ed. acome is ay. e included Neutral 4	be cc high in or 5	errect concerning the interest that you would ea enough excluded (identifying reason) from the pool of Very Clear Excellent Too Difficult

Validation Panel Round 1, Group 1 Online Survey (continued)

ara and Joshua iust had a baby. They received	mon	ev as	baby	aifts	and	want to put it away for the baby's education.
nich of the following tends to have the highest	growt	h ove	er pen	ods o	of tim	ne as long as 18 years?
A checking account.						
Stocks. A U.S. Governement savings bond.						
A savings account.						
Prefer not to answer						
on reviewing the question shows, do you feel it	taha	ıld ba	inclu	dod	n or	evoluted (identifying reason) from the people
estions for the final basic financial literacy asses	smen	t test?		laea	in or	excluded (identifying reason) from the pool of
	1	2 1	Veutra	al 4	5	
Language: Very Unclear	0	0	0	0	0	Very Clear
Wording: Poor	0	0	0	0	0	Excellent
Difficulty of Question: Too Fasy	0	0	0	0	0	Too Difficult
b and Mary are the same age. At age 25 Mary	v bega	an sav	/ing \$	52,000	0 a y	ear while Rob saved nothing. At age 50, Rob
b and Mary are the same age. At age 25 Mary alized that he needed money for retirement and ay are both 75	v bega d start	an sav ed sa	ving \$ aving	\$2,000 \$4,00	0 a y 00 pe	ear while Rob saved nothing. At age 50, Rob er year while Mary kept saving her \$2,000. No
b and Mary are the same age. At age 25 Mary alized that he needed money for retirement and ay are both 75 They would each have the same amount beca	/ bega d start	an sat ted sa	ving \$ aving out av	\$2,000 \$4,00	0 a y 00 pe	ear while Rob saved nothing. At age 50, Rob er year while Mary kept saving her \$2,000. No ly the same.
b and Mary are the same age. At age 25 Mary alized that he needed money for retirement and ay are both 75 They would each have the same amount beca Rob, because he saved more each year.	o bega d stan	an sav ed sa	ving \$ aving out av	\$2,000 \$4,00	0 a y 00 pe	ear while Rob saved nothing. At age 50, Rot er year while Mary kept saving her \$2,000. No ly the same.
b and Mary are the same age. At age 25 Mary alized that he needed money for retirement and ay are both 75 They would each have the same amount beca Rob, because he saved more each year. Mary, because she has put away more money Mary, because her money has grown for a	y bega d start ause i y. Iong	an sa led sa they p er tin	ving \$ aving pout av	52,000 \$4,00 vay e com	0 a y 00 pe xacti pour	ear while Rob saved nothing. At age 50, Rot er year while Mary kept saving her \$2,000. N ly the same. nd interest.
b and Mary are the same age. At age 25 Mary alized that he needed money for retirement and ay are both 75 They would each have the same amount beca Rob, because he saved more each year. Mary, because she has put away more money Mary, because her money has grown for a Don't know	r bega I stan ause I y. Iong	nn sau ed sa they p er tin	ving \$ aving out av ne at	62,000 \$4,00 vay e com ,	0 a y 00 pe xacti pour	ear while Rob saved nothing. At age 50, Rob er year while Mary kept saving her \$2,000. No ly the same. nd interest.
b and Mary are the same age. At age 25 Mary alized that he needed money for retirement and ay are both 75 They would each have the same amount beca Rob, because he saved more each year. Mary, because she has put away more money Mary, because her money has grown for a Don't know Prefer not to answer	r bega I stan ause I y. Iong	an sau red sa they p er tin	ving \$ aving put av	52,000 \$4,00 vay e com ;	0 a y 00 pe xacti pour	ear while Rob saved nothing. At age 50, Rob er year while Mary kept saving her \$2,000. No ly the same. nd interest.
b and Mary are the same age. At age 25 Mary lized that he needed money for retirement and are both 75 They would each have the same amount beca Rob, because he saved more each year. Mary, because she has put away more money Mary, because her money has grown for a Don't know Prefer not to answer on reviewing the question above, do you feel it estions for the final basic financial literacy asses	r bega d stan ause l y. long t shoi smen	an sa bed sa they p er tin uld be t test?	ving \$ aving ne at e inclu	52,000 \$4,00 way e com , uded i	0 a y 00 pe xxacti pour in or	ear while Rob saved nothing. At age 50, Rob er year while Mary kept saving her \$2,000. No ly the same. nd interest. excluded (identifying reason) from the pool c
b and Mary are the same age. At age 25 Mary lized that he needed money for retirement and any are both 75 They would each have the same amount beca Rob, because he saved more each year. Mary, because he saved more each year. Mary, because her money has grown for a Don't know Prefer not to answer on reviewing the question above, do you feel it estions for the final basic financial literacy asses	r bega d start ause v. Iong t shou ssmen 1	an san ed sa they p er tin uld be t test? 2 1	ving \$ aving pout av ne at a inclu Neutra	52,000 \$4,00 way e com , uded i	0 a y 00 pe xxacti pour in or 5	ear while Rob saved nothing. At age 50, Rob er year while Mary kept saving her \$2,000. No ly the same. nd interest. excluded (identifying reason) from the pool o
b and Mary are the same age. At age 25 Mary alized that he needed money for retirement and ay are both 75 They would each have the same amount beca Rob, because he saved more each year. Mary, because her saved more each year. Mary, because she has put away more money Mary, because her money has grown for a Don't know Prefer not to answer on reviewing the question above, do you feel it estions for the final basic financial literacy asses Language: Very Unclear	y bega d stan ause f y. Iong t shou smen 1	an sau eed sa they p er tin uld be t test? 2 1	ving § aving put av ne at § inclu Neutra	52,000 \$4,00 way e com, uded i al 4	0 a y 00 pe xacti pour 5	ear while Rob saved nothing. At age 50, Rob er year while Mary kept saving her \$2,000. No ly the same. nd interest. excluded (identifying reason) from the pool o
b and Mary are the same age. At age 25 Mary lized that he needed money for retirement and ay are both 75 They would each have the same amount beca Rob, because he saved more each year. Mary, because he saved more each year. Mary, because she has put away more money Mary, because her money has grown for a Don't know Prefer not to answer on reviewing the question above, do you feel it estions for the final basic financial literacy asses Language: Very Unclear Wording: Poor	y bega d start ause t y. Iong t shou smen 1	an sau they p er tin uld be t test?	ving \$ aving out av ne at ; inclu Neutra	62,000 \$4,00 way e com	0 a y 00 pe xxacti pour 5 0	ear while Rob saved nothing. At age 50, Rob er year while Mary kept saving her \$2,000. No ly the same. nd interest. excluded (identifying reason) from the pool o Very Clear Excellent

Validation Panel Round 1, Group 1 Online Survey (continued)

nder which of the following circumstances would nd repay it with future income? When you need to buy a car to get a much When you really need a week vacation. When some clothes you like go on sale. When the interest on the loan is greater than t	d it be bette	e fina er paj teres	ncially ying j e t you e	ben b. aet o	eficia n voi	al to you to borrow money to buy something n ur savings.
Don't know	110 111	0,00	, you	jor o	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	a ouvingo.
Prefer not to answer						
oon reviewing the question above, do you feel it uestions for the final basic financial literacy asses	sment	uld be t test?	e inclu	ded i	n or	excluded (identifying reason) from the pool of
	1	2	Neutra	4	5	
Language: Very Unclear		0	0	0	0	Very Clear
Wording: Poor	0	0	0	\bigcirc	\bigcirc	No Bias
Difficulty of Question: Too Easy	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Too Difficult
hich of the following statements best describes Your credit record can be checked once a y You cannot see your credit record. All credit records are the property of the U.S.	; your r ear fo Gove	right or fro	t to ch ee. ent and	eck y I acc	our/our/	credit history for accuracy? is only available to the FBI and Lenders.
Thich of the following statements best describes Your credit record can be checked once a y You cannot see your credit record. All credit records are the property of the U.S. You can only check your record for free if your Don't know Prefer not to answer	; your r ear fe Gove are t	right or fro mme ume	t to ch ee. ent and d dowr	eck y I acc n for	/our ess credi	credit history for accuracy? is only available to the FBI and Lenders. t based on a credit report.
Thich of the following statements best describes Your credit record can be checked once a y You cannot see your credit record. All credit records are the property of the U.S. You can only check your record for free if you Don't know Prefer not to answer pon reviewing the question above, do you feel it jestions for the final basic financial literacy assess	s your rear fo Gove are t shou sment	or fro or fro ume ume	t to ch ee. ent and d dowr e inclu ?	eck y I acc n for ded i	/our ess credi n or	credit history for accuracy? is only available to the FBI and Lenders. it based on a credit report. excluded (identifying reason) from the pool of
Thich of the following statements best describes Your credit record can be checked once a y You cannot see your credit record. All credit records are the property of the U.S. You can only check your record for free if you Don't know Prefer not to answer pon reviewing the question above, do you feel it uestions for the final basic financial literacy assess	s your rear fo Gove are t shou sment	r right or fro ume uld be t test 2	t to ch ee. ent and d down e inclu ? Neutra	eck j lacc n for ded i	vour ess credi n or 5	credit history for accuracy? is only available to the FBI and Lenders. it based on a credit report. excluded (identifying reason) from the pool of
Thich of the following statements best describes Your credit record can be checked once a y You cannot see your credit record. All credit records are the property of the U.S. You can only check your record for free if your Don't know Prefer not to answer pon reviewing the question above, do you feel it uestions for the final basic financial literacy assess Language: Very Unclear	s your rear fo Gover are to sment	or fromme urme uld be t test	t to ch ee. ent and d dowr e inclu ? Neutra	l acc n for ded i	vour ess credi n or 5	credit history for accuracy? is only available to the FBI and Lenders. t based on a credit report. excluded (identifying reason) from the pool of Very Clear
Chich of the following statements best describes Your credit record can be checked once a y You cannot see your credit record. All credit records are the property of the U.S. You can only check your record for free if you Don't know Prefer not to answer pon reviewing the question above, do you feel it uestions for the final basic financial literacy assess Language: Very Unclear Wording: Poor	s your rear fo Gove. are t t shou sment	r right or fro mme urme urme t test 2	t to ch ee. ent and d down e inclu ? Neutra	l acc n for ded i	vour ess credi n or 5 0	credit history for accuracy? is only available to the FBI and Lenders. t based on a credit report. excluded (identifying reason) from the pool of Very Clear Excellent
Ihich of the following statements best describes Your credit record can be checked once a y You cannot see your credit record. All credit records are the property of the U.S. You can only check your record for free if you Don't know Prefer not to answer pon reviewing the question above, do you feel it uestions for the final basic financial literacy assess Language: Very Unclear Wording: Poor Difficulty of Question: Too Fasy	s your rear fi Gove a are t t shou sment	r right	t to ch ee. ent and d down e inclu ? Neutra	I acconfor	vour ess credi	credit history for accuracy? is only available to the FBI and Lenders. it based on a credit report. excluded (identifying reason) from the pool of Very Clear Excellent Too Difficult

Validation Panel Round 1, Group 1 Online Survey (continued)

Bank A offers monthly compounding and Bank B offers yearly compounding. Both banks offer the same interest rate. In your opinion, which bank would you choose if you wanted a higher return?

- a. Bank A.
- b. Bank B.
- c. Both banks offer the same return..
- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

		1	2	Neutra	al 4	5	
	Language: Very Unclear	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Very Clear
	Wording: Poor	0	\bigcirc	\bigcirc	0	\bigcirc	Excellent
	Difficulty of Question: Too Easy	0	0	0	0	\bigcirc	Too Difficult
Comments:							

You are likely to obtain the lowest interest on loans with _____?

- a. Credit card companies.
- b. Banks/credit unions.
- c. Check-cashing firms.
- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded (identifying reason) from the pool of questions for the final basic financial literacy assessment test?

	1	2	Ne	eutra	I 4	5	
Language: Very Unclear	\bigcirc	\bigcirc)	\bigcirc	\bigcirc	\bigcirc	Very Clear
Wording: Poor	\bigcirc	\bigcirc	$\mathbf{)}$	\bigcirc	\bigcirc	\bigcirc	Excellent
Difficulty of Question: Too Easy	\bigcirc	\bigcirc	$\mathbf{)}$	\bigcirc	\bigcirc	\bigcirc	Too Difficult

1

Validation Panel Round 1, Group 1 Online Survey (continued)

Self-assessed Financial Literacy Questions

Financial Literacy Self-Assessment Questions:

Reviewing the following financial literacy self-assessment questions, do you feel each should be included in or excluded (identifying reason) from the self-assessment portion of the final basic financial literacy assessment test?

I am pretty good at math.

On a scale of 1 to 7, with 1 meaning Strongly Disagree and 7 meaning Strongly Agree, how do you rate yourself with regard to this statement?

- 1. Strongly Disagree
- 2. Disagree
- 3. Slightly Disagree
- 4. Neither Disagree nor Agree
- 5. Slightly Agree
- 6. Agree
- 7. Strongly Agree
- Include
- Exclude-Not relevant

1,

- Exclude-Other
- Comments:

I am good at dealing with day-to-day financial matters, such as checking accounts, credit and debit cards, mortgages, installment payments, and tracking expenses.

On a scale of 1 to 7, with 1 meaning Strongly Disagree and 7 meaning Strongly Agree, how do you rate yourself with regard to this statement?

- 1. Strongly Disagree
- 2. Disagree
- 3. Slightly Disagree
- 4. Neither Disagree nor Agree
- 5. Slightly Agree
- 6. Agree
- 7. Strongly Agree

Include

Validation Panel Round 1, Group 1 Online Survey (continued)

	Exclude-Not relevant		
	Exclude-Other		
	Comments:		
	h		
am	n good at managing money, budge	ng and planning my expenditi	ures.
n b t	a scale of 1 to 7, with 1 me his statement?	ning Strongly Disagree a	and 7 meaning Strongly Agree, how do you rate yourself with re
5	Strongly Disagree		
	Disagree Slightly Disagree		
Ň	Veither Disagree nor Agree		
. 8	Slightly Agree Agree		
Ś	Strongly Agree		
	Include		
	Exclude-Not relevant		
	Exclude-Other		
	Comments:		
	Commenta.		
	1.		
am	n good at managing credit cards a	d debt like car loans or a moi	rtgage.
	8		"a"o"
n t	a scale of 1 to 7, with 1 me his statement?	ning Strongly Disagree a	and 7 meaning Strongly Agree, how do you rate yourself with re-
5	Strongly Disagree		
	Strongly Disagree Disagree Slightly Disagree		
N CO L	Strongly Disagree Disagree Slightly Disagree Neither Disagree nor Agree		
CON CON CON	Strongly Disagree Disagree Slightly Disagree Veither Disagree nor Agree Slightly Agree		
S L S N S A S	Strongly Disagree Disagree Slightly Disagree Neither Disagree nor Agree Slightly Agree Agree Strongly Agree		
S L S N S A S	Strongly Disagree Disagree Slightly Disagree Veither Disagree nor Agree Slightly Agree Agree Strongly Agree		
S L S A S A S	Strongly Disagree Disagree Slightly Disagree Veither Disagree nor Agree Slightly Agree Agree Strongly Agree		
SLSNSAS	Strongly Disagree Disagree Slightly Disagree Neither Disagree nor Agree Slightly Agree Agree Strongly Agree		
	Strongly Disagree Disagree Slightly Disagree Veither Disagree nor Agree Slightly Agree Agree Strongly Agree Include Exclude-Not relevant		
	Strongly Disagree Disagree Slightly Disagree nor Agree Slightly Agree Agree Strongly Agree Include Exclude-Not relevant Exclude-Other		
	Strongly Disagree Disagree Slightly Disagree Neither Disagree nor Agree Slightly Agree Agree Strongly Agree Include Exclude-Not relevant Exclude-Other Comments:		
	Strongly Disagree Disagree Slightly Disagree Neither Disagree nor Agree Slightly Agree Agree Strongly Agree Include Exclude-Not relevant Exclude-Other Comments:		
	Strongly Disagree Disagree Slightly Disagree nor Agree Slightly Agree Agree Strongly Agree Include Exclude-Not relevant Exclude-Other Comments:		

Validation Panel Round 1, Group 1 Online Survey (continued)

annuities).

On a scale of 1 to 7, with 1 meaning Strongly Disagree and 7 meaning Strongly Agree, how do you rate yourself with regard to this statement?
 Strongly Disagree Disagree Slightly Disagree Neither Disagree nor Agree Slightly Agree Agree Agree Strongly Agree
Exclude-Not relevant
Exclude-Other
Comments:
4
I understand insurance products.
On a scale of 1 to 7 with 1 meaning Strongly Disagree and 7 meaning Strongly Agree, how do you rate yourself with regard
to this statement?
 Strongly Disagree Disagree Slightly Disagree Neither Disagree nor Agree Slightly Agree Agree Agree Strongly Agree
Exclude-Not relevant
Exclude-Other
Comments:
h.
On a scale of 1 to 7, with 1 meaning Very Low and 7 meaning Very High, how would you assess your overall financial knowledge?
1. Very Low
2. Low 3. Slightly Low
4. Average 5. Slightly High 6. High

7. Very High

Validation Panel Round 1, Group 1 Online Survey (continued)

Include		
Exclude-Not relevant		
Exclude-Other		
Comments:		
1.		

Validation Panel Round 1, Group 2 Online Survey

Hello Panel Expert,

Thank you for completing Expert Panel Round 1 and participating in this valuable financial literacy research.

In this second round of review, you will be evaluating 32 financial literacy knowledge questions in the pool for the basic financial literacy instrument and seven financial literacy self-assessment questions. The estimated time to complete the review is 30 to 45 minutes. Please complete your review as soon as possible. Your input will determine the questions included in future rounds of review by other members of the expert panel. For your convenience, survey allows you to stop at any point. It will automatically save and later return to the place you stopped. If you have any questions or need assistance, please feel free to contact me at <u>pjdahmen@mail.usf.com</u> or <u>941-962-1211</u>.

INSTRUCTIONS

As you read each financial literacy question and possible answers (correct answer is bold), identify whether you believe the question should be included or excluded from an instrument used to determine a basic financial literacy score. The goal of this round is to reduce the number of questions in the pool. You are reviewing 50% of the pool, therefore, please feel free to be critical in your selection of the questions to include. Only included question you strongly feel should be included; all other questions should be excluded.

If you determine the question should be included, please identify the financial literacy content category or categories you feel it addresses. The categories for content areas are:

Financial Literacy Category	Topics Included in Category
Money basics	Compound interest, Time value of money, purchasing power/inflation, spending, short-term savings
Borrowing	APR, finance charges, loan terms, credit scores; type of borrowing (e.g., education, home, vehicles)
Building assets	Long-term-savings accounts/Certificates of deposit, stocks, bonds, mutual funds, annuities
Protecting assets	Investment product risk (stocks vs. bonds vs. mutual funds), investment diversity, insurance protection

Following the financial literacy knowledge questions are seven financial literacy self-assessment questions. Please read each question and evaluate if:

1. it has appropriate for language (whether the question is simple, clear and easy to understand),

2. the wording is adequate (whether the question contains wording that is relevant and free from bias, in other words, it portrays a neutral viewpoint). Evaluation is on the following scale:

	Very Unclear	Unclear	Neutral	Clear	Very Clear	
Language	juage 1		3	4	5	
	Poor	Needs Improvement	Neutral	Good	Excellent	
Wording	1	2	3	4	5	

Please feel free to comment on any of the question included in this round.

Validation Panel Round 1, Group 2 Online Survey (continued)

Suppose that in the year 2015, your income has doubled and prices of all goods have doubled too. In 2015, how much will you be able to buy with your income?

- a. More than today
- b. The same
- c. Less than today
- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	gory or Cate addi	gories this resses		
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	0					
Comments:							
							h
If you were to invest \$1,000 in a stor money.	ck mut	ual fund	d, it woul	d be possib	le to have	e less than \$	1,000 when you withdraw your
a. True b. False c. Don't know d. Prefer not to answer							
Upon reviewing the question above, basic financial literacy assessment test addresses.	do you ? If yo	feel it ou feel i	should be it should	e included i be included	n or exclu , please :	ided from the select the ca	pool of questions for the final tegory or categories you feel it
	Inc ques	lude stion?	Cate	gory or Cate add	gories this resses	s question	
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	

This question should be included in the basic financial literacy assessment test.

Comments:

Validation Panel Round 1, Group 2 Online Survey (continued)

High inflation means that the cost of living is increasing rapidly.

- a. True
- b. False
- c. Don't know
- d. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Include question?		Category or Categories this question addresses				
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	\bigcirc					

Comments:

It is usually possible to reduce the risk of investing in the stock market by buying a wide range of stocks and shares.

- a. True
- b. False
- c. Don't know
- d. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Include question?		Category or Categories this question addresses				
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	0					

Comments:

Validation Panel Round 1, Group 2 Online Survey (continued)

The figure that best reflects a mutual fund's performance over a period of years is

a. its current yield.

- b. the total of dividends and capital gains it has paid.
- c. its total return.
- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

No	Money Basics	Borrowing	Building Assets	Protecting Assets	

- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Include question?		Category or Categories this question addresses			
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets
This question should be included in the basic financial literacy assessment test.	0	\bigcirc				

Validation Panel Round 1, Group 2 Online Survey (continued)

To ensure that some of your retirement savings will not be subject to income tax upon withdrawal, you would contribute to:

- a. A Traditional IRS or Individual Retirment Account
- b. A Roth IRA
- c. A 401(k) plan
- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	ory or Cate addi			
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	0					
Comments:							

- b. Longest repayment term
- c. Shortest repayment
- d. Don't know

e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Include question?		Category or Categories this question addresses			
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets
This question should be included in the basic financial literacy assessment test.	0	0				
Validation Panel Round 1, Group 2 Online Survey (continued)

If you always pay the full balance on your credit card, which of the following is least important?

- a. Annual interest rate
- b. Annual fee
- c. Line of credit
- d. Don't know

assessment test.

e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	ory or Cate add	gories this resses	question	
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	0					
Comments:							
Which type of mortgage would allow	a first-t	time ho	mebuyer	to qualify f	or the hig	hest loan am	ount?
a. Fixed-rate mortgage b. Adjustable-rate mortgage							
c. Reverse mortgage d. Don't know 							
Upon reviewing the question above, basic financial literacy assessment test addresses	do you ? If yo	feel it s u feel i	should be t should l	included in the included	n or exclu , please s	ded from the elect the cat	pool of questions for the fina egory or categories you feel
	Incl	udo	Cata	any or Coto	acrice this	question	
	ques	tion?	Categ	add	resses	question	
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy	0	0					**

Validation Panel Round 1, Group 2 Online Survey (continued)

If you signed a twelve-month lease for \$300 month but never occupied the apartment, you legally owe the landlord

- a. your security deposit.
- b. your first month's rent of \$300.
- c. your twelve month's rent of \$3.600.
- d. nothing.
- e whatever the landlord requires.
- f. Don't know
- g. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Include question?		Categ	gory or Cate addi	gories this resses	question		
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets		
This question should be included in the basic financial literacy assessment test.	0	0						
Comments:								
								/;
The most liquid asset is								
 a. money in a certificate of deposit. b. money in a checking account. c. a car. d. an investment account. e a house (real estate). f. Don't know g. Prefer not to answer 								
Upon reviewing the question above,	do you	feel it	should be	included ir	n or exclu	ded from the	e pool of questions for the	fina

basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

Include question? Category or Categories this question addresses

Validation Panel Round 1, Group 2 Online Survey (continued)

	res	No	Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	0	0		0	۲	
Comments:							
Which of the following investments re	equires	that y	ou keep y	our money	invested	for a specifie	ed period or face an early
vithdrawal penalty?							
 Certificate of deposit. Checking account that pays interaction. Government savings bond. 	est.						
 Money market mutual fund. Passbook savings account. 							
f. Don't know g. Prefer not to answer							
Jpon reviewing the question above	do vou	feel it	should be	included in	n or exclu	ded from the	pool of questions for the fir
vasic financial literacy assessment test	? If you	u feel i	t should b	be included	, please s	elect the cat	egory or categories you fee
duiesses.							
	1						
	Incluques	ude tion?	Categ	gory or Cate add	gories this resses	question	
	Incl ques Yes	ude tion? No	Categ Money Basics	gory or Cate addi Borrowing	gories this resses Building Assets	question Protecting Assets	
This question should be included in the basic financial literacy assessment test.	Inclu ques Yes	ude tion? No	Categ Money Basics	gory or Cate add Borrowing	gories this resses Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	Incliques Yes	ude tion? No	Categ Money Basics	gory or Cate add Borrowing	gories this resses Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test. Comments:	Inclu ques Yes	ude tion? No	Categ Money Basics	gory or Cate add Borrowing	gories this resses Building Assets	e question Protecting Assets	
This question should be included in the basic financial literacy assessment test.	Incluques Yes	ude tion? No	Categ Money Basics	gory or Cate add Borrowing	gories this resses Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test. Comments:	Incluques Yes	ude tion? No	Categ Money Basics	gory or Cate add Borrowing	gories this resses Building Assets	e question Protecting Assets	
This question should be included in the basic financial literacy assessment test. Comments:	Incla ques Yes	ude tion? No	Categ Money Basics	gory or Cate add Borrowing	gories this resses Building Assets	rquestion Protecting Assets	
This question should be included in the basic financial literacy assessment test. Comments:	Incluques Yes	No	Categ Money Basics	gory or Cate add Borrowing	gories this resses Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test. Comments:	Incluques Yes	ude tion? No	Categ Money Basics	ory or Cate add Borrowing	gories this resses Building Assets	rincipal.	
This question should be included in the basic financial literacy assessment test. Comments: With compound interest, you earn int	Incluques Yes	No	Categ Money Basics	as well as a	gories this resses Building Assets	rincipal.	
This question should be included in the basic financial literacy assessment test. Comments: <i>With compound interest, you earn int</i> a. True J. False 2. Don't know	Incluques Yes	n your	Categ Money Basics	as well as a	gories this resses Building Assets	rincipal.	
This question should be included in the basic financial literacy assessment test. Comments: <i>Mith compound interest, you earn int</i> a. True 2. False 2. Don't know 3. Prefer not to answer	Incluques Yes	No	Categ Money Basics	as well as a	gories this resses Building Assets	rincipal.	

Validation Panel Round 1, Group 2 Online Survey (continued)

addresses.						
	include question?	Categ	gory or Cate add	gories this resses	question	
	Yes No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0 0					
Comments:						
Autual funds pay a guaranteed rate o a. True 3. False 5. Don't know 1. Prefer not to answer	of return.					
Jpon reviewing the question above, pasic financial literacy assessment test addresses.	do you feel it ? If you feel	should be it should I	e included ir be included	n or exclu , please s	ded from the select the cat	pool of questions for the fin- egory or categories you feel
	Include question?	Categ	gory or Cate add	gories this esses	question	
	Yes No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0 0					
Comments:						
Repeatedly refinancing your home m	nortgage over	a short pe	eriod of time	e results i	n added fee a	and points that further increa
a. True). False 5. Don't know d. Prefer not to answer						
Jpon reviewing the question above,	do you feel it	should be	included in	or exclu	ded from the	pool of questions for the fin
addresses.	r ii you ieei	it should i	be included	please s	select the cat	egory or categories you teel

Validation Panel Round 1, Group 2 Online Survey (continued)

	question?	ad	dresses		
	Yes No	Money Basics Borrowin	g Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.		0 0	٢	0	
Comments:					
Using extra money in a bank saving	s account to p	ay off high interest	rate credi	t card debt is	a good idea.
a. True b. False					
c. Don't know d. Prefer not to answer					
Upon reviewing the question above, basic financial literacy assessment test addresses.	do you feel it ? If you feel i	should be included it should be include	in or exclu d, please	uded from the select the cate	pool of questions for the fina egory or categories you feel i
	Include question?	Category or Cat ad	egories thi dresses	s question	
	Yes No	Money Basics Borrowin	g Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0 0				
Comments					
comments.					
Creditors are required to tell you the	APR that you	will pay when you	get a loan.		
a. True b. False c. Don't know d. Prefer not to answer					
Upon reviewing the question above, basic financial literacy assessment test addresses.	do you feel it ? If you feel i	should be included it should be include	in or exclu d, please	uded from the select the cate	pool of questions for the fina egory or categories you feel i
	Include question?	Category or Cat ad	egories thi dresses	s question	

Validation Panel Round 1, Group 2 Online Survey (continued)

	Yes No	Money Basics Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0 0				5
Comments:					
Your credit rating is not affected by I	how much yo	u charge on your cre	dit cards.		
a. True b. False					
s. Don't know d. Prefer not to answer					
Jpon reviewing the question above, basic financial literacy assessment test addresses.	do you feel it ? If you feel	should be included i it should be included	n or exclu I, please s	ded from the select the cat	pool of questions for the fina egory or categories you feel
	Include question?	Category or Cate add	gories this resses	question	
	Yes No	Money Basics Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0 0				
Comments:					
· · · · · · · · · · · · · · · · · · ·			<i>r.</i> .		
Your creait rating is not arrected by r a. True	now much you	u charge on your cre	alt caras.		
b. Faise c. Don't know d. Prefer not to answer					
Upon reviewing the question above,	do you feel it	should be included i it should be included	n or exclu I, please s	ded from the select the cat	pool of questions for the fin-
basic financial literacy assessment test	? If you reel				5 I S S S S
basic financial literacy assessment test addresses.	Include	Category or Cate	gories this	question	

Validation Panel Round 1, Group 2 Online Survey (continued)

This second as the state in the second								
This question should be included in the basic financial literacy assessment test.	0	0						
Comments:								
Taxes affect how you should invest	your m	oney.						
n . True D. False								
:. Don't know 1. Prefer not to answer								
LTERNATIVE WORDING								
Taxes do not affect how you should	invest	your m	oney.					
a. True			1000-1000-1000					
o. False								
C. Don't know								
с. Don't know d. Prefer not to answer								
2. Don't know d. Prefer not to answer Jpon reviewing the question above, pasic financial literacy assessment test	do you ? If yo	feel it u feel i	should be t should b	included in be included	n or exclu	ded from the patent	bool of que	stions for the fina egories you feel i
<i>c. Don't know</i> <i>d. Prefer not to answer</i> Upon reviewing the question above, basic financial literacy assessment test addresses.	do you ? If yo	feel it i u feel i	should be t should t	included ir be included	n or exclu , please s	ded from the pelect the cate	bool of que gory or cat	stions for the fina egories you feel i
d. Prefer not to answer J. Prefer not to answer Jpon reviewing the question above, basic financial literacy assessment test addresses.	do you ? If yo Incl ques	feel it : u feel i ude tion?	should be t should t Categ	included ir be included gory or Cate addr	n or exclu , please s gories this resses	ded from the pelect the cate	bool of que gory or cat	stions for the fina egories you feel i
2. Don't know d. Prefer not to answer Jpon reviewing the question above, basic financial literacy assessment test addresses.	do you ? If yo Incl ques Yes	feel it : u feel i ude tion? No	should be t should t Categ Money Basics	included ir be included gory or Cater addr Borrowing	n or exclu , please s gories this resses Building Assets	ded from the p select the cate question Protecting Assets	bool of que gory or cat	stions for the fina egories you feel i
C. Don't know d. Prefer not to answer Joon reviewing the question above, basic financial literacy assessment test addresses. This question should be included in the basic financial literacy assessment test.	do you ? If yo Incl ques Yes	feel it i u feel i ude tion? No	should be t should b Categ Money Basics	included ir be included gory or Cate addr Borrowing	or exclu please s gories this resses Building Assets	ded from the p select the cate question Protecting Assets	bool of que	stions for the fina egories you feel i
 Dont know Prefer not to answer Jpon reviewing the question above, pasic financial literacy assessment test addresses. This question should be included in the basic financial literacy assessment test.	do you ? If yo Incl ques Yes	feel it : u feel i ude tion? No	should be t should b Categ Money Basics	e included ir pe included gory or Cate addr Borrowing	or exclu please s gories this resses Building Assets	ded from the pelect the cate equestion Protecting Assets	bool of que gory or cal	stions for the fina egories you feel i
2. Dont know d. Prefer not to answer Jopon reviewing the question above, i basic financial literacy assessment test addresses. This question should be included in the basic financial literacy assessment test. Comments:	do you ? If yo Incl ques Yes	feel it : u feel i ude tion? No	should be t should t Categ Money Basics	e included ir pe included gory or Cates addr Borrowing	or exclu please s gories this resses Building Assets	ded from the period of the cate	bool of que	stions for the fina egories you feel i
2. Don't know d. Prefer not to answer Jpon reviewing the question above, basic financial literacy assessment test addresses. This question should be included in the basic financial literacy assessment test. Comments:	do you ? If yo Incl ques Yes	feel it i u feel i ude tion? No	should be t should b Categ Money Basics	e included ir be included gory or Cate addr Borrowing	a or exclu please s gories this resses Building Assets	ded from the peelect the cate	bool of que	stions for the fina egories you feel i
2. Dont know d. Prefer not to answer Jpon reviewing the question above, i basic financial literacy assessment test addresses. This question should be included in the basic financial literacy assessment test. Comments:	do you ? If yo lncl ques Yes	feel it : u feel i ude tion? No	should be t should t Categ Money Basics	e included ir pe included gory or Cates addr Borrowing	o or exclu please s gories this resses Building Assets	ded from the period of the cate	bool of que	stions for the fina egories you feel i
2. Don't know d. Prefer not to answer Jpon reviewing the question above, pasic financial literacy assessment test addresses. This question should be included in the basic financial literacy assessment test. Comments:	do you ? If yo lncl ques Yes	feel it i u feel i ude tition? No	should be t should b Categ Money Basics	e included ir be included gory or Cate addr Borrowing	n or exclu please s gories this resses Building Assets	ded from the peelect the cate	bool of que	stions for the fina egories you feel i
 Don tknow Prefer not to answer Upon reviewing the question above, ibasic financial literacy assessment test addresses. This question should be included in the basic financial literacy assessment test. Comments: 	do you ? If yo Incl ques Yes	feel it : u feel i ude tion? No	should be t should t Categ Money Basics	e included ir pe included gory or Cates addr Borrowing	a or exclu please s gories this resses Building Assets	ded from the period of the categories of the cat	pool of que	stions for the fina egories you feel i
2. Don't know d. Prefer not to answer Jpon reviewing the question above, pasic financial literacy assessment test addresses. This question should be included in the basic financial literacy assessment test. Comments:	do you ? If yo lncl ques Yes	feel it i u feel i ude tion? No	should be t should b Categ Money Basics	e included ir be included gory or Cate addr Borrowing	a or exclu please s gories this resses Building Assets	ded from the peelect the cate	bool of que	stions for the fina egories you feel i
2. Don tenow d. Prefer not to answer Joon reviewing the question above, i basic financial literacy assessment test addresses. This question should be included in the basic financial literacy assessment test. Comments: Which of the following instruments is	do you ? If yo lncl ques Yes	feel it i u feel i ude tion? No	should be t should b Categ Money Basics	e included ir pe included gory or Cates addr Borrowing	ending?	ded from the period of the categories of the cat	pool of que	stions for the fina egories you feel i
Dent know d. Prefer not to answer Upon reviewing the question above, basic financial literacy assessment test addresses. This question should be included in the basic financial literacy assessment test. Comments: Mhich of the following instruments is a. Debit card. Contribute card.	do you ? If yo lncl ques Yes 	feel it i u feel i ude tion? No	should be t should b Categ Money Basics	e included ir be included gory or Cate addr Borrowing	ending?	ded from the period of the categories of the cat	bool of que	stions for the fina egories you feel i
 Don trivinow Deport know Prefer not to answer Upon reviewing the question above, basic financial literacy assessment test addresses. This question should be included in the basic financial literacy assessment test. Comments: Comments: Mhich of the following instruments is a. Debit card. Certificate of deposit. Cash. 	do you ? If yo Incl ques Yes	feel it i u feel i ude tion? No	should be t should b Categ Money Basics	e included ir pe included gory or Cates addr Borrowing	ending?	ded from the period of the categories of the cat	bool of que	stions for the fina egories you feel i

Validation Panel Round 1, Group 2 Online Survey (continued)

			230,00					
This question should be included in the basic financial literacy assessment test.	0	0						
comments:								
axes affect how you should invest	your me	oney.						
True								
. False . Don't know I. Prefer not to answer								
LTERNATIVE WORDING								
Taxes do not affect how you should	l invest j	your m	oney.					
n. True								
 False Don't know Prefer not to answer 								
Upon reviewing the question above, basic financial literacy assessment tes addresses.	do you t? If yo	feel it u feel i	should be t should l	included in the included	n or exclu , please s	ded from the p select the cate	ool of questi gory or categ	ions for the fin gories you fee
	ques	tion?	Cateç	add	gones uns resses	question		
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets		
This question should be included in the basic financial literacy assessment test.	0	0						
Comments:								
								1
Which of the following instruments i	s NOT t	ypicall	y associa	ted with sp	ending?			
Which of the following instruments i a. Debit card. b. Certificate of deposit. 2. Cash.	is NOT t	ypicall <u></u>	y associa	ted with sp	ending?			

Validation Panel Round 1, Group 2 Online Survey (continued)

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Inclu quest	ude tion?	Categ	ory or Cate add	gories this resses	question	
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	0					
Comments:							
		,					
Many people put aside money to tak emergencies, in which of the followir	e care o ng forms	of unex s would	expected e	xpenses. li LEAST ben	f Juan an nefit to the	d Elva have em if they n	money put aside for eeded it right away?
Many people put aside money to tak emergencies, in which of the followir a. Invested in a down payment on b. Checking account c. Stocks d. Savings account e. Don't know	te care of ng forms n the ho	of unex s would o use	kpected e d it be of i	xpenses. li LEAST ben	f Juan an nefit to the	d Elva have om if they n	e money put aside for eeded it right away?
Many people put aside money to tak emergencies, in which of the followir a. Invested in a down payment or b. Checking account c. Stocks d. Savings account e. Don't know f. Prefer not to answer	te care of ng forms n the ho	of unex would buse	kpected e	xpenses. II LEAST ben	f Juan and lefit to the	d Elva have	e money put aside for eeded it right away?
Many people put aside money to tak emergencies, in which of the followir a. Invested in a down payment on b. Checking account c. Stocks d. Savings account e. Don't know f. Prefer not to answer Upon reviewing the question above, basic financial literacy assessment test addresses.	do you f	of unex s would o use feel it s u feel i	cpected e d it be of i should be t should b	xpenses. Ii LEAST ben included ir be included	f Juan an hefit to the n or exclu , please s	d Elva have om if they n ded from th select the c	e money put aside for eeded it right away? e pool of questions for the fi ategory or categories you fee
Many people put aside money to tak emergencies, in which of the followir a. Invested in a down payment on b. Checking account c. Stocks d. Savings account e. Don't know f. Prefer not to answer Upon reviewing the question above, basic financial literacy assessment test addresses.	do you f do you f ? If you lnclu quest	of unex s would ouse feel it s u feel i ude tion?	cpected e d it be of i should be t should b	xpenses. Il LEAST ben included ir pe included jory or Cate add	f Juan ann hefit to the n or exclu , please s gories this resses	d Elva have om if they n ded from th select the c	e money put aside for eeded it right away? e pool of questions for the fi ategory or categories you fee
Many people put aside money to tak emergencies, in which of the followir a. Invested in a down payment on b. Checking account c. Stocks d. Savings account e. Don't know f. Prefer not to answer Upon reviewing the question above, basic financial literacy assessment test addresses.	do you f do you f ? If you quest Yes	feel it suffeel it suf	should be catego Money Basics	xpenses. Il LEAST beri included ir pe included gory or Cate add Borrowing	f Juan ann hefit to the n or exclu , please s gories this resses Building Assets	d Elva have am if they n ded from th select the c s question Protecting Assets	e money put aside for eeded it right away? e pool of questions for the fi ategory or categories you fee
Many people put aside money to tak emergencies, in which of the followir a. Invested in a down payment or b. Checking account c. Stocks d. Savings account e. Don't know f. Prefer not to answer Upon reviewing the question above, i basic financial literacy assessment test addresses.	do you f do you f ? If you yes	of unexs s would use feel it : u feel i u feel i No	should be t should be t should t Categ Money Basics	included ir be included ir ory or Cate add Borrowing	f Juan ann hefit to the n or exclu , please s gories this resses Building Assets	d Elva have em if they n ded from th select the c s question Protecting Assets	e money put aside for eeded it right away? e pool of questions for the fi ategory or categories you fee

If you had a savings account at a bank, which of the following would be correct concerning the interest that you would earn on this account?

a. Earnings from savings account interest may not be taxed.

Validation Panel Round 1, Group 2 Online Survey (continued)

- b. Income tax may be charged on the interest if your income is high enough
- c. Sales tax may be charged on the interest that you earn.
- d. You cannot earn interest until you pass your 18th birthday.
- e. Don't know f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	ory or Cate add	gories this resses	question			
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets			
This question should be included in the basic financial literacy assessment test.	0	0					-		
omments:									
 Which of the following tends to have A checking account. Stocks. A U.S. Government savings bor A savings account. Don't know Prefer not to answer Jpon reviewing the question above, asic financial literacy assessment test 	the hig nd. doyou ? Ifyo	hest gr feel it s u feel it	should be t should l	er periods o	f time as n or exclu	long as 18 ye	ears? e pool c tegory	of questic or categ	ons for the fi
ddresses.	Incl	ude tion?	Categ	gory or Cate add	gories this resses	s question			
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets			
This question should be included in the basic financial literacy assessment test.	0								
Comments:									

Rob and Mary are the same age. At age 25 Mary began saving \$2,000 a year while Rob saved nothing. At age 50, Rob

Validation Panel Round 1, Group 2 Online Survey (continued)

- b. Income tax may be charged on the interest if your income is high enough
- c. Sales tax may be charged on the interest that you earn.
- d. You cannot earn interest until you pass your 18th birthday.
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	ory or Categ addr	gories this resses	question
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets
This question should be included in the basic financial literacy assessment test.	0	0				

Comments:

Sara and Joshua just had a baby. They received money as baby gifts and want to put it away for the baby's education. Which of the following tends to have the highest growth over periods of time as long as 18 years?

- a. A checking account.
- b. Stocks.
- c. A U.S. Government savings bond.
- d. A savings account.
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets		
	0						
							//

Rob and Mary are the same age. At age 25 Mary began saving \$2,000 a year while Rob saved nothing. At age 50, Rob

Validation Panel Round 1, Group 2 Online Survey (continued)

realized that he needed money for retirement and started saving \$4,000 per year while Mary kept saving her \$2,000. Now they are both 75

- a. They would each have the same amount because they put away exactly the same.
- b. Rob, because he saved more each year.
- c. Mary, because she has put away more money.
- d. Mary, because her money has grown for a longer time at compound interest.
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Category or Categories addresses			s this question s	
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	0					
Comments:							

Under which of the following circumstances would it be financially beneficial to you to borrow money to buy something now and repay it with future income?

- a. When you need to buy a car to get a much better paying job.
- b. When you really need a week vacation.
- c. When some clothes you like go on sale.
- d. When the interest on the loan is greater than the interest you get on your savings.
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Category or Categories this question addresses			Category or Categories this que addresses		
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets		
This question should be included in the basic financial literacy assessment test.	0	0						

Comments:

Validation Panel Round 1, Group 2 Online Survey (continued)

Which of the following statements best describes your right to check your credit history for accuracy?

- a. Your credit record can be checked once a year for free.
- b. You cannot see your credit record.
- c. All credit records are the property of the U.S. Government and access is only available to the FBI and Lenders.
- d. You can only check your record for free if you are turned down for credit based on a credit report.
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Include question?		Categ	Category or Categories this question addresses			
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	0					
Comments:							

If you are behind on your debt payments and go to a responsible credit counseling service such as the Consumer Credit Counseling Services, what help can they give you?

- a. They can cancel and cut up all of your credit cards without your permission.
- b. They can get the federal government to apply your income taxes to pay off your debts.
- c. They can work with those who loaned you money to set up a payment schedule that you can meet.
- d. They can force those who loaned you money to forgive all your debts.
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Include question?		Category or Categories this question addresses				
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	0					

Comments:

Validation Panel Round 1, Group 2 Online Survey (continued)

Barbara has just applied for a credit card. She is an 18-year-old high school graduate with few valuable possessions and no credit history. If Barbara is granted a credit card, which of the following is the most likely way that the credit card company will reduce ITS risk?

- a. It will make Barbara's parents pledge their home to repay Karen's credit card debt.
- b. It will require Barbara to have both parents co-sign for the card.
- c. It will charge Barbara twice the finance charge rate it charges older cardholders.
- d. It will start Barbara out with a small line of credit to see how she handles the account.
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Include question?		Category or Categories this question addresses				
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	0					

Comments:

Bank A offers monthly compounding and Bank B offers yearly compounding.	Both banks offer the same interest rate.	In
your opinion, which bank would you choose if you wanted a higher return?		

- a. Bank A.
- b. Bank B.
- c. Both banks offer the same return ..
- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

Category or Categories this question addresses				
Money Borrowing Building Protecting Basics Assets Assets				

This question should be included in

Validation Panel Round 1, Group 2 Online Survey (continued)

Comments:								
								1.
Variant Black as a laster the laster to strategic to				2				
a. Credit card companies.	n loans wit	h						
 b. Banks/credit unions. c. Check-cashing firms. 								
d. Don't know e. Prefer not to answer								
I non reviewing the question above	do vou fe	ol it s	hould be i	ncluded in	or exclu	led from the no	ol of questions for the	final
basic financial literacy assessment test addresses.	? If you	feel it	should be	included,	please s	elect the catego	ory or categories you for	el it
	Inclue	de on?	Catego	ry or Categ addr	ories this	question		
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets		
This question should be included in the basic financial literacy assessment test.	0	0						
	2							
Comments:								
Comments:								
Comments:								
Comments:								1,
Comments:								1,
Comments: F-assessed Financial Literacy Questions								1
Comments: f-assessed Financial Literacy Questions Financial Literacy Self-Asse	ssmen	nt Qu	estions					h
Comments: F-assessed Financial Literacy Questions Financial Literacy Self-Asse	ssmen	nt Qu	estions	:				h
Comments: F-assessed Financial Literacy Questions Financial Literacy Self-Asse Reviewing the following financial literacy sel reason) from the self-assessment portion	SSMEN If-assessmen of the fin	nt Que	estions stions, do y	ou feel eac	h should t ment test?	be included in or	excluded (identifying	1,
Comments: Fassessed Financial Literacy Questions Financial Literacy Self-Asse Reviewing the following financial literacy sel reason) from the self-assessment portion <i>I am pretty good at math.</i>	ssmen If-assessmen a of the fin	nt Qu	estions stions, do y c financial li	: ou feel eac	h should t ment test?	be included in or	excluded (identifying	1,

Validation Panel Round 1, Group 2 Online Survey (continued)

 Strongly Disagree Disagree Slightly Disagree Neither Disagree nor Agree Slightly Agree Agree Strongly Agree 		·		
Language: Very Unclear	1 2 M	Neutral 4	5	Very Clear
Wording: Poor		0 0	0	Excellent
I am good at dealing with day-to-day financial matters, st tracking expenses. On a scale of 1 to 7, with 1 meaning Strongly L to this statement? 1. Strongly Disagree 2. Disagree 3. Slightly Disagree 4. Neither Disagree nor Agree 5. Slightly Agree 6. Agree 7. Strongly Agree	uch as checking Disagree and	accounts, cre 7 meaning	dit ar	nd debit cards, mortgages, installment payments, and ngly Agree, how do you rate yourself with regai
	121	Neutral 4	5	Ven Clear
Languages Many Hardson		- COLD - COLD		VEIV CIER
Language: Very Unclear Wording: Poor	0 0	0 0	\circ	Excellent

Validation Panel Round 1, Group 2 Online Survey (continued)

On a scale of 1 to 7, with 1 meaning Strongly Disagree and 7 meaning Strongly Agree, how do you rate yourself with regard to this statement?
1. Strongly Disagree 2. Disagree 3. Slightly Disagree 4. Neither Disagree nor Agree 5. Slightly Agree 5. Agree 7. Strongly Agree
1 2 Neutral 4 5
Language: Very Unclear Wording: Poor
Comments:
 I am good at managing credit cards and debt like car loans or a mortgage. On a scale of 1 to 7, with 1 meaning Strongly Disagree and 7 meaning Strongly Agree, how do you rate yourself with regard to this statement? 1. Strongly Disagree 2. Disagree 3. Slightly Disagree nor Agree 5. Slightly Agree
6. Agree 7. Strongly Agree 1 2 Neutral 4 5
Language: Very Unclear
Wording: Poor
Comments:

Appendix L

Self-assessment Financial Literacy Questions

How strongly do you agree or disagree with the following statements? Use the scale below to note your agreement.

- Strongly Disagree
 Disagree
 Slightly Disagree
 Neither Agree nor Disagree
 Slightly Agree
 Agree
 Strongly Agree
- 1. I am pretty good at math.¹
- 2. I am good at dealing with day-to-day financial matters, such as checking accounts, credit and debit cards, mortgages, installment payments, and tracking expenses.^{1 2}
- 3. I am good at managing money, budgeting and planning my expenditures.³
- 4. I am good at managing credit cards and debt like car loans or a mortgage.³
- 5. I understand investment products (stocks, bonds and mutual funds) and retirement accounts (IRAs, 401Ks and annuities).³
- 6. I understand the stock market reasonably well.²
- 7. I understand insurance products.³

On a scale from 1 to 7, where 1 means very low and 7 means very high,

8. How would you assess your overall financial knowledge?¹

Notes.

¹ Lusardi and Mitchell (2011a, 2011b, 2013, 2014) subjective financial literacy questions used in NFCS, HRS, ALP, DHS, and OECD.

² Hsu (2011) subjective question used in CogEcon (Cognitive Economics Survey).

³ Finke and Huston (2014) subjective questions used in FLAT assessment.

Appendix M

Validation Panel Round 2 Online Surveys

Validation Panel Round 2, Group 1 Online Survey

Hello Panel Expert,

Thank you for completing Expert Panel Round 1 and participating in this valuable financial literacy research.

In this second round of review, you will be evaluating 32 financial literacy knowledge questions in the pool for the basic financial literacy instrument and seven financial literacy self-assessment questions. The estimated time to complete the review is 30 to 45 minutes. Please complete your review as soon as possible. Your input will determine the questions included in future rounds of review by other members of the expert panel. For your convenience, survey allows you to stop at any point. It will automatically save and later return to the place you stopped. If you have any questions or need assistance, please feel free to contact me at <u>pjdahmen@mail.usf.com</u> or <u>941-962-1211</u>.

INSTRUCTIONS

As you read each financial literacy question and possible answers (correct answer is bold), identify whether you believe the question should be included or excluded from an instrument used to determine a basic financial literacy score. The goal of this round is to reduce the number of questions in the pool. You are reviewing 50% of the pool, therefore, please feel free to be critical in your selection of the questions to include. Only included question you strongly feel should be included; all other questions should be excluded.

If you determine the question should be included, please identify the financial literacy content category or categories you feel it addresses. The categories for content areas are:

Financial Literacy Category	Topics Included in Category
Money basics	Compound interest, Time value of money, purchasing power/inflation, spending, short-term savings
Borrowing	APR, finance charges, loan terms, credit scores; type of borrowing (e.g., education, home, vehicles)
Building assets	Long-term-savings accounts/Certificates of deposit, stocks, bonds, mutual funds, annuities
Protecting assets	Investment product risk (stocks vs. bonds vs. mutual funds), investment diversity, insurance protection

Following the financial literacy knowledge questions are seven financial literacy self-assessment questions. Please read each question and evaluate if:

1. it has appropriate for language (whether the question is simple, clear and easy to understand),

2. the wording is adequate (whether the question contains wording that is relevant and free from bias, in other words, it portrays a neutral viewpoint). Evaluation is on the following scale:

	Very Unclear	Unclear	Neutral	Clear	Very Clear
Language	1	2	3	4	5
	Poor	Needs Improvement	Neutral	Good	Excellent
Wording	1	2	3	4	5

Please feel free to comment on any of the question included in this round.

Validation Panel Round 2, Group 1 Online Survey (continued)

Financial Literacy Assessment Questions:

Suppose you had \$100 in a savings account and the interest rate was 2% per year. After 5 years how much do you think you would have in the account if you left the money to grow?

a. More than \$102

- b. Exactly \$102
- c. Less than \$102
- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	jory or Categ addr	gories this esses	question
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets
This question should be included in the basic financial literacy assessment test.	0	0				
Comments:						

Imagine that the interest rate on your savings account was 1% per year and inflation was 2% per year. After 1 year, would you be able to buy:

- a. More than today
- b. Exactly the same
- c. Less than today
- d. Don't know
- e. Prefer not not answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	Category or Categories this question addresses					
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets			
This question should be included in the basic financial literacy assessment test.	0	0	0	0	0	0			
Comments:									

Validation Panel Round 2, Group 1 Online Survey (continued)

An annuity is a financial product that pays a lump sum when you die.

- a. True
- b. False
- c. Don't know
- d. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	gory or Cate addi	gories this resses	question	
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	0					
Comments:							

- a. True
- b. False
- c. Don't know
- d. Prefer not to say

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	ory or Categ addr	gories this resses	question		
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets		
This question should be included in the basic financial literacy assessment test.	0	0						
omments:								

Validation Panel Round 2, Group 1 Online Survey (continued)

Yes						
	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
0	0					-
gher	month	nly payme	ents than a	30-year r	nortgage, bu	t the total interest paid ove
u fee	l it sho	uld be includ	luded in or e	excluded f	rom the pool	of questions for the final basi
u fee el it Inclu	l it sho should ude ion?	uld be incl be includ Categ	luded in or e ed, please s ory or Cate addr	excluded f select the o gories this esses	rom the pool category or ca question	of questions for the final basi ategories you feel it addresse
u fee eel it Inclu quest Yes	l it sho should ude tion? No	uld be incl be includ Categ Money Basics	luded in or e ed, please s lory or Cate addr Borrowing	excluded f elect the gories this esses Building Assets	rom the pool category or ca question Protecting Assets	of questions for the final basi ategories you feel it addresse
u fee el it Inclu juest Yes	l it sho should ude tion? No	uld be includ be includ Categ Money Basics	luded in or e ed, please s ory or Cater addr Borrowing	excluded f select the gories this esses Building Assets	rom the pool of category or category or category or category or category or category of the ca	of questions for the final basi ategories you feel it addresse
u fee eel it Inclu juest Yes	l it sho should ude tion? No	uld be includ be includ Categ Money Basics	luded in or e ed, please s ory or Cate addr Borrowing	excluded f leelect the o gories this esses Building Assets	rom the pool of category or category or category or category or category or category or category of the second sec	of questions for the final basi ategories you feel it addresse
u fee eel it Inclu quest Yes	l it sho should ude ion? No	uld be includ be includ Categ Money Basics	luded in or e ed, please s lory or Categ addr Borrowing	excluded f lelect the gories this esses Building Assets	rom the pool category or ca question Protecting Assets	of questions for the final basi ategories you feel it addresse
u fee eel it Inclu quest Yes	l it sho should ude ion? No	uld be includ be includ Categ Money Basics	luded in or e ed, please s ory or Cate addr Borrowing	excluded f elect the gories this esses Building Assets	rom the pool of category or category or category or category or category or category or category of the catego	of questions for the final basi ategories you feel it addresse
u fee eel it Inclu quest Yes	l it sho should ude tion? No	uld be includ be includ Categ Money Basics	luded in or e ed, please s ory or Cate, addr Borrowing	excluded f elect the gories this esses Building Assets	rom the pool of category or category or category or category or category of the category of th	of questions for the final basi ategories you feel it addresse
ig	gher	gher month	gher monthly payme	gher monthly payments than a	gher monthly payments than a 30-year r	gher monthly payments than a 30-year mortgage, bu

Validation Panel Round 2, Group 1 Online Survey (continued)

	Incl	ude tion?	Categ	gory or Categ addr	gories this resses	question	
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	0	0	0	0	0	-
Comments:							
Which bank account is likely to pay	the higi	hest in	terest rate	e on money	saved?		
Which bank account is likely to pay	the higi	hest in	terest rate	e on money	saved?		
Which bank account is likely to pay b. Savings account b. Six month CD or certificate	the higi	hest in	terest rate	e on money	saved?		
Which bank account is likely to pay b. Savings account c. Six month CD or certificate c. Three year CD b. Don't know	the higi	hest in	terest rate	e on money	saved?		
Which bank account is likely to pay a. Savings account b. Six month CD or certificate c. Three year CD b. Don't know b. Prefer not to answer	the higi	hest in	terest rate	e on money	saved?		
Which bank account is likely to pay b. Savings account c. Six month CD or certificate c. Three year CD b. Don't know c. Prefer not to answer	the higi	hest in	terest rate	e on money	saved?		
Which bank account is likely to pay b. Savings account b. Six month CD or certificate c. Three year CD c. Don't know b. Prefer not to answer Joon reviewing the question above, do nancial literacy assessment test? If yo	the high	hest in el it sho	terest rate	e on money luded in or e led, please s	saved? excluded f	rom the pool category or ca	of questions for the final bas ategories you feel it address
Which bank account is likely to pay a. Savings account b. Six month CD or certificate c. Three year CD c. Don't know b. Prefer not to answer Joon reviewing the question above, do nancial literacy assessment test? If yo	the high	el it should	terest rate	e on money luded in or e led, please s	saved? excluded f	rom the pool category or ca	of questions for the final bas ategories you feel it address
Which bank account is likely to pay a. Savings account b. Six month CD or certificate b. Three year CD c. Don't know b. Prefer not to answer Jpon reviewing the question above, do nancial literacy assessment test? If yo	the higi o you fee ou feel it Incl ques	hest in el it sho should ude tion?	terest rate build be includ be includ Categ	e on money luded in or e led, please s gory or Cates addr	saved? excluded f elect the gories this esses	rom the pool category or ca question	of questions for the final bas ategories you feel it address
Which bank account is likely to pay b. Savings account c. Six month CD or certificate c. Three year CD c. Don't know c. Prefer not to answer Upon reviewing the question above, do nancial literacy assessment test? If yo	b you fee bu feel it ques Yes	hest in el it sho s should ude tion? No	terest rate ould be incl be includ Categ Money Basics	e on money luded in or e led, please s gory or Cate; addr Borrowing	saved? excluded f elect the gories this esses Building Assets	rom the pool category or ca question Protecting Assets	of questions for the final bas ategories you feel it address
Which bank account is likely to pay b. Savings account c. Six month CD or certificate c. Three year CD d. Don't know d. Don't know prefer not to answer Upon reviewing the question above, do nancial literacy assessment test? If you This question should be included in the basic financial literacy assessment test.	byou fee ou feel it ques Yes	el it sho should ude tion?	terest rate	luded in or e led, please s gory or Cates addr Borrowing	saved? excluded f elect the gories this esses Building Assets	rom the pool category or ca question Protecting Assets	of questions for the final bas ategories you feel it address
Which bank account is likely to pay a. Savings account b. Six month CD or certificate c. Three year CD b. Don't know b. Prefer not to answer Upon reviewing the question above, do nancial literacy assessment test? If you This question should be included in the basic financial literacy assessment test.	by you feed by feed it uncl ques Yes	hest in should ude tion? No	terest rate	luded in or e led, please s gory or Cate Borrowing	saved? excluded f elect the gories this esses Building Assets	rom the pool category or ca question Protecting Assets	of questions for the final bas ategories you feel it address
Which bank account is likely to pay a. Savings account b. Six month CD or certificate c. Three year CD b. Don't know c. Prefer not to answer Joon reviewing the question above, do nancial literacy assessment test? If you This question should be included in the basic financial literacy assessment test. Comments:	byou feed ou feed it ques Yes	el it sho should ude No	terest rate	luded in or e led, please s gory or Cate, addr Borrowing	saved? excluded f elect the gories this esses Building Assets	rom the pool category or ca e question Protecting Assets	of questions for the final bas ategories you feel it address
Which bank account is likely to pay a. Savings account b. Six month CD or certificate c. Three year CD b. Don't know b. Prefer not to answer Upon reviewing the question above, do nancial literacy assessment test? If you This question should be included in the basic financial literacy assessment test. Comments:	byou fee ou feel it ques Yes	el it sho should ude No	uld be incl be includ Catego Money Basics	luded in or e led, please s gory or Cates addr Borrowing	saved? excluded f elect the gories this esses Building Assets	rom the pool category or ca question Protecting Assets	of questions for the final bas ategories you feel it address
Which bank account is likely to pay A. Savings account C. Six month CD or certificate Three year CD Don't know Prefer not to answer Upon reviewing the question above, do nancial literacy assessment test? If yo This question should be included in the basic financial literacy assessment test. Comments:	byou fee ou feel it ques Yes	el it sho should ude ttion? No	uld be includ be includ Categ Money Basics	luded in or e led, please s gory or Categ addr Borrowing	saved? excluded f elect the gories this esses Building Assets	rom the pool category or ca question Protecting Assets	of questions for the final bas ategories you feel it address
Which bank account is likely to pay a. Savings account b. Six month CD or certificate c. Three year CD b. Don't know b. Prefer not to answer Joon reviewing the question above, do nancial literacy assessment test? If you This question should be included in the basic financial literacy assessment test. Comments:	byou fee ou feel it lncl ques Yes	el it sho s should ude ttion?	terest rate	luded in or e led, please s gory or Cate, Borrowing	saved? excluded f elect the gories this esses Building Assets	rom the pool category or ca question Protecting Assets	of questions for the final bas ategories you feel it address
Which bank account is likely to pay a. Savings account b. Six month CD or certificate c. Three year CD b. Don't know c. Prefer not to answer Joon reviewing the question above, do nancial literacy assessment test? If your This question should be included in the basic financial literacy assessment test. Comments:	byou fee ou feel it ques Yes	el it sho should ude tion? No	terest rate	luded in or e led, please s gory or Cate addr Borrowing	saved? excluded f elect the gories this esses Building Assets	rom the pool category or ca question Protecting Assets	of questions for the final bas ategories you feel it address
Which bank account is likely to pay Savings account Six month CD or certificate Three year CD Don't know Prefer not to answer Upon reviewing the question above, do nancial literacy assessment test? If you This question should be included in the basic financial literacy assessment test. Comments:	byou fee ou feel it ques Yes	el it sho should ude No	terest rate	luded in or e led, please s gory or Cate, addr Borrowing	saved? excluded f elect the gories this esses Building Assets	rom the pool category or ca e question Protecting Assets	of questions for the final bas ategories you feel it address

Validation Panel Round 2, Group 1 Online Survey (continued)

	Incluquesti	de on?	Categ	ory or Cate addi	gories this esses	question	
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	•					-
'omments:							
he main advantage of a 401(k) plar	n is that i	it:					
The main advantage of a 401(k) plan . Provides a high rate of return with . Allows you to shelfter retirmen	n is that i h little ris nt saving	it: :k is fron	ı taxatio	n			
The main advantage of a 401(k) plant . Provides a high rate of return with . Allows you to shelfter retirment . Provides a well diversified mix of . Don't know	n is that i h little ris nt saving f investm	it: sk is fron ient as	1 taxatio sets	n			
The main advantage of a 401(k) plan a. Provides a high rate of return with b. Allows you to shelfter retirment c. Provides a well diversified mix of l. Don't know b. Prefer not to answer	n is that i h little ris nt saving f investm	it: sk is fron bent as	1 taxatio sets	n			
The main advantage of a 401(k) plan b. Provides a high rate of return with b. Allows you to shelfter retirment c. Provides a well diversified mix of d. Don't know b. Prefer not to answer Joon reviewing the question above, do nancial literacy assessment test? If you	n is that i h little ris nt saving f investm o you feel ou feel it s	it: sk is fron bent as it shoud	n <i>taxatio</i> sets ild be inc be includ	n luded in or e ed, please s	excluded f	rom the pool category or ca	of questions for the final ba ategories you feel it addres
The main advantage of a 401(k) plan a. Provides a high rate of return with b. Allows you to shelfter retirment c. Provides a well diversified mix of d. Don't know b. Prefer not to answer Jpon reviewing the question above, do nancial literacy assessment test? If you	n is that i th little ris nt saving f investm o you feel ou feel it s Inclu	it: is from is from it shou should de	n <i>taxatio</i> sets Ild be inc be includ Categ	n luded in or e ed, please s jory or Cate	excluded f select the o	rom the pool category or ca	of questions for the final ba ategories you feel it addres
The main advantage of a 401(k) plan a. Provides a high rate of return with b. Allows you to shelfter retirment c. Provides a well diversified mix of d. Don't know b. Prefer not to answer Jpon reviewing the question above, do inancial literacy assessment test? If you	n is that i h little ris nt saving f investm o you feel ou feel it s Inclu questi Ves	it: is from eent as it should should de on?	n taxatio sets Ild be inc be includ Categ Money	n luded in or e ed, please s jory or Cate, addu Borrowing	excluded f select the gories this esses Building	rom the pool category or ca question Protecting	of questions for the final ba ategories you feel it addres
The main advantage of a 401(k) plan p. Provides a high rate of return with p. Allows you to shelfter retirment p. Provides a well diversified mix of 1. Don't know p. Prefer not to answer Upon reviewing the question above, do nancial literacy assessment test? If you This question should be included in	n is that i h little ris nt saving f investm o you feel ou feel it s Inclui questi Yes	it: sk is fron eent as it should should de on? No	n taxatio sets Ild be inc be includ Categ Money Basics	n luded in or e ed, please s lory or Cate addi Borrowing	excluded f select the gories this esses Building Assets	rom the pool category or ca question Protecting Assets	of questions for the final ba ategories you feel it addres
The main advantage of a 401(k) plan a. Provides a high rate of return with b. Allows you to shelfter retirment b. Provides a well diversified mix of c. Don't know b. Prefer not to answer Jpon reviewing the question above, do nancial literacy assessment test? If you This question should be included in the basic financial literacy assessment test.	n is that i h little ris t saving f investm o you feel ou feel it s Inclui questi Yes	it: sk ss from eent as it should should de on? No	n taxatio sets Idd be inc be includ Categ Money Basics	n luded in or e ed, please s gory or Cates addu Borrowing	excluded f elect the or gories this esses Building Assets	rom the pool category or ca question Protecting Assets	of questions for the final ba ategories you feel it addres
The main advantage of a 401(k) plan Provides a high rate of return with Allows you to shelfter retirment Provides a well diversified mix of Don't know Prefer not to answer Jpon reviewing the question above, do nancial literacy assessment test? If you This question should be included in the basic financial literacy assessment test.	n is that i th little ris t saving f investm o you feel ou feel it s Inclu questi Yes	it: sk sent as it shou should de on? No	n taxatio sets IId be inc be includ Categ Money Basics	n luded in or e ed, please s lory or Cate, addi Borrowing	excluded f select the gories this esses Building Assets	rom the pool category or ca question Protecting Assets	of questions for the final ba ategories you feel it addres
The main advantage of a 401(k) plan a. Provides a high rate of return with b. Allows you to shelfter retirment c. Provides a well diversified mix of t. Don't know b. Prefer not to answer Joon reviewing the question above, dd nancial literacy assessment test? If you This question should be included in the basic financial literacy assessment test.	n is that i h little ris t saving f investm o you feel ou feel it s Inclui questi Yes	it: sk sfrom eent as it should de on? No O	n taxatio sets Idd be includ Categ Money Basics	n luded in or e ed, please s lory or Cates addr Borrowing	excluded f select the gories this esses Building Assets	rom the pool category or ca question Protecting Assets	of questions for the final ba ategories you feel it addres
The main advantage of a 401(k) plan Provides a high rate of return with Allows you to shelfter retirment Provides a well diversified mix of Don't know Prefer not to answer Ipon reviewing the question above, do nancial literacy assessment test? If you This question should be included in the basic financial literacy assessment test.	n is that i h little ris t saving f investm o you feel ou feel it s linclui questi Yes	it: is from s from ent as it should de on? No	n taxatio sets IId be inc be includ Categ Money Basics	n luded in or e ed, please s ory or Cate, addr Borrowing	excluded f iselect the o gories this resses Building Assets	rom the pool category or ca question Protecting Assets	of questions for the final ba ategories you feel it addres
The main advantage of a 401(k) plan Provides a high rate of return with Provides a well diversified mix of Don't know Prefer not to answer Upon reviewing the question above, do nancial literacy assessment test? If you This question should be included in the basic financial literacy assessment test. Comments:	n is that i h little ris t saving f investm o you feel ou feel it s Inclui questi Yes	it: it shound hould No	n taxatio sets IId be includ Categ Money Basics	n luded in or e ed, please s gory or Cate addr Borrowing	excluded f select the gories this esses Building Assets	rom the pool category or ca question Protecting Assets	of questions for the final ba ategories you feel it addres
The main advantage of a 401(k) plan Provides a high rate of return with Allows you to shelfter retirment Provides a well diversified mix of Don't know Prefer not to answer Upon reviewing the question above, do nancial literacy assessment test? If you This question should be included in the basic financial literacy assessment test.	n is that i h little ris t saving f investm o you feel ou feel it s linclui questi Yes	it: is from s from ent as it should on? No	n taxatio sets IId be includ Categ Money Basics	n luded in or e ed, please s gory or Cate, addu Borrowing	excluded f select the f gories this resses Building Assets	rom the pool category or ca question Protecting Assets	of questions for the final ba ategories you feel it addres
The main advantage of a 401(k) plan Provides a high rate of return with Allows you to shelfter retirment Provides a well diversified mix of Don't know Prefer not to answer Upon reviewing the question above, do nancial literacy assessment test? If you This question should be included in the basic financial literacy assessment test.	n is that i h little ris t saving f investm o you feel ou feel it s Inclui Yes	it: sk is from eent as it should on? No	n taxatio sets IId be inc be includ Categ Money Basics	n luded in or e ed, please s gory or Cate addi Borrowing	excluded f eelect the o gories this esses Building Assets	rom the pool category or ca question Protecting Assets	of questions for the final ba ategories you feel it addres
The main advantage of a 401(k) plan Provides a high rate of return with Allows you to shelfter retirment Provides a well diversified mix of Don't know Prefer not to answer Ipon reviewing the question above, do nancial literacy assessment test? If you This question should be included in the basic financial literacy assessment test.	n is that i h little ris t saving f investm o you feel ou feel it s Inclui questi Yes	it: is from solution it should de on? No	n taxatio sets IId be includ Categ Money Basics	n luded in or e ed, pleases gory or Cate addi Borrowing	excluded f select the gories this resses Building Assets	rom the pool category or ca question Protecting Assets	of questions for the final ba ategories you feel it addres

Validation Panel Round 2, Group 1 Online Survey (continued)

	Include question?	Cateç	gory or Cate addi	gories this esses	question		
	Yes No	Money Basics	Borrowing	Building Assets	Protecting Assets		
This question should be included in the basic financial literacy assessment test.	0 0					-	
Comments:							
							,
n overdraft							
 occurs when you srite a \$1,000 c is a stop-payment order written b 	lollar check wl y the payee.	nen you h	ave \$500 ir	n your acc	count.		
c. will result in fines. d. all of the above.							
 c. will result in fines. d. all of the above. both A and C. Don't know 							
 c. will result in fines. d. all of the above. e both A and C. f. Don't know g. Prefer not to answer 							
 will result in fines. all of the above. both A and C. Don't know Prefer not to answer Jpon reviewing the question above, do inancial literacy assessment test? If you have a series of the se	o you feel it sho ou feel it should	uld be inc	luded in or e	excluded f	rom the pool o	of questions for the final b tegories you feel it addre	oasic
 will result in fines. all of the above. both A and C. Don't know Prefer not to answer Jpon reviewing the question above, do inancial literacy assessment test? If you 	o you feel it sho ou feel it should	uld be inc be includ	luded in or e led, please s	excluded f	rom the pool o category or ca	of questions for the final b tegories you feel it addre	oasic esses
 will result in fines. all of the above. both A and C. Don't know Prefer not to answer Jpon reviewing the question above, do inancial literacy assessment test? If you 	o you feel it sho ou feel it should Include	uld be inc be includ	luded in or e led, please s gory or Cate	excluded f select the	rom the pool of category or ca	of questions for the final b tegories you feel it addre	asic esses
 will result in fines. all of the above. both A and C. Don't know Prefer not to answer Jpon reviewing the question above, do inancial literacy assessment test? If you have a set of the s	o you feel it sho ou feel it should Include question?	uld be inc be includ Categ	luded in or e led, please s gory or Cates addi	excluded f select the gories this esses Building	rom the pool of category or ca question	of questions for the final b ategories you feel it addre	oasic esses
 will result in fines. all of the above. both A and C. Don't know Prefer not to answer Jpon reviewing the question above, do inancial literacy assessment test? If you have a substant test for the substant test in the substant test is a substant test in the substant test is a substant test in the substant test is a substant test in the substant test in the substant test is a substant test in the substant test is a substant test.	o you feel it sho bu feel it should Include question? Yes No	uld be inc be includ Categ Money Basics	luded in or e led, please s gory or Cate addi Borrowing	excluded f elect the gories this esses Building Assets	rom the pool of category or ca question Protecting Assets	of questions for the final b tegories you feel it addre	oasic esses
2. will result in fines. 3. all of the above. a both A and C. 5. Don't know g. Prefer not to answer Jpon reviewing the question above, do inancial literacy assessment test? If yo This question should be included in the basic financial literacy assessment test.	o you feel it sho bu feel it should Include question? Yes No	uld be includ be includ Categ Money Basics	luded in or e led, please s gory or Cate add Borrowing	excluded f select the gories this resses Building Assets	rom the pool of category or ca e question Protecting Assets	of questions for the final b tegories you feel it addre	pasic
2. will result in fines. 3. all of the above. a both A and C. 5. Don't know g. Prefer not to answer Jpon reviewing the question above, do inancial literacy assessment test? If yo This question should be included in the basic financial literacy assessment test. Comments:	o you feel it sho bu feel it should Include question? Yes No	uld be includ be includ Categ Money Basics	luded in or e led, please s gory or Cate add Borrowing	excluded f select the gories this resses Building Assets	rom the pool of category or ca question Protecting Assets	of questions for the final b tegories you feel it addre	oasic osses
 will result in fines. all of the above. both A and C. Don't know Prefer not to answer Jpon reviewing the question above, do inancial literacy assessment test? If you have a set of the basic financial literacy assessment test. This question should be included in the basic financial literacy assessment test. Comments:	o you feel it sho bu feel it should question? Yes No	uld be includ be includ Categ Money Basics	luded in or e led, pleases gory or Cate add Borrowing	excluded f select the esses Building Assets	rom the pool of category or ca e question Protecting Assets	of questions for the final b ategories you feel it addre	oasic esses
 will result in fines. all of the above. both A and C. Don't know Prefer not to answer Upon reviewing the question above, do inancial literacy assessment test? If yo This question should be included in the basic financial literacy assessment test. Comments:	o you feel it sho bu feel it should Include question? Yes No	uld be includ be includ Categ Money Basics	luded in or e led, pleases gory or Cate add Borrowing	excluded f select the resses Building Assets	rom the pool of category or ca o question Protecting Assets	of questions for the final b tegories you feel it addre	oasic ssses
 will result in fines. all of the above. both A and C. Don't know Prefer not to answer Jpon reviewing the question above, do inancial literacy assessment test? If yo This question should be included in the basic financial literacy assessment test. Comments:	byou feel it should u feel it should question? Yes No	uld be includ be includ Categ Money Basics	luded in or e led, pleases gory or Cate; addi Borrowing	excluded f select the esses Building Assets	rom the pool of category or ca e question Protecting Assets	of questions for the final b ategories you feel it addre	pasic
 will result in fines. all of the above. both A and C. Don't know Prefer not to answer Jpon reviewing the question above, do inancial literacy assessment test? If yo This question should be included in the basic financial literacy assessment test. Comments:	o you feel it sho bu feel it should Include question? Yes No	uld be includ be includ Categ Money Basics	luded in or e led, please s gory or Cate add Borrowing	excluded f select the esses Building Assets	rom the pool of category or ca question Protecting Assets	of questions for the final b ategories you feel it addre	pasic
 will result in fines. all of the above. both A and C. Don't know Prefer not to answer Jpon reviewing the question above, do inancial literacy assessment test? If yo This question should be included in the basic financial literacy assessment test. Comments:	o you feel it sho bu feel it should Include question? Yes No	uld be includ be includ Categ Money Basics	luded in or e led, please s gory or Cate, addi Borrowing	excluded f select the gories this esses Building Assets	rom the pool of category or ca e question Protecting Assets	of questions for the final b ategories you feel it addre	pasic posses
 will result in fines. all of the above. both A and C. Don't know Prefer not to answer Upon reviewing the question above, do inancial literacy assessment test? If you have a set of the basic financial literacy assessment test. This question should be included in the basic financial literacy assessment test. Comments:	o you feel it sho bu feel it should question? Yes No	uld be includ be includ Categ Money Basics	luded in or e led, pleases gory or Cate add Borrowing	excluded f select the esses Building Assets	rom the pool of category or ca a question Protecting Assets	of questions for the final b ategories you feel it addre	pasic passes

Validation Panel Round 2, Group 1 Online Survey (continued)

- e none of the above.
- f. Don't know
- g. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	Category or Categories this question addresses				
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets		
This question should be included in the basic financial literacy assessment test.	0	0						

Comments:

Which of the following statements is TRUE about the annual percentage rate (APR)?

- a. APR is the actual rate of interest paid over the life of the loan.
- b. APR is expressed as a percentage on an annual basis.
- c. APR is a good measure of comparing loan costs.
- d. APR takes into account all loan fees.
- e All of the above.
- f. Don't know
- g. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Category or Categories this question addresses				
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	0					

Comments:

Validation Panel Round 2, Group 1 Online Survey (continued)

You can receive your credit report from

- a. a credit union.
- b. a commercial bank.
- c. the Better Business Bureau.
- d. a credit bureau.
- e a professor.
- f. Don't know
- g. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Category or Categories this question addresses			
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets
This question should be included in the basic financial literacy assessment test.	0	\bigcirc				

Comments:

if you co-sign a loan for a friend, then

- a. you become responsible: for the loan payments if your friend defaults.
- b. it means that your friend cannot receive the loan by himself.
- c. you are entitled to receive part of the loan.
- d. both A and B.
- e Both A and C.
- f. Don't know
- g. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Include question?		Category or Categories this question addresses			
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets
This question should be included in the basic financial literacy assessment test.	0	\bigcirc				

Comments:

Validation Panel Round 2, Group 1 Online Survey (continued)

The earlier you start saving for retirement, the more money you will have because the effects of compounding interest increase over time.

- a. True
- b. False
- c. Don't know
- d. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	ory or Cate addr	gories this resses	question	
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	\bigcirc					

Comments:

You should have an emergency fund that covers two to six months of your expenses.

- a. True
- b. False
- c. Don't know

d. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	jory or Cate addi	gories this resses	question
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets
This question should be included in the basic financial literacy assessment test.	0	0				

Comments:

Validation Panel Round 2, Group 1 Online Survey (continued)

If you have a savings account at a bank, you may have to pay taxes on the interest you earn.

a. True

b. False

c. Don't know d. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	ory or Cates addr	gories this resses	question	
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	\bigcirc					

Comments:

If the interest rate on an adjustable-rate mortgage loan goes up, your monthly mortgage payments will also go up.

- a. True
- b. False
- c. Don't know
- d. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	ory or Cates addı	gories this esses	question	
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	\bigcirc					

Comments:

Validation Panel Round 2, Group 1 Online Survey (continued)

The finance charge on your credit card statement is what you pay to use credit.

a. True

- b. False
- c. Don't know
- d. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	jory or Categ addr	gories this esses	question
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets
This question should be included in the basic financial literacy assessment test.	0	\bigcirc				

Comments:

Your credit report includes employment data	, your payments history,	any inquiries made by creditors	, and any pubic record
information.			

a. True

b. False

c. Don't know

d. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	ory or Cate addr	gories this resses	question	
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	\bigcirc					

Comments:

Validation Panel Round 2, Group 1 Online Survey (continued)

If you expect to carry a balance on your credit card, the APR is the most important thing to look at when comparing credit card offers.

a. True

- b. False
- c. Don't know
- d. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	ory or Categ addr	gories this resses	question
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets
This question should be included in the basic financial literacy assessment test.	0	\bigcirc				

Comments:

Making payments late on your bills can make it more difficult to take out a loan.

- a. True
- b. False
- c. Don't know

d. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	ory or Categ addr	gories this resses	question
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets
This question should be included in the basic financial literacy assessment test.	0	\bigcirc	0	0	\bigcirc	\bigcirc

Comments:

Validation Panel Round 2, Group 1 Online Survey (continued)

If you buy certificates of deposit, savings bonds, or Treasury bills, you can earn higher returns than on a savings account, with little or no added risk.

a. True

- b. False
- c. Don't know
- d. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	ory or Categ addr	gories this resses	question	
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	\bigcirc					

Comments:

Which of the following types of investment would best protect the purchasing power of a family's savings in the event of a sudden increase in inflation?

- a. A 10-year bond issued by a corporation.
- b. A certificate of deposit at a bank.
- c. A twenty-five year corporate bond.
- d. A house financed with a fixed-rate mortgage.
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	ory or Cates addr	gories this esses	question
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets
This question should be included in the basic financial literacy assessment test.	0	\bigcirc				

Validation Panel Round 2, Group 1 Online Survey (continued)

hich of the following statements is	NOT cor	rect a	about mo	st ATM (Au	tomated	Teller Machi	ne) cards?	
You can generally get each 24 h		,	is out mo		tomatou.		no) culuci	
You can generally obten in forma You can get cash anywhere in You must have a bank account to Don't know Prefer not to answer	tion conce the world o have an	eming d with a ATN	g your ba n no fee. 1 card.	nk balance	at an AT	M machine.		
oon reviewing the question above, do ancial literacy assessment test? If yo	o you feel i ou feel it sł	it shou hould	uld be inc be includ	luded in or e ed, please s	excluded f	from the pool category or cate	of questions for the fin ategories you feel it ac	al basic
, , , , ,	Includ	de	Categ	ory or Categ	gories this	question	<u> </u>	
	Yes 1	No	Money Basics	Borrowing	Building Assets	Protecting Assets		
his question should be included in		_	_	_			-	
ssessment test.	0	0						
ssessment test.								
ssessment test.		0						
ssessment test.								
omments:	0	0			J			
omments:	0				J			
imments:	0				J			
imments:	0							
omments:								
ebecca has saved \$12,000 for her	college e	xpens	ses by w	prking part-	ime. Her	plan is to st	art college next year	and sh
ebecca has saved \$12,000 for her beds all of the money she saved. V	college e.	xpens he fo	ses by we	orking part-the safest	ime. Her place for	plan is to st her college	art college next year money?	and sh
ebecca has saved \$12,000 for her seds all of the money she saved. V Locked in her closet at home. Stocks.	college e; Vhich of t	xpens he fo	ses by we	orking part-t the safest	ime. Her place for	plan is to st her college	art college next year money?	and sh
ebecca has saved \$12,000 for her seds all of the money she saved. V Locked in her closet at home. Stocks. Corporate bonds.	college e. Which of t	xpens he fo	ses by we	orking part-i the safest	ime. Her place for	plan is to st her college	art college next year money?	and sh
ebecca has saved \$12,000 for her beeds all of the money she saved. V Locked in her closet at home. Stocks. Corporate bonds. A bank savings account. Don't know	college ex	xpens he fo	ses by we	orking part-t the safest	ime. Her place for	plan is to st her college	art college next year money?	and sh
ebecca has saved \$12,000 for her seds all of the money she saved. V Locked in her closet at home. Stocks. Corporate bonds. A bank savings account. Don't know Prefer not to answer	college ex	xpens he fo	ses by we	orking part-the safest	ime. Her place for	r plan is to st r her college i	art college next year money?	and sh
ebecca has saved \$12,000 for her beeds all of the money she saved. V Locked in her closet at home. Stocks. Corporate bonds. A bank savings account. Don't know Prefer not to answer	college e. Which of t	xpens he fou	ses by we llowing is	orking part-i the safest	ime. Her place for	plan is to st her college	art college next year money? of questions for the fin	and sh
ebecca has saved \$12,000 for her seeds all of the money she saved. V Locked in her closet at home. Stocks. Corporate bonds. A bank savings account. Don't know Prefer not to answer pon reviewing the question above, do nancial literacy assessment test? If yo	College e Which of t	xpens he for hould	uld be includ Categ	orking part-i the safest	time. Her place for excluded f	r plan is to st her college from the pool category or ca s question	art college next year money? of questions for the fin ategories you feel it ac	and sh al basic
ebecca has saved \$12,000 for her seds all of the money she saved. V Locked in her closet at home. Stocks. Corporate bonds. A bank savings account. Don't know Prefer not to answer soon reviewing the question above, do iancial literacy assessment test? If yo	college ex Which of t by you feel i lnclud questic Yes 1	xpens he for hould de No	ses by we llowing is uld be includ Categ Money Basics	orking part-i the safest luded in or e ed, please s ory or Cate addr Borrowing	ime. Her place for excluded f elect the gories this esses Building Ascete	r plan is to st her college from the pool category or ca s question Protecting	art college next year money? of questions for the fin ategories you feel it ac	and sh

Validation Panel Round 2, Group 1 Online Survey (continued)

Scott and Eric are young men. Each he same salary. Scott has borrowed kely to pay the lowest finance charg	n has a good c d \$6,000 to tak ge?	redit history. They v ke a foreign vacatior	vork at the n. Eric has	e same com s borrowed \$	pany and make approximat 66,000 to buy a car. Who is
Eric will pay less because the o	car is collater	al for the loan.			
. They will both pay the same beca	ause the rate i	s set by law.			
. Scott will pay less because peop . They will both pay the same beca	le who travel o ause they hav	overseas are better . e almost identical fil	nsks. nanciak k	backgrounds	
. Don't know	,			U	
Prefer not to answer					
pon reviewing the question above, do	you feel it sho	uld be included in or	excluded t	from the pool	of questions for the final bas
		Category or Cate	agories this	s question	alegones you leer it address
	question?	add	lresses	- 4	
	Yes No	Money Basics Borrowing	Building Assets	Protecting Assets	
This question should be included in		Dublob	/ 100010	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	
he basic financial literacy	\circ \circ				
the basic financial literacy assessment test.	0 0				
the basic financial literacy assessment test.	0 0				
the basic financial literacy assessment test. Comments:	0 0				
the basic financial literacy assessment test. Comments:	0 0				
the basic financial literacy assessment test. 'omments:	0 0				
the basic financial literacy assessment test. comments:	0 0				
he basic financial literacy assessment test. omments:	0 0				
the basic financial literacy assessment test. omments:	0 0				
the basic financial literacy assessment test. omments:	0 0				
the basic financial literacy assessment test. omments:	0 0				
the basic financial literacy assessment test. fomments: Vhich of the following statements is	true?				
the basic financial literacy assessment test. comments: Which of the following statements is Banks and other lenders share	true?	story of their borro	wers with	each other	r and are likely to know o
the basic financial literacy assessment test. comments: Which of the following statements is Banks and other lenders share oan payments that you have miss	true? the credit his	story of their borro	wers with	n each other	r and are likely to know o
the basic financial literacy assessment test. fomments: Which of the following statements is Banks and other lenders share ban payments that you have miss People have so many loans it is Your bad loan payment cord wi	true? the credit his sed. very unlikely t	story of their borro	wers with	n each other	r and are likely to know o another bank. er bank for a loan
he basic financial literacy assessment test. omments: Vhich of the following statements is . Banks and other lenders share oan payments that you have miss . People have so many loans it is . Your bad loan payment record wi . If you missed a payment more th	true? the credit his sed. very unlikely t th one bank w tan 2 years ag	story of their borro hat one bank will kn ill not be considered o, it cannot be cons	wers with ow your h l if you ap idered in	n each other history with a ply to anoth a loan decis.	r and are likely to know o another bank. er bank for a loan. ion.
he basic financial literacy assessment test. omments: /hich of the following statements is . Banks and other lenders share ban payments that you have miss . People have so many loans it is . Your bad loan payment record wi . If you missed a payment more th . Don't know	true? the credit his sed. very unlikely t th one bank w ban 2 years ag	story of their borro hat one bank will kn ill not be considered o, it cannot be cons	wers with now your h l if you ap idered in	n each other history with a loan decis	r and are likely to know o another bank. er bank for a loan. ion.
he basic financial literacy assessment test. omments: /hich of the following statements is Banks and other lenders share ban payments that you have miss People have so many loans it is Your bad loan payment record wil If you missed a payment more th Don't know Prefer not to answer	true? the credit his sed. very unlikely t th one bank w an 2 years ag	story of their borro hat one bank will kn ill not be considered o, it cannot be cons	wers with how your h l if you ap idered in	n each other history with a holy to anoth a loan decis	r and are likely to know o another bank. er bank for a loan. ion.
he basic financial literacy assessment test. omments: Which of the following statements is Banks and other lenders share ban payments that you have miss People have so many loans it is Your bad loan payment record with Sour bad loan payment more the Don't know Prefer not to answer pon reviewing the question above, do nancial literacy assessment test? If you	true? the credit his sed. very unlikely t th one bank w ian 2 years ag	story of their borro hat one bank will kn ill not be considered o, it cannot be cons uld be included in or l be included, please	wers with the your h i f you ap idered in excluded select the	n each other history with a history with a loan decise from the pool category or c	r and are likely to know o another bank. er bank for a loan. ion. of questions for the final bas ategories you feel it address
he basic financial literacy assessment test. omments: Vhich of the following statements is Banks and other lenders share ban payments that you have miss People have so many loans it is : Your bad loan payment record wi If you missed a payment more th Don't know Prefer not to answer pon reviewing the question above, do nancial literacy assessment test? If you	true? the credit his sed. very unlikely t th one bank w ban 2 years ag o you feel it should lnclude question?	story of their borro hat one bank will kn ill not be considered o, it cannot be cons uld be included in or be included, please Category or Cate add	wers with ow your h if you ap idered in select the excluded select the resses	n each other history with a ply to anoth a loan decis. from the pool category or c s question	r and are likely to know o another bank. er bank for a loan. ion. of questions for the final bas categories you feel it address
the basic financial literacy assessment test. Comments: Which of the following statements is Banks and other lenders share oan payments that you have miss People have so many loans it is Your bad loan payment record wi I If you missed a payment more th Don't know Prefer not to answer Upon reviewing the question above, do nancial literacy assessment test? If you	true? the credit his sed. very unlikely t th one bank w an 2 years ag o you feel it should u feel it should Include question? Yes No	story of their borro hat one bank will kn ill not be considered o, it cannot be consi uld be included in or l be included, please Category or Cate add Money Basics Borrowing	wers with the wyour h i f you ap idered in excluded select the egories this resses Building Assets	a each other history with a history with a loan decise from the pool category or c s question Protecting Assets	r and are likely to know o another bank. er bank for a loan. ion. of questions for the final bas rategories you feel it address

Validation Panel Round 2, Group 1 Online Survey (continued)

Comments:					
/hich of the following credit card use I charge the same amount per year	ers is likely to on their carc	o pay the GREATES	r dollar ar	mount in finan	ce charges per year, if the
Jessica, who pays at least the min Vera, who generally pays off her co Megan, who always pays off her crea Erin, who only pays the minime Don't know Prefer not to answer	nimum amou credit card in dit card bill in Im amount	nt each month and m full but, occasionally full shortly after she re each month.	ore, wher will pay ceives it.	n she has the the minimum	money. when she is short of cash
pon reviewing the question above, do nancial literacy assessment test? If you	you feel it sho u feel it shoul	ould be included in or e d be included, please s	excluded fi elect the o	rom the pool of category or cat	fquestions for the final basic egories you feel it addresse
	Include question?	Category or Category addr	Category or Categories this question addresses		
	Yes No	Money Basics Borrowing	Building Assets	Protecting Assets	
his question should be included in ne basic financial literacy ssessment test.	0 0				
omments:					
o you think buying stock in a single	company is	safer than buving sto	in sev	eral different	companies?
Yes.					
No					
Dep# know					
Don't know Prefer not to answer					
Don't know Prefer not to answer pon reviewing the question above, do nancial literacy assessment test? If you	you feel it sho u feel it shoul	ould be included in or e d be included, please s	excluded fi	rom the pool of category or cat	f questions for the final basic egories you feel it addresse
Don't know Prefer not to answer pon reviewing the question above, do nancial literacy assessment test? If you	you feel it sho u feel it shoul Include question?	ould be included in or e d be included, please s Category or Categ addr	excluded fi elect the o gories this esses	rom the pool of category or cat question	f questions for the final basic egories you feel it addresses
. No. . Don't know . Prefer not to answer lpon reviewing the question above, do nancial literacy assessment test? If you	you feel it sho u feel it shoul Include question? Yes No	ould be included in or e d be included, please s Category or Categ addr Money Basics Borrowing	excluded fi elect the o gories this esses Building Assets	rom the pool of category or cat question Protecting Assets	f questions for the final basic egories you feel it addresses

Validation Panel Round 2, Group 1 Online Survey (continued)

Comments:
If-assessed Financial Literacy Questions
Financial Literacy Self-Assessment Questions:
Reviewing the following financial literacy self-assessment questions, do you feel each should be included in or excluded (identifying reason) from the self-assessment portion of the final basic financial literacy assessment test?
I am pretty good at math.
On a scale of 1 to 7, with 1 meaning Strongly Disagree and 7 meaning Strongly Agree, how do you rate yourself with regard to this statement?
 Strongly Disagree Disagree Slightly Disagree Neither Disagree nor Agree Slightly Agree Agree Agree Strongly Agree
1 2 Neutral 4 5
Language: Very Unclear
Wording: Poor
Comments:
h
I am good at dealing with day-to-day financial matters, such as checking accounts, credit and debit cards, mortgages, installment payments, and tracking expenses.
On a scale of 1 to 7, with 1 meaning Strongly Disagree and 7 meaning Strongly Agree, how do you rate yourself with regard to this statement?
1. Strongly Disagree 2. Disagree
Validation Panel Round 2, Group 1 Online Survey (continued)

 Slightly Disagree Neither Disagree noi Slightly Agree Agree Agree Strongly Agree 	r Agree				
		12	Neutral	4 5	
	In the second se	Very Clear Excellent			
Comments:					
					1
On a scale of 1 to 7, w to this statement? 1. Strongly Disagree 2. Disagree 3. Slightly Disagree 4. Neither Disagree noi 5. Slightly Agree 6. Agree 7. Strongly Agree	ith 1 meaning Strongly Di: r Agree	sagree ar	nd 7 mea	ning Stro	ongly Agree, how do you rate yourself with regard
		1 2	Neutral	4 5	
	Language: Very Unclear Wording: Poor			0	Very Clear Excellent
Comments:					
					4
I am good at managing crea	dit oande and dobt like oan loan				

Validation Panel Round 2, Group 1 Online Survey (continued)

 Strongly Disagree Disagree Slightly Disagree Neither Disagree nor Agree Slightly Agree Agree Agree Strongly Agree 		
	1 2 Neutral 4 5	Very Clear
Wording: Poor	0 0 0 0 0	Excellent
Comments:		
I understand investment products (stocks, bo annuities). On a scale of 1 to 7, with 1 meaning Strongly Di to this statement? 1. Strongly Disagree 2. Disagree	onds and mutual funds) a isagree and 7 meaning Stro	and retirement accountants (IRAs, 401Ks and ongly Agree, how do you rate yourself with regard
 Slightly Disagree Neither Disagree nor Agree Slightly Agree Slightly Agree Agree Strongly Agree 		
Language: Ven/ Unclear	1 2 Neutral 4 5	Very Clear
Wording: Poor	0 0 0 0 0	Excellent
	1	

Validation Panel Round 2, Group 1 Online Survey (continued)

1 Strongly Disagree 2 Disagree 3 Slightly Disagree or Agree 5 Slightly Agree 6 Agree 7 Strongly Agree 1 2 Neutral 4 5 Language: Very Unclear Wording: Poor 0 0 0 0 0 0 Very Clear Excellent Comments: On a scale of 1 to 7, with 1 meaning Very Low and 7 meaning Very High, how would you assess your overall finance knowledge? 1 Very Low 2 Low 3 Slightly Low 4 Average 5 Slightly Low 4 Average 1 2 Neutral 4 5 Language: Very Unclear 1 2 Neutral 4 5 Language: Very Unclear 1 2 Neutral 4 5 Very Low 2 Low 3 Slightly Low 4 Average 5 Slightly High 6 High 7. Very High 1 2 Neutral 4 5 Language: Very Unclear Wording: Poor 1 2 Neutral 4 5 Language: Very Unclear Wording: Poor 1 2 Neutral 4 5 Language: Very Unclear Wording: Poor 1 Very Clear Excellent Excellent	financial	rree ee nor Agree Agree Language: Very Unclear Wording: Poor	2 Neutral 4 5 Image: Second system Image: Second system Very Clear Image: Second system Image: Second system Excellent	
6. Agree 7. Strongly Agree 1 2 Neutral 4 5 Language: Very Unclear Wording: Poor Very Clear Excellent Comments: Image: Very Low and 7 meaning Very High, how would you assess your overall finance knowledge? 1 Very Low 2. Low Slightly Low 4. Average Slightly High 7. Very High 1 2. Language: Very Unclear Wording: Poor Very Clear 1 2. Neutral 4 5. Slightly High Very Unclear Wording: Poor Very Clear	I financial	e Language: Very Unclear Wording: Poor	2 Neutral 4 5 Very Clear Excellent	
1 2 Neutral 4 5 Language: Very Unclear Wording: Poor 0 0 0 0 Excellent Comments: On a scale of 1 to 7, with 1 meaning Very Low and 7 meaning Very High, how would you assess your overall finance knowledge? 1. Very Low 2. Slightly Low 3. Slightly Low 3. Slightly Low 4. Average 3. Slightly High 3. Slightly High 3. Slightly High 3. Slightly Low 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 0 1 0 </th <th>I financial</th> <th>Language: Very Unclear Wording: Poor</th> <th>2 Neutral 4 5 Very Clear Excellent</th> <th></th>	I financial	Language: Very Unclear Wording: Poor	2 Neutral 4 5 Very Clear Excellent	
Language: Very Unclear Wording: Poor Comments: On a scale of 1 to 7, with 1 meaning Very Low and 7 meaning Very High, how would you assess your overall finance knowledge? 1. Very Low 2. Low 3. Slightly Low 4. Average 5. Slightly High 6. High 7. Very High 1. 2. Neutral 4. 5 Language: Very Unclear Wording: Poor 0. 0. 0. 0. Very Clear Excellent	l financial	Language: Very Unclear Wording: Poor	Very Clear Excellent	
Wording: Poor Excellent Comments: On a scale of 1 to 7, with 1 meaning Very Low and 7 meaning Very High, how would you assess your overall finance knowledge? 1. Very Low 2. Low 3. Slightly Low 4. Average 5. Slightly High 6. High 7. Very High 1 2. Neutral 4 5 Language: Very Unclear Wording: Poor 0 0 0 0 0 0 0 0 0 0 1 2. Neutral 4 5	l financial	Wording: Poor	Excellent	
Comments: On a scale of 1 to 7, with 1 meaning Very Low and 7 meaning Very High, how would you assess your overall financ knowledge? 1. Very Low 2. Low 3. Slightly Low 4. Average 5. Slightly High 6. High 7. Very High 1 2 Neutral 4 5 Language: Very Unclear Wording: Poor	l financial			
On a scale of 1 to 7, with 1 meaning Very Low and 7 meaning Very High, how would you assess your overall financ knowledge? 1. Very Low 2. Low 3. Slightly Low 4. Average 5. Slightly High 6. High 7. Very High 1 2 Neutral 4 5 Language: Very Unclear Wording: Poor	l financial			
On a scale of 1 to 7, with 1 meaning Very Low and 7 meaning Very High, how would you assess your overall financ knowledge? 1. Very Low 2. Low 3. Slightly Low 4. Average 5. Slightly High 6. High 7. Very High 1 2 Neutral 4 5 Language: Very Unclear Wording: Poor	l financial			
On a scale of 1 to 7, with 1 meaning Very Low and 7 meaning Very High, how would you assess your overall financ knowledge? 1. Very Low 2. Low 3. Slightly Low 4. Average 5. Slightly High 6. High 7. Very High 1 2 Neutral 4 5 Language: Very Unclear Wording: Poor	l financial			
On a scale of 1 to 7, with 1 meaning Very Low and 7 meaning Very High, how would you assess your overall financ knowledge? 1. Very Low 3. Slightly Low 4. Average 5. Slightly High 6. High 7. Very High 1 2 Neutral 4 5 Language: Very Unclear Wording: Poor	l financial			
On a scale of 1 to 7, with 1 meaning Very Low and 7 meaning Very High, how would you assess your overall financ knowledge? 1. Very Low 2. Low 3. Slightly Low 4. Average 5. Slightly High 6. High 7. Very High 1 2 Neutral 4 5 Language: Very Unclear Wording: Poor	l financial			
On a scale of 1 to 7, with 1 meaning Very Low and 7 meaning Very High, how would you assess your overall financ knowledge? 1. Very Low 2. Low 3. Slightly Low 4. Average 5. Slightly High 6. High 7. Very High 1 2 Neutral 4 5 Language: Very Unclear Wording: Poor 0 0 Very Clear Excellent	l financial			
1. Very Low 2. Low 3. Slightly Low 4. Average 5. Slightly High 6. High 7. Very High 1 2 Neutral 4 5 Language: Very Unclear Wording: Poor		7, with 1 meaning Very Low and 7 m	ning Very High, how would you assess your o	verall financial
2. Low 2. Low 3. Slightly Low 4. Average 5. Slightly High 6. High 7. Very High 1 2 Neutral 4 5 Language: Very Unclear Wording: Poor				
1 2 Neutral 4 5 Slightly High 7. Very High Language: Very Unclear Wording: Poor				
1 2 Neutral 4 5 Language: Very Unclear Wording: Poor				
1 2 Neutral 4 5 Language: Very Unclear Image: Very Unclear Wording: Poor Image: Very Unclear Image: Very Unclear Image: Very Unclear Image: Very Unclear Wording: Poor Image: Very Unclear Image: Very Unclear Image: Very Unclear Image: Very Unclear Wording: Poor Image: Very Unclear Image: Very Unclear Image: Very Unclear Image: Very Unclear				
1 2 Neutral 4 5 Language: Very Unclear Image: Very Unclear Image: Very Unclear Image: Very Unclear Image: Very Unclear Wording: Poor Image: Very Unclear Image: Very Unclear Image: Very Unclear Image: Very Unclear				
Language: Very Unclear Image: Very Clear Wording: Poor Image: Very Clear Excellent			2 Neutral 4 5	
Wording: Poor		Language: Very Unclear	Very Clear	
		Wording: Poor	Excellent	

Validation Panel Round 2, Group 2 Online Survey

Hello Panel Expert,

Thank you for completing Expert Panel Round 1 and participating in this valuable financial literacy research.

In this second round of review, you will be evaluating 32 financial literacy knowledge questions in the pool for the basic financial literacy instrument and seven financial literacy self-assessment questions. The estimated time to complete the review is 30 to 45 minutes. Please complete your review as soon as possible. Your input will determine the questions included in future rounds of review by other members of the expert panel. For your convenience, survey allows you to stop at any point. It will automatically save and later return to the place you stopped. If you have any questions or need assistance, please feel free to contact me at <u>pjdahmen@mail.usf.com</u> or <u>941-962-1211</u>.

INSTRUCTIONS

As you read each financial literacy question and possible answers (correct answer is bold), identify whether you believe the question should be included or excluded from an instrument used to determine a basic financial literacy score. The goal of this round is to reduce the number of questions in the pool. You are reviewing 50% of the pool, therefore, please feel free to be critical in your selection of the questions to include. Only included question you strongly feel should be included; all other questions should be excluded.

If you determine the question should be included, please identify the financial literacy content category or categories you feel it addresses. The categories for content areas are:

Financial Literacy Category	Topics Included in Category
Money basics	Compound interest, Time value of money, purchasing power/inflation, spending, short-term savings
Borrowing	APR, finance charges, loan terms, credit scores; type of borrowing (e.g., education, home, vehicles)
Building assets	Long-term-savings accounts/Certificates of deposit, stocks, bonds, mutual funds, annuities
Protecting assets	Investment product risk (stocks vs. bonds vs. mutual funds), investment diversity, insurance protection

Following the financial literacy knowledge questions are seven financial literacy self-assessment questions. Please read each question and evaluate if:

1. it has appropriate for language (whether the question is simple, clear and easy to understand),

2. the wording is adequate (whether the question contains wording that is relevant and free from bias, in other words, it portrays a neutral viewpoint). Evaluation is on the following scale:

	Very Unclear	Unclear	Neutral	Clear	Very Clear
Language	1	2	3	4	5
	Poor	Needs Improvement	Neutral	Good	Excellent
Wording	1	2	3	4	5

Please feel free to comment on any of the question included in this round.

Validation Panel Round 2, Group 2 Online Survey (continued)

Suppose that in the year 2015, your income has doubled and prices of all goods have doubled too. In 2015, how much will you be able to buy with your income?

- a. More than today
- b. The same
- c. Less than today
- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	ory or Cate addi	gories this resses	question
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets
This question should be included in the basic financial literacy assessment test.	0	\bigcirc				

Comments:

If you were to invest \$1,000 in a stock mutual fund, it would be possible to have less than \$1,000 when you withdraw your money.

- a. True
- b. False
- c. Don't know
- d. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	Include question?		Category or Categories t addresses			
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	\bigcirc					

Comments:

Validation Panel Round 2, Group 2 Online Survey (continued)

High inflation means that the cost of living is increasing rapidly.

- a. True
- b. False
- c. Don't know
- d. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	ory or Cate addr	gories this esses	question
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets
This question should be included in the basic financial literacy assessment test.	0	0				

Comments:

It is usually possible to reduce the risk of investing in the stock market by buying a wide range of stocks and shares.

- a. True
- b. False
- c. Don't know
- d. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	ory or Cates addr	gories this esses	question
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets
This question should be included in the basic financial literacy assessment test.	0	0				

Comments:

Validation Panel Round 2, Group 2 Online Survey (continued)

The figure that best reflects a mutual fund's performance over a period of years is

a. its current yield.

- b. the total of dividends and capital gains it has paid.
- c. its total return.
- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	ory or Cate add	gories this resses	question		
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets		
This question should be included in the basic financial literacy assessment test.	0	0						
Comments:								
f your assets increase by \$5,000 ar	nd your	liabiliti	es decrea	se by \$3,0	00, your i	net worth wou	ıld	
a. Increase by \$2,000 b. Increase by \$8,000 c. Increase by \$3,000								

- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	ory or Cates addr	gories this resses	question	
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	0					

Validation Panel Round 2, Group 2 Online Survey (continued)

To ensure that some of your retirement savings will not be subject to income tax upon withdrawal, you would contribute to:

- a. A Traditional IRS or Individual Retirment Account
- b. A Roth IRA
- c. A 401(k) plan
- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	gory or Cate add	gories this resses	question		
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets		
This question should be included in the basic financial literacy assessment test.	0	0					m ;	
Comments:								
To reduce the total finance costs participation of the total finance costs and the tot	id over	the life	e of an au	to loan, yoι	i should d	hoose a loai	n with the	

- a. Lowest monthly payment
- b. Longest repayment term
- c. Shortest repayment
- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Include question?		Category or Categories this question addresses			
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets
This question should be included in the basic financial literacy assessment test.	0	0				

Validation Panel Round 2, Group 2 Online Survey (continued)

If you always pay the full balance on your credit card, which of the following is least important?

a. Annual interest rate

- b. Annual fee
- c. Line of credit
- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Include question?		Category or Categories this question addresses			
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets
This question should be included in the basic financial literacy assessment test.	0	0				

Comments:

Which type of mortgage would allow a first-time homebuyer to qualify for the highest loan amount?

- a. Fixed-rate mortgage
- b. Adjustable-rate mortgage
- c. Reverse mortgage
- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Include question?		Category or Categories this question addresses				
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	\bigcirc					

Validation Panel Round 2, Group 2 Online Survey (continued)

If you signed a twelve-month lease for \$300 month but never occupied the apartment, you legally owe the landlord

- a. your security deposit.
- b. your first month's rent of \$300.
- c. your twelve month's rent of \$3.600.
- d. nothing.
- e whatever the landlord requires.
- f. Don't know
- g. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Include question?	Category or Categories this question addresses
	Yes No	Money Borrowing Building Protecting Assets Assets
This question should be included in the basic financial literacy assessment test.	0 0	
Comments:		
The most liquid asset is		
 a. money in a certificate of deposit. b. money in a checking account. c. a car. d. an investment account. e a house (real estate). f. Don't know g. Prefer not to answer 		
Upon reviewing the question above, or basic financial literacy assessment test addresses.	do you feel it ? If you feel	should be included in or excluded from the pool of questions for the final it should be included, please select the category or categories you feel it
	Include question?	Category or Categories this question addresses

addresses

Validation Panel Round 2, Group 2 Online Survey (continued)

This question should be included in the basic financial literacy		No	Money Basics	Borrowing	Building Assets	Protecting Assets	
assessinent lest.	0	0	0	0	0	0	
Comments:							
Which of the following investments	requires	that y	ou keep y	our money	invested	for a specifie	ed period or face an early
vithdrawal penalty?							
 Checking account that pays integet Government savings bond. 	rest.						
 Money market mutual fund. Passbook savings account. 							
. Don't know g. Prefer not to answer							
Jpon reviewing the question above,	do you	feel it	should be	included in	n or exclu	ded from the	pool of questions for the fi
asic financial literacy assessment tes addresses.	t? If yo	ou feel	it should t	be included	, please s	elect the cate	egory or categories you fee
	Incl	lude	Categ	ory or Cate	gories this	question	
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	•	0					
Comments:							
With compound interest, you eam ir	nterest c	on your	r interest,	as well as	on your p	rincipal.	
With compound interest, you earn ir 1 . True	nterest c	on your	r interest,	as well as	on your p	rincipal.	
<i>With compound interest, you eam ir</i> 1. True 1. False 2. Don't know 1. Prefer not to answer	nterest c	on your	r interest,	as well as	on your p	rincipal.	

Validation Panel Round 2, Group 2 Online Survey (continued)

addresses.									
	incl ques	ude tion?	e Category or Categories this question n? addresses						
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets			
This question should be included in the basic financial literacy assessment test.	0	0							
Comments:									
									h
Mutual funds pay a guaranteed rate	of returi	η.							
a. True b. Faise c. Don't know d. Prefer not to answer									
Upon reviewing the question above, basic financial literacy assessment test addresses.	do you ? If yo	feel it s u feel it	should be t should b	included ir be included	or exclu please s	ded from the elect the cat	pool of que egory or ca	stions for t tegories yo	he final u feel if
	Incl ques	ude tion?	Categ	jory or Cate addi	gories this resses	question			
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets			
This question should be included in the basic financial literacy assessment test.	0	0							
Comments:									
									11
Repeatedly refinancing your home rr your debt.	ortgage	e over a	a short pe	riod of time	e results i	n added fee a	and points t	hat further	increas
a. True b. False c. Don't know d. Prefer not to answer									
Upon reviewing the question above, basic financial literacy assessment test addresses.	do you ? If yo	feel it s u feel it	should be t should b	included ir be included	or exclu please s	ded from the elect the cat	pool of que egory or ca	stions for t tegories yo	he fina u feel i
	Incl	ude	Categ	ory or Cate	gories this	question			

Validation Panel Round 2, Group 2 Online Survey (continued)

	question?		addi	esses		
	Yes No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.		0	0		0	
Comments:						
						,
						//
Using extra money in a bank savings	s account to p	ay off hig	h interest r	ate credit	card debt is	a good idea.
a. True b. False c. Don't know						
d. Prefer not to answer						
Upon reviewing the question above, obscience of the past of the pa	do you feel it s ? If you feel i	should be t should l	included ir	n or exclu , please s	ded from the select the cat	pool of questions for the fina egory or categories you feel i
	Include question?	Categ	gory or Cate addi	gories this esses	question	
	Yes No	Money Basics	Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0 0					
Comments						
Comments.						
						,
						″
Creditors are required to tell you the	APR that you	will pay v	when you g	et a loan.		
a. True b. False c. Don't know d. Prefer not to answer						
Upon reviewing the question above, obside the set basic financial literacy assessment test addresses.	do you feel it : ? If you feel i	should be t should l	e included ir be included	n or exclu , please s	ded from the select the cat	pool of questions for the fina egory or categories you feel i
	Include question?	Categ	gory or Cate addi	gories this esses	question	

Validation Panel Round 2, Group 2 Online Survey (continued)

	Yes No	Money Basics Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0 0				
Comments:					
Nour andit ration is not affected by			dit oordo		
Your creat rating is not affected by f a. True b False	iow much you	r charge on your cre	an caras.		
2. Don't know d. Prefer not to answer					
Upon reviewing the question above, basic financial literacy assessment test addresses.	do you feel it s ? If you feel i	should be included t should be include	in or exclu d, please s	ded from the select the cat	pool of questions for the fina egory or categories you feel
	Include question?	Category or Cat ad	egories this dresses	question	
	Yes No	Money Basics Borrowin	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0 0				
Comments:					
Nour andit rating is not affected by	ow much you	obarra an vaur ar	dit oordo		
a. True	iow mach you	charge on your cre	fuit carus.		
b. False c. Don't know d. Prefer not to answer					
Upon reviewing the question above, basic financial literacy assessment test addresses.	do you feel it s ? If you feel i	should be included t should be include	in or exclu d, please s	ded from the select the cat	pool of questions for the fina egory or categories you feel
	Include question?	Category or Cat	egories this dresses	question	
	,	Money	Building	Protecting	

Validation Panel Round 2, Group 2 Online Survey (continued)

		NO	Basics	Borrowing	Assets	Assets		
This question should be included in the basic financial literacy assessment test.	0	0						
Comments:								
Taxes affect how you should invest	vour ma	onev.						
a. True	,	,, .						
). ⊢aise 2. Don't know d. Prefer not to answer								
ALTERNATIVE WORDING								
Taxes do not affect how you should	invest y	our m	oney.					
n. True). False). Don't know 1. Prefer not to answer								
	douou	feel it s	abould be	the standard for		dod from the	ool of ques	tions for the fir
Upon reviewing the question above, basic financial literacy assessment test addresses.	? If yo	u feel i	t should be	e included in de included) or exclu , please s	select the cate	egory or cat	egories you fee
Upon reviewing the question above, basic financial literacy assessment test addresses.	If yo	u feel i ude tion?	t should be	ory or Cate add	n or exclu , please s gories this resses	select the cate	egory or cat	egories you fee
Upon reviewing the question above, basic financial literacy assessment test addresses.	? If yo Incl ques Yes	ude tion?	Categ Money Basics	gory or Cate addi	n or exclu , please s gories this resses Building Assets	elect the cate question Protecting Assets	egory or cat	egories you fee
Upon reviewing the question above, basic financial literacy assessment test addresses. This question should be included in the basic financial literacy assessment test.	? If yo Incl ques Yes	ude tion? No	Money Basics	ancluded in oe included gory or Cate addi Borrowing	n or exclu , please s gories this resses Building Assets	elect the cate	egory or cat	egories you fee
Upon reviewing the question above, basic financial literacy assessment test addresses. This question should be included in the basic financial literacy assessment test.	? If yo Incl ques Yes	ude tion? No	Money Basics	gory or Cate addr Borrowing	gories this resses Building Assets	Protecting Assets	gory or cat	egories you fee
Upon reviewing the question above, basic financial literacy assessment test addresses. This question should be included in the basic financial literacy assessment test.	? If yo Incl ques Yes	ude tion? No	Money Basics	gory or Cate add	n or exclu , please s gories this resses Building Assets	Protecting Assets	gory or cat	egories you fee
Upon reviewing the question above, basic financial literacy assessment test addresses. This question should be included in the basic financial literacy assessment test. Comments:	? If yo Incl ques Yes	u feel i ude tion? No	Money Basics	gory or Cate add	n or exclu , please s gories this resses Building Assets	Protecting Assets	gory or cat	egories you fee
Upon reviewing the question above, basic financial literacy assessment test addresses. This question should be included in the basic financial literacy assessment test. Comments:	? If yo Incl ques Yes	u feel i ude tion? No	Money Basics	gory or Cate add	n or exclu , please s gories this resses Building Assets	Protecting Assets	gory or cat	egories you fee
Upon reviewing the question above, i basic financial literacy assessment test addresses. This question should be included in the basic financial literacy assessment test. Comments:	Yes	u feel i ude tion? No	Money Basics	a included in poe included gory or Cate add Borrowing	n or exclu , please s gories this resses Building Assets	Protecting Assets	gory or cat	egories you fee
Upon reviewing the question above, i basic financial literacy assessment test addresses. This question should be included in the basic financial literacy assessment test. Comments:	Yes	u feel i ude tion? No	Money Basics	ancluded in per included gory or Cate add Borrowing	or exclu please s gories this resses Building Assets	Protecting Assets	gory or cat	egories you fee
Upon reviewing the question above, i basic financial literacy assessment test addresses. This question should be included in the basic financial literacy assessment test. Comments: Which of the following instruments is a. Debit card.	 ? If yo Incl ques Yes • NOT t 	u feel i ude tion? No	y associa	ancluded in poe included gory or Cate add Borrowing	ending?	Protecting Assets	gory or cat	egories you fee
Upon reviewing the question above, ibasic financial literacy assessment test addresses. This question should be included in the basic financial literacy assessment test. Comments: Which of the following instruments is a. Debit card. b. Certificate of deposit. c. Cash.	 If yo Incl ques Yes NOT t 	u feel i ude tion? No	y associa	e included ir poe included gory or Cate add Borrowing	ending?	Protecting Assets	gory or cat	egories you fee

Validation Panel Round 2, Group 2 Online Survey (continued) Yes No Basics Borrowing Assets Assets This question should be included in the basic financial literacy \bigcirc \bigcirc assessment test. Comments: Taxes affect how you should invest your money. a. True b. False c. Don't know d. Prefer not to answer ALTERNATIVE WORDING Taxes do not affect how you should invest your money. a. True b. False c. Don't know d. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses. Include Category or Categories this question question? addresses Money Basics Borrowing Building Protecting Assets Assets Yes No This question should be included in the basic financial literacy \bigcirc \bigcirc assessment test. Comments: Which of the following instruments is NOT typically associated with spending? a. Debit card. b. Certificate of deposit. c. Cash.

- d. Credit card.
- e. Don't know
- f. Prefer not to answer

Validation Panel Round 2, Group 2 Online Survey (continued)

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses. Include Category or Categories this question question? addresses Money Borrowing Building Protecting Yes No Assets Assets This question should be included in the basic financial literacy \bigcirc \bigcirc assessment test. Comments: Many people put aside money to take care of unexpected expenses. If Juan and Elva have money put aside for emergencies, in which of the following forms would it be of LEAST benefit to them if they needed it right away? a. Invested in a down payment on the house b. Checking account Stocks С. d. Savings account e. Don't know f. Prefer not to answer Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses. Include Category or Categories this question auestion? addresses Money Basics Borrowing Assets Assets Yes No Basics Assets Assets This question should be included in the basic financial literacy assessment test Comments: If you had a savings account at a bank, which of the following would be correct concerning the interest that you would earn on this account?

a. Earnings from savings account interest may not be taxed.

Validation Panel Round 2, Group 2 Online Survey (continued)

- b. Income tax may be charged on the interest if your income is high enough
- c. Sales tax may be charged on the interest that you earn.
- d. You cannot earn interest until you pass your 18th birthday.
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	lude stion?	Cateç	ory or Cate addr	gories this resses	question			
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets			
This question should be included in the basic financial literacy assessment test.	0	0							
Comments:									
Sara and Joshua just had a baby. Th Which of the following tends to have	hey rec the hig	eived r ghest g	noney as rowth ove	baby gifts a r periods of	and want f time as	to put it awa long as 18 y	y for the ba ears?	aby's edu	catior
 a. A checking account. b. Stocks. c. A U.S. Government savings box 	nd.								

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Category or Categories this question addresses					
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets		
This question should be included in the basic financial literacy assessment test.	0	\bigcirc						

Comments:

d. A savings account.e. Don't knowf. Prefer not to answer

Rob and Mary are the same age. At age 25 Mary began saving \$2,000 a year while Rob saved nothing. At age 50, Rob

Validation Panel Round 2, Group 2 Online Survey (continued)

- b. Income tax may be charged on the interest if your income is high enough
- c. Sales tax may be charged on the interest that you earn.
- d. You cannot earn interest until you pass your 18th birthday.
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Include question?		Category or Categories this question addresses			
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets
This question should be included in the basic financial literacy assessment test.	0	\bigcirc				

Comments:

Sara and Joshua just had a baby. They received money as baby gifts and want to put it away for the baby's education. Which of the following tends to have the highest growth over periods of time as long as 18 years?

- a. A checking account.
- b. Stocks.
- c. A U.S. Government savings bond.
- d. A savings account.
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Category or Categories this question addresses						
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets			
This question should be included in the basic financial literacy assessment test.	0	0							
Comments:									
									1,
Rob and Mary are the same age. At	age 25	Mary I	began sav	ving \$2,000	a year w	hile Rob sav	ed nothing. At	age 50, R	lob

Validation Panel Round 2, Group 2 Online Survey (continued)

realized that he needed money for retirement and started saving \$4,000 per year while Mary kept saving her \$2,000. Now they are both 75

- a. They would each have the same amount because they put away exactly the same.
- b. Rob, because he saved more each year.
- c. Mary, because she has put away more money.
- d. Mary, because her money has grown for a longer time at compound interest.
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

YesNoMoney BasicsBorrowingBuilding AssetsProtecting AssetsThis question should be included in the basic financial literacy assessment test.Image: Image:		Incl ques	ude tion?	Categ	Category or Categories this question addresses					
This question should be included in the basic financial literacy assessment test.		Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets			
	This question should be included in the basic financial literacy assessment test.	0	0							

Comments:

Under which of the following circumstances would it be financially beneficial to you to borrow money to buy something now and repay it with future income?

- a. When you need to buy a car to get a much better paying job.
- b. When you really need a week vacation.
- c. When some clothes you like go on sale.
- d. When the interest on the loan is greater than the interest you get on your savings.
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	ory or Cate add	gories this resses	question
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets
This question should be included in the basic financial literacy assessment test.	0	0				
			1			

Comments:

Validation Panel Round 2, Group 2 Online Survey (continued)

Which of the following statements best describes your right to check your credit history for accuracy?

- a. Your credit record can be checked once a year for free.
- b. You cannot see your credit record.
- c. All credit records are the property of the U.S. Government and access is only available to the FBI and Lenders.
- d. You can only check your record for free if you are turned down for credit based on a credit report.
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Cateç	Category or Categories this question addresses		
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets
This question should be included in the basic financial literacy assessment test.	0	0				
Comments:						

If you are behind on your debt payments and go to a responsible credit counseling service such as the Consumer Credit Counseling Services, what help can they give you?

- a. They can cancel and cut up all of your credit cards without your permission.
- b. They can get the federal government to apply your income taxes to pay off your debts.
- c. They can work with those who loaned you money to set up a payment schedule that you can meet.
- d. They can force those who loaned you money to forgive all your debts.
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Include question?		Category or Categories this question addresses				
	Yes	Yes No		Borrowing	Building Assets	Protecting Assets	
This question should be included in the basic financial literacy assessment test.	0	\bigcirc					

Comments:

Validation Panel Round 2, Group 2 Online Survey (continued)

Barbara has just applied for a credit card. She is an 18-year-old high school graduate with few valuable possessions and no credit history. If Barbara is granted a credit card, which of the following is the most likely way that the credit card company will reduce ITS risk?

- a. It will make Barbara's parents pledge their home to repay Karen's credit card debt.
- b. It will require Barbara to have both parents co-sign for the card.
- c. It will charge Barbara twice the finance charge rate it charges older cardholders.
- d. It will start Barbara out with a small line of credit to see how she handles the account.
- e. Don't know
- f. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

	Incl ques	ude tion?	Categ	jory or Cate addi	gories this esses	question
	Yes	No	Money Basics	Borrowing	Building Assets	Protecting Assets
This question should be included in the basic financial literacy assessment test.	0	\bigcirc				
Comments:						

Bank A offers monthly compounding and Bank B offers yearly compounding. Both banks offer the same interest rate. In your opinion, which bank would you choose if you wanted a higher return?

- a. Bank A.
- b. Bank B.
- c. Both banks offer the same return..
- d. Don't know
- e. Prefer not to answer

Upon reviewing the question above, do you feel it should be included in or excluded from the pool of questions for the final basic financial literacy assessment test? If you feel it should be included, please select the category or categories you feel it addresses.

Include question?	Category or Categories this question addresses
 Yes No	Money Basics Borrowing Building Protecting Assets Assets

This question should be included in

Validation Panel Round 2, Group 2 Online Survey (continued)

the basic financial literacy assessment test.	0 0						
Comments:							
							1.
You are likely to obtain the lowest interest on	loans with		?				
 a. Credit card companies. b. Banks/credit unions. c. Check-cashing firms. d. Don't know e. Prefer not to answer 							
Upon reviewing the question above, or basic financial literacy assessment test? addresses.	do you feel it s ? If you feel i	should be i t should be	ncluded in included,	or exclue please s	ded from the p elect the cate	bool of questions for t gory or categories yo	he final u feel it
	Include question?	Catego	ory or Categ addr	jories this esses	question		
	Yes No	Money Basics	Borrowing	Building Assets	Protecting Assets		
This question should be included in the basic financial literacy assessment test.	0 0						
Comments:							
							1
elf-assessed Financial Literacy Questions							
Financial Literacy Self-Asse	ssment Qı	estions	:				
Reviewing the following financial literacy set	f-assessment quart	estions do v	ou feel eac	h should l	e included in	or excluded (identifyin	a
reason) from the self-assessment portion	of the final bas	ic financial li	teracy assess	sment test?	in the function of the functio	excluded (identifyin	5
I am pretty good at math.							
On a scale of 1 to 7, with 1 meaning	Strongly Disa	gree and 7	' meaning	Strongly .	Agree, how do	you rate yourself wit	th reaard

Validation Panel Round 2, Group 2 Online Survey (continued)

 Disagree Slightly Disagree Neither Disagree Nightly Agree Agree Agree Strongly Agree 	nor Agree			
	Language: Very Unclear	1 2 Neutra	al 4 5	Very Clear
	Wording: Poor	\circ \circ \circ	0 0	Excellent
Comments:				
l am good at dealing wi tracking expenses.	th day-to-day financial matters, su	uch as checking acco	unts, credit ar	// nd debit cards, mortgages, installment payments, and
I am good at dealing wi tracking expenses. On a scale of 1 to 7 to this statement?	th day-to-day financial matters, su , with 1 meaning Strongly E	uch as checking acco Disagree and 7 me	unts, credit ar əaning Stroi	nd debit cards, mortgages, installment payments, and ngly Agree, how do you rate yourself with rega
I am good at dealing wi tracking expenses. On a scale of 1 to 7 to this statement? 1. Strongly Disagree 2. Disagree 3. Slightly Disagree 4. Neither Disagree 5. Slightly Agree 6. Agree 7. Strongly Agree	th day-to-day financial matters, st , with 1 meaning Strongly E e nor Agree	uch as checking acco Disagree and 7 me	unts, credit ar eaning Stro	// nd debit cards, mortgages, installment payments, and ngly Agree, how do you rate yourself with regai
I am good at dealing wi tracking expenses. On a scale of 1 to 7 to this statement? 1. Strongly Disagree 2. Disagree 3. Slightly Disagree 4. Neither Disagree 5. Slightly Agree 6. Agree 7. Strongly Agree	th day-to-day financial matters, su , with 1 meaning Strongly E e nor Agree	uch as checking acco	unts, credit ar eaning Stro	nd debit cards, mortgages, installment payments, and ngly Agree, how do you rate yourself with rega
I am good at dealing wi tracking expenses. On a scale of 1 to 7 to this statement? 1. Strongly Disagree 2. Disagree 3. Slightly Disagree 4. Neither Disagree 5. Slightly Agree 6. Agree 7. Strongly Agree	th day-to-day financial matters, su , with 1 meaning Strongly E nor Agree Language: Very Unclear	uch as checking account Disagree and 7 me 1 2 Neutra	unts, credit ar eaning Stron al 4 5	nd debit cards, mortgages, installment payments, and ngly Agree, how do you rate yourself with regai
I am good at dealing wi tracking expenses. On a scale of 1 to 7 to this statement? 1. Strongly Disagree 2. Disagree 3. Slightly Disagree 4. Neither Disagree 5. Slightly Agree 6. Agree 7. Strongly Agree	th day-to-day financial matters, st , with 1 meaning Strongly E nor Agree Language: Very Unclear Wording: Poor	uch as checking accor Disagree and 7 me 1 2 Neutra	unts, credit ar eaning Stron al 4 5 0 0 0	nd debit cards, mortgages, installment payments, and ngly Agree, how do you rate yourself with regar Very Clear Excellent

Validation Panel Round 2, Group 2 Online Survey (continued)

On a scale of 1 to 7, with to this statement?	1 meaning Strongly Disa	gree and 7 me	eaning Stro	ongly Agree, how do you rate yourself with rega
 Strongly Disagree Disagree Slightly Disagree Neither Disagree nor A Slightly Agree Agree Strongly Agree 	gree			
		1 2 Neutra	al 4 5	
L	anguage: Very Unclear	0 0 0	0 0	Very Clear
	Wording: Poor	0 0 0	0 0	Excellent
Comments:				
I am good at managing credit	cards and debt like car loans o	r a mortgage.		
On a scale of 1 to 7, with to this statement?	1 meaning Strongly Disa	gree and 7 me	eaning Stro	ongly Agree, how do you rate yourself with rega
 Strongly Disagree Disagree Slightly Disagree Neither Disagree nor A Slightly Agree 	gree			
6. Agree 7. Strongly Agree				
		1 2 Neutra	al 4 5	
L	anguage: Very Unclear	0 0 0	0 0	Very Clear
	Wording: Poor	0 0 0	0 0	Excellent
Comments:				

Validation Panel Round 2, Group 2 Online Survey (continued)

 Slightly Low Average Slightly High High High Very High 			
	Language: Very Unclear Wording: Poor	1 2 Neutral 4	5 Very Clear Excellent
Comments:			

Appendix N

Verification Panel Online Survey

Verification Panel member,

Thank you for agreeing to serve as a member of the Verification Panel for my dissertation research. Your input will determine the final assessment instrument.

In the verification round of review, you will be evaluating 26 knowledge questions for the basic financial literacy assessment instrument and five self-assessed financial literacy questions. The final instrument will be administered to U.S. adults age 18 and over using an online survey.

The estimated time to complete the review is approximately 30 minute. Please complete your review as soon as possible. The survey allows for you to stop and come back, so please feel free to break when necessary. You will be able to save the survey and return where you left off. If you have any questions or need assistance, please feel free to contact me at pjdahmen@mail.usf.com or 941-962-1211.

INSTRUCTIONS

Please read each knowledge questions and possible answers (correct answer is bold). Then, respond to the following statements:

- This question should be included in a basic financial literacy assessment instrument. (Agree or disagree)
- The language used in this question is . . . (From very unclear to very clear, evaluate whether the question is simple, clear and easy to understand),
- The wording of this question is . . . (From poor to excellent, evaluate whether the question contains wording that is
 relevant and free from bias. In other words, it portrays a neutral viewpoint)
- The difficulty rating of this question is . . . (Difficulty definitions are below)
- · The financial literacy content category (categories) is (are) appropriate. (Definitions of categories are below)

	1	2	<u>3</u>	4	5
Include question	Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree
Language	Very Unclear	Unclear	Neutral	Clear	Very Clear
Wording	Poor	Needs Improvement	Neutral	Good	Excellent
Difficulty Rating	Too Easy	Easy	Average	Difficult	Too Difficult
Content Category	Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree

Please evaluate each question on the following scale:

Difficulty ratings are as follows:

Rating	Description
Very Easy	Common Knowledge, most adults would know this.
Easy	An adult would have acquired this financial literacy knowledge in the course normal K-12 education.
Neutral	Average, requires some understanding of financial literacy and learning beyond K-12.
Difficult	Requires a level of education specifically directed at financial literacy concepts.
Very Difficult	Advance financial literacy concepts learned through focused financial courses (i.e. financing or investing courses).

The categories for content areas are:

Verification Panel Online Survey (continued)

Financial Literacy Category	Topics Included in Category
Money basics	Compound interest, Time value of money, purchasing power/inflation, spending, short-term savings
Borrowing	APR, finance charges, loan terms, credit scores; type of borrowing (e.g., education, home, vehicles)
Building assets	Long-term-savings accounts/Certificates of deposit, stocks, bonds, mutual funds, annuities
Protecting assets	Investment product risk (stocks vs. bonds vs. mutual funds), investment diversity, insurance protection

Please feel free to comment in the areas provided. When rating 1 or 2, providing reasons and/or suggestion for possible improvement are welcomed.

Knowledge Questions

If you had \$100 in a savings account and the interest rate was 2% per year. After 5 years how much do you think you would have in the account if you left the money to grow?

a. More than \$102

- b. Exactly \$102
- c. Less than \$102
- d. Don't know
- e. Prefer not to answer

Categories this question addresses: Money Basics & Building Assets

Upon reviewing the question above, please note your responses to the statements below. Please comment or provide your suggestion for items you rate 1 or 2.

	Plea	ise ra	te the	follov	wing:	Comments
	1	2	3	4	5	
This question should be included in a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	
Language used in the question is 1= Very Unclear - 5 = Very Clear	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The wording of the question is 1=Poor - 5 = Excellent	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The difficulty rating of the question is 1=Very Easy - 5=Very	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	

Verification Panel Online Survey (continued)

Difficult The Financial Literacy content category (categories) is (are) appropriate. 1=Strongly Disagree - 5 = Strongly Agree	•	0	0	0	0	
Imagine that the interest you be able to buy:	rate o	n you	ır sav	rings	ассоі	as 1% per year and inflation was 2% per year. After 1 year, w
a. More than today b. Exactly the same c. Less than today d. Don't know e. Prefer not not answer						
Categories this question	addres	sses:	Mon	ey B	asics	
Upon reviewing the questi suggestion for items you ra	on abo ate 1 or	ve, pl r 2.	lease	note <u>r</u>	your re	ses to the statements below. Please comment or provide your
	Plea	ase ra	te the	e follo	wing 5	Comments
This question should be included in a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree	•	0	0	0	0	
Language used in the question is 1= Very Unclear - 5 = Very Clear	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The wording of the question is 1=Poor - 5 = Excellent	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The difficulty rating of		\bigcirc	\bigcirc	\bigcirc	\bigcirc	
the question is 1=Very Easy - 5=Very Difficult						

In the year 2016, the Johnson's family income will double and prices of all goods and service will double too. In 2016, how much will the Johnson family be able to buy with their income?

a. More than 2015

b. The same

c. Less than 2015

d. Don't know

e. Prefer not to answer

Categories this question addresses: Money Basics

Upon reviewing the question above, please note your responses to the statements below. Please comment or provide your suggestion for items you rate 1 or 2.

Verification Panel Online Survey (continued)

	Plea	se ra	te the	follov	wing:	Comments
	1	2	3	4	5	
This question should be included in a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	
Language used in the question is 1= Very Unclear - 5 = Very Clear	0	\bigcirc	\bigcirc	\bigcirc	0	
The wording of the question is 1=Poor - 5 = Excellent	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The difficulty rating of the question is 1=Very Easy - 5=Very Difficult	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The Financial Literacy content category (categories) is (are) appropriate. 1=Strongly Disagree - 5 = Strongly Agree	0	0	\bigcirc	0	0	

An annuity is a financial product that pays a lump sum when you die.

a. True **b. False**

c. Don't know d. Prefer not to answer

Categories this question addresses: Money Basics & Building Assets

Upon reviewing the question above, please note your responses to the statements below. Please comment or provide your suggestion for items you rate 1 or 2.

	Plea	se rat	te the	follov	ving:	Comments
	1	2	3	4	5	
This question should be included in a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	
Language used in the question is 1= Very Unclear - 5 = Very Clear	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The wording of the question is 1=Poor - 5 = Excellent	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The difficulty rating of the question is 1=Very Easy - 5=Very Difficult	0	\bigcirc	\bigcirc	\bigcirc	0	
The Financial Literacy content category						

Verification Panel Online Survey (continued)

(categories) is (are) appropriate. 1=Strongly Disagree - 5 = Strongly Agree		0	0	0	0	
If you were to invest \$1, money. a. True	000 in	a sto	ck m	utual	fund,	it would be possible to have less than \$1,000 when you withdraw you
b. False c. Don't know d. Prefer not to answer						
Categories this question	addres	sses:	Mon	iey Ba	asics	& Building Assets
suggestion for items you ra	ate 1 or	2.	to the	faller	wing	
	1	2	3	4	wing. 5	Comments
This question should be included in a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	
Language used in the question is 1= Very Unclear - 5 = Very Clear	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The wording of the question is 1=Poor - 5 = Excellent	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The difficulty rating of the question is 1=Very Easy - 5=Very Difficult	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The Financial Literacy content category (categories) is (are) appropriate. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	
A 15-year mortgage typic life of the loan will be les a. True b. False c. Don't know d. Prefer not to answer Category this guestion a	cally ress.	equire	es hig	her n	nonthl	y payments than a 30-year mortgage, but the total interest paid over t
Upon reviewing the question a	ion abo	ve, p	lease	note y	your re	sponses to the statements below. Please comment or provide your
suggestion for tients you f	Plea	se ra	te the	follo	wing:	Comments
	1	2	3	4	5	
This question should be included in a basic						

Verification Panel Online Survey (continued)

financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0		
Language used in the question is 1= Very Unclear - 5 = Very Clear	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The wording of the question is 1=Poor - 5 = Excellent	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The difficulty rating of the question is 1=Very Easy - 5=Very Difficult	0	0	\bigcirc	\bigcirc	\bigcirc	
The Financial Literacy content category (categories) is (are) appropriate. 1=Strongly Disagree - 5 = Strongly Agree	0	\bigcirc	0	0	0	

Suppose you owe \$1,000 on your credit card and the interest rate you are charged is 20% per year compounded annually. If you didn't make any payments on the credit card, at this interest rate, how many years would it take for the amount you owe to double?

a. 2 years b. Less than 5 years

c. 5 to 10 years

d. More than 20 years

c. Don't know

d. Prefer not to answer

Category this question addresses: Borrowing

Upon reviewing the question above, please note your responses to the statements below. Please comment or provide your suggestion for items you rate 1 or 2.

	Plea	se ra	te the	follov	wing:	Comments
	1	2	3	4	5	
This question should be included in a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	
Language used in the question is 1= Very Unclear - 5 = Very Clear	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The wording of the question is 1=Poor - 5 = Excellent	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The difficulty rating of the question is 1=Very Easy - 5=Very Difficult	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The Financial Literacy content category (categories) is (are) appropriate. 1=Strongly Disagree -	0	0	0	0	0	

Verification Panel Online Survey (continued)

5 = Strongly Agree						
It is possible to reduce the	he risk	of in	vesti	ng in	the st	ock market by buying a wide range of stocks and shares.
a. True b. False c. Don't know d. Prefer not to answer						
Categories this question	addres	sses:	Mon	ey Ba	asics,	Building Assets, & Protecting Assets
Upon reviewing the questi suggestion for items you ra	on abo ate 1 or	ve, pl 2.	ease	note y	our res	sponses to the statements below. Please comment or provide your
	Plea	ise ra	te the	follo	ving:	Comments
	1	2	3	4	5	
This question should be included in a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	
Language used in the question is 1= Very Unclear - 5 = Very Clearar.	0	\bigcirc	\bigcirc	\bigcirc	0	
The wording of the question is 1=Poor - 5 = Excellent	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The difficulty rating of the question is 1=Very Easy - 5=Very Difficult	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The Financial Literacy content category (categories) is (are) appropriate. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	
Net worth is equal to						
a. Total assets b. Total asets plus liabil c. Total assets minus l d. Don't know e. Prefer not to answer	ities liabilit	ies				
Category this question a	ddress	es: N	loney	/ Bas	ics	
Upon reviewing the questi suggestion for items you ra	on abo ate 1 or	ve, pl 2.	ease	note y	our res	sponses to the statements below. Please comment or provide your
		Plea fol	se rat Ilowin	te the ig:.		Comments
	1	2	3	4	5	
This question should be included in a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly	0	0	0	0	0	

Continued on the next page

Verification Panel Online Survey (continued)

Agree						
Language used in the question is 1= Very Unclear - 5 = Very Clear	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The wording of the question is 1=Poor - 5 = Excellent	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The difficulty rating of the question is 1=Very Easy - 5=Very Difficult	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The Financial Literacy content category (categories) is (are) appropriate. 1=Strongly Disagree - 5 = Strongly Agree	0	\bigcirc	\bigcirc	\bigcirc	0	
Retirement savings will r	not be a	subje	ect to	feder	al inco	me tax upon withdrawal when the contribution is made to:
a. A Traditional IRS or I b. A Roth IRA c. A 401(k) plan	ndividu	ial Re	etirme	ent A	ccount	
 d. Don't know e. Prefer not to answer Categories this question Upon reviewing the question 	addres	sses: ve, pl	Mon	ey Ba note y	asics &	Building Assets
 d. Don't know e. Prefer not to answer Categories this question Upon reviewing the questi suggestion for items you re 	addres on abo ate 1 or	ve, pl	Mon lease	ey Ba note y	asics &	Building Assets ponses to the statements below. Please comment or provide your
 d. Don't know e. Prefer not to answer Categories this question Upon reviewing the question suggestion for items you reviewed to a second sec	addres on abo ate 1 or Plea 1	sses: ve, pl 2. se rat	Mon lease te the 3	ey Ba note y follov 4	asics & your res wing: 5	Building Assets ponses to the statements below. Please comment or provide your Comments
 d. Don't know e. Prefer not to answer Categories this question Upon reviewing the questisuggestion for items you range to a suggestion for items you range to a suggestion a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree 	addres on abo ate 1 or Plea 1	ve, pl 2. se rat 2	Mon lease 1 te the 3	ey Ba note y follov 4	vour res	Building Assets sponses to the statements below. Please comment or provide your Comments .
 d. Don't know e. Prefer not to answer Categories this question Upon reviewing the question uggestion for items you raise This question should be included in a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree Language used in the question is 1= Very Unclear - 5 = Very Clear 	addres on abo ate 1 or Plea 1	sses: ve, pl 2. sse rat 2	Mon te the 3	ey Banote y follov	vour resolution of the second	Building Assets sponses to the statements below. Please comment or provide your Comments
 d. Don't know e. Prefer not to answer Categories this question Upon reviewing the question uggestion for items your This question should be included in a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree Language used in the question is 1= Very Unclear - 5 = Very Clear The wording of the question is 1=Poor - 5 = Excellent. 	address on abo ate 1 or Plea 1	ve, pl 2. sse rat 2	Mon lease r te the 3	ey Ba follov 4	ving: 5	Building Assets ponses to the statements below. Please comment or provide your Comments
 d. Don't know e. Prefer not to answer Categories this question Upon reviewing the question upon reviewing the question suggestion for items you reviewing the question for items you reviewed in a basic financial literacy assessment assessment astrument. 1=Strongly Disagree - 5 = Strongly Agree Language used in the question is 1=Very Unclear - 5 = Very Clear The wording of the question is 1=Poor - 5 = Excellent. The difficulty rating of the question is 1=Very Difficult 	address on abo I or Plea 1	sses: ve, pl 2. sse rat 2	Mon lease f te the 3	ey Ba note y follov 4	ving: 5	Building Assets sponses to the statements below. Please comment or provide your Comments

If your assets increase by \$5,000 and your liabilities decrease by \$3,000, your net worth would

Verification Panel Online Survey (continued)

D (
e. Prefer not to answer						
Categories this question	addres	ses:	Mon	ey Ba	asics	& Protecting Assets
Jpon reviewing the quest	ion abo	ve, pl	ease 1	note y	our re	sponses to the statements below. Please comment or provide your
uggestion for items you r		2.	te the	follo	vina [.]	Comments
	1	2	3	4	5	Commenta
This question should be included in a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	
Language used in the question is 1= Very Unclear - 5 = Very Clear	0	\bigcirc	\bigcirc	0	0	
The wording of the question is 1=Poor - 5 = Excellent	0	0	0	0	0	
The difficulty rating of the question is 1=Very Easy - 5=Very Difficult	0	\bigcirc	0	0	0	
The Financial Literacy content category (categories) is (are) appropriate. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	
	ce cos	ts pa	id ov	er the	e life d	of an auto loan, you should choose a loan with the
To reduce the total finan a. Lowest monthly payn b. Longest repayment to c. Shortest repayment d. Don't know e. Prefer not to answer Categories this question Jpon reviewing the questi suggestion for items you r	nent erm addres ion abo ate 1 or	sses: ve, pl 2.	Mon ease 1	ey Ba 10te y	asics /our re	& Borrowing sponses to the statements below. Please comment or provide your
To reduce the total finan a. Lowest monthly payn b. Longest repayment to c. Shortest repayment d. Don't know e. Prefer not to answer Categories this question Upon reviewing the questi suggestion for items you r	addres ion abo ate 1 or Plea	sses: ve, pl 2. se rat	Mone ease r	ey Ba note y follov	asics /our re wing:	& Borrowing sponses to the statements below. Please comment or provide your Comments
To reduce the total finan a. Lowest monthly payn b. Longest repayment to c. Shortest repayment d. Don't know e. Prefer not to answer Categories this question Upon reviewing the questi suggestion for items you r	addres ion abo ate 1 or Plea	ve, pl 2. se rat	Mone ease r te the 3	ey Ba note y follov 4	asics /ourre wing: 5	& Borrowing sponses to the statements below. Please comment or provide your Comments .
To reduce the total finan a. Lowest monthly payn b. Longest repayment to c. Shortest repayment d. Don't know e. Prefer not to answer Categories this question Upon reviewing the questi suggestion for items you r This question should be included in a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree	address ion abo ate 1 or Plea 1	sses: ve, pl 2. se rat 2	Mone ease r te the 3	ey Banote y follow	vour review vour reconstruction of the second secon	& Borrowing sponses to the statements below. Please comment or provide your Comments .

Verification Panel Online Survey (continued)

question is 1=Poor - 5 = Excellent	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
The difficulty rating of he question is =Very Easy - 5=Very Difficult	0	0	0	0	0
he Financial Literacy ontent category ategories) is (are) ppropriate. =Strongly Disagree - = Strongly Agree	٢	0	0	0	0
Annual fee Line of credit Don't know Prefer not to answer ategories this question	addres	sses:	Mon	ey Ba	asics &
on reviewing the questi	on abo	ve nl	ease	note y	Jour res
oon reviewing the questi ggestion for items you r	on abo ate 1 or Plea	ve, pl 2.	te the	follov	vour re:
on reviewing the questi ggestion for items you r	on abo ate 1 or Plea 1	ve, pl 2. ise ra 2	te the	follov 4	vour re: wing: 5
Ipon reviewing the questi aggestion for items you n This question should be included in a basic inancial literacy assessment nstrument. 1=Strongly Disagree - 5 = Strongly Agree	on abo ate 1 or Plea 1	ve, pl ·2. ise ra 2	te the	follov 4	wing: 5
pon reviewing the questi iggestion for items you n This question should be included in a basic inancial literacy assessment nstrument. 1=Strongly Disagree - 5 = Strongly Agree Language used in the question is 1= Very Jnclear - 5 = Very Clear.	on abo ate 1 or Plea 1	ve, pl 2. 	te the 3	follov 4	wing: 5
pon reviewing the questi iggestion for items you n fhis question should be included in a basic inancial literacy issessment nstrument. 1=Strongly Disagree - 5 = Strongly Disagree - 5 = Strongly Disagree - 5 = Strongly Disagree - 5 = Strongly Disagree - 5 = Very Disagree - 5 = Very Clear - 5 = Very Clear. The wording of the juestion is 1=Poor - 5 = Excellent	Plea 1	ve, pl 2. 	te the 3	follov 4	wing: 5
Ipon reviewing the questi aggestion for items you not be included in a basic inancial literacy assessment nstrument. 1=Strongly Disagree - 5 = Strongly Agree Language used in the question is 1= Very Jinclear - 5 = Very Clear. The wording of the question is 1=Poor - 5 = Excellent The difficulty rating of he question is I=Very Easy - 5=Very Difficult	Plea 1 0	ve, pl 2. sse rar 2 0	te the 3		vour re:

Your net worth is

- a. the difference between your expenditures and income.
 b. the difference between your liabilities and assets.
 c. the difference between your cash inflow and outflow.
 d. the difference between your bank borrowing and savings.
Verification Panel Online Survey (continued)

e none of the above. f. Don't know

- g. Prefer not to answer

Category this question addresses: Money Basics

Upon reviewing the question above, please note your responses to the statements below. Please comment or provide your suggestion for items you rate 1 or 2.

	Plea	ise ra	te the	follo	wing:	Comments
	1	2	3	4	5	
This question should be included in a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	
Language used in the question is 1= Very Unclear - 5 = Very Clear	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The wording of the question is 1=Poor - 5 = Excellent	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The difficulty rating of the question is 1=Very Easy - 5=Very Difficult	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The Financial Literacy content category (categories) is (are) appropriate. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	
 a. a credit union. b. a commercial bank. c. the Better Business E d. a credit bureau. e a professor. f. Don't know g. Prefer not to answer Categories this question Upon reviewing the question suggestion for items you review you you you you you you you you you you	addression abo	sses: ve, pl 2.	Born	owing) your re	sponses to the statements below. Please comment or provide you
	1	2	3	4	5	
This question should be included in a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	
Language used in the question is 1= Very Unclear - 5 = Very Clear	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	

Verification Panel Online Survey (continued)

The wording of the question is 1=Poor - 5 = Excellent	0	\bigcirc	0	\bigcirc	0	
The difficulty rating of the question is 1=Very Easy - 5=Very Difficult	0	0	0	0	0	
The Financial Literacy content category (categories) is (are) appropriate. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	

With compound interest, you earn interest on your interest, as well as on your principal.

a. True

b. False c. Don't know d. Prefer not to answer

Categories this question addresses: Money Basics, Building Assets, & Protecting Assets

Upon reviewing the question above, please note your responses to the statements below. Please comment or provide your suggestion for items you rate 1 or 2.

	Plea	ise ra	te the	follow	wing:	Comments
	1	2	3	4	5	
This question should be included in a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree	۲	0	0	0	0	
Language used in the question is 1= Very Unclear - 5 = Very Clear	0			0	0	
The wording of the question is 1=Poor - 5 = Excellent	0	\bigcirc	0	0	\bigcirc	
The difficulty rating of the question is 1=Very Easy - 5=Very Difficult	0	\bigcirc	0	\bigcirc	0	
The Financial Literacy content category (categories) is (are) appropriate. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	

The earlier you start saving for retirement, the more money you will have because the effects of compounding interest increase over time.

a. True b. False c. Don't know d. Prefer not to answer

Categories this question addresses: Money Basics & Building Assets

Verification Panel Online Survey (continued)

Upon reviewing the question above, please note your responses to the statements below. Please comment or provide your suggestion for items you rate 1 or 2.

	Plea	se ra	te the	follov	ving:	Comments
	1	2	3	4	5	
This question should be included in a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	
Language used in the question is 1= Very Unclear - 5 = Very Clear	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The wording of the question is 1=Poor - 5 = Excellent	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The difficulty rating of the question is 1=Very Easy - 5=Very Difficult	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The Financial Literacy content category (categories) is (are) appropriate. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	

You should have an emergency fund that covers two to six months of your expenses.

a. True b. False c. Don't know d. Prefer not to answer

Categories this question addresses: Money Basics & Building Assets

Upon reviewing the question above, please note your responses to the statements below. Please comment or provide your suggestion for items you rate 1 or 2

	Plea	se ra	te the	follov	ving:	Comments
	1	2	3	4	5	
This question should be included in a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	
Language used in the question is 1= Very Unclear - 5 = Very Clear	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The wording of the question is 1=Poor - 5 = Excellent	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The difficulty rating of the question is 1=Very Easy - 5=Very Difficult	0	0	0	\bigcirc	\bigcirc	
The Financial Literacy						

Verification Panel Online Survey (continued)

you have a savings acc	count	at a t	bank,	you	may h	ave to pay taxes on the interest you earn.
True False Don't know Prefer not to answer						
ategories this question	addres	sses:	Mon	ey Ba	asics	Building Assets
pon reviewing the question	on abo	ve, pl	lease	note y	your re	ponses to the statements below. Please comment or provide your
ggestion for items you ra	te 1 or Plea	:2. ise rat	te the	follo	wina:	Comments
	1	2	3	4	5	
his question should e included in a basic nancial literacy ssessment nstrument. 1=Strongly bisagree - 5 = Strongly gree	0	0	0	0	0	
anguage used in the uestion is 1= Very Inclear - 5 = Very Clear	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
he wording of the uestion is 1=Poor - = Excellent	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
he difficulty rating of ne question is =Very Easy - 5=Very Difficult	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
he Financial Literacy ontent category categories) is (are) ppropriate. =Strongly Disagree - = Strongly Agree	0	0	0	0	0	
he Financial Literacy ontent category categories) is (are) ppropriate. = Strongly Disagree - = Strongly Agree the interest rate on an a True False Don't know	○ adjusta	able-i	⊂ rate r	mortg	age lo	an goes up, your monthly mortgage payments will also go up.
Prefer not to answer ategories this question	addres	sses:	Mon	ey Ba	asics	Building Assets
oon reviewing the questio ggestion for items you ra	on abo te 1 or	ve, pl 2.	lease	note y	your re	ponses to the statements below. Please comment or provide your
	Plea	ise ra	te the	follo	wing:	Comments

Verification Panel Online Survey (continued)

instrument. 1=Strongly Disagree - 5 = Strongly Agree						
Language used in the question is 1= Very Unclear - 5 = Very Clear	0	0	0	0	0	
The wording of the question is 1=Poor - 5 = Excellent	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The difficulty rating of the question is 1=Very Easy - 5=Very Difficult	0	\bigcirc	\bigcirc	0	0	
The Financial Literacy content category (categories) is (are) appropriate. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	

Repeatedly refinancing your home mortgage over a short period of time results in added fee and points that further increase your debt.

- **a. True** b. False c. Don't know
- d. Prefer not to answer

Categories this question addresses: Borrowing & Building Assets

Upon reviewing the question above, please note your responses to the statements below. Please comment or provide your suggestion for items you rate 1 or 2.

		Plea fol	se rat Ilowin	te the		comments
	1	2	3	4	5	
This question should be included in a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	
Language used in the question is 1= Very Unclear - 5 = Very Clear	0	\bigcirc	0	0	0	
The wording of the question is 1=Poor - 5 = Excellent	0	\bigcirc	\bigcirc	\bigcirc	0	
The difficulty rating of the question is 1=Very Easy - 5=Very Difficult	0	\bigcirc	\bigcirc	0	0	
The Financial Literacy content category (categories) is (are) appropriate. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	

Verification Panel Online Survey (continued)

Your credit report includes employment data, your payments history, any inquiries made by creditors, and any pubic record information.

a. True

- b. False
- c. Don't know d. Prefer not to answer

Categories this question addresses: Money Basics & Borrowing

Upon reviewing the question above, please note your responses to the statements below. Please comment or provide your suggestion for items you rate 1 or 2.

	Plea	se ra	te the	follov	wing:	Comments
	1	2	3	4	5	
This question should be included in a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	
Language used in the question is 1= Very Unclear - 5 = Very Clear	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The wording of the question is 1=Poor - 5 = Excellent	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The difficulty rating of the question is 1=Very Easy - 5=Very Difficult	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The Financial Literacy content category (categories) is (are) appropriate. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	

If you buy certificates of deposit, savings bonds, or Treasury bills, you can earn higher returns than on a savings account, with little or no added risk.

a. True

- b. False
- c. Don't know

d. Prefer not to answer

Category this question addresses: Building Assets

Upon reviewing the question above, please note your responses to the statements below. Please comment or provide your suggestion for items you rate 1 or 2.

	Plea	se ra	te the	follov	ving:	Comments
	1	2	3	4	5	
This question should be included in a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	
Language used in the question is 1= Very						

Verification Panel Online Survey (continued)

Unclear - 5 = Very Clear	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The wording of the question is 1=Poor - 5 = Excellent	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The difficulty rating of the question is 1=Very Easy - 5=Very Difficult	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The Financial Literacy content category (categories) is (are) appropriate. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	
Don't know Prefer not to answer Category this question a	ddress	es: N	loney	/ Bas	ics	
 Don't know Prefer not to answer Category this question a Jpon reviewing the questing gestion for items you reviewing the syncestic terms and terms are syncestic terms and terms and terms are syncestic terms are syncesteres are syncestic terms are syncestic terms are syncestic ter	ddress ion abo ate 1 or	es: M ve, pl	loney ease	/ Bas note y	i cs 7 our re	esponses to the statements below. Please comment or provide you
Don't know Prefer not to answer ategory this question a pon reviewing the questi aggestion for items you re	ddress ion abo ate 1 or Plea	es: N ve, pl 2. ise ra	foney lease t te the	/ Bas note y follov	ics /our re wing:	esponses to the statements below. Please comment or provide you Comments
 Don't know Prefer not to answer Category this question a Upon reviewing the question gestion for items you not be included in a basic Financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree 	ddress ion abo ate 1 or Plea 1	es: N ve, pl 2. ise ra 2	foney lease to te the 3	y Bas note y follow 4	vour rewing:	esponses to the statements below. Please comment or provide you Comments .
 Don't know Prefer not to answer Category this question a Upon reviewing the question gestion for items you not be included in a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree Language used in the question is 1= Very Unclear - 5 = Very Clear 	ddress ion abo ate 1 or Plea 1	es: N ve, pl 22. sse ra 2	Aoney ease f te the 3	/ Bas note y follow 4	ics vour re wing: 5	esponses to the statements below. Please comment or provide you Comments .
 Don't know Prefer not to answer Category this question a Upon reviewing the question gestion for items you measure the second second	ddress ion abo ate 1 or Plea 1	es: N ve, pl 2. se ra 2	Money lease 1 te the 3	y Bas note y follow 4	vour review vour r	esponses to the statements below. Please comment or provide you Comments
 Don't know Prefer not to answer Category this question a Upon reviewing the question Upon reviewing the question is 1= Very Upon reviewing the question Upon reviewing the question	ddress ion abo ate 1 or Plea 1	es: N ve, pl 2. sse ra 2 0	foney lease f te the 3	y Bas note y follow 4	ics vour re ving: 5 0 0 0 0 0 0 0 0 0 0 0 0 0	esponses to the statements below. Please comment or provide you Comments

Bank A offers monthly compounding and Bank B offers yearly compounding. Both banks offer the same interest rate. In your opinion, which bank would you choose if you wanted a higher return?

a. Bank A.

Verification Panel Online Survey (continued)

b. Bank B.

- c. Both banks offer the same return..
- d. Don't know
- e. Prefer not to answer

Category this question addresses: Building Assets

Upon reviewing the question above, please note your responses to the statements below. Please comment or provide your suggestion for items you rate 1 or 2.

	Plea	ise ra	te the	follo	wing	Comments
	1	2	3	4	5	
This question should be included in a basic financial literacy assessment instrument. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	
Language used in the question is 1= Very Unclear - 5 = Very Clear	0	\bigcirc	0	\bigcirc	\bigcirc	
The wording of the question is 1=Poor - 5 = Excellent.	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The difficulty rating of the question is 1=Very Easy - 5=Very Difficult	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
The Financial Literacy content category (categories) is (are) appropriate. 1=Strongly Disagree - 5 = Strongly Agree	0	0	0	0	0	
Do you think buying stoc a. Yes. b. No. c. Don't know d. Prefer not to answer	k in a	single	e con	npany	y is sa	fer than buying stock in several different companies?
Categories this question	addres	ses:	Build	ding A	Assets	& Protecting Assets

Upon reviewing the question above, please note your responses to the statements below. Please comment or provide your suggestion for items you rate 1 or 2.

	Plea	ise ra	te the	follov	wing	Comments
	1	2	3	4	5	
This question should be included in a basic financial literacy assessment instrument.	0	0	0	0	0	
Language used in the question is clear.Language used in the question is 1= Very Unclear - 5 = Very Clear	0	0	0	0	0	
The wording of the question is 1=Poor -	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	

Verification Panel Online Survey (continued)

5 = Excellent.										
The difficulty rating the question is 1=Very Easy - 5=V Difficult	of ery	0 0								
The Financial Liter content category (categories) is (are appropriate. 1=Strongly Disagre 5 = Strongly Agree	acy) ee -	0 (
f-Assessment Questio	ns									
Please read each Then, respond to t This question The language The wording o bias, in other word	of the foll he followi should b used in t of this que s,	owing ng sta e inclu his qu estion	self-a teme ided i estio is	asses ents: in a b n is . . (e ^v	sme asic (i valua	nt financi financial evaluate tte wheth	al knowled literacy as whether the er the ques	ge questionsessment gessment geguestion stion conta	ons (five questic instrument. n is simple, clea ains wording tha	ons) and possible answers. r and easy to understand), t is relevant and free from
	neutrai v	lewpoi			2		2	4	5	Г
Include cuestion	Steensky	Discourse			2 Diago		3 Novinal	4	Strongly Agree	_
Language	Vomul	Inglaar	~		Unal	and and a set	Neutral	Clean	Strongly Agree	_
Wanding	very t	nclear	_	Unclear			Neutral	Clear	Fueellent	_
I am comfortable with installment payments, On a scale of 1 to to this statement? 1. Strongly Disagre 2. Disagree 3. Slightly Disagree 5. Slightly Agree 6. Agree 7. Strongly Agree	my knowled and trackin 7, with 1 ee e nor Agn	dge dea g expen meani ee	ling w ses. Ing S	as production of the following of the fo	ivide ⊳-to-da ly Dis	ny financial sagree ar	matters, suc	h as checkin ng Strong,	ng accounts, credit i	and debit cards, mortgages, o you rate yourself with regard
	'	lease	rate t		lowin L	ig: 5	(comments		
This question shou be included in finan literacy self- assessment instrument.	ıld ncial	0 0								
Language used in question is 1= Ve Unclear - 5 = Very Clear	the ery	0								
The wording of the										

Verification Panel Online Survey (continued)

am comfortable with my kno	wledge i	manag	ging m	oney,	budget	ng and planning my expenditures.
On a scale of 1 to 7, with to this statement?	n 1 me	aning	Stro	ongly	Disag	ree and 7 meaning Strongly Agree, how do you rate yourself with reg
. Strongly Disagree 2. Disagree 3. Slightly Disagree 4. Neither Disagree nor A 5. Slightly Agree 7. Strongly Agree	lgree					
	Plea	se ra	te the	follo	wing:	Comments
	1	2	3	4	5	
This question should be included in financial literacy self- assessment instrument.	0	0	0	0	0	
Language used in the question is 1= Very Unclear - 5 = Very Clear	0	\bigcirc	\bigcirc	\bigcirc	0	
The wording of the question is 1=Poor - 5 = Excellent. am comfortable with my kno On a scale of 1 to 7, with	wledge i) manag aning	ging cr	•edit co	ards ar	d debt like car loans or a mortgage. ree and 7 meaning Strongly Agree, how do you rate yourself with reg
The wording of the question is 1=Poor - 5 = Excellent. am comfortable with my kno On a scale of 1 to 7, with o this statement? . Strongly Disagree Disagree . Slightly Disagree . Neither Disagree . Neither Disagree . Agree	wledge i n 1 med Agree	manag	ging cr	edit co	ards ar Disag	d debt like car loans or a mortgage. ree and 7 meaning Strongly Agree, how do you rate yourself with reg
The wording of the question is 1=Poor - 5 = Excellent. am comfortable with my kno On a scale of 1 to 7, with o this statement? . Strongly Disagree . Disagree . Disagree . Neither Disagree . Neither Disagree . Agree . Strongly Agree . Strongly Agree	wledge n n 1 me. Agree	aning	tring cr tring cr	oredit co	o ards ar Disag	d debt like car loans or a mortgage. ree and 7 meaning Strongly Agree, how do you rate yourself with reg
The wording of the question is 1=Poor - 5 = Excellent. am comfortable with my kno On a scale of 1 to 7, with o this statement? 1. Strongly Disagree 2. Disagree 3. Slightly Disagree 4. Neither Disagree 5. Slightly Agree 6. Agree 7. Strongly Agree	wledge r n 1 med Agree Plea	o mana <u>s</u> aning se ra 2	o stro	oredit ca ongly follow	O ards ar Disag Wing: 5	d debt like car loans or a mortgage. ree and 7 meaning Strongly Agree, how do you rate yourself with reg Comments
The wording of the question is 1=Poor - 5 = Excellent. am comfortable with my kno On a scale of 1 to 7, with o this statement? . Strongly Disagree 2. Disagree 2. Slightly Disagree 3. Slightly Agree 3. Slightly Agree 5. Slightly Agree 5. Strongly Agree 7. Strongly Agree This question should be included in financial literacy self- assessment instrument.	 wiedge n n 1 med Agree Plea 1 	aning	te the 3	 redit ca nngly follor 4 	o ards ar Disag wing: 5	d debt like car loans or a mortgage. ree and 7 meaning Strongly Agree, how do you rate yourself with reg Comments
The wording of the question is 1=Poor - 5 = Excellent. am comfortable with my kno Dn a scale of 1 to 7, with o this statement? Strongly Disagree Disagree Sightly Disagree Neither Disagree on A Sightly Disagree Neither Disagree Sagree Agree Strongly Agree This question should be included in financial literacy self- assessment instrument. Language used in the question is 1= Very Unclear - 5 = Very Clear	wiedge n n 1 mee Agree 1	sse ra 2	e stro	eredit co ngly follor 4	wing: 5	d debt like car loans or a mortgage. ree and 7 meaning Strongly Agree, how do you rate yourself with reg Comments

Verification Panel Online Survey (continued)

annuities).

On a scale of 1 to 7, with 1 meaning Strongly Disagree and 7 meaning Strongly Agree, how do you rate yourself with regard to this statement?

1. Strongly Disagree 2. Disagree 3. Slightly Disagree 4. Neither Disagree nor Agree 5. Slightly Agree 6. Agree 7. Strongly Agree Please rate the following: Comments 1 2 3 4 5 This question should be included in financial literacy self- \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc assessment instrument. Language used in the question is.... 1= Very Unclear - 5 = Very \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc Clear The wording of the question is.... 1=Poor -5 = Excellent. \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc

On a scale of 1 to 7, with 1 meaning Very Low and 7 meaning Very High, how would you assess your overall financial knowledge?

- 1. Very Low
- 2. Low
- 3. Slightly Low
- 4. Average
- 5. Slightly High
- 6. High
- 7. Very High

	Please rate the following:				ving:	Comments
	1	2	3	4	5	
This question should be included in financial literacy self- assessment instrument.	0	0	0	0	0	
Language used in the question is 1= Very Unclear - 5 = Very Clear	\bigcirc	\bigcirc	\bigcirc	\bigcirc	0	
The wording of the question is 1=Poor - 5 = Excellent.	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	

Appendix O

Field Test Communication

From: Pearl Para <<u>pjdahmen@mail.usf.edu</u>> Date: Tue, Sep 22, 2015 at 3:48 PM Subject: SBDC Small Business Financial Literacy Research To: <<u>smallbusinessowner.address></u>

Hello <Name.,

You are receiving this email because you have been one of my consulting clients at the Florida SBDC at USF. I am currently working on my Doctorate at USF and I feel you could provide valuable input to my dissertation research.

I am currently a part-time SBDC employee, while I work on my dissertation: Financial Literacy Continuing Professional Education Cognitive Needs Assessment for Florida Small Business Owners.

As a part of this research, I am developing a basic financial literacy assessment survey. The results of the research will be to provide the SBDC with valuable data about the educational needs of small business owners related to financial literacy.

The survey is ready for field-testing. In other words, I am looking for small business owners to take the survey and provide feedback. You would be a perfect candidate for this phase of my research. I hope you will agree to participate.

Participation entails completing the online survey of 30 questions, and giving your feedback (in the form of a few short comments) at the end of the survey. The estimated time to complete the survey is less than 20 minutes. The results of the survey will be anonymous, and will go into statistical totals. I will be happy to share the results of the field test (included the correct answers), as well as the final research results, if you are interested.

Will you be a part of this valuable research? Please respond to this email either way. Once I receive your response, the link to the survey will be provided to you.

Thank you in advance for your consideration.

Best regards,



Pearl Dahmen Para, CPA

Consultant FSBDC at the University of South Florida Muma College of Business 1101 Channelside Drive, Suite 210 Port Tampa Bay Building Tampa, Fl. 33602-3613 813.905.5819 pjdahmen@mail.usf.edu www.SBDCTampaBay.com

Appendix P

Field Test Online Survey

Introduction
Thank you for your participation in this research (USF IRB 22494) regarding financial literacy. Information gathered from this study may be utilized to assist small business organizations and governmental agencies guide financial literacy education. Because of your background and expertise, your participation and input are invaluable.
This assessment survey is a part of a dissertation. The survey consists of two sections: 20 knowledge questions, and five self-assessments questions. Completing it should take approximately than 15 minutes.
Once you complete each section of the survey, you will be asked to give your feedback. Consider the understandability and difficulty of the questions when giving feedback. Please add any comments you feel would be helpful related to any question.
Your participation is voluntary and uncompensated. No identifying information is being gathered and your participation is anonymous and confidential. There are no known benefits or risks to this research. The data will be kept for a period of 5 years.
If, at any time, you have questions regarding your participation or issues with this website, please contact the Principal Investigator, Pearl Dahmen Para, at pjdahmen@mail.usf.edu.
Demographic Questions
Please respond to the following demographic questions:
What is your gender?
O Male
Other (Please Specify)
What year were you born?
How many years have you owned your current small business?
0 - 2 years
3 - 5 years
○ 5 - 10 years
10 - 20 years
Over 20 years

Field Test Online Survey (continued)

Not counting yourself, how many employees are there in your current business?

No employees

Less than 5 employees

More than 5 employees

What was the last year of education you completed?

- Did not complete high school
- O High school graduate regular high school or GED
- Some college
- College graduate
- Post graduate education

Knowledge Questions

Please respond to the following financial literacy questions.

You should have an emergency fund that covers three to six months of your expenses.

- True
- False
- Don't know
- Prefer not to answer

With compound interest, interest is earned on interest in addition to the original deposit or loan.

- True
- False
- On't know
- Prefer not to answer

The earlier you start saving for retirement, the more money you will have because the effects of compounding interest increase over time.

- True
- False
- On't know
- Prefer not to answer

Field Test Online Survey (continued)
Bank A offers monthly compounding and Bank B offers yearly compounding. Both banks offer the same interest rate. In your opinion, which bank would you choose if you wanted a higher return?
Bank A
O Bank B
Both banks offer the same return.
O Don't know
Prefer not to answer
Annually, you can receive a free credit report from
a credit union.
a commercial bank.
the Better Business Bureau
a credit bureau.
On't know
Prefer not to answer
Your credit report includes your outstanding credit balances, payments history, any inquiries made by creditors, but, does not include employment data and pubic record information. True False
To reduce the total interest or finance costs paid over the life of a loan, you should choose a loan with the
Lowest monthly payment
Longest repayment term
Shortest repayment term
On't know
Prefer not to answer
If you always pay the full balance on your credit card, which of the following is least important?
Annual interest rate
O Annual fee
Credit Limit

Field Test Online Survey (continued)
O Don't know
O Prefer not to answer
Suppose you owe \$1,000 on your credit card and the interest rate you are charged is 20% per year compounded annually. If you didn't make any payments on the credit card, at this interest rate, how many years would it take for the amount you owe to double?
2 years
C Less than 5 years
◯ 5 to 10 years
O More than 20 years
O Don't know
O Prefer not to answer
To ensure that some of your retirement savings will not be subject to income tax upon withdrawal, you would contribute to: A Traditional IRA (Individual Retirement Account)
A Roth IRA
○ A 401(k) plan
O Don't know
Prefer not to answer
If you have a savings account at a bank and do not withdraw any money, you will not have to pay taxes on the interest you earn.
O True
◯ False
O Don't know
Prefer not to answer
A 15-year mortgage typically requires higher monthly payments than a 30-year mortgage, but the total interest paid over the life of the loan will be less.
O True
◯ False
On't know
O Prefer not to answer
If the interest rate on an adjustable-rate mortgage loan goes up, your monthly mortgage payments will not go up.

 True False Don't know Prefer not to answer Repeatedly refinancing your home mortgage over a short period of time results in added fees and points that further increase your debt. True False Don't know Prefer not to answer If you buy certificates of deposit (CDs), savings bonds, or Treasury bills, you can earn higher returns than on a savings account, with little or no added risk. True False Don't know Prefer not to answer Do you think buying stock in a single company is safer than buying stock in several different companies? Yes No Don't know Prefer not to answer 	
 False Don't know Prefer not to answer Repeatedly refinancing your home mortgage over a short period of time results in added fees and points that further increase your debt. True False Don't know Prefer not to answer If you buy certificates of deposit (CDs), savings bonds, or Treasury bills, you can earn higher returns than on a savings account, with little or no added risk. True False Don't know Prefer not to answer Do you think buying stock in a single company is safer than buying stock in several different companies? Yes No Don't know Prefer not to answer 	
 Don't know Prefer not to answer Repeatedly refinancing your home mortgage over a short period of time results in added fees and points that further increase your debt. True False Don't know Prefer not to answer If you buy certificates of deposit (CDs), savings bonds, or Treasury bills, you can earn higher returns than on a savings account, with little or no added risk. True False Don't know Prefer not to answer Do you think buying stock in a single company is safer than buying stock in several different companies? Yes No Don't know Prefer not to answer 	
 Prefer not to answer Repeatedly refinancing your home mortgage over a short period of time results in added fees and points that further increase your debt. True False Don't know Prefer not to answer If you buy certificates of deposit (CDs), savings bonds, or Treasury bills, you can earn higher returns than on a savings account, with little or no added risk. True False Don't know Prefer not to answer Do you think buying stock in a single company is safer than buying stock in several different companies? Yes No Don't know Prefer not to answer 	
Repeatedly refinancing your home mortgage over a short period of time results in added fees and points that further increase your debt. True False Don't know Prefer not to answer If you buy certificates of deposit (CDs), savings bonds, or Treasury bills, you can earn higher returns than on a savings account, with little or no added risk. True False Don't know Prefer not to answer	
 True False Don't know Prefer not to answer If you buy certificates of deposit (CDs), savings bonds, or Treasury bills, you can earn higher returns than on a savings account, with little or no added risk. True False Don't know Prefer not to answer Do you think buying stock in a single company is safer than buying stock in several different companies? Yes No Don't know Prefer not to answer 	
 False Don't know Prefer not to answer If you buy certificates of deposit (CDs), savings bonds, or Treasury bills, you can earn higher returns than on a savings account, with little or no added risk. True False Don't know Prefer not to answer Do you think buying stock in a single company is safer than buying stock in several different companies? Yes No Don't know Prefer not to answer 	
 Don't know Prefer not to answer If you buy certificates of deposit (CDs), savings bonds, or Treasury bills, you can earn higher returns than on a savings account, with little or no added risk. True False Don't know Prefer not to answer Do you think buying stock in a single company is safer than buying stock in several different companies? Yes No Don't know Prefer not to answer 	
 Prefer not to answer If you buy certificates of deposit (CDs), savings bonds, or Treasury bills, you can earn higher returns than on a savings account, with little or no added risk. True False Don't know Prefer not to answer Do you think buying stock in a single company is safer than buying stock in several different companies? Yes No Don't know Prefer not to answer 	
If you buy certificates of deposit (CDs), savings bonds, or Treasury bills, you can earn higher returns than on a savings account, with little or no added risk. True False Do n't know Prefer not to answer No No Prefer not to answer Prefer not to answer	
 False Don't know Prefer not to answer Do you think buying stock in a single company is safer than buying stock in several different companies? Yes No Don't know Prefer not to answer 	\$
 False Don't know Prefer not to answer Do you think buying stock in a single company is safer than buying stock in several different companies? Yes No Don't know Prefer not to answer 	
 Don't know Prefer not to answer Do you think buying stock in a single company is safer than buying stock in several different companies? Yes No Don't know Prefer not to answer 	
 Prefer not to answer Do you think buying stock in a single company is safer than buying stock in several different companies? Yes No Don't know Prefer not to answer 	
Do you think buying stock in a single company is safer than buying stock in several different companies? Yes No Don't know Prefer not to answer	
 No Don't know Prefer not to answer 	
 Don't know Prefer not to answer 	
O Prefer not to answer	
It is possible to reduce the risk of losing your investment when investing in the stock market by buying a wide range of stocks and shares.	
◯ True	
O False	
On't know	
Prefer not to answer	
If you were to invest \$1,000 in a stock mutual fund, it would be possible to have less than \$1,000 when you withdraw y money.	our

To what extent do you agree

eld Test Online Survey	y (continu	ed)					
False							
On't know							
Prefer not to answer							
Your net worth is							
the difference between your e	expenditure and	income.					
the difference between your c	ash inflow and	outflow.					
the difference between your b	ank borrowing	and savings.					
The difference between your	liabilities and as	ssets.					
none of the above.							
On't know							
Prefer not to answer							
If your assets increase by \$5,00	00 and your lia	bilities decrea	ase by \$3,00	0, your net worl	h would		
Increase by \$3,000							
Increase by \$8,000							
On't know							
Prefer not to answer							
Please provide any feedback or	r comments or	the knowled	ge questions	portion of this	survey.		
elf-Assessment Questions							
For the following five questions,	, rate (or asse	ss) your belief	or confiden	ce on each of th	ne statements	S.	
I am comfortable with my knowl cards, mortgages, installment p	ledge to deal v ayments, and	vith day-to-da tracking expe	y financial m nses.	atters, such as	checking acc	counts, cred	it and debi
	Strongly Disagree	Disagree	Slightly Disagree	Neither Disagree or Agree	Slightly Agree	Agree	Strongly Agree

Field Test Online Survey (continued)

		\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
	,						
am comfortable with my know	ledge to mana	ge money, bi	udgeting and p	olanning my e	expenditures.		
	Strongly		Slightly	Neither Disagree or	Slightly		Strongly
	Disagee	Disagree	Disagree	Agree	Agree	Agree	Agree
o what extent do you agree or disagree with the following statement.	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
am comfortable with my know	ledge to mana	ge credit car	ds and debt lik	ke car loans d	or a mortgage.		
			0	Neither			<u>.</u>
	Disagree	Disagree	Disagree	Agree or	Agree	Agree	Strongly Agree
o what extent do you agree or disagree with the following tatement.	0	0	0	0	0	0	0
understand investment produce 01Ks, and annuitites).	cts (for exampl Strongly Disagree	e stocks, bor Diagree	nds, and mutu Slightely Disagree	al funds) and Neither Disagree or Agree	Slightly Agree	counts (suc	h as IRAs, Strongly Agree
o what extent do you agree or disagree with the following tatement.	0	0	0	0	0	0	0
Consider your overall financial	knowledge,						
	I						
	Very Low.	Low	Slightly Low	Average	Slightly High	High	Very High

Appendix Q

Final Review Panel Email Communication

From: Pearl Para [mailto:pjdahmen@mail.usf.edu] Sent: Sunday, October 18, 2015 12:10 PM To: Pearl Dahmen Subject: Financial Literacy Research - Assessment Final Review
Hello,
Thank again for agreeing to be part of the expert panels assisting in the development of financial literacy assessment instrument that will be used in my dissertation research.
The instrument has been field-tested and is ready for the final review. If you would like to be part of the final review panel and have availability between now and Friday, October 23 rd , your input is invaluable and appreciated.
The instrument consists of 20 knowledge questions and five self-assessment questions, plus five demographic questions. It takes less than 15 minutes to complete.
If you are interested, the assessment is available online at http://usf.qualtrics.com/SE/?SID=SV_8eKm1uN7jLndif or you can review the attached PDF copy of the survey. Please consider to the follow questions:
 Did you have any problems with a question? If so, which one? What was the problem? Does everything read clearly? Including the directions? Did you find any issues with the flow or order of the questions? If so, explain. Is there anything you feel could improve the survey? Do you feel the results of this survey will be an appropriate assessment of an individual's basic financial literacy? Please respond with your feedback and comments at the end of the survey or via email.
NOTE: This survey is mobile friendly and can be accessed with a smart phone or tablet as well as a computer or laptop.
Your assistance and support is greatly appreciated. Please let me know if you would like to know the final results of my research.
Best regards,
Pearl Dahmen Para University of South Florida PhD Candidate

Appendix R

Final Review Panel Online Survey

Introduction

Thank you for your participation in this research (USF IRB 22494) regarding financial literacy. Information gathered from this study may be utilized to assist small business organizations and governmental agencies guide financial literacy education. Because of your background and expertise, your participation and input are invaluable.

This assessment survey is a part of dissertation research. The survey consists of three sections: 5 demographic questions, 20 knowledge questions, and five self-assessments questions. Completing it takes less than 15 minutes.

Your participation is voluntary and uncompensated. No identifying information is being gathered and your participation is anonymous and confidential. There are no known benefits or risks to this research. The data will be kept for a period of 5 years.

Additionally, information gathered will ONLY be used for research related to financial literacy within this study. Your participation will not cause you to receive any emails unrelated to this survey.

Your email will not be connected to the results of this assessment in any way. If you are interest in the final results of the study, please register your email separately at the end of the survey.

If, at any time, you have questions regarding your participation or issues with this website, please contact the Principal Investigator, Pearl Dahmen Para, at pjdahmen@mail.usf.edu.

Demographic Questions

Please respond to the following demographic questions:

D1. What is your gender?

- Male
- Female

Other (Please Specify)

D2. What year were you born?

D3. How many years have you owned your current small business?

- 0 2 years
- 3 5 years
- 5 10 years
- Over 10 Years
- Not in business yet

Final Review Panel Online Survey (continued)

D4. Not counting yourself, how many employees are there in your current business?
No employees
O Less than 5 employees
O More than 5 employees
D5. What was the last year of education you completed?
Did not complete high school
O High school graduate - regular high school or GED
Some college
College graduate
O Post graduate education
Knowledge Questions
Please respond to the following financial literacy questions.
Q1. You should have an emergency fund that covers three to six months of your expenses.
O True
◯ False
O Don't know
O Prefer not to answer
Q2. With compound interest, interest is earned on interest in addition to the original deposit or loan.
◯ True
◯ False
O Don't know
O Prefer not to answer
Q3. The earlier you start saving for retirement, the more money you will have because the effects of compounding interest increase over time.
◯ True
◯ False
O Don't know

Final Review Panel Online Survey (continued)

	Prefer not to answer
Q4. In yo	Bank A offers monthly compounding and Bank B offers yearly compounding. Both banks offer the same interest rate. our opinion, which bank would you choose if you wanted a higher return?
\bigcirc	Bank A
\bigcirc	Bank B
\bigcirc	Both banks offer the same return.
\bigcirc	Don't know
\bigcirc	Prefer not to answer
Q5.	Annually, you can receive a free credit report from
\bigcirc	a credit union.
\bigcirc	a commercial bank.
\bigcirc	the Better Business Bureau
\bigcirc	a credit bureau.
\bigcirc	Don't know
\bigcirc	Prefer not to answer
Q6. but,	Your credit report includes your outstanding credit balances, payments history, and any inquiries made by creditors, does not include employment data and pubic record information. True
\bigcirc	False
\bigcirc	Don't know
\bigcirc	Prefer not to answer
Q7.	To reduce the total interest or finance costs paid over the life of a loan, you should choose a loan with the
\bigcirc	Lowest monthly payment
\bigcirc	Longest repayment term
\bigcirc	Shortest repayment term
\bigcirc	Don't know
\bigcirc	Prefer not to answer
Q8.	If you always pay the full balance on your credit card, which of the following is least important?
\bigcirc	Annual interest rate
\bigcirc	

Final Review Panel Online Survey (continued)

Annual fee
Credit Limit
O Don't know
Prefer not to answer
Q9. Suppose you owe \$1,000 on your credit card and the interest rate you are charged is 20% per year compounded annually. If you didn't make any payments on the credit card, at this interest rate, how many years would it take for the amount you owe to double?
○ 2 years
C Less than 5 years
○ 5 to 10 years
O More than 20 years
O Don't know
Prefer not to answer
Q10 To ensure that some of your retirement savings will not be subject to income tax upon withdrawal, you would contribute to:
A Traditional IRA (Individual Retirement Account)
O A Roth IRA
○ A 401(k) plan
O Don't know
Prefer not to answer
Q11. If you have a savings account at a bank and do not withdraw any money, you will not have to pay taxes on the interest you earn.
◯ True
◯ False
O Don't know
O Prefer not to answer
Q12. A 15-year mortgage typically requires higher monthly payments than a 30-year mortgage, but the total interest paid over the life of the loan will be less.
O True
○ False
O Don't know
Prefer not to answer

Final Review Panel Online Survey (continued)

True	
False	
On't kno	W
O Prefer not	t to answer
Q14. Repeatincrease you	tedly refinancing your home mortgage over a short period of time results in added fees and points that further r debt.
True	
False	
🔵 Don't kno	w
O Prefer not	t to answer
Q15. If you b savings acco	buy certificates of deposit (CDs), savings bonds, or Treasury bills, you can earn higher returns than on a unt, with little or no added risk.
🔘 True	
False	
On't kno	w
Prefer not	t to answer
Q16. Do you	think buying stock in a single company is safer than buying stock in several different companies?
Yes	
🔘 No	
On't kno	W
Prefer not	t to answer
Q17. It is position and sl	ssible to reduce the risk of losing your investment when investing in the stock market by buying a wide range hares.
True	
False	
🔘 Don't kno	W
Prefer not	t to answer

Final Review Panel Online Survey (continued)

your money										
C True										
O False										
O Don't know										
Prefer not to answer										
0										
Q19. Your net worth is	vpondituro ond	incomo								
the difference between your ca	the difference between your cash inflow and outflow.									
	the difference between your bank borrowing and savings.									
	adilities and as	ssets.								
 Doubt la pove. 										
Don't know										
Prefer not to answer										
 Q20. If your assets increase by Increase by \$2,000 Increase by \$3,000 Increase by \$8,000 Don't know Don't know Prefer not to answer Self-Assessment Questions For the following five questions, S1. I am comfortable with my kn debit cards, mortgages, installm	\$5,000 and y rate (or asse nowledge to d ent payments	ss) your belief leal with day-t	f or confidence o-day finance expenses.	\$3,000, your n	et worth wou ne statement ch as checkir	ts. ng accounts,	credit and			
	Strongly		Slightly	Neither Disagree or	Slightly		Strongly			
To what extent do you agree	Disagree	Disagree	Disagree	Agree	Agree	Agree	Agree			
or disagree with the above	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc			
รเสเยทยาแ.										
S2. I am comfortable with my ki	nowledge to n	nanage mone	y, budgeting	and planning n Neither	ny expenditu	res.				

Final Review Panel Online Survey (continued)

	Strongly Disagee	Disagree	Slightly Disagree	Disagree or Agree	Slightly Agree	Agree	Strongl Agree
To what extent do you agree or disagree with the following statement.	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
S3. I am comfortable with my	knowledge to m	nanage cred	it cards and de	ebt like car lo	ans or a mortga	age.	
	Strongly Disagree	Disagree	Slightly Disagree	Neither Disagree or Agree	Slightly Agree	Agree	Strongly Agree
To what extent do you agree or disagree with the following statement.	0	\bigcirc	0	\bigcirc	0	0	0
S4. I understand investment p IRAs, 401Ks, and annuitites).	roducts (for exa	ample stocks	s, bonds, and r	nutual funds) and retiremen	t accounts	(such as
	Strongly Disagree	Diagree	Slightely Disagree	Neither Disagree or Agree	Slightly Agree	Agree	Strongl Agree
To what extent do you agree or disagree with the following statement.	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
S5. Consider your overall fina	ncial knowledge	Э,					
	Very Low.	Low	Slightly Low	Average	Slightly High	High	Very Hig
How would you assess your	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	0
overall financial knowledge?							0
overall financial knowledge?							0

Appendix S

Final Financial Literacy Instrument

Thank you for your participation in this research (USF IRB 22494) regarding financial literacy. Information gathered from this study may be utilized to assist small business organizations and governmental agencies guide financial literacy.
education. Because of your background and expertise, your participation and input are invaluable.
This assessment survey is a part of dissertation research. The survey consists of three sections: Five demographic questions, 22 knowledge questions, and five self-assessments questions. Completing it takes less than 15 minutes.
Your participation is voluntary and uncompensated. No identifying information is being gathered and your participation is anonymous and confidential. There are no known benefits or risks to this research. The data will be kept for a period of 5 years.
Additionally, information gathered will ONLY be used for research related to financial literacy within this study. Your participation will not cause you to receive any emails unrelated to this survey.
Your email will not be connected to the results of this survey in any way. If you are interest in the final results of the study, please provide your email separately at the end of the survey.
If, at any time, you have questions regarding your participation or issues with this website, please contact the Principal Investigator, Pearl Dahmen Para, at pjdahmen@mail.usf.edu.
Demographic Questions
Please respond to the following demographic questions:
What is your gender?
O Male
○ Female
Other (Please Specify)
What year were you born?
How many years have you owned your current small business?
O 0 - 2 years
3 - 5 years
○ 6 - 10 years
Over 10 years

Final Financial Literacy Instrument (continued)

Not counting yourself, how many employees are there in your current business?	
No employees	
Less than 5 employees	
O More than 5 employees	
What is the highest level of education you completed?	
Did not complete high school	
O High school or GED	
O Associates Degree	
O Bachelor's Degree	
O Master's Degree or higher	
Knowledge Questions	
Please respond to the following financial literacy questions:	
It is recommended that you have an emergency fund that covers three to six months of personal expenses.	
O True	
○ False	
O Don't know	
Prefer not to answer	
Compound interest is:	
The same as simple interest	
Only important for short-terms savings or loans	
Interest earned on interest in addition to the original deposit or loan	
O Don't know	
Prefer not to answer	
The earlier you start regularly saving for retirement, the more money you can accumulated because the effects of compounding interest over time.	
O True	
○ False	
O Don't know	
\odot	

Prefer not to answer
Bank A offers monthly compounding and Bank B offers yearly compounding. Both banks offer the same interest rate. In
your opinion, which bank would you choose if you wanted a higher return?
Both banks offer the same return
O Bank A
O Bank B
O Don't know
Prefer not to answer
If you are monitoring or checking your personal credit score or history, you can receive a free credit report annually from:
O A credit union
O A commercial bank
◯ The Better Business Bureau
O A credit bureau
O Don't know
O Prefer not to answer
Your credit report includes your outstanding credit balances, payments history, and any inquiries made by creditors. It does not include employment data and public record information.
◯ True
◯ False
O Don't know
O Prefer not to answer
If your goal was to reduce the total interest or financing costs paid over the life of a loan, which of the following would accomplish this:
Constant
Longest repayment term
Shortest repayment term
O Don't know
O Prefer not to answer
If you always pay the full balance on your credit card, which of the following is of least concern?
Annual interest rate

 Annual fee
O Credit Limit
O Don't know
Prefer not to answer
To ensure that some of your retirement savings will not be subjected to income tax upon withdrawal, you would contribute to:
A Traditional IRA (Individual Retirement Account)
A 401(k) plan
A 15-year mortgage typically requires higher monthly payments than a 30-year mortgage, but the total interest paid over the life of the loan will be less.
◯ True
◯ False
O Don't know
Prefer not to answer
If the interest rate on an adjustable rate methods lean account your monthly markage normants will not as up
True True
Prefer not to answer
Repeatedly refinancing your home mortgage over a short period of time results in added fees and interest rate points that further increase your debt.
◯ True
○ False
O Don't know
O Prefer not to answer
If you buy certificates of deposit (CDs), savings bonds, or Treasury bills, you can earn higher returns than on a savings account, with little or no added risk.

True
O False
O Don't know
Prefer not to answer
Do you think buying stock in a single company is safer than buying stock in several different companies?
○ Yes
○ No
O Don't know
Prefer not to answer
It is possible to reduce the risk of losing your investment when investing in the stock market by buying a wide range of stocks and shares.
◯ True
◯ False
O Don't know
Prefer not to answer
If you were to invest \$1,000 in a stock mutual fund, it would be possible to have less than \$1,000 when you withdraw your money.
O True
◯ False
O Don't know
Prefer not to answer
Your net worth is:
The difference between your cash inflow and outflow
The difference between your bank borrowing and savings
O Your income minus expenditures
Your assets minus liabilities
O None of the above
O Don't know
Prefer not to answer

Final Financial Literacy Instrument (continued)

If your assets increase by \$5,000 and your liabilities decrease by \$3,000, your net worth would:

- Increase by \$2,000
- Increase by \$3,000
- Increase by \$8,000
- On't know
- Prefer not to answer

Block 4

If you had \$100 in a savings account and the interest rate was 2% per year. After 5 years how much do you think you would have in the account if you left the money to grow?

More than \$102

Exactly \$102

- Less than \$102
- On't know
- Prefer not to answer

Imagine that the interest rate on your savings account was 1% per year and inflation was 2% per year. After 1 year, would you be able to buy:

- More than today
- Exactly the same
- Less than today
- On't know
- Prefer not to answer

If interest rates rise, what will typically happen to bond prices?

- They will rise
- They will fall
- They will stay the same
- There is no relationship between bond prices and the interest
- On't know
- Prefer not to answer

Buying a company stock usually provides a safer return than a stock mutual fund.

Final Financial Literacy Instrument (continued)

True

- False
- On't know
- Prefer not to answer

Self-Assessment Questions

For the following five questions, rate (or assess) your belief or confidence on each of the statements.

I am comfortable with my knowledge to deal with day-to-day financial matters, such as checking accounts, credit and debit cards, mortgages, installment payments, and tracking expenses.

	Strongly Disagree	Disagree	Slightly Disagree	Neither Disagree or Agree	Slightly Agree	Agree	Strongly Agree
To what extent do you agree or disagree with the above statement.	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

I am comfortable with my knowledge to manage money, budgeting, and planning my expenditures.

	Strongly Disagee	Disagree	Slightly Disagree	Neither Disagree or Agree	Slightly Agree	Agree	Strongly Agree
To what extent do you agree or disagree with the following statement.	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

I am comfortable with my knowledge to manage credit cards and debt like car loans or a mortgage.

	Strongly Disagree	Disagree	Slightly Disagree	Neither Disagree or Agree	Slightly Agree	Agree	Strongly Agree	
To what extent do you agree or disagree with the following statement.	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	

I understand investment products (for example stocks, bonds, and mutual funds) and retirement accounts (such as IRAs, 401Ks, and annuities).

	Strongly Disagree	Diagree	Slightely Disagree	Neither Disagree or Agree	Slightly Agree	Agree	Strongly Agree
To what extent do you agree or disagree with the following statement.	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Very Low	Low	Slightly Low	Average	Slightly High	High	Very High
\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
	Very Low	Very Low Low	Very Low Low Slightly Low	Very Low Low Slightly Low Average Image: Ima	Very Low Low Slightly Low Average Slightly High Image: Imag	Very Low Low Slightly Low Average Slightly High High Image:

Appendix T

Participants' Email Communications

Participant Initial Email

From: Florida SBDC Network <<u>no-reply@floridasbdc.org</u>> Date: Wed, Nov 4, 2015 at 2:23 PM Subject: Florida Small Business Financial Literacy Continuing Professional Education Study To: <<u>smallbusinessowner.address></u>

Dear Florida SBDC client,

As a small business owner, and an active client of the Florida SBDC Network, your input and support is requested to assist in academic research (USF IRB22494) related to Florida Small Business Financial Literacy. Your participation is strictly voluntary and uncompensated. It will help us assess the financial literacy continuing professional education needs of small business owners in Florida as well as aid in defining programs to support your needs. The estimated time to complete the survey is less than 15 minutes.

All information gathered **WILL BE strictly confidential** and responses will remain **ANONOMYOUS**. You will never be asked to give your name or other identifying information. Additionally, information gathered will ONLY be used for research related to financial literacy within this study. Your participation will not cause you to receive any emails unrelated to this survey. If you are interested in receiving the final results of the study, please send an email to <u>pjdahmen@mail.usf.edu</u>.

This research is being conducted for final requirements in PhD studies at the University of South Florida. It is aimed at understanding the complex phenomena of financial literacy as it relates to small business owners in Florida. The intended audiences for the results include the dissertation committee, FSBDCN, ASBDC, SBA, SCORE, the Florida Legislature, and the President's Advisory Council on Financial Capacity.

To participate in this research, please click on the link below to access the survey. Please note that this survey is mobile friendly and can be accessed with a smartphone or tablet as well as a computer or laptop.

Link to survey: http://usf.az1.qualtrics.com/SE/?SID=SV_aa6QAelivhqP177

Thank you for your participation.

Best regards,



FSBDC at the University of South Florida Muma College of Business 1101 Channelside Drive, Suite 210 Port Tampa Bay Building Tampa, Fl. 33602-3613 813.905.5819 www.SBDCTampaBay.com
Participant 2nd Reminder Email

From: Florida SBDC Network <<u>no-reply@floridasbdc.org</u>> Date: Wed, Nov 18, 2015 at 2:17 PM Subject: Reminder Florida Small Business Financial Literacy Continuing Professional Education Study To: <smallbusinessowner> Dear Florida SBDC client, You have already received an email inviting you to participate in the Florida Small Business Financial Literacy research (USF IRB 22494). If you have completed the survey, thank you. If you have not, please take less than 15 minutes to assist in gathering this valuable information that will help profile Florida small business financial literacy, as well as aid in defining education and programs to support your needs. You were selected because you are a small business owner and an active client of the Florida SBDC Network. Your participation is strictly voluntary, and will be greatly appreciated. All information gathered WILL BE strictly confidential and responses will remain ANONOMYOUS. You will never be asked to give your name or other identifying information. Additionally, information gathered will ONLY be used for research related to financial literacy within this study. Your participation will not cause you to receive any emails unrelated to this survey. If you are interested in receiving the final results of the study, please send an email to pjdahmen@mail.usf.edu. This research is being conducted for final requirements in PhD studies at the University of South Florida. It is aimed at understanding the complex phenomena of financial literacy as it relates to small business owners in Florida. The intended audiences for the results include the dissertation committee, FSBDCN, ASBDC, SBA, SCORE, the Florida Legislature, and the President's Advisory Council on Financial Capacity. To participate in this research, please click on the link below to access the survey. Please note that this survey is mobile friendly and can be accessed with a smartphone or tablet as well as a computer or laptop. Link to survey: http://usf.az1.qualtrics.com/SE/?SID=SV_aa6QAelivhqP177 Thank you for your participation. Best regards, FSBDC at the University of South Florida Muma College of Business MERIC 1101 Channelside Drive, Suite 210 Port Tampa Bay Building Tampa, Fl. 33602-3613 813.905.5819 www.SBDCTampaBay.com Hosted by

Participant 3rd Reminder Email



Participant Final Initial Email

From: Florida SBDC Network <<u>no-reply@floridasbdc.org</u>> Date: Tue, Dec 15, 2015 at 9:39 AM Subject: Final reminder Florida Small Business Financial Literacy Study To: pjdahmen@mail.usf.edu

Dear Florida SBDC client,

If you have completed the FSBDCN Florida Financial Literacy survey, **thank you. The response to this survey has been very good. We are at 80% of the needed survey responses.** If you have not responded, please do so now to help us achieve our goal.

We appreciate your valuable time and hope you will take less than 15 minutes to assist us in gathering this valuable information that will help profile small business owner's financial literacy in Florida as well as aid in defining education and programs to support your needs.

You were selected because you are a small business owner and an active client of the Florida SBDC Network. Your participation is strictly voluntary, and will be greatly appreciated.

All information gathered **WILL BE strictly confidential** and responses will remain **ANONOMYOUS**. You will never be asked to give your name or other identifying information. Additionally, information gathered will ONLY be used for research related to financial literacy within this study. Your participation will not cause you to receive any emails unrelated to this survey. If you are interested in receiving the final results of the study, please send an email to <u>pjdahmen@mail.usf.edu</u>.

This research is being conducted for final requirements in PhD studies at the University of South Florida. It is aimed at understanding the complex phenomena of financial literacy as it relates to small business owner's in Florida. The intended audiences for the results include the dissertation committee, FSBDCN, ASBDC, SBA, SCORE, the Florida Legislature, and the President's Advisory Council on Financial Capacity.

Please note that this survey is mobile friendly and can be accessed with a smartphone or tablet as well as a computer or laptop.

Link to survey: http://usf.az1.qualtrics.com/SE/?SID=SV_aa6QAelivhqP177

Thank you for your participation.

Best regards,



FSBDC at the University of South Florida Muma College of Business 1101 Channelside Drive, Suite 210 Port Tampa Bay Building Tampa, Fl. 33602-3613 813.905.5819 www.SBDCTampaBay.com

Appendix U

IRB Approval Letter



IRB Approval Letter (continued)

Exempt (i.e., adding a focus group, access to identifying information, adding a vulnerable population, or an intervention), these changes require a new application. However, administrative changes, including changes in research personnel, do not warrant an amendment or new application. Given the determination of exemption, this application is being closed in ARC. This does not limit your ability to conduct your research project. We appreciate your dedication to the ethical conduct of human subject research at the University of South Florida and your continued commitment to human research protections. If you have any questions regarding this matter, please call 813-974-5638. Sincerely, Ar CAn Kristen Salomon, Ph.D., Vice Chairperson USF Institutional Review Board

Appendix V

Summary of Sources and Permissions Used to Obtain Financial Literacy Questions for Dissertation Study

	0	December /	Veen	Demoiseien	
Ŧ	Survey	Researcher/ Author(s)	Year	Permission	URL Reference
1	Investing IQ	Volpe, Chen, & Pavliko (1996)	1996	What's Your Investing IQ? Money Forecast 1993, Time, Inc. Public document access	http://maagblog.ysu .edu/financialliterac y/files/2009/03/volp e4.pdf
2	Survey of Personal Financial Literacy	Chen & Volpe (1998)	1998	Email sent 5-2-2015 Response Received 5-31-2015	
3	Survey of Consumers	Hilgert, Hogarth & Beverly (2003)	2002	Sponsored by Federal Reserve Board, Department of Treasury, data collected by the NORC of the University of Chicago National Opinion Research Center	http://www.federalr eserve.gov/econres data/scf/aboutscf.ht m
					http://www.norc.org /About/Pages/our- history.aspx
4	HRS	Lusardi & Mitchell (2007c)	2004	HRS Public Files and Data sets sponsored by the National Institute on Aging (grant number NIA U01AG009740) and conducted by the University of Michigan	http://hrsonline.isr.u mich.edu/index.php ?p=regcou

*Acronym defined in table below.

Continued on the next page

Summary of Sources and Permissions Used to Obtain Financial Literacy Questions for Dissertation Study (continued)

#	Survey*	Researcher/ Author(s)	Year	Permission	URL Reference
5	DHS	van Rooij, Lusardi & Alessie (2005)	2005	Data from special financial literacy module publicly available	http://www.nber.org /papers/w17339
6	ALP	Lusardi & Mitchell (2007b)	2006	Support of Social Security Administration	http://www.rand.org /labor/centers/finan cial-literacy.html
7	TNS Global	Lusardi & Tufano (2009)	2007	National Bureau of Economic Research NBER Working Paper	http://www.nber.org /papers/w14808
8	Jump\$tart	Mandell (2008)	2008	Merrill Lynch Foundation underwriting costs of study (2008 study)	http://www.jumpstar t.org/survey.html
9	CogEcon	Delavande, Rohwedder & Willis (2008); Hsu (2011)	2008	Public access through HRS part of the Unified Studies of Cognition (CogUSC) led by cognitive psychologist Jack McArdle at the University of Southern California The Cognitive Economics Project is funded by the National Institute on Aging, grant NIA P01 AG026571.	http://ebp- projects.isr.umich.e du/CogEcon/survey s.html
		(-)			http://ebp- projects.isr.umich.e du/CogEcon/index. html

Continued on the next page

Summary of Sources and Permissions Used to Obtain Financial Literacy Questions for Dissertation Study (continued)

#	Survey*	Researcher/ Author(s)	Year	Permission	URL Reference
10	NFCS	Lusardi (2011)	2009	The National Financial Capability Study (NFCS) was funded by the FINRA Investor Education Foundation and conducted by Applied Research and Consulting.	The National Financial Capability Study (NFCS) was funded by the FINRA Investor Education Foundation
11	WBG	World Bank (2009)	2009	Financial support for work provided by United States Agency for Interanational Development (USAID)	http://web.worldban k.org/WBSITE/EXT ERNAL/TOPICS/E XTFINANCIALSEC TOR/0,,contentMD K:22766495~page PK:148956~piPK:2 16618~theSitePK:2 82885.00.html
12	FLAT	Finke & Huston (2014)	2011	Email sent 5-2-2015 Response received 5-28-2015	
13	OECD	Atkinson & Messy (2011)	2012	OECD account allows to access-Account info attached	

Continued on next page

Summary of Sources and Permissions Used to Obtain Financial Literacy Questions for Dissertation Study (continued)

#	Survey*	Researcher/ Author(s)	Year	Permission	URL Reference
14	USIS	Alhenawi & Elkhal (2013)	2013	Email sent 5-2-2015 Response received 5-2-2015	
*Acro	nym defined	in table below			

Acronym defined in table below.

Key to Pool of Financial Literacy Questions Source Codes: Source, Number of Questions (n), and References

Code	Source	n	Reference
ALP	SurveyAmerican Life Panel	23	Lusardi & Mitchell, 2007; RAND American Life Panel, 2014
CogE	Cognitive Economics Journal Article	24	Hsu, 2011
CV	Journal Article	31	Chen & Volpe, 1998
DHS	Survey—De Nederlandsche Bank Household Survey	16	van Rooij, Lusardi, & Alessie, 2005
FH	Journal Article	16	Finke & Huston, 2014
FLN	Journal Article	13	Fernande, Lynch, & Netemeyer, 2014
HHB	Journal Article	28	Hilgert, Hogarth, & Beverly, 2003
HRS	SurveyHealth and Retirement Study	3	Agnew, 2013; Beckmann, 2013; Gale & Levine, 2010; Lusardi, 2012; Lusardi & Mitchell, 2008, 2011; Lusardi, Mitchell, & Curto, 2010
J\$	Jump\$tart	31	JumpStart Coalition, 2007; Mandell, 2006, 2008; Mandell & Klein, 2007, 2009
K1	Journal Article	20	Knolls & Houts, 2012
K2	Journal Article	6	Kunovskayaet al., 2014
NFCS	SurveyNational Financial Capabilities Survey	5	Bumcrot et al., 2013; Fernandes et al., 2014; Hastings, Madrian, & Skimmyhorn, 2012; Lachance, 2014; Lusardi, 2011; Mottola, 2013; Nicolini et al., 2013; Scheresberg, 2013; Simms, 2014

Continued on the next page

Key to Pool of Financial Literacy Questions Source Codes: Source, Number of Questions (n), and References (continued)

Code	Source	n	Reference
OEDC	SurveyOrganization for Economic Co-operation and Development	8	Atkinson & Massey, 2011
TNS	SurveyTaylor, Nelson, Sofres Global Survey	3	Lusardi & Tufano, 2009
USIS	University of Southern Indiana Study	7	Alhenawi & Elkhal, 2013
VCP	Journal Article	10	Volpe et al., 1996
WBG	SurveyWorld Bank Group	4	World Bank, 2009

Email Communication for Permission from Dr. H. Chen

	Pearl Para <pearlpara@gmail.com></pearlpara@gmail.com>
Fwd: Please forward to Dean Haiyang Chen: P literacy assessment questions 1 message	ermission to use financial
Pearl Para <pjdahmen@mail.usf.edu> To: Pearl Para <pearlpara@gmail.com></pearlpara@gmail.com></pjdahmen@mail.usf.edu>	Wed, Jan 27, 2016 at 10:17 AM
Forwarded message From: Chen, Haiyang < <u>chenha@marshall.edu</u> > Date: Sun, May 31, 2015 at 5:29 PM Subject: Re: Please forward to Dean Haiyang Chen: Permission to To: Pearl Para < <u>pjdahmen@mail.usf.edu</u> >	o use financial literacy assessment questions
Dear Pearl,	
Thank you for your interest in our work. You can use the survey.	
Haiyang	
From: Pearl Para <pjdahmen@mail.usf.edu> Sent: Sunday, May 31, 2015 2:52 PM To: Chen, Haiyang Subject: Fwd: Please forward to Dean Haiyang Chen: Permission</pjdahmen@mail.usf.edu>	to use financial literacy assessment questions
Dr. Chen,	
Thank you for speaking with me last Thursday regarding permission your 1998 article "An Analysis of Personal Financial Literacy Amo	on to use the financial literacy questions from ng College Students".
This email is to confirm our conversation. Please replay so that I committee.	am have evidence for my dissertation
Again, thank you.	
Pearl Dahmen Para 941-962-1211	
Forwarded message From: Pearl Para <pjdahmen@mail.usf.edu<mailto:pjdahmen@ma Date: Fri, May 8, 2015 at 10:36 AM Subject: Please forward to Dean Haiyang Chen: Permission to use To: bittinger@marshall.edu<mailto:bittinger@marshall.edu></mailto:bittinger@marshall.edu></pjdahmen@mail.usf.edu<mailto:pjdahmen@ma 	ail.usf.edu>> e financial literacy assessment questions
Dr. Chen,	
I am writing to request permission to include the financial literacy	

Continued on next page

Email Communication for Permission from Dr. H. Chen (continued)

assessment questions developed for the 1998 *Financial Services Review *article *An Analysis of Personal Financial Literacy Among College Students *as part of a pool of questions, which will be used to develop an instrument for my research focusing on the financial literacy continuing education needs of small business owners in Florida.

As a PhD candidate at the University of South Florida (Tampa, FL) pursuing dissertation, I plan to develop the instrument for my research from questions tested in previous research. The permission I am seeking is to include these questions in a pool of questions from several surveys. After thorough review and evaluation processes, if questions are retained for the final instrument, permission will include using the questions as part of the final survey.

If you would like further information regarding my research, please feel free to contact me via email or phone (941-962-1211<tel:%28941-962-1211>). I will also be happy to share the results of my research once it has been completed. Let me know if you have an interest hearing from me when my research is completed.

Thank you in advance for your consideration in this matter and response to this email.

Best regards,

Pearl Dahmen Para PhD Candidate University of South Florida Tampa, FL 941-962-1211<tel:941-962-1211>

Email Communication for Permission from Dr. S. Huston

Gmail"	Pearl Para <pearlpara@gmail.com></pearlpara@gmail.com>
Fwd: Permission to use (Financial Literacy Assessmin dissertation research	nent Test) FLAT questions
Pearl Para <pjdahmen@mail.usf.edu> To: Pearl Para <pearlpara@gmail.com></pearlpara@gmail.com></pjdahmen@mail.usf.edu>	Wed, Jan 27, 2016 at 10:19 AM
Forwarded message From: Pearl Para <pjdahmen@mail.usf.edu> Date: Sun, May 31, 2015 at 2:47 PM Subject: Re: Permission to use (Financial Literacy Assessment Test) FLA To: "Huston, Sandra" <sandra.huston@ttu.edu></sandra.huston@ttu.edu></pjdahmen@mail.usf.edu>	NT questions in dissertation research
Thank you.	
On Thu, May 28, 2015 at 1:53 PM, Huston, Sandra < <u>sandra.huston@ttu.e</u> Yes, I think this would be fine.	edu> wrote:
Sent from my iPhone	
On May 28, 2015, at 7:32 AM, "Pearl Para" <pjdahmen@mail.usf.edu></pjdahmen@mail.usf.edu>	wrote:
Forwarded message From: Pearl Para <pjdahmen@mail.usf.edu> Date: Fri, May 8, 2015 at 9:36 AM Subject: Permission to use (Financial Literacy Assessment Test research To: sandra.huston@ttu.edu, michael.finke@ttu.edu, vickie.hamp</pjdahmen@mail.usf.edu>) FLAT questions in dissertation
Hello,	
I am writing to request permission to include the FLAT (Financial questions developed in the Texas Tech University Financial Litera as part of a pool of questions that will be used to develop an instr on the financial literacy continuing education needs of small busin	Literacy Assessment Test) cy Assessment Project in 2011 ument for my research focusing ess owners in Florida.
As a PhD candidate at the University of South Florida (Tampa, Fl develop the instrument for my research from questions tested in p permission I am seeking is to include the FLAT questions in a poor surveys. After thorough review and evaluation processes, if quest instrument, permission will include using the questions as part of t	-) pursuing dissertation, I plan to revious research. The of questions from several ions are retained for the final the final survey.
If you would like further information regarding my research, please	e feel free to contact me via

Email Communication for Permission from Dr. S. Huston (continued)

email or phone (941-962-1211). I will also be happy to share the results of my research once it has been completed. Let me know if you have an interest hearing from me when the research is complete.
Thank you in advance for your consideration in this matter and response to this email.
Best regards,
Pearl Dahmen Para
PhD Candidate
University of South Florida
Tampa, FL
941-962-1211

Email Communication for Permission from Dr. Alhenawi

GMail [®]	Pearl Para <pearlpara@gmail.com></pearlpara@gmail.com>
Fwd: Permission to use financial literacy asse	essment questions
Pearl Para <pjdahmen@mail.usf.edu> To: Pearl Para <pearlpara@gmail.com></pearlpara@gmail.com></pjdahmen@mail.usf.edu>	Wed, Jan 27, 2016 at 10:20 AM
From: Pearl Para <pjdahmen@mail.usf.edu> Date: Sun, May 3, 2015 at 2:14 PM Subject: Re: Permission to use financial literacy assessment que To: "Alhenawi, Yasser" <ya22@evansville.edu></ya22@evansville.edu></pjdahmen@mail.usf.edu>	stions
Dr. Alhenawi,	
Thank you.	
Pearl Para	
On Sat, May 2, 2015 at 9:08 PM, Alhenawi, Yasser <ya22@evan< td=""><td>sville.edu> wrote:</td></ya22@evan<>	sville.edu> wrote:
Pearl-	
You may employ the questions in our paper as long citation.	as you use proper reference and
Dr. Alhenawi	
From: Pearl Para <pjdahmen@mail.usf.edu> Sent: Saturday, May 2, 2015 4:44 PM To: Alhenawi, Yasser; kelkhal@usi.edu Subject: Permission to use financial literacy assessment quest</pjdahmen@mail.usf.edu>	ions
I am writing to request permission to include the fin developed for the 2013 Financial Services Review ar Households: Knowledge vs. Long- term Financial Pla which will be used to develop an instrument for my literacy continuing education needs of small busines	ancial literacy assessment questions ticle Financial Literacy of U.S. anning as part of a pool of questions, research focusing on the financial as owners in Florida.

Email Communication for Permission from Dr. Alhenawi (continued)

As a PhD candidate at the University of South Florida (Tampa, FL) pursuing dissertation, I plan to develop the instrument for my research from questions tested in previous research. The permission I am seeking is to include these questions in a pool of questions from several surveys. After thorough review and evaluation processes, if questions are retained for the final instrument, permission will include using the questions as part of the final survey.

If you would like further information regarding my research, please feel free to contact me via email or phone (941-962-1211). I will also be happy to share the results of my research once it has been completed. Let me know if you have an interest hearing from me at that time.

Thank you in advance for your consideration in this matter and response to this email.

Best regards,

Pearl Dahmen Para

PhD Candidate

University of South Florida

Tampa, FL

941-962-1211

About the Author

Earning her undergraduate degree in accounting at the University of Dayton in Ohio, Pearl J. Dahmen has been a Certified Public Accountant for over 35 years. Pearl entered the University of South Florida's graduate program in 2009, completing her M.A. in Adult Education. With a passion for learning and teaching, she entered the University of South Florida adult education Ph.D. program in 2011.

For nearly two decades, as owner of the small business Pearl Financial Services, Pearl was dedicated to supporting and consulting small business owners. After selling her business, Pearl became a certified consultant for the Small Business Development Center at the University of South Florida. Her experiences working with small business owners and observing their need for improved financial literacy skills inspired this research.

Presently, Pearl is employed by Vertex, Inc., a tax software company, where in additional to her technical responsibilities, she develops and delivers adult learning for workshops, conferences, webinars, and eLearning.